Crisis and Stability: The Global Financial Crisis, British Broadsheet Press and the Politics of Ideational Reversion

William John Vittery

Doctor of Philosophy

University of York

Politics

December 2015

Abstract

By analysing UK media narrations surrounding the global financial crisis, this thesis presents a critical engagement with existing constructivist institutionalist literature. Through the application of a 'dynamic tracing' methodology to British broadsheet newspaper discourse from 2007-10, the thesis reveals three significant, and interconnected, dynamics. Firstly, it highlights the existence of 'ideational reversion', whereby after a short period of flux through late-2008 and early-2009, prominent discourses by and large returned to the pre-crisis status quo ante. By analysing the pre-crisis, crisis, and post-crisis discourse holistically, a notably higher degree of overall ideational stability is found than the existing literature suggests would be the case. Secondly, it is demonstrated that ideational disjuncture within media commentary was effectively 'siloed' in the financial sector, meaning that the perception of crisis did not challenge broader conceptualisations of the neo-liberal economy. Thirdly, the impact of such reversion and siloing was to provide a greater social source of legitimacy, or strategic advantage, to orthodox austerity narratives than to their Keynesian alternative. On the back of these observations, conceptual extensions are put forward that involve developing a greater focus on the 'stickiness' of pre-existing discourse through crisis periods.

Table of Contents

Abstract	
Table of Contents	3
Acknowledgements	6
Author's declaration	8
Chapter 1: Introduction to the Thesis	g
Chapter 2. Constructivism, the British Broadsheet Press, and the Global Financial Cris	
2.1 Introduction	13
2.2 Constructivism and the Ontological Middle Ground	14
2.3 Structure versus Agency for Constructivists	20
2.4 Strategically Selective Contexts, Strategic Actors, and the Financial Crisis	22
2.5 Crisis and Strategic-Relational Action	23
2.6 The 'Realist Constructivist' Agenda and crisis analysis	26
2.7 The Relevance of Constructivism to Crisis	28
2.8 Conclusions	29
Chapter 3. Constructivism, the British Broadsheet Press, and the Global Financial Cris	
3.1 Introduction	30
3.2 Setting the Scene: Constructivism and the New Institutional Turn	31
3.3 Institutionalist Studies of Crisis	34
3.4 The Constructivist Institutionalist Alternative: Prospects and Weaknesses	36
3.5 Constructivist Responses to the Global Financial Crisis	42
3.6 When is a crisis not a crisis?	44
3.7 Discursive Institutionalism and Media Narration	46
3.8 Methodology	48
3.9 Conclusions	53
Chapter 4: Contextualising the Global Financial Crisis	55
4.1 Introduction	55
4.2 Introducing the Keynesian and Orthodox Paradigms	56
4.3 The Great Moderation and the Events of the Financial Crisis	60
4.31 The Leverage Economy	61

4.32 Current Account Imbalances	62
4.33 Monetary Consensus	63
4.4 The Onset of Crisis and its Political Response	64
4.5 The Economic Logic of Stimulus in the Current Context	66
4.6 The Economic Case for Austerity in the Current Context	69
4.7 Ideal Types and Their Political Usage	71
4.71 The Role of Uncertainty	
4.72 Stimulating Growth	73
4.73 Economic Priorities	74
4.74 A Case of Contestability	74
4.8 Conclusions	75
Chapter 5: Charting the Pre-Crisis Economic Discourse	76
5.1 Introduction	76
5.2 The Size and Role of the State	77
5.3 Fiscal Responsibility and State Capacity	81
5.4 Drivers of Economic Growth	87
5.5 The Tax Regime	89
5.6 Conclusions	91
Chapter 6: Political Economic Discourses of the Crisis Period: August 2008 – January 20	109 94
6.1 Introduction	94
6.2 A crisis of capitalism or a crisis of finance?	96
6.3 Corporate Capitalism in Crisis?	99
6.4 'No room for manoeuvre' and orthodoxy v stimulus	100
6.5 Medium term decisions	
6.6 Cures beyond fiscal policy	107
6.7 Conclusions	110
Chapter 7: Ideational Reversion: January 2009 – May 2010	113
7.1 Introduction	113
7.2 'No room for manoeuvre' and its critics	115
7.21 'No room for manoeuvre' and the 2009 Budget	116
7.22 The alternative discourse; maintaining the need for stimulus	120
7.3 The Late 2009 – May 2010 period	125
7.4 Public sector cuts and deficit reduction as a pre-requisite for economic growth	130
7.5 Stimulus, spending cuts and political trust	133
7.6 Non-state foundations of economic growth	139
7.7 Spending cuts and public services	144
7.8 Conclusions	146

Chapter 8: Conclusions	149
8.1 Introduction	149
8.2 Overview of the Major Conclusions from the Research	150
8.3 Further Research	156
Abbreviations	158
Bibliography	159

Acknowledgements

Naturally, I have a great number of people to thank for assisting me in the development of this project and, perhaps even more importantly, keeping me sane(ish) along the way!

Firstly, to my supervisor Dr Liam Clegg who has provided considerable assistance, guidance and support throughout the course of my research. His close attention to detail, and in particular his ability to bring out the key messages clearly and concisely has been invaluable in compiling the final written version of this PhD.

My other major academic acknowledgement goes to Dr Michael Neu, who first inspired me and showed belief in me in my academic studies. My ability to think critically and express such thought through my writing was improved considerably by the support and advice he offered as a tutor, and I would not have begun the journey of postgraduate study without having been inspired by him. He has continued to be an unofficial mentor, and more importantly a friend throughout my time at university.

I would also like to thank the ESRC for funding my thesis, and the Library of Congress in Washington, DC, which hosted me for three months of study in 2014. I would like to thank the International Student House in Washington for providing me with accommodation during that stay, and to the countless friends who I made during those three months. The experience has given me friendships with people from every corner of the globe, many of whom I will now be free to visit again without the burden of thesis deadlines weighing on my shoulders.

On a more general note, I would like to thank the University of York's and in particular its Politics Department as a whole for hosting me, and for the resources provided that allowed me to undertake and complete the thesis. In particular, the warmth and humour of Lisa Webster, Liz O'Brien have considerably increased the enjoyment and decreased the stress of studying at postgraduate level. I also owe a debt of thanks to the other departmental staff who I have had the pleasure of working alongside during my time at York: Sue Leatt, Chris Clegg, Karen Saxton, Fran Roach and Caroline Carfrae.

I owe particular gratitude to the community of PhD students in the politics department at York, and those who I met in Washington, for the support network which provides vital shoulders to lean on during the inevitable moments of depression which accompany a thesis. More importantly than this, and far more important than the thesis itself, are the friendships which have developed in such

a context. Particular thanks go to John Mellors, Paul Tobin, Peg Murray-Evans, Adam Fusco, Fay Farstad, Elizabeth Thorson, Ben Chwistek and Robin Jervis, alongside too many others to name here.

Likewise I owe thanks to all the other important people in my life who have provided me with distractions, support, motivation and help whenever needed. Becca Armstrong, David Armstrong, Nikita Dunn, Gavin Fullstone, Ami Gell, Thomas Mutton, Michael Pirrie, Aimee Walton, Dean Winfield and Marc Winfield have all given me more than I can possibly find words to thank them for.

Chloë Ashley, Nick Hall, Adam Plumbley, Guy Thomas, Carly Wilkinson and the wider 'Thursday...' group have likewise given me important academic and non-academic support throughout my undergraduate and post-graduate studies, for which I am incredibly thankful.

Importantly, I must also thank my family, who have supported me throughout my life, and without whom none of this would have been possible. My parents, Kath and Tony, provided me with the perfect start in life and instilled the importance of education in me. Whilst my father did not live long enough to see me begin my thesis, I can only he would be proud of my achievement. And to my siblings, Helen, Susan and Edward, who have always put up with an annoying little brother with remarkable patience. Ed's activism in 2003 first piqued my interest in politics as a teenager, and perhaps therefore represents the first step on the journey I complete below.

And finally, special thanks go to the most important person in my life. This thesis is dedicated to Emma Vittery, who I had the honour of marrying whilst undertaking my PhD studies. In my happiest, and my darkest, moments, you have shone a light on my life that makes everything else pale in significance.

Author's declaration

I, William Vittery, declare that this thesis is a presentation of original work and I am the sole author. This work has not previously been presented for an award at this, or any other, University. All sources are acknowledged as References.

Chapter 1: Introduction to the Thesis

This thesis aims to make both empirical and conceptual developments relating to the understanding of economic crisis and political change. In particular, it seeks to make clarifications and extensions on the issue of the ability of crises to create the conditions for changing economic paradigms, something that is frequently cited by the growing constructivist institutionalist literature. The purpose of this introduction is to provide a brief summary of the intended argument of the thesis as a whole, and an overview of how each chapter will contribute to the elaboration of this overall goal.

The central research puzzle animating the thesis is one of explaining political continuity of economic policy in Britain after the global financial crisis, when much of the literature on economic crises suggests that they are a time ripe for paradigmatic change. In particular, it seeks to utilise constructivist institutionalist thinking, and to seek to develop this theory by suggesting that with a couple of revisions it is capable of discussing both political change and continuity with equal comfort. The major research questions to be addressed, then, are:

- 1. To what extent did the ideas underpinning political economy in Britain actually shift as a result of the global financial crisis?
- 2. To what extent can we observe a structural vacuum in which there was genuine uncertainty and a generalised sense of economic crisis?
- 3. To what extent did continuity of ideas constitute an important source of social legitimacy for ideas of austerity?

Firstly, the thesis intends to advance empirically our understanding of crises situations by exploring crisis discourse and narratives in detail to explore the extent of change that takes place through a crisis. Whilst existing works focusing on crisis and political change have tended to approach newspaper analysis as a supporting or subsidiary medium for focus, this thesis intends to utilise more systematically such documents as a primary focus of study, and ones that are justified in providing a proxy for the wider public debate on economic policy. The thesis will therefore analyse pre-crisis political economic discourse in the UK during 2007 and early 2008, and track the development of the prominent discourses through the crisis and post-crisis context to provide a rich detail of continuity and change over time.

The thesis as a whole seeks to work within the constructivist approach to political analysis, but highlights the need for greater specificity within this literature in relation to political paradigms and their breakdown in a time of crisis. It argues that constructivist institutionalism is of great value in

utilising discourse analysis to understand shifting economic ideas, but that it has to date been underutilised in explaining political continuity. Very briefly, constructivism holds that institutions tend to
maintain a political paradigm for understanding economic developments and creating policy
solutions to social problems, until a crisis arises which the existing paradigm cannot explain. When
this occurs, the structural power the existing paradigms holds is eradicated, creating a vacuum in
which new ideas can germinate and come to replace the previous narrative. As such, it is expected
that economic crises are moments of major turmoil and change, with narratives that are able to
appeal to a wide audience and explain the nature of the crisis likely to prevail. The Keynesian shift
during the Great Depression and the rise of monetarism in the 1970s-80s are seen as two such major
turning points where one policy paradigm was replaced by another. However, whilst in theory the
Global Financial Crisis would appear to be another such episode suited to such major paradigmatic
shifts, the success of the 'austerity' narrative in Britain highlights that continuity, rather than
change, characterises this crisis. Rather than being replaced in the aftermath of the crisis, neoliberal
orthodoxy has maintained its dominance within Western thinking and governance.

Therefore, whilst constructivist institutionalism has added a commendable level of depth and theoretical intricacy to these examples of major paradigmatic shift, it has not yet been so regularly applied to examples of ideational continuity, or in Hall's (1989) terms 'first' and 'second order' ideational change. To date historical institutionalism has been more commonly utilised in such instances. However, if we are to accept constructivist institutionalism's critique of historical institutionalism, covered in chapter 3, then one important extension of constructivist institutionalism must be to expand its focus and provide explanations for continuity as well as change. The purpose of this thesis is to take a first step in this direction. As will be stated at various points below, then, the purpose of the thesis is to work within the constructivist institutionalist tradition with the intention of developing and extending the literature, rather than merely critiquing it from without. These extensions will be both theoretical and empirical. Theoretically, it will suggest that a more 'realist' approach to discourse analysis is of value in explaining ideational contexts, and that crisis narratives need not permeate the entire political economic sphere but rather can be restricted to certain sections of the economy. Empirically, it will show that there is a greater amount of ideational 'reversion' in post-crisis ideas than has been identified to date.

The thesis will take the constructivist idea of a structural vacuum very seriously, however, and uses this as a justification for the research framework. It works within a constructivist mindset by suggesting that a time of crisis is indeed a very public political context, and therefore one in which public debate and the importance of ideas is of much more paramount importance than during times of economic and political stability. As such, it is argued that the public debate surrounding economic policy has a significant degree of rhetorical power in shaping the acceptability or 'social source of legitimacy' of different policy approaches. The primary research focus is on the broadsheet press as a valuable resource for analysis, for it provides a representative microcosm of the 'public debate' more broadly. By focusing on the ideas circulating the public debate of the broadsheet media, we can explain the more conventionally 'institutionalist' outcomes of the

acceptability or otherwise of Keynesian and orthodox economic policies. This research agenda is justified in more detail in chapter 3.

To expand on the methodological implications of this, through a 'dynamic tracing' methodology, starting in 2007 and finishing in 2010, the empirical chapters discover a process of 'ideational reversion' whereby the central ideas of the pre-crisis period re-emerged once the immediate danger of financial collapse had passed. With existing discourse highlighting a need for reduced government involvement in the economy and warning of government over-spending and a burgeoning public sector, in the context of budget deficits created by the crisis these ideas helped to justify fiscal consolidation as a policy approach over its Keynesian stimulative alternative. A significant finding within this is that blame for the crisis was effectively siloed within the financial sector, thus leaving the primary ideas governing the corporate sector un-touched. Thus a greater level of ideational continuity is observable compared to what existing constructivist literature would prepare us for. The effect of this greater than expected level of stability is that the orthodox paradigm had a greater social source of legitimacy than its Keynesian alternative in the aftermath of the crisis. The existence of this strategic advantage in the public imagination helps to explain its political success.

These empirical contributions in turn lead to two notable theoretical conclusions from the thesis. Firstly, the analysis of broadsheet discourse demonstrates that crises are not to be considered structural vacuums within which existing narratives are completely de-stabilised, as has previously been understood. Rather, crises appear to be episodes of confusion but also of some continuity, in which the pre-existing discursive context is still vitally important in shaping the likely policy outcome. Whilst the failure of the dominant pre-crisis paradigm in the existing literature is taken as representing a structural vacuum in which a new idea will come to dominate, here it is suggested that this is actually a more sociological process in which the description of failure from an external political analyst does not necessitate the public perception of the same thing; thus because academics highlight the crisis as demonstrating the failure of the neo-liberal financialisation growth model, does not necessitate that it is viewed in this way by the public, and press, at large.

The structure of the thesis to achieve these conclusions is set out as follows:

Chapter Two discusses the intellectual context of the development of constructivist as a unique approach to the understanding of politics. As such, the general meta-theoretical tenets of constructivist thought are justified against the more prevalent rationalist approach to academic political explanation. However, the chapter demonstrates the particular brand of 'realist constructivist' thought which will be applied in this instance; an approach based in IR scholarship but which I feel has lessons for domestic constructivist political analysis.

Chapter three focuses more explicitly on crises and constructivist institutionalist scholarship, highlighting the advances to date but importantly also the unanswered questions and potential theoretical weaknesses in the existing work. Here I focus on the overly-determinist thinking of

existing constructivist scholarship and the reliance in the post-crisis literature on more structural explanations for stability. I seek to apply strategic-relational thinking to allow for a research design that clears these existing hurdles whilst remaining explicitly constructivist. From this the dynamic tracing methodology of the thesis is justified; a focus on the broadsheet media in the pre-, mid- and post-crisis context is presented as a way of tracing closely ideational development, stability and change through differing economic conditions.

Chapter four provides an overview of the economic context of the thesis as a whole. It provides a review of the development of orthodox monetarist and Keynesian schools of economic thought, and the significance of crises to these bodies of work. Secondly it provides an overview of the economic context of the Great Moderation period which begun in the 1980s and was ended by the Global Financial Crisis. The existing tenets of political economic governance are therefore described. The chapter also provides an overview of the economic events of the financial crisis itself, providing the economic detail which the discourses covered in the empirical chapters attached themselves too. Finally, the chapter highlights the potential utility of both orthodox and Keynesian logic in the crisis context. In doing so, the chapter suggests that neither provides a more 'scientifically correct' understanding of the crisis, and hence justifies an essentially sociological and discursive approach to crisis resolution.

The empirical chapters are split into three parts. Chapter Five covers the period from the beginning of 2007 until the onset of the major heart of the crisis in late summer 2008. The purpose here is to provide an overview of the major pre-crisis discourses of the British press so that the extent of continuity and change within these can be gauged in the mid- and post-crisis periods. Chapter Six covers the mid-crisis period of the autumn and early winter of 2008-09, in which the period of uncertainty was at its height. Here it is explained that the severity of the crisis was sufficient for much of the pre-crisis concern with fiscal deficits to be side-lined, but that there was never a complete ideational reconfiguration even during this period. Chapter Seven covers the period from early 2009 until the General Election of 2010, when the sense of a 'fiscal crisis' became most prominent in the British press. It demonstrates the significance of the pre-crisis discourses to this issue and the methods of crisis resolution in general, thus suggesting that the austerity discourse held a strategic advantage over its rival.

Finally, chapter eight provides a summation of the major empirical and theoretical findings and conclusions. From this, a discussion is presenting relating to the major contributions of the thesis as a whole, and the suggestion of further study to extend its findings and contribute more widely to constructivist discussions of political continuity and change.

Chapter 2. Constructivism, the British

Broadsheet Press, and the Global

Financial Crisis: An Overview

2.1 Introduction

The Global Financial Crisis of 2007-08 generated deep questions over the nature and sustainability of the economic and political order of the Western world. We are now at a sufficient distance from the core of the crisis to take stock of the key points of continuity and change that have followed this point of disjuncture. Recent works have suggested that, in the realm of economic management, a fundamental continuity has been displayed in the post-2007 period (e.g. Moschella and Tsingou, 2013; Broome et al, 2012; McCarty, 2012). Whilst some governmental innovation has been observed to help mitigate the effects of the crisis (Clift and Woll, 2012), on a wider macroeconomic level there has not been a 'third order' change in ideas (Hall, 1993). Therefore rather than fore-telling a 'return of the state' and a major re-configuration of the relative powers of the state vis-à-vis the market, as many social democrats predicted, the impact of the financial crisis appears to have actually consolidated and entrenched the classically 'neo-liberal' pre-crisis role of the state in economic life (Gamble, 2013). This thesis aims to explore and explain such patterns of continuity and change in UK economic policy debate through the immediate pre-, mid- and post-crisis years of 2007-10.

Through the thesis as a whole, I critically draw upon established constructivist political economy literature to accomplish this goal. I offer a number of extensions to this literature that stem from my analysis of the UK case study. I specifically suggest that broadsheet media reporting constitutes a valuable focus of analysis as a means of exploring the constitution of ideas in political economic landscapes. By holistically reviewing prominent discourse across mainstream broadsheet reporting, I demonstrate the existence of two key trends. Firstly, I demonstrate a process if ideational reversion whereby the 'austerity' discourse that emerged from 2009 represented a return to the pre-

¹ The term 'the broadsheet newspaper media' is somewhat cumbersome, and hence with the exception of the methodology section where closer precision is necessary, I will shorten this to 'the media' for the rest of the thesis.

crisis status quo ante, in which criticism of excessive spending and public sector inefficiency had predominated across many outlets. Second, supporting initial findings from Pirie (2012), I demonstrate that ideational change associated with the crisis was effectively siloed; while commentators accepted that extraordinary monetary interventions were justifiable and necessary, other beliefs about the appropriate role and policies of the state in relation to the corporate sector remained in place

Through this chapter, I provide an overview of the explanatory potential of constructivist research in general, in order to justify the approach taken in this thesis. The thesis in general intends to broadly work within the constructivist intellectual tradition in order to suggest developments for its understanding and contextualisation of the issues of crisis and change. This chapter will begin by contrasting constructivist logic to those of the most extreme elements on the 'ontological spectrum', rationalism and post-structuralism. Further, it will highlight the importance, in particular, of the constructivist theoretical work on structure and agency which will be utilised in the thesis as a whole.. I will also adopt insights from the 'realist constructivist' school which I argue help to clarify this approach further. This more general justification of a constructivist approach to the topic will lead in to the following chapter, where a more detailed explanation of the approach I will adopt is reached.

2.2 Constructivism and the Ontological Middle Ground

Constructivism as a school of thought developed, and is still developing, largely as a result of frustrations at some of the enduring weaknesses of existing scholarship of positivist, rationalist and institutionalist nature (see Morrow, 1994; Levi, 1997; Scott, 2000; Shepsle, 2006). 'Rational Choice Theory', adopting a belief in the application of positivist economic methods to political analysis, had become central to the study of political outcomes by the mid-1980s, with William Riker calling it the greatest advancement made to political science (Riker, 1990). However, whilst its ability to offer statistical measurements and measurable outcomes gives such an approach a surface rigour which helps explain its popularity, the strength and reflectiveness of such research on political reality has increasingly been questioned, along with the theoretical logic underpinning it. The 'Comtean' foundationalist ontology (Matlary, 1997: 202) of positivism began to be questioned; thus also questioning the capability of political 'scientists' to truly capture the 'truth' about the 'real' world around us. As the spate of apostrophisation in the previous sentence suggests, constructivists take issue with the widespread belief in political academia that we can make truth-claims about the social world which are truly independent of our own interpretation of it, i.e. there is a 'constructed nature of our claims to knowledge' (Bevir and Rhodes, 2003: 3; also Fierke, 2005: 7; Guzzini,

2005). Facts, for constructivists, are therefore not discovered in the social world so much as created (Collingwood, 1965: 99). The obsession with creating a 'science of the political' in order to justify political studies has been most starkly criticised by Connolly as 'a wish to escape the political. It emerges...as a desire to rationalise public life, placing a set of ambiguities and contestable orientations under the control of a settled system of understandings and priorities' (Connolly, 1993: 213). There is a small irony therefore that the public attempt to 'instrumentally' rationalise (Dryzek, 1994: 5) the understanding of the political is in this sense as doomed to fail as political attempts to rationalise the public (Scott, 1998).

The weakness of the assumption of rationality has been critiqued in detail on several occasions, but is worth reviewing in brief. The central lines of rationalist reasoning follow three steps: actors have perfect knowledge of the potential outcomes of their potential actions; actors are always self-interested and never motivated by normative or other non-material factors, and, consequently, only one course of action can be defined as rational - that which will increase the actor's material interests, and; political outcomes can therefore be explained by the material interests of relevant actors (Elster 1982: 453-82).

Whilst the focus on the 'microfoundations' of political processes has certain appeal, and can help to explain the often disjointed behaviour of democratic institutions (Green and Shapiro, 1994: 4), several flaws are notable within the assumption of perfect foresight and pure material interest. The empirical record of this approach has been severely critiqued, with Hayward citing the 'portentous claims, methodological obsession and paltry performance' (Hayward, 1986: 8) that came with political science acting like the pure sciences. Likewise, Krehbiel's review of rationalist studies on legislative politics found that 'empirical successes are as yet difficult to identify' (1988: 259). In short, therefore, the prediction of human rationality and a micro-focus for political analysis does not seem to have adequately explained a great deal of the political events it has been most interested in. As we shall see in the following chapter, rationalist assumptions of crisis solution have been equally flawed.

On the theoretical level, too, however, the assumption of rationality and material interest can be questioned. Certainly, there are instances where the material interests of parties can be adequately measured; the interest of a business in lobbying for a lower corporate tax rate, for instance. However, even within the field of economics from which rational choice has taken considerable inspiration, the idea of rational actors having perfect foresight has been rigorously contested. Minsky (1977: 24) described the "tendency to transform doing well into a speculative investment boom (as) the basic instability in the capitalist system" as a development of Keynes' notion of herd mentality. Whilst constructivism is a relatively recent political school of thought, the critique of such approaches to rationality has deeper intellectual bases. Weber's very typology of political action depends on the significance of non-rational components of behaviour such as tradition, for instance. Constructivism could also be said to have roots in Durkheimian and to an extent Gramscian philosophy along similar lines (Allcock, 2000: 219; Hopf, 1998: 177; Ruggie, 1998: 857).

In such an environment, actors rely upon internalised assumptions rather than facts to rationalise their action (Denzau and North, 1994; Bell and Hindmoor, 2014). It is in particular suggested by Abdelal that in times of crisis, actors' shorthand assumptions attain an increased degree of importance (Abdelal, 2007: 197). Financial crises, by their very nature unexpected ruptures to the pre-existing stability, create a considerable amount of uncertainty over policy goals and methods. As such, in these contexts, the suggestion of a single rational response to crisis is particularly questionable. As such, a legitimate avenue of political inquiry is not on asking which collection of interests explains an outcome, but to take a step back and consider why certain policies, programmes or goals were deemed rational in the first place.

Likewise in relation to the public acceptability of policy proposals, content alone cannot be taken to best explain the success or not of a political message (Hall, 1993: 280). Rather it is the 'communicative environment' that helps to legitimise certain approaches over others (Payne, 2001: 39). Where material interests cannot be objectively calculated, constructivism seeks to understand the social construction of knowledge that led to certain political avenues being pursued. This need not suggest that empirics are insignificant, nor that any explanation could be considered of equal rationality. Rather, where rational choice theory sees a dualism of rational and irrational, constructivism sees a continuum with different approaches taking in different evidence to put forward its case, with some strategically more likely to succeed in democratic appeal than others. Rationalism must therefore be understood as a 'contingent product of its time' (Abdelal, 2007: 218). Rather than rationality being a pre-ordained and un-measured assumption, therefore, it becomes an area of political contestability and academic analysis.

Constructivism therefore utilises this different approach to rationality to suggest a switch of focus of study from being one seeking to explain why different policy outcomes occurred, and doing so by explaining the rational interests of the agents involved, to asking why the rational agents acted as they did, and how the construed their circumstances to lead to the power relationship developing as it did. It is in this context that the term 'depth ontology' in relation to constructivism is most clear; for rather than assuming rational interests and merely explaining political outcomes through the power relations of the relevant actors, constructivist study seeks to understanding the power relations *and* the construction of interests which explain them.

If the rationalist approach to political analysis is deemed too reductionist, empiricist and behavioural, what exactly does constructivism offer as an alternative to its ontological approach to political behaviour and institutions? Rationalism can be broadly defined as ontologically foundational, i.e. believing in a world that exists objectively and independently from our knowledge of it (Sanders, 2010: 32), whilst poststructuralism occupies the alternative end of the spectrum by being stridently anti-foundational in the belief that all knowledge and understanding of the world is inherently subjective. Constructivists believe in a 'minimal foundationalism' (Hoffman, 1991:170); that our access to the social world is essentially mediated through socially constructed norms, i.e. they seek to 'denaturalise' the social world. As Hacking (1999) highlights, a proliferation of constructivist work has highlighted (and often mis-highlighted) how a wide variety of concepts that

are taken for granted in the modern world need not actually be seen as permanent or inevitable. Moreover, this is often matched by an implicit normative suggestion by constructivists that these perceived structures should actually be replaced (Ibid.: 6). Thus constructivism has often attempted 'to empirically discover and reveal how the institutions and practices and identities that people take as natural, given, or matter of fact, are in fact the product of human agency, or social construction' (Hopf, 1998: 182; also Béland and Cox, 2011: 11, Bevir and Rhodes, 2003: 3, Pouliot, 2007: 367).

In ontological terms this approach therefore suggests a certain scepticism towards a fully knowable social world, but not one that reduces all social knowledge to relativism. The social world is certainly not fully knowable in the positivist sense; rather it is the social constructions which take on this persona of being infinite and eternal truths which are of importance to scholars of the political. In Cox's words, 'social constructivists argue that beliefs and preferences of individuals cannot be deduced from preconceived assumptions about human nature but rather they are constructed in a social environment where the beliefs and preferences held by others members of the community constitute the basis for what is deemed to be socially valued or preferred' (Cox, 2001: 473). Whilst much knowledge is social, this does not make agents free to completely imagine and hence develop new paradigms; ideational structures 'define the meaning and identity of the individual actor and the patterns of appropriate... activity engaged in by those individuals' (Boli et al, 1989: 12). As a result, then, constructivist has placed the study of ideas as central to political explanation, with a reduced role for the previously pre-dominant study of interests.

The response that constructivist literature received from the dominant positivist mainstream was mixed. Whilst in the field of IR in particular the lack of explanatory power in existing theories to explain the seismic end of the Cold War led to a certain amount of self-reflection amongst positivists, there was still some scepticism towards constructivism generally. Keohane admitted that 'reflectivist' accounts had highlighted some of the theoretical weaknesses within neo-realism and rationalist theory (1989: 174), and even turned towards the use of ideas himself (Goldstein and Keohane, 1993). However, he criticised the lack of a clear operationalisable constructivist research agenda (Keohane, 1989: 173). More pertinent, however, was the clear lack of actual transformation positivism's basic epistemological views in relation to the constructivist challenge.

Alongside King and Verba, Keohane still proclaimed that a worthy scholarly study should 'make a specific scientific contribution to an identifiably scholarly literature by increasing our collective ability to construct verified scientific explanations of some aspect of the world' (King, Keohane and Verba, 1994: 15), i.e. that a science of the political is still possible. Moreover even this turn to ideas was essentially still framed in terms of rational choice (Blyth, 1997: 229), highlighting a remarkable lack of 'learning' on the part of the positivist mainstream from the critiques that constructivism had provided. Keohane and others were robustly criticised for this clear lack of understanding of the fundamental challenges that had been put before them (Ashley and Walker, 1990: 266; Der Derian, 1990: 295-6; George and Campbell, 1990: 288; Walker 1993: 81-103). Rationalism and positivism therefore tend still to rely more upon empirical generalisation

and the potential for quasi-scientific explanation as a justification of their work than any notably strongly held ontological beliefs (Shepsle, 2006: 32-33).

Constructivism can certainly claim to have played a notable role in the development of ideas as a legitimate focus of study for political scholars however, alongside the new institutionalists. It has not completely transformed the scholarly landscape, but has rather seen a more 'quiet cataclysm' (Mueller, 1995), in which perceptions of the explanatory power of political study have subtly changed, with an increased leaning towards more qualitative and social aspects of change than had previously been the case. Importantly, it has caused all scholars to be more explicitly reflexive about the assumptions than underlie their work. The turn towards ideas and social construction is flexible enough to have permeated a wide variety of different research foci also, providing an ever-richer research output.

In essence therefore this ontological challenge is quite straightforward, with the two schools representing starkly conflicting worldviews; constructivists critique the positivist belief that the empirical can be separated from the philosophical (George and Campbell, 1990: 288). Epistemologically, with human behaviour no longer being explained by its 'natural' constituency of utility maximisation based on pre-ordained economic interests, constructivism has focused more on the social elements of knowledge, notably on ideas (Abdelal, Blylth and Parsons, 2010: 2; Bell and Hindmoor, 2014). This focus analyses the causal role they can play independent of the interests of those that created them, going against what positivism, historical materialist Marxism or post-structuralism would ascertain (Béland, 2009; Béland and Cox, 2011; Carstensen, 2011; Hay, 2011). Indeed as well as being independent of interests, ideas are also deemed important in their very creation. As such we cannot understand what interests are without fully understanding the ideational environment in which they were forged.

It must be noted, however, that constructivism is by no means a unified school of thought and a great deal of theoretical variation exists within it. Most significantly, there is a split between 'thinner' and 'thicker' constructivisms which place different weights on the accessibility of a 'real' world and the independence of material interests from the ideas which narrate their meaning. For 'thinner' constructivists we can still provide an objective analysis of the world because even though agents' understandings are inherently social, our analysis of them needn't be (see Adler, 1997).² Under such an approach, the scholar can claim to objectively analyse others' subjectivity (Pouliot, 2007). 'Thicker' constructivists would doubt that we could provide such an objective understanding however, for even our own understandings of events may colour our empirical focus and hence the scholar also to an extent replicates the social reality which constitutes them. For thinner constructivists, material power has a crucial causal role independent of ideas and discourse, whilst for thicker constructivists the distinction between 'the material and the ideational (is) meaningless, because there is no 'real' material, word (sic) independent of discourse, or the narratives or traditions that shape outcomes, or, to put it another way, there is no extra-discursive realm' (Marsh,

_

Thin constructivism therefore follows a causal epistemology whilst thicker constructivism provides a more relativist one.

2009: 680). Realist variants also exist, with Marsh suggesting that 'real' material structures independent of agents' understandings of them exist and exert a causal influence on political processes. The commonalities across these approaches involve the modesty which they claim over absolute scientific social knowledge; as Moravcsik notes, history rarely fully vindicates any theory (Moravcsik, 1998: 12). One of constructivism's core aims is not to proscribe and deduct too many universal 'scientific' variables onto given social situations.

To varying degrees both thinner and thicker variants of constructivism puts a notable explanatory weight behind the idea of norm- and ideational-based political behaviour (Parsons, 2010: 80-81), to offer added value over a rationalist approach which can only explain through an appeal to a single rational interest. However, it should be noted that the leading exponents of crisis research from constructivist positions have tended to fall on the side of thicker constructivisms to date (Marsh, 2009: 681-687). Whilst constructivist institutionalism is not dependent on a 'thick' approach, it does lend itself most easily to largely ideationalist approaches to political outcomes, and as it has been currently constituted does theorise crisis in particular as an area of study where the ideational gains primacy over the material because of the uncertainty it creates. This thesis will, however, suggest that benefits can be gained from adopting a more realist approach to ideas in section 2.7, methodologically and analytically at least. Generally, however, the thesis seeks to work within a generally constructivist framework in order to make suggestions for development within rather than critiquing its theoretical premises from without.

We have seen, then, that constructivist scholarship occupies a middle ground between rationalism and post-modernist approaches. The implications for the study of the political economy of the credit crunch and financial crisis are quite straightforward. Given the constructivist implication that preferences need not always be (indeed, perhaps rarely ever are) fixed, and that interests are not simply dependent on objective calculation but to an extent are dependent on the social climate in which they exist (Price and Reus-Smit, 1998: 261). It suggests that the ideas of those with power matter, and these cannot be equated with interests alone (Hay, 2011). It is the contention of this study that the discursive environment, the social understandings attached to 'real world' events play a role in shaping a reaction to them. Thus this guides us naturally towards a research design focused on seeking to explain how the ideational context of the financial crisis may have led to certain policy approaches being deemed favourable to others. However, whilst ideas have been argued above to matter, the issue of how, where and when they matter most must also be discussed. In order to do this, and expand in more detail on the theoretical promise of constructivism, a discussion of its approach to the issue of structure and agency is necessary.

2.3 Structure versus Agency for Constructivists

By occupying the ontological middle-ground, the constructivist framework facilitates the production of pragmatic accounts that combine a focus on the interplay between material and ideational forces in shaping outcomes. In addition, the constructivist approach also allows for the interplay between structure and agency to be effectively interrogated. In the paragraphs below I provide an overview of such work and its implications for the research of this thesis. Despite the centrality of the question in political study over whether agents or structures are key to understanding outcomes, the problem of observability of the social world means that no scientific answer can ever be given to it. Thus a variety of approaches based in different philosophical beliefs about the nature of the social world have produced different analyses of the issue. A dichotomous split between two extremes can here be identified: intentionalist analysis focuses entirely on the motivations, actions, desires and beliefs of the agents in question to provide the explanation of the consequences and actions in question; and structuralist analysis which highlights the all-important role of the context in framing the events in question (see Hay, 2002: 96-100). However, few but the most extreme scholars would suggest that political action can be simply defined in terms of one of these two approaches, and therefore it is a common consensus within political academia that we must look for explanations that involve at least to some extent both elements of structure and agency.

Essentially the argument that constructivism makes is that both extremes of what I call the 'ontological spectrum' suffer from an inherent structuralism which gives no scope for humans to actually possess strategies independent of either their personal, or society's, interests (Ibid.: 103-104). In the case of positivism these are boiled down to the material interests of those involved, whilst with postmodernism Foucault's term 'power-knowledge' explains the prism in which all are subjected to thinking (Foucault, 1977). 'Research traditions such as rational choice, postmodernism and...large parts of constructivism, which occupy endpoints in the agent-structure debate, have life easy' argues Checkel (1999: 558), in that they can ignore the 'messy middle ground' where norms, discourse, language and material capabilities interact with motivations, social learning and preferences' (Ibid.). This has allowed modern middle-ground theorists to develop on the classic agent-structure debate, arguing that neither agents nor structures are fully independent entities. Rather, the two are mutually and dialectically constitutive of each other; this frees us from assuming that agents aren't capable of creating ideas or strategies outside of their institutional context whilst still keeping in place an understanding of the restrictions that historically prior actions have had on their scope to institute change. We might, however, not think of these structural restrictions solely in legally formal manners, but also of the creation of norms which achieve

general acceptance and therefore shape future strategies. Certain critiques of constructivism have argued that it swings the pendulum too far towards the power of agency (Bell, 2011: 884), and in the case of Blyth's assertion that 'in moments of uncertainty (crises) generated by the failure of existing institutions, institutional choice becomes undetermined by structure' (Blyth, 2007: 761), this is perhaps the case. The relationship between structure and agency is not one we can solve empirically based on the newer understandings of the complex inter-relationships between the two (Carlsnaes, 1992: 250), and the debates are therefore somewhat ontological. There are three central theories, two of which are arguably compatible with constructivist ontology.

Metaphors abound in these debates. The morphogenetic approach of Margaret Archer uses that of DNA to explain the relationship between structure and agency; two branches which are interlinked (over time as opposed to space), and therefore suggesting an ontological separation of structure and agency (Archer, 1989, 1995, 1998). Such an account is essentially realist and does not therefore provide scope for an understanding of the constitution of agents per se. Rather structures are to be seen as outside of agents, obstacles to be overcome. Giddens' structuration theory, popular amongst constructivists (Glarbo, 1999 takes it for granted as the constructivist methodology of choice, and Wendt (1999: 165) relies on it to), provides the metaphor of a coin; structure and agency are bound together as with the DNA example, but as they are two sides of the same coin we cannot see both at the same time (Giddens 1979, 1984). Structure and agency are ontologically conjoined in this theory, but still have an element of independence. Moreover by taking an empirical approach of 'methodological bracketing', although we cannot analyse the respective input of both combined in a given moment, it is possible to analyse one or the other in any given event. As stated, this is a common and popular approach of constructivists to date, something I consider slightly problematic given the criticism of this approach by Hay (2002) and Bieler and Morton (2001: 7-9). The strategic-relational approach of Jessop (1996) and Hay (2002) cures some of these problems. Firstly it notes that the 'distinction between structure and agency is a purely analytical one' (Hay, 2002: 127). Neither is truly 'real' in this sense since neither can exist without the other. It is here that constructivism differentiates itself most clearly from critical realism, given the denial of structures as causal entities whose existence we can only understand through the observation of their effects.³ Given this it chooses to replace the terms 'structure' and 'agency' altogether and instead to talk of 'strategic agents' and 'strategically selective contexts', i.e. agents have certain aims, based in part upon their social constitution, but their ability to meet these aims is determinant on the context, which will favour some courses of action over others.

The metaphor of choice here is that of a duo-metallic alloy. Whilst in a specific situation we know there to be a combination of both elements, what we observe is the final combination of the

_

³ It is here that McAnulla (2005) posited the charge of 'actualism' against constructivists, for denying the realm of the 'real'. Whether one is preferable to the other is essentially an ontological question in which the answer lies in the eye of the beholder. As Hay's (2005) response highlights, the accusation that actualism is bad on the basis that it isn't sufficiently realist doesn't ultimately prove very persuasive.

two, making it impossible to ascertain exactly what 'proportion' of either has led to what we find before us. The idea that social construction plays a large role in creating agential identity poses the problematic epistemological question of 'to what extent is identity "pre-" and "post-social" (see Zehfuss, 2001)?' However again this suggests an ontological separability of agents and structures which arguably the strategic-relational approach dissolves. Ultimately we can only see the outcome of the fusing of the two, not their relative parts.

The challenges of resorting to structural explanations of crisis resolution can be seen through a brief analysis of two such works. Both Ashbee (2011) and Armingeon (2012) have pointed to structural differences in the British and American political systems to explain their different fiscal paths in the post-crisis context. Briefly, Ashbee highlights the pork-barrel approach of the American system and the greater immunity of the British Prime Minister to the business lobby to explain the differentiation. Armingeon outlines institutional veto points as the vital factor in influencing whether counter-cyclical spending will occur. Two central problems are visible here. Firstly the two analyses reach very different conclusions, with Ashbee highlighting the structural capacity for Britain to implement austerity without hindrance, whilst Armingeon places the UK as a counter-cyclical spender for exactly the same reason. But more significantly the appeal to 'static' institutional factors cannot explain why a dynamic shift took place within the British, and to an extent the American, response to the crisis. Britain in particular reverted very quickly from a more stimulative approach to one characterised by austerity. Such variation suggests a role for ideas and discourse in shifting acceptability for different policies. Indeed, as the above theoretical discussion implies anyway, such structures 'do not come with an instruction sheet' (Blyth, 2003).

2.4 Strategically Selective Contexts, Strategic Actors, and the Financial Crisis

Given the ontological question of whether structure and agency really exist independent from one another, Hay argues that, from a meta-theoretical standpoint which he doesn't explicitly call constructivist, but which we can term as such, what we actually see in political outcomes is not so much an inter-twining of the two, but a more complex alloy of their dialectical inter-relationship, in which a 'true' assessment of the relative component of each is not possible. Rather than discussing structure and agency at all, then, he argues it is preferable to talk in terms of 'strategically selective contexts', and 'strategic actors' (Hay, 2002: 128). The implication of this for the causal chains we draw as political analysts is therefore significant, albeit highly complex.

The key impact of this dissolving of the ontological dualism of structure and agency in the strategic-relational approach is to make the two inter-dependent, and hence, in its most simple terms, to change the terminology of the debate surrounding the relative significance of context and conduct in the political arena (Ibid.: 129). Thus we cease to seek the relative significance of

structures and agents on political outcomes, but look at the political context and ask which strategies are conducive in such an environment. But, importantly, the context does not make a certain strategy inevitable; it may be that an unelectable political party take up a strategy that in other circumstances might be favoured but fail regardless, i.e. the context does not pre-determine the outcome as structuralist approaches do (Ibid.: 128). The central appeal of such an approach is that it prevents a political analysis which accidentally slips into structural or agential explanations, but highlights the political contingency of individual situations and importantly provides a crucial role for ideas in that structural constraints 'do not come with an instruction sheet' (Blyth, 2002: 8). The fact that political outcomes in each situation are a specific outcome of the agents who existed within it and the constraints they found upon themselves may be considered overly simplistic, perhaps even blindingly obvious. But as Hay (2002: 129) suggests, such 'sociological truisms' are a sign of strength, not weakness. 'Good political analysis is often a case of stating and re-stating that which is obvious but all too rarely reflected upon' (Ibid.), he continues.

2.5 Crisis and Strategic-Relational Action

Below I begin to operationalise this theoretical starting point to justify the focus of this thesis on analysing media discourse as an essential component of the strategic context of the global financial crisis.

The shift towards thinking of politics as involving strategic contexts and situated agents is very instructive in relation to our expectation of political change in the midst of a crisis, and yet perhaps has been somewhat overlooked by the constructivist desire to explain endogenous political change in such an environment. For rather than the failure of ideas necessitating political change, as constructivist theory seems to suggest, what we see when we apply this strategic-relational logic is that a financial crisis actually presents a context in which certain strategies are selectively favoured over others, depending on the political circumstances of the polity in which it occurs in. Thus, crudely, we could suggest that the financial crisis either represents a context in which policy stability and a lack of change are favoured, or a context in which change is favoured, likely leading to a rupture in the policy norm – an episode of punctuated equilibrium. The central point remains that when we apply strategic-relational thinking to the issue, we cannot suggest that the context determines a specific outcome. Thus we must move beyond the suggestion that crisis necessitates change. It is important to note, however, that such a suggestion is not at odds with constructivist thinking to date, but arguably a more full application of its logic to crisis episodes.

Again, for the purposes of clarity, this thesis explicitly wishes to place itself within this theoretical heritage that constructivism's founding authors have created. But a central contention of the theoretical argument of this thesis is that these major meta-theoretical tenets of constructivist analysis of political reality *could be more clearly applied to the issue of crisis and change*, and indeed that to date much of this work has failed to apply these insights fully enough into its empirical analysis. Thus, the insights of selective contexts and situated and strategic actors help to explain why a crisis need not necessarily lead to widespread change. For whilst, in the issue of a financial crisis, the context can be confidently stated to be selective towards change on some level the above discussion of my definition of 'crisis' suggests that this should be the case - the nature of the change that takes place depends on the inter-relationship of the strategic actors who come to fill this context.

But such a suggestion risks a reductionism in which the only element of contextual selectivity is in the need for some level of change, and thus, as with other constructivist work on the issue of crisis, might perhaps tip the balance of explanation too far towards agential interpretation and too far away from strategic selective contingency. There are other contextual factors which must be considered when interpreting the discourse of political actors, however. Most notable amongst these are electoral pressures. The debate over the 'correct' economic course to take in a financial crisis, and of the real meaning of that crisis in the first place, does not happen in a vacuum. Electoral pressures greatly influence how different agents choose to publicly discuss events, either passing the blame to external factors or utilising the opportunity to blame the existing government for the crisis and thus increase the electoral appeal of the opposition, depending on one's relationship to political power. Thus the political argument over ideas ceases to simply be a case of getting a majority of people to think of the crisis is one's own terms, and generating sufficient external credibility for these ideas, but an important dialectical relationship in which political agents seek to neutralise the arguments of their opponents. Put in such a context, radical political change in a democratic society suddenly appears much less likely, as no party wishes to be seen to be deviating too far from the 'average' voter.

The role of the media in shaping perceptions about the relative political discourses of the major parties, and indeed of the issues which lie behind them, is therefore a subject worthy of study. With much of the work of discursively-focused scholars focusing on political and institutional elites, discourse, whilst communicative, is largely top-down directionally. The analytical focus is therefore largely on how change takes place through a discursive re-framing of interests within epistemic communities, policy communities, policy networks, etc. A role for the media does exist to an extent in such analysis, for they may play a role in making certain discourses acceptable or not, but often there is assumed to be relatively little outside light shone on the ideational understandings of elites. I do not wish to question the value of such work here; it is often a fruitful and logical progression of analysis. However, for two reasons I wish to take a slightly different starting point from more conventional discursive approaches to politics. Firstly, the very public nature of the discussion about the financial crisis, a necessary outcome of the confusion and shock it caused, places the

significance of the media on a higher level than is often the case in terms of political ideas. This is especially the case if we take seriously Blyth's starting position of a crisis being an episode of 'Knightian uncertainty'. Many political decisions frankly take place without a great deal of media analysis of the ideas behind them, hence the acceptable decision in such circumstances to focus on political elites and policy networks. However, for the wider macro picture which I cover in this thesis, the media play a much greater role in political discourse, and indeed play a more generative role in the context of public acceptability than is normally the case. In this instance, then, the 'puzzling' about policy necessities was an unusually public affair.

Secondly, the decision to focus on the media more than politicians' statements comes from adopting a more realist-constructivist position as outlined by Kreb and Jackson (2007) in relation to discourse and ideas. Justified in more detail below, their work argues that scholars should be less focused on the *true* ideas of political agents, but rather more on which ideas are publicly acceptable, and which are not. If certain policy frames are accepted by the public, then they will force politicians to utilise them whether they are believed in or not. When ideational contestation takes place, whichever frame comes to gain public acceptability will be adopted as the new policy norm. Thus, to operationalise such a perspective, we become less interested in what politicians actually *say*, but more on which elements of public arguments gain traction and force strategic actors to either adopt them or challenge them. Hence, the media becomes a suitable unit of analysis; if a certain frame of understanding becomes sufficiently dominant publicly, then the space for political parties to oppose it becomes limited, and thus it increases the chances of it being adopted as a policy frame. It is therefore through the lens of the contestation of ideas in the media that we can achieve an understanding of why certain policy positions became acceptable and others didn't.

Thus the question of whether the media is a generative political actor in its own right or not ceases to be important, something again that I feel is a suitable and logical development of realist constructivist, and strategic-relational, theory. Whether the media's analysis is a result of structures or agents cannot be known, it is an alloy of various elements. Thus again whether the media's view causes political views, or simply reflects them, is not something that can be easily empirically identified. However by analysing the extent to which certain ideas gain traction in the media realm, we can begin to understand how they limit the discursive space for politicians to follow an alternate course.

It is not to place the media necessarily above political elites in terms of their significance in sculpting public opinion and political acceptability, but given that the broadsheet press provide a vital medium through which political discourse reaches the public, the way in which this lens reflects the light of political discourse provides academic appeal. Again, this is not to state that this thesis will produce a 'better' understanding of political discourse than one focused purely on the political elite, or indeed of a study that combined both. Rather, due to the necessary limits on the scope of this project, the media have been chosen as a case study from which it is hoped that theoretical developments will spring; other ontological and epistemological positions on the nature

of the political world and how best to access it may well produce different, but equally valid, conclusions.

2.6 The 'Realist Constructivist' Agenda and crisis analysis

'Realist constructivism' has to date largely been a sub-group of academic discussion within the field of international relations, attempting to bridge some of the barriers between classical realism and constructivist scholarship (see Barkin, 2003). Whilst much of the discussion about the compatibility or incompatibility of realist and constructivist conceptions of state interests and the balance of power are not relevant to the domestic level, the attempt by Krebs and Jackson (2007) in particular to highlight the power of rhetoric in creating and resolving disputes has some potential applications to constructivist institutionalist theory in relation to domestic discursive contexts. One of their central concerns with existing constructivist scholarship in IR, but arguably one that applies also to constructivist political economy, is that it creates 'unanswerable questions about actors' true motives' (2007: 36). The works of Payne (2001) Finnemore (1996) and Finnemore and Sikkink (1998) all highlight 'persuasion' as being a key tool in transforming ideas into norms. However, from a more realist constructivist perspective, the constraints of working in a social science mean that we are incapable of measuring or assessing where 'persuasion' of the validity of arguments has truly taken place (Krebs and Jackson, 2007: 40), and thus need to focus not so much on what individual policy actors really believe, for we cannot be certain of this from their public statements, but on the social processes that create outcomes (Ibid.: 40-41).

Much critique on constructivist work stems from this questioning of whether ideas are an independent causal variable, or merely a tool used by competing interests to promote their agenda (See Buller and Lindstrom, 2013: 405-406). Ideas are a 'Janus-faced' academic concept with which to grapple: 'sometimes actors' beliefs guide their actions, sometimes apparent beliefs only rationalize strategies chosen for other reasons' (Parsons, 2002: 49). Utilising a realist perspective helps to dissolve this issue: rather than stating that ideas either were or weren't the cause of action, paradigm change, continuity etc., from a realist strategic-relational viewpoint we can suggest that the existence of certain discourses created a context which limited the boundaries of acceptable policy development or change.

Rhetoric, then, works not by 'persuading one's opponents of the rectitude of one's stance, but by denying them the rhetorical materials out of which to craft a socially sustainable rebuttal' (Krebs and Jackson, 2007: 42). Whilst their application of this logic implies known preferences and the use of rhetoric as a tool of power and not simply one of social puzzling, the more 'realist' understanding of the power of political discourse can be applied to the domestic sphere to advance

our conceptual grasp of the dynamics of a crisis context and the causes of continuity and change. For the key element here is not that social actors have 'internalised a particular set of normative principles', but on whether certain arguments are 'socially sustainable' (Ibid.: 57; 42 respectively). The link here with the notion of a strategically selective context is clear; we are not attempting to gauge how many people were persuaded by the logic of austerity over stimulus, but of the extent to which the arguments for either were 'socially sustainable' within the strategically selective context(s) of the developing financial crisis. Whilst it is impossible to make a general law within constructivist scholarship to delineate the limits of what is socially sustainable, the key insight here is that methodologically we should consider how the arguments made for and against certain policy positions generate credibility within the public realm, rather than seek to discover whether the ideas of the orthodoxy school had been sufficiently internalised by a wide enough variety of actors to offer a causal explanation of the continuity in policy that emanated from the financial crisis.

Again, I feel such an approach justifies a focus on the media as a locus of economic and political debate. More centrally, however, it highlights the futility of the question of whether the media shape political actors' preferences or whether politicians shape the media debate. For realist constructivists, this is not causally important. As long as a frame of understanding pre-dominates the public debate about an issue, then it becomes socially unsustainable to act against it. Whether the media generated that norm of understanding or whether politicians did is impossible to decipher in the same way the relative weight of agential and structural causes of outcomes are; it becomes necessary to dissolve the question entirely to offer an account that has analytical tractability alongside epistemological consistency. Indeed the central argument of the thesis will postulate that the common frames of understanding of the media did not shift over the crisis period, contra to expected constructivist beliefs, and this provided the foundation upon which the orthodox austerity approach became socially sustainable in a way that the Keynesian approach didn't in the immediate post-crisis period, as it relied upon the pre-crisis frames of reference to a much greater extent than the Keynesian school was able to.

A critique of such an approach to discourse is that, as Marsh (2009) suggests, it overstates the causal role of ideas compared to interests and structures. This may well be true, and I do not actually wish to challenge such a claim. For it is the major suggestion of this thesis that even if we are working within a constructivist approach to crisis which posits Knightian uncertainty, then we can still question the validity of its explanatory power. In other words, whilst Marsh critiques the theory from 'outside', by questioning its beliefs regarding the material and the ideational, I wish to critique it from 'within' by suggesting that even within its own terms, there are developments to be made. Insufficient space in unfortunately available here to discuss in wider detail whether or not the critique that constructivism over-states the role of ideas is a strong one of not, ultimately it lies outside of the focus of this study.

2.7 The Relevance of Constructivism to Crisis

In relation to crisis politics therefore, what makes constructivism the most suitable school of political thought for undertaking a study? Firstly the uncertainty which crises create makes problematic the positivist understanding of perfect information in regard to outcomes and the role of material interest in shaping political opinions. In this sense the political impact of an economic crisis is not 'economically given, but politically orchestrated' (Hay, 2010: 447). Whilst rationalist account of crisis-solution would point to the materially-given interests involved and the way in which they went about changing policy, a constructivist account would find some of this problematic. For, as Blyth and Hay have succinctly argued, crises are not objective and external events; rather they are lived experiences, the popular narratives of which affect the ways they are understood and the ability of ideas to simplify and 'cure' them (Blyth, 2002: 9; Hay, 1999: 338; Hay, 2010: 447). Those in the public sphere have the power to influence both public sentiment and economic confidence (Hassdorf, 2005; also Gill, 2003: 16) with a popular construction of events. A coalition analysis approach which highlighted how different resources were mobilised to support a policy therefore focuses 'on how a coalition is able to reach its goals', but in so doing 'overlooks the question of how those goals are defined and articulated' (Cox, 2001: 469). Whilst much of the literature suggests that the onset of a critical juncture gives scope for ideational change, the experience of our current crisis suggests that this is not by any means pre-destined. Ideas do not magically change by themselves, but require agential intervention to modify them (Béland, 2005; Berman, 1998). This mixture of ontological and epistemological theorising makes a constructivist approach offer notable promise for the study of crises. This is not to say that constructivism is a perfect theory of political action; below I discuss some of the existing issues that need improving within the field.

What questions does constructivism ask of crisis politics? As stated above it problematises materialist accounts of change, and it therefore asks what ideational formations can be used to explain change alongside material factors (Blyth, 2002: 10). More work is needed on exactly the processes by which ideas shape policy and influence agents however, and this is something for the methodology section to address. The turn to discourse advocated below however is, I feel, a critically under-utilised one in terms of understanding the power of ideas. This means a much greater constructivist emphasis on context, history, policy networks and the media than traditional accounts would include.

2.8 Conclusions

This chapter has placed the development of constructivism in its historical context, and sought to demonstrate the alternative it has offered to mainstream rationalism on account of its different meta-theoretical starting positions. Through this concluding section, I briefly review the core conceptual points of entry that I take from the approach in to the study of financial crisis in the UK.

Firstly, the chapter has highlighted and justified a social approach to political analysis which questions rationalist assumptions about objective reality, human nature, and the foundations of knowledge. As such, the constructivist focus on ideas and discourse as important variables in political explanation has been detailed. Following on from this meta-theoretical starting point, I continued by outlining how this feeds into an approach to the structure/agency issue which attempts to dissolve the 'Gordian knot' (Bieler and Morton, 2007) of existing discussion on the topic. The strategic-relational approach, which stems from constructivist thought on social reality and identity, shifts the discussion towards one of focusing on the strategic context within which action takes place, and the situated actors who undertake it. On top of this, by borrowing insights from the realist constructivist school of IR, we can focus on crisis and crisis-resolution as a more sociological process in which we can observe ideas and discourse and how these provided a social source of legitimacy for certain policy approaches. Whilst much existing constructivist work has sought to understand why actors believed the ideas that they did, the central insight from realist constructivism is that research should focus to a greater extent on how language frames, creates and limits the options open to political actors.

From these points of entry, I develop two main extensions to the existing constructivist approach through this thesis whilst continuing to work broadly within this intellectual and theoretical tradition. These are, firstly, that crises display a far greater level of ideational continuity than the school of thought has to date recognised. Following from this observation, I will argue that epistemic uncertainty may not be a sufficient cause of ideational change and thus that a new type of conversation needs to develop within constructivism about the situations in which continuity and change take place, for the purpose of nuancing and advancing the theory in a way that enables it to explain both more satisfactorily.

Chapter 3. Constructivism, the British

Broadsheet Press, and the Global

Financial Crisis: The Extensions

3.1 Introduction

Whilst the previous chapter set out the theoretical underpinnings of constructivism and its approaches to crisis and central questions relating to structure and agency, this chapter proceeds by analysing more specifically the institutionalist analyses of crisis to date, and their major developments and limitations. The central implication of this, crucial to the overall argument of the thesis, is that we need not see crisis situations as determining widespread paradigm change, but rather that the central political formations of the immediate pre-crisis era are more pervasive to the ways in which crises are explained, thus creating a greater level of ideational continuity than has hitherto been suggested.

I begin by critically engaging with the other 'new institutionalist' theories of the state, which alongside constructivist institutionalism have been commonly deployed to explain crisis resolution by modern scholars. Whilst the previous chapter introduced constructivism by comparing it with other ontologies, here we compare and contrast it to other 'middle way' institutionalisms (Hall and Taylor, 19996). The chapter will then proceed to focus more explicitly on constructivist institutionalist scholarship, particularly the works of Blyth and Hay who have most explicitly laid the foundations of this scholarly approach to ideational change. Their central pre- and post-crisis works on the subject will be discussed, a whilst their approach to crisis narratives is deemed highly valid and relevant, two critiques are made which this thesis will seek to develop; either overdetermine the likelihood of change, or fall back into structural arguments to explain continuity. This thesis will seek to further constructivist institutionalism by expanding its focus to moments of ideational stability and hence seek to re-frame constructivist institutionalist discussions of crises away from simply seeking to explain where crises create change but, importantly, also by focusing on where they do not. The outcome, it is intended, will be a more rounded discussion of ideas and crisis. By focusing on the 'social source' of governing legitimacy through an analysis of the public narration of crisis, the chapter will go on to justify the dynamic tracing methodology adopted as one capable of explaining continuity within a constructivist approach to political analysis.

3.2 Setting the Scene: Constructivism and the New Institutional Turn

It was established in the opening chapter that constructivist literature occupies an ontological middle ground between rationalism and post-structuralism. However, rather than emerging into an intellectual vacuum, the constructivist approach emerged in to a field populated by other prominent attempts to explore the complexities of the social world from similarly pragmatic positions. In the study of political economic governance, various forms of institutional analysis in particular have provided frameworks to guide investigation in relation to crisis resolution. Thus as well as justifying itself against more rationalist approaches to politics ontologically, this thesis must also defend constructivism against the various alternative 'new institutionalist' approaches to ideas and change. In the paragraphs below, I first introduce each of the main schools of institutionalism and explore relevant contributions made to the study of on the politics of crisis. Whilst this thesis intends to work within the constructivist institutionalist school, there are insights from the historical and sociological institutionalisms which remain of value, to the extent that some crossover is still notable.

Historical institutionalism, as the name suggests, highlights the importance of history and the structural constraints that come with the continuation of a practice over a period of time. 'Path dependence' is therefore the most common phrase used to explain this expected outcomes with historical institutionalism; the institutional context, once a practice has been accepted, will structure future actions towards the same practice even if external conditions change. The institutional decisions of political actors are not to be understood simply by an explanation of the rational self-interest, but also by the way in which institutions mould those within them towards seeing issues in certain ways. The example of the QWERTY keyboard is often used to demonstrate such an approach in its simplest form (Mahoney, 2000). Although ergonomically the layout of this keyboard is far from optimal for typing efficiently (and indeed some computer experts use different layouts), the fact that the first computers to be designed for mass-production and consumption featured such a keyboard made it hard for any future companies to choose any other given layout, for people will naturally align themselves towards buying a product that they are already comfortable with the mechanisms of (Mahoney, 2000: 515).

Thus a path dependence is created in which a structure becomes accepted and thus influences all future actions in that particular sphere. Note that although the keyboard cannot be considered optimal, it is also not sub-optimal to the extent that people find it impossible to use. Returning this metaphor to institutional politics, it is worth noting then that not any idea can be uploaded and become hegemonic, only those that have sufficient explanatory traction to be supported.

Constructivists, of course, would problematise the notion of 'explanatory traction', suggesting that this itself is socially constructed to a certain extent. Whilst historical institutionalists use ideas as an explanatory variable similar to constructivists, the issue of how exactly certain ideas become accepted in the first place is less clear in historical institutionalist scholarship, whilst constructivism places this at the centre of its research agenda.

The question of ruptures and the creation of hegemonies is therefore a central problem for historical institutionalists. Such accounts are strong at explaining continuity: once policy paradigms have been institutionalised within a polity, we can expect to see responses to future events shaped by this paradigm, and problem-solving likely to take place within the confines of its intellectual basis. Given that institutions are generally inert to change, such an approach suggests that policy continuity and equilibrium become the norm. Naturally, though, such an approach finds change harder to analyse and explain than continuity, as by definition to explain change we need to be able to explain a rupture in existing institutional constraints or understandings (although see Bell (2011) for an opposite conclusion). The absence of research on paradigm shift within this tradition, or questioning of the extent to which external events can problematise institutionalised ideas, was perhaps a major motivation in the creation of a more explicitly constructivist approach to ideas and politics. Nonetheless the two schools have as many similarities as differences, and the findings of this thesis in relation to ideational continuity through crisis could also easily be viewed through a historical institutionalist lens.

Sociological institutionalism bears a similarly close resemblance to constructivism, given its greater emphasis on socialising and norms on political outcomes. However it tends towards a similar, if less concrete, view of the normalising effects of institutions on actor behaviour as does historical institutionalism, transplanting the structure of history and formal political procedures for socialising 'logics of appropriateness' (March and Olsen, 1998: 951; also Börzel and Risse, 2003). The essence of this, as with historical institutionalism, is therefore a somewhat static view of institutions and political behaviour. The forces of continuity here are the logics of appropriateness which institutions embed in their agents; socialising them into following given policy paradigms. Thus again, as agents internalise such norms and act accordingly, we expect to see continuity in political practice. Both schools therefore provide theoretical explanations of why political behaviour will remain the same or conform to certain pre-existing formal, normative or discursive standards (Hay and Wincott, 1998; Wincott, 2004), but fail to offer a strong explanation of political change (Hay, 2006: 58). Indeed given the lack of scope for political change in these theories, they tend to rely on exogenous variables to explain it. That said, the approach I offer here is in many ways a sociological-constructivist one: in highlighting the continuity of central ideas, norms and values through a crisis I reach conclusions that may well be compatible with a form of sociological institutionalism. This highlights that although some differences in theoretical approach do certainly exist within the new institutionalisms, their similarities are often overlooked because of this. Whilst

academia is rife with petty sectarianism over relatively minor meta-theoretical differentiations, I will allow the reader to arrive at their own conclusion about the natural placement of this work.

Historical institutionalism is inclined towards an analysis of structure and agency as analytically separable entities (see, for instance, Thelen and Steinmo, 1992: 10) which the constructivist approach has seriously questioned the logic of (Hay, 2006: 58). This dynamic interplay between structure and agency, or structured contexts and strategic situated actors as they become, highlights a further development of constructivist theory on the existing institutionalisms. The existing institutions are poor at explaining political change, and are better suited to equilibrium conditions (Hay, 2011: 66). For instance, given the historical institutionalist focus on the development of structures which make agents increasingly bound into path-dependence over time, how can this school understand sharp breaks or ruptures in institutions? Likewise with sociological institutionalism, if actors are influenced by social norms, what causes these to change suddenly? And finally for rational choice institutionalism, if agents act upon material self-interest then why do these suddenly change in times of crisis? These schools become dependent on exogenous factors to explain such developments (Skocpol, 1979; Tilly, 1994). Meanwhile by making structures notably less reified, albeit still with the power to shape agents' opinions, constructivist institutionalism has a view on politics which is better capable of understanding endogenous development.

One central issue here relates to the utilisation of ideas as an object of study by the different new institutionalisms. Historical and rational choice institutionalisms have tended towards using ideas rather 'instrumentally and functionally, rather than as progressive extensions of their research programs. They reduce ideas to "filler" to shore up these already existing research programs rather than treat them as objects of investigation in their own right' (Blyth, 1997: 229). A critique of rationalism was made in the previous chapter; hence we need only briefly extend this to rationalist institutionalism. Because of its individualism and belief in singular rational calculation, thus ideas must also merely be an element of such rationality. Thus ideas are not distinct of their own right, but merely a function of pre-existing interest. The attempt by such a school to utilise ideas to help solve problems within their existing research framework is therefore undone by failing to offer an independent causal role for such ideas. The same ultimately applies to historical institutionalism's approach, with Blyth critiquing Sikkink's (1991) and Hall's (1986; 1989; 1993) works as treating ideas as 'the most important factor (which) 'do not ultimately determine the outcome' (Blyth, 1997: 235). Significantly, a critique is also made of the elitism of such suggestions (Ibid.; also Campbell, 1998): relatively insulated policy elite decide on policy developments within given policy paradigms, and occasionally these change at a critical juncture. At times such an approach is suitable, but given the very public nature of the need for ideational deliberation at a time of crisis, a more discourse-oriented approach is of validity in questioning the ability to new ideas to be uploaded to the policy level at times of crisis. Discursive institutionalism therefore encompasses a greater focus than other institutionalisms in seeking such explanations.

It is important, also, to note that we should not shift seamlessly between historical or sociological and constructivist institutionalism depending on whether we are facing a continuous or dynamic

context (Schmidt, 2008: 316). The fact that constructivism places ideas as a more central explanatory variable independent of material factors is what leads me towards favouring such an approach⁴. Indeed such an approach allows for a closer examination of both continuity and change (Schmidt, 2011: 107). Regardless, constructivist institutionalism has to date largely been used to explain and theorise change, whereas this thesis seeks to suggest that it could also be used to explain continuity. My preference for labelling myself within the constructivist institutionalist tradition also stems from the tendency of constructivism to engage with a wider set of actors than historical institutionalism, where the focus is on the internal evolution of institutions. As will be justified below, crisis and the public deliberation it stimulates requires a focus beyond merely institutional actors but to the narration of such a rupture to the public. Whilst first and second order policy evolution can often take place without a great deal of public deliberation, the onset of a crisis whereby the third order ideas of macroeconomic policy are being materially challenged requires a greater level of public engagement than historical or sociological institutionalisms classically engage with. It is here that a more discursive approach, elaborated on below, intends to offer a broader analysis of policy choice. It includes institutional ideas and norms but also analyses the temporal and hence dynamic impact of discursive deliberation within the political domain, which in turn will influence and sculpt economic policy decisions.

3.3 Institutionalist Studies of Crisis

To operationalise some of the problems with the various institutionalisms highlighted above, we can look specifically to their research outputs in relation to crises. In particular, a variety of work exists utilising these positions in relation to the last major western financial collapse: the Wall Street Crash and the subsequent Great Depression.

Rationalist institutionalist analyses have taken a 'corporatist' approach to explaining crisis politics (notably Amenta and Poulson, 1996; Gourevitch, 1984, 1986). Given the implied rationalism, the assumption made of the key vested interests who lobbied the state is that their interests were known and fixed. With existing interests weakened by the crisis, a space was opened for new coalitions to form. Gourevitch's analysis suggests that the New Deal, something very out of kilter with conventional American economic and social governance, was caused by the unusual coalition of agriculture and industry combining together to get financial assistance and producing a common policy response (Gourevitch, 1986: 33). This work fits into the American political

_

⁴ Hay (2006:61) notes, however, that although historical institutionalism has often taken a rationalist approach to ideas through a suggestion of material self-interest, there is an ontological diversity within the school, with some less clearly rationalist. It is here that the boundaries blur, leading to Bell (2011) asking if a separate constructivist institutionalism is actually necessary.

tradition of assuming polyarchy in relation to power questions in politics (Dahl: 1961). Politics is taken as an open field for different interests to compete against each other, and those that successfully create the necessary lobbying power tend to get their way. There is also an appealing logic to it, after all the very term 'New Deal' implies some sort of bargaining to which these authors make reference as their central explanatory factor. However, a number of questions can be posed of this approach from a constructivist perspective.

Firstly is assumes that all interests are known and caused by material conditions. The constructivist would problematise this. As was stated above, adjudicating what is in one's real interest is not always possible, given the inherent uncertainty surrounding issues of economic confidence. Even if a group's interests remain fixed therefore, we must make scope for the ideational context which has brought this about. Essentially therefore the constructivist takes as a social construct what the rationalist assumes is natural. Therefore there are other research questions to ask, such as why such a position was deemed to be in an agents' interest. It is also a position of dubious relation to the historical evidence. Whilst Gourevitch has pointed to the coalitional alignments in the US to explain policy change, Dobbin (1993) has shown that no such shifts in alignment can be identified in the UK, which also saw several policy shifts during the Depression. It also fails to take into account the often quite differing responses that nations produce for crises; corporatism simply assumes that a couple of different options are available depending on which coalition succeeds, whilst the history of economic policy is much more diverse. It therefore fails to ask what ideas different coalitions have had and why.

Gourevitch's work focuses on 'state strategies', asking how states respond to crises and why (Smith, 1993: 351). The explanation for different outcomes often cites the different production profile of the country, and the economic preferences of major societal actors. Again, however, we are faced with the question of known and fixed interests. Whilst Gourevitch's work is prefaced with commendable modesty in relation to the capacity to make simple models explaining state behaviour, ultimately the question of why interests were perceived in the way they were requires explanation. As he notes in a hint towards a more ideational explanation, 'the most puzzling question of all is why the Labour Party did not take up demand stimulus' as a policy suggestion in the 1930s (Gourevitch, 1984: 121). By suggesting that the desire to fit within elite opinion, which at the time excluded the likes of Keynes, was part of the strategic approach of Ramsey MacDonald (Ibid.), we get a sense of the significance of political contexts in shaping the perceived acceptability of different strategies.

Historical Institutionalist scholarship of the Depression has produced an abundance of research. As stated above the core explanatory factor in the development of policies is whether ideas fit into institutional structures. Like constructivism, it takes ideas as important, but they can only work if they fit the specific national institutions. In the Great Depression therefore, it is argued that the policy responses chosen by different countries reflected the cultures of the institutions that were in charge of them. There is much of interest here, and particularly impressively it provides an

allowance for diversity that stricter theories of rationalism do not permit. However, it fails in the case of the Great Depression to give a strong understanding of the radical nature of the shift that the New Deal epitomised. How can institutional structures be the core explanatory variable for a policy programme that tore up such structures and replaced them, one which was a complete departure from the ideational norms of American governance since the nation's founding?

Historical institutionalism therefore explains moderate policy developments or equilibrium much better than large changes in policy. Given that the current financial crisis has so far not produced a new orthodoxy in terms of economic policy (Hodson and Mabbett, 2009), a historical institutionalist perspective could easily explain the stability of policy paradigms by recourse to the existing economic neo-liberalism of the Anglo-Saxon countries. Whilst the lack of a fundamental shift in economic policy paradigms has puzzled both political analysts (Hay, 2014) and economists (Stiglitz, 2010), it is doubtful that institutional stasis and inertia alone can be used as the explanatory factor for this, especially when we compare different case studies such as Britain and the USA where different policy approaches were adopted. It also creates a question of why economic failure led to a widespread shift in the institutional conception of economic priorities in the Great Depression but not now? Again, I feel there is justification here for focusing on how the public debate about policy priorities was undertaken in order to help explain this process. Indeed, given the expectation of policy change for constructivists in times of crisis, the constructivist institutionalist school is also open for re-examination in the face of political developments. The next section develops on this important issue by discussing constructivist institutionalist work to date, and the questions that the existing crisis asks of it.

3.4 The Constructivist Institutionalist Alternative: Prospects and Weaknesses

The above section intended to spell out some of weaknesses of the existing 'institutionalisms', and the scope for a more ideational analysis of crisis moment in particular. However, to recap briefly, constructivists take issue with the structuralism and lack of scope for endogenous institutional change within these theories, not to mention the familiar positivist criticisms of Rational Choice Institutionalism. They also stand accused of downplaying the role of ideas in politics, and also failing to acknowledge the open-ended influence that material factors can have on ideational developments (Hay, 2011). As such ideas should not simply be seen as a by-product of material interests, but as something much more causally independent. The utility of constructivist institutionalism in explaining change is not to say that this is the only context where it is relevant, but rather than it allows for a more fully rounded analysis of the interplay of structure and agency, material and ideational regardless of context. Constructivist institutionalist studies of crisis to date have added valuable depth to the issue of the role of ideas in policy change, and it is explicitly

within this academic tradition that this thesis intends to make its mark. However, as with any nascent intellectual paradigm, it has created some questions and challenges. Some of these stem naturally from its desire to focus on change from a less structural perspective, meaning that it has not perhaps yet had time to reflect more widely on continuity and change. This thesis seeks to undertake this progression to further the school of thought.

Constructivist work to date has sometimes conflated crisis and change to the extent that one is often taken as a key determinant of the other, leading to charges of tautology and determinism. By analysing briefly the principal works of constructivist institutionalist literature to date, this section seeks to portray some of the weaknesses of these works which the thesis will seek to expand upon and provide solutions to. In brief, these are: that the analysis of existing works has focused insufficiently on the crisis period itself; that they would benefit from widening their scope from elites to the wider public debate; and that these works were initially overly determinist of change. This thesis will seek to make methodological developments on the first two points to ensure an approach that does not fall into the trap of the third, whilst maintaining much of the constructivist flavour which I have argued thus far to offer compelling analytical purchase to the issue of political change. The following section will demonstrate that whilst post-crisis reviews of constructivist institutionalism have already sought to overcome some of these problems, the outcome is a situation where the links between crisis and ideational continuity and change remain contested, and would benefit from closer examination. Thus a dynamic tracing methodology focused on public debates will be justified for use in this thesis.

Let us start with perhaps the most notable book of the modern constructivist school, Mark Blyth's (2002) Great Transformations: Economic Ideas and Institutional Change in the Twentieth Century. The book intends to build on Polanyi's work on the 'double movement', which he used to explain political change over time, noting that whilst this is an impressive addition to the field, it is guilty of using comparative statics (Blyth, 2002: 7) in order to make its case. This therefore falls into the trap of post hoc ergo propter hoc, i.e. studying two different cases in different time periods and producing an exogenous factor to explain the change from one to another (Ibid: 8). Whilst exogenous factors can indeed lead to change, Blyth notes that this is reliant on endogenous agents interpreting these events, for they are not objectively identifiable in their own right. Blyth's focus is therefore on endogenous ideational change by actors within institutions, which explains their development and changes over time. It is therefore intended to produce a dynamic, or 'sequential' (Ibid: 18), as opposed to static analysis to the issue, something that is a very welcome development. He takes issue with some of the existing ideational work of the historical institutionalist school, notably those of Skocpol and Weir, and Hall (Skocpol and Weir, 1985; Hall, 1989; 1993) for failing to provide a model of change in which ideas are genuinely transformative, but instead suggesting that only ideas that fit with existing institutional arrangements can succeed in being uploaded into policy (Blyth, 2002: 22).

Blyth's work represents an important introduction of a constructivist analysis of institutional change, and provides a compelling constructivist account of paradigmatic change. But as always with notable academic work it throws open a lot of questions, about both constructivism itself and institutional ideational change. Firstly whilst constructivism proclaims to be most interested in 'critical realignments' (Gourevitch, 1986: 32-33) 'organic crisis' (Gill, 1995: 400), 'creedal periods' (Blyth, 1997: 245) 'windows of opportunity' (Kingdon, 1984) or 'policy de-legitimation' (Goldstein and Keohane, 1993: 12) depending on your diction of choice, relatively little attention is actually paid to the crisis point in *Great Transformations* (Hay, 2006: 70-71). Whilst the study is still sequential, relatively little attention is paid to the most crucial time period of all; the period in which no ideational framework was dominant. The picture that is painted is therefore at times one that is guilty of teleology, with the assumption cast on Keynesianism in the 1930's or monetarism in the 1970's as ideologies which were homogenous in their nature and in their popular understandings, and arguably therefore pre-ordained in their success.

One of the central assumptions within Blyth's account of institutional change is the suggestion that the scale of the uncertainty in a crisis context is sufficient to blur agents' perception of their interests. This 'Knightian' level of uncertainty is argued to be sufficient to nullify all structural barriers to change, something that, as Hay points out, is not clear that crises necessarily do (Hay, 2004: 209-210). Given that there is some lack of clarity in Blyth's analysis about how crises are resolved in such epistemic uncertainty, or why it blocks some to identify their interests with new narratives, such an approach 'need(s) to be able to tell us rather more about the determinants, internal dynamics and narration of the crisis itself' (Ibid.: 210). In this sense, one of the more notable critiques of Blyth's work is that it overstates the role of the ideational at the expense of the material (Marsh, 2009). It is this critique which lies at the heart of what this study hopes to be able to build upon, for if we take away the starting assumption that crisis renders existing discourses null, then we are left with a question of what determines whether ideational change will happen or not. By taking this as the focus of study, rather than as an assumption, the British case study will demonstrate a high level of ideational stability and continuity during the financial crisis, providing empirical support for Hay's suggestion. The thesis does, however, take very seriously the idea of a moment of 'Knightian uncertainty' but the empirical evidence suggests that in the global financial crisis this window was very short. Hence, the crisis did not lead to ideational change comparable to the 1930s.

Berman's (2006, 2013) approach is similar in its invoking of a type of Polanyian double movement. She posits that the rise and fall of ideas follow a two-stage process: firstly where a failure of an existing paradigm creates a demand for new ideas; and secondly once this political space has opened, a supply of new ideas is provided which vie for dominance (Berman, 2013: 227). This, as Berman notes (Ibid.: 228) also leads us to the question of why some crises lead to a paradigm shift, as the Great Depression eventually did, and others apparently don't, as is the case with the current crisis. A society may enter the first phase but not proceed to the second because of the lack of development of an attractive alternative or because of the structural power of the

defenders of the status quo, she summaries (Ibid.).⁵ Likewise, Van Hooren *et al* (2014) Capiccia and Keleman (2007) and Soifer (2012) suggest that crises are necessary but not sufficient causes of paradigm change. Van Hooren *et al* note that in fact such change is the exception, rather than the rule. This leads to a need for close attention within case studies to offer suggestions as to why change or continuity occurred. This thesis finds empirical support for these suggestions, and suggests that they need not be at odds with constructivist institutionalism. By widening our definitions of crisis, the constructivist institutionalism of Blyth can be utilised to understand a much wider array of political contexts than those of the type debated in *Great Transformations*.

This therefore returns us to the importance of closely studying internal discursive dynamics so as to begin to answer why a change of ideas did not seem to flourish in a new economic context. At times Blyth's empirical work seems to forget his own nuanced theoretical work on the differentiation between 'real' and perceived interests and adopts conventional materialist explanations. Rather, I feel it is better to ask how certain strategies were constructed so as to give the perception of being in most people's interests. Furthermore it seems at times as if Blyth's focus is purely elite-focused (see Blyth, 2007), but if it is only elite ideas that are important, how does this fit in to 'patronising' argument quoted above? The links between elites and the public therefore need to be more explicitly explored. Placing a greater focus on the narration of messages from elites to the public, through the media, as I do below, intends to build on his approach.

The suggestion that the existing set of ideas could not continue within the crisis context of the Great Depression also seems to ignore much of the historical evidence on the uptake of new ideas. Whilst it is easy to view the Depression as a simple switch from one set of failed ideas to a new set with better explanatory and popular appeal, the relatively slow switch towards Keynesianism questions this. Roosevelt's election did not take place until three years after this apparent failure of ideas, of course, and even then the policies of the Hoover and Roosevelt administrations defy the simple economic dichotomy between orthodox and Keynesian. Thus a greater than appreciated period of social puzzling took place, with insufficient focus on continuity and change within this, or how this fitted with political strategies to create viable policy suggestions.

This returns us therefore to the relative under-focus of the actual critical juncture itself as the core subject of study. Not enough attention is paid by Blyth to the actual dynamics of discourse development within these periods, and he therefore fails to offer a deep understanding of how a certain framing of events came to win greater support than others. It is important not to teleologically assume that Keynesianism was a natural outcome of the Depression, it was actually a dynamic debate which 'took a certain amount of learning – economic, intellectual and political' (Gourevitch, 1986: 153), before the outcome was finalised. Likewise there is a historical problem

For Baker (2012) and Hall (2013), time can also be added to this list of reasons for continuity.

⁶ 'Social Learning' is a term many constructivists find highly problematic, in that it suggests that it is possible to learn better policy responses from objective analyses of former ones. I prefer the term 'social puzzling' (Wincott, 2004: 360) to show that this process is not quite so unidirectional and objective.

with suggesting a single homogenous Keynesianism, for ideas are 'contradictory, ambiguous and open for contestation' at all times, not just during crisis (Carstensen, 2011: 598). Whilst his argument that crisis problematises agents' ideas and interests is intriguing, he fails to detail the dynamics that lead to ideational shifts in times of crisis. For as Hay has noted, a major difference between our current crisis and that of the Winter of Discontent is that our current crisis lacks 'a clear public discourse of crisis, linking our present economic woes to a systemic failure of economic policy' (Hay, 2010: 447). Our current crisis and that of the Great Depression therefore have key commonalities in this sense, making the discursive dimension by which a new idea comes to take hold in this 'problematised' state of affairs all the more important. This is both an empirical and theoretical critique; on the empirical level not enough research has actually been done into the dynamics of discursive conflict within the period of critical juncture, and on the theoretical level it fails to offer a detailed epistemologically-backed account of the role of power and epistemic communities in shaping discourse development. To these theoretical questions I will return later.

The other central work on the construction of crisis refers to the British case in the 1970s; Colin Hay's work on the Winter of Discontent (Hay, 2010). Hay's theoretical position is largely similar, pointing out that we cannot explain it through material explanations alone and must focused on how it was 'lived and experienced', and what 'collective understandings' developed as to the nature of the crisis (Ibid.: 447). He points to the empirical difference between the material development and ideational development of the time, pointing out that the largest economic crises of the times actually occurred before and after the central political crisis which led to the fall of the Labour government (Ibid.: 464). He makes a similar empirical observation to Blyth's central thesis of the 1930s, suggesting that a particular construction of events became a public norm, in this case Thatcher's argument that the events were being brought about by an overload of the state, and as such that the most suitable cure was increased privatisation and monetarism (Ibid.: 465, also Hay, 2001). He deems its success as a construction not to be the result of its inherent accuracy, but in its ability to 'resonate with, and make sense of, the events and experiences of the winter of 1978-79' (Hay, 2010: 465). Therefore whilst the economic difficulties Britain faced in the period provided 'exceptional raw materials' to craft a narrative like Thatcher's, it did not in and of itself guarantee that such a narrative would be successful (Ibid.).

As with Blyth's work, this argument throws up almost as many questions as it answers however. In some cases these are methodological. For instance, whilst the theoretical suggestion that the lived experience and the construction of crisis being important are ones I support, there is arguably insufficient empirical evidence showing that these frames of understanding increased in popularity in the period leading up to the political crisis. A key question, given this interest in the popular narration of crisis, is that of why it came about when it did. What were the differences in terms of the discursive environment between the pre- and mid-crisis periods? It is not quite clear whether a consistent set of ideas was utilised by monetarists and these became more popular as economic events became worse, or whether the ideas evolved to better capture the public mood. A more

detailed focus on the dynamics of ideas and their extent of change through a crisis is the major development this thesis seeks to make beyond Hay's work.

Where though, does this leave the constructivist approach to crisis politics? The theoretical work, as outlined above, certainly offers a strong rebuke to some of the more conventional understandings of ideational developments, and the theoretical suggestion in both Hay's and Blyth's work that material events do not necessitate certain ideational positions 'naturally' but only when they develop enough social support is one that this study seeks to be based upon. The relative under-focus in this work on the actual crisis period itself and the internal dynamics here that explain change, or indeed continuity, calls for a more detailed dynamic tracing of frames, norms and understandings within crisis episodes to help explain in more contextual detail the way in which new ideas entered the political discourse and found resonance, or likewise failed to penetrate existing discursive trends. As such, the dynamic tracing methodology outlined below will seek to provide this greater detail so as to open up a constructivist discussion on the extent to which crises really problematise existing frameworks. The central constructivist political writings tend to focus also on elites primarily as the adjudicates of the value of new ideas; whilst I do not explicitly choose to critique such an approach, I feel in the context of problematised interests and ideas there is greater value to be had in focusing on the public deliberation over such ideas. As such, whilst media narrations are often included within the prior analyses, this is often done sporadically and could benefit from more sustained and systematic research.

To an extent, the first contributions of Blyth and Hay also seem to over-determine the necessity of change in a crisis situation. The apparent failure of an existing paradigm to explain developing events and the ability of others to do so and hence become predominant has an appealing theoretical logic because of its clarity and simplicity. However, what I feel is less certain is that all media discussion takes place neatly within such frameworks, and that only ever one singular framework is in place at any one time. Given the often sociological nature of constructivism's approach to politics, this neatness is perhaps therefore actually problematic. The failure of one set of events, I argue, does not necessitate that the associating paradigm is objectively failing and doomed to be replaced; if the crisis is not understood as a crisis of that paradigm but as of something else, then the 'failed' paradigm may not be replaced. In other words, to expect crisis to be debated publicly in terms of the failure of a wide-arching set of ideas such as 'neo-liberalism' on the basis that we as academics view it as highlighting contradictions within such a set of ideas, is a flawed starting point of analysis. By focusing more holistically on the discourses of state and economy through a crisis we can hope to observe in greater detail the extent to which these are problematised by crisis events, and thus the extent to which crises may remedied by new ideas or existing ones. One of the major arguments of the empirical chapters will be that for the most part the idea of 'crisis' itself was largely confined to the financial sector, and hence the demand for change stayed largely within this sector also. An important starting point for much CI literature to date has been to focus on 'generalised' crises, as opposed to specific ones. However, this thesis seeks to extend this discussion by disaggregating the issue of political economy into its two major spheres: financial and corporate. The empirical chapters find that the sense of failure of ideas did not percolate from one to the other, and hence the scope for wider change was more muted.

3.5 Constructivist Responses to the Global Financial Crisis

One of the challenges of writing about constructivism and its position on crisis and change is that, almost ironically, this has been a somewhat moving target since the financial crisis began. The suitability of the financial crisis for analysis in constructivist terms and indeed on the type of reflection on constructivism itself which I am partially engaged in here, has meant that the nature of the constructivism being engaged with in this research has changed as new works have been added to the field. Naturally, such developments have advanced the working definitions of constructivist institutionalism from their original beginnings. The 'classical' views of crisis and change, most notably from Blyth and Hay, analysed above, demonstrated the need to explore the link between the two, and made some preliminary theoretical suggestions about the links herein.

The failure of the financial crisis to lead to widespread political economic change in the UK and other western countries, which indeed have often seen a solidification of the existing paradigm, has caused both authors, as well as others, to reflect on constructivist institutionalism itself in a way that added further nuance to its explanatory potential. Many of the constructive criticisms of constructivism which this thesis therefore intended to engage with have therefore already begun to be ironed out during the course of the research undertaken. This is undoubtedly a positive development, of course, and I believe there has been a subtle but important development of the constructivist institutionalist theory that has come out of the process of this self-reflection on the part of its principal authors. It has not made the research irrelevant, however, as the analysis of continuity and change in public discourses through the crisis period, I believe, complement these theoretical developments and allow us to flesh them out more notably. The following section analyses the nature of these contemporary theoretical developments and what they mean for the understanding of crisis and its political implications.

One of the constructive criticisms outlined above suggested that existing constructivism overstated the extent of the 'structural vacuum' that crisis generated. In other words, it was suggested that the assumption that all the structural impediments to a change in paradigm simply were eradicated by the failure of that paradigm to predict economic events was too idealistic and simplistic. Whilst not always explicitly stated in these terms, two of Blyth's post-crisis academic contributions have in a number of ways accepted this position and produced a number of structural impediments which may have continued to play a role in shaping post-crisis policy. In his 2013 contribution to the legacy of Hall's policy paradigm debate (Blyth, 2013b: 208-210), five (largely structural) challenges to paradigm change are cited as having helped maintain the orthodoxy of the Great Moderation or neoliberal period:

- 1. Time 'it is too early to expect the accumulated wisdom of an entire generation to be trashed and replaced'.
- 2. A crisis in one paradigm doesn't necessarily mean that there is another one waiting in the wings to replace it (see also Baumgartner, 2013: 253; Berman, 2013: 228; Gamble, 2013: 55).
- 3. The economics profession is still largely structured towards the shibboleths of the precrisis period and there is academic pressure to comply.
- 4. Those in power were too firmly wedded to the ideas, and were not open to change. They also used this power to decide which evidence was of importance and chose to focus on government debt (see also Baumgartner, 2013: 252).
- 5. There was strong institutional, academic, media and financial weight behind these ideas, compared to the 1970s when a number of these respected voices were against the status quo.

Within these lie some significant arguments about the nature of crisis and change. Firstly, and perhaps most importantly, it demonstrates an understanding that the entire discursive force behind a dominant paradigm cannot disappear overnight, even if the paradigm fails on its own terms to understand the developing economic events. The agents, institutions and discourses that have supported it are slow, if not inert, to change. This creates a research agenda focused on exactly how vulnerable existing paradigms are to change within a crisis, which this thesis seeks to explore more fully. One problem, however, is the apparent shift away from constructivist notions of strategicrelational behaviour and towards more conventionally structuralist or intentionalist arguments. Blyth's re-assessment seems to demonstrate a re-emergence of both within his beliefs on policy change. Within both is a continued elitism which I feel is questionable. Firstly it suggests a neat elite groupthink in which the political-corporate-media nexus decide which evidence to present to the public and hence receive passive compliance from them to continue their policy experiments. I am not convinced that sufficient evidence is presented of this ubiquity of elite thinking. Secondly the return of structuralist explanations question the theoretical ability of crises to cause epistemic uncertainty, as highlighted in Blyth's earlier work. There is a danger here then that the theoretical baby of constructivist institutionalism is thrown out with the global financial bath water.

It does, however, highlight the contingency of change on the ideational conditions which can allow for it. Whilst previously it was suggested that crisis necessitated a change in paradigm because the failure of one necessitated its removal, this was arguably an overly-Bayesian approach

to the issue of 'failure'. For if there is no true scientific measure of the success of a paradigm, and what determines this is actually a more sociological process, then likewise there can be no objective measure of its failure. Thus crisis cannot *necessitate* a change in paradigm unless we either view the discursive and sociological approach to evidence as paradoxically scientific, or we define crisis purely in terms of change, which I return to below. What can be concluded from Blyth's contributions then is that an empirical focus is needed on ideational stability and instability through crisis, so that more clear formulations can be made about exactly how economic ruptures are likely to problematise particular discursive formations and create, or not, the conditions for new ideas to flourish. The British context laid out in this study suggests that greater political stability is indeed apparent than is often presumed, and that the particular pre-crisis context created a strategic advantage for the discourse of austerity over that of stimulus.

At this point it is worth noting the contributions of Colin Hay to the theoretical questions which the financial crisis stimulated. Perhaps his most telling addition to the theoretical literature comes in his chapter on the 'crisis of politics in the politics of crisis' (Hay, 2014). Here he astutely notes that the fact that the narration of crisis has been paradigm-reinforcing rather than paradigm-threatening 'is not something that the existing literature, in its typical (and perhaps understandable) concern to link crises with episodes of paradigm succession, prepares us well' (Hay, 2014: 65). However his conclusion is in many ways quite different to that of Blyth. For, rather than questioning the more structural factors which may have limited the capacity of a crisis to produce change, he sticks (linguistically and etymologically, very logically) to the perception that crisis can only be considered as such at a moment of 'decisive intervention' (Ibid.: 77). Thus the financial crisis cannot be viewed as a crisis as it did not lead to such a decisive change, and is better viewed simply as an accumulation of contradictions in the existing paradigm (perhaps 'ailment', to maintain the medical metaphor) that has not yet led to its crisis. As stated above, the etymological consistency of this is very hard to challenge, but again it leads to the criticism that the study of crisis is somewhat tautological if we only define it in terms that justify itself. Below, I turn to the literature on strategically selective contexts to attempt to dissolve this definitional issue of crisis.

3.6 When is a crisis not a crisis?

It may be beneficial here to take a small detour to return to the question of what exactly we consider a crisis to be in the first place, for implicit within the whole discussion of the impact of crisis on political change is the assumption that the global financial crisis was actually a crisis in the

first place. On one level, my concern with Hay's suggestion that this was an accumulation of contradictions rather than a full-blown crisis is somewhat normative. I am troubled by the suggestion that the western world coming to within thirty minutes of the collapse of its entire financial system, to which every citizen and government was deeply interconnected, as somehow being definitionally insufficient to be deemed worthy of the term 'crisis'. Importantly, however, the above discussion of strategically selective contexts suggests that the issue of what actually constitutes a 'crisis' can be unproblematically resolved and dissolved, or at least rendered analytical as opposed to causal. For, as stated above, if no political situation pre-determines action, there can be no such thing as a crisis necessitating change. Whilst this seems to run counter to Hay's definition of crisis,⁷ it need not necessarily be read as a critique of it. This leaves us with two different ways of defining crisis.

Firstly it could be defined as a period of upheaval or uncertainty, in this instance economic, caused by a rupture in an existing model. Secondly, however, it could also become a post-hoc heuristic tag used by scholars to group episodes of similar characteristics, for the ease of their collective understanding. This follows Hay's conception more closely, but crucially rather than being an independent variable upon which change is dependent, it becomes a category used to describe, rather than explain, common changes. For such a view, given that crisis can therefore be relatively easily identified – if there's punctuated equilibrium there must have been a crisis, if there hasn't then there wasn't – the issue of whether the financial crisis was indeed a crisis can be relatively easily solved; it cannot have been due to the lack of major change. For the former view, on the other hand, the global financial crisis was indeed a crisis, albeit one that didn't lead to change. Either way, the discussion, when put in these terms, dissolves the importance of the term 'crisis' as an important explanatory discussion in its own right, and merely leaves us with the more important issue of determining why a strategically selective context did not favour change in this period of upheaval. I believe a more solid definition of crisis could be conceived in the following terms:

For something to be considered a 'crisis', previously unforeseen internal contradictions to an existing model must have become visible to the extent it becomes universally understood that there is something unstable or unsustainable about the status quo which has created this problem to be solved.

On first appearance, such a definition might be accused of the same tautology that Rogers lays at the door of Hay; after all it comes close to pre-determining that change must take place, and that 'decisive intervention' is therefore required. However, it does not pre-determine that the change that takes place must be a paradigm-shift; the policies that are created to 'solve' a crisis may not actually rectify the contradictions that led to it, but as long as they are discursively believed to do

continuity is no longer possible.

⁷ And arguably Blyth's too: although his take on exactly what constitutes a 'crisis' in the first place is a little unclear, he appears to situate it as a context in which something has clearly "gone wrong" in which economic ideas can explain "what is to be done" (2002: 10). This follows a similar etymological process as Hay's definition of a critical event in which

so, this is irrelevant. Change may therefore be cosmetic, or indeed the full nature of the crisis and its implications may be conceived in a way that is dependent on a specific reading of the crisis that may or may not really deal with these contradictions. Due to the ontological diversity of the human mind, constructivism takes as a crucial starting point that multiple understandings can flow from the same empirical events. When we apply this logic to a crisis, we can see that there is never a single way to interpret, and therefore devise a strategy for curing, it. There is no objective existence of a 'real' or 'true' set of causes for the crisis or how society should interpret it.

The definition therefore allows, I believe, for sufficient sociological space not to pre-determine specific change or paradigm-shifts, and therefore incorporates my argument above that paradigm-shift need not be a Bayesian process in which an existing policy paradigm is objectively proved to have failed and thus be replaced with something that can objectively address the contradictions that led to the crisis. Rather, a wider space is opened through which the interpretation of exactly what the contradiction mean in terms of the existing paradigm, the extent of change necessary to rectify them, and the success of them, is a sociological and political, rather than purely empirical, process. It also incorporates Hay's (2014: 77) concern that 'crisis' itself has become an over-used term in political discourse in recent decades, muddying the waters of the analysis of it.

The above definition does not allow for a crisis to be one purely because a political actor uses the term for rhetorical purposes; it must have some basis in ongoing events that a consensus of instability has arrived relatively quickly. However, the definition admittedly has blurry edges in terms of what exactly constitutes a consensus on instability, and how exactly this is formed. Despite this, I utilise it, for as with the argument about early constructivist institutionalism and crisis, a definition which is too prescriptive in the social sciences is likely to have faults within it. The focus of study, therefore, under such an understanding, ceases to be a question of 'what determines whether a crisis leads to change?' and instead becomes 'what determines whether major economic ruptures leads to change?' The added benefit of such an eradication of 'crisis' as a major conceptual discussion is that it tackles Rogers' (2012) critique of Hay's conception of crisis as tautological.

3.7 Discursive Institutionalism and Media Narration

To date, prominent constructivist analyses of crisis events and their institutionalist forerunners have tended to focus heavily on political elites as the key unit of analysis. Following the insights of Schmidt and others, through this thesis I have sought to explore the wider discursive context in which elite discourse is located by analysing the media narration of the UK financial crisis. In the

paragraphs below, I explore the ways in which this focus can extend our understanding of dynamics of continuity and change in political economic crises.

As has been noted by Seabrooke, much scholarship about ideas in politics focuses on elites actors (Seabrooke, 2006: xi). Discourse and ideas, where covered, tend to revolve around the ways in which political agents narrate to the public. Discursive Institutionalism, as conceptualised by Schmidt, for instance, covers the 'coordinative' and 'communicative' elements of elite discourse (Schmidt, 2008: 303; 2011: 3). Others focus on advocacy coalitions, epistemic communities or policy networks to analyse the spread of ideas through elites (Haas, 1992; Sabatier and Jenkins Smith, 1993; Keck and Sikkink, 1998). Much of the work on 'policy paradigms' focuses on the way elites develop, maintain or change ideas about economic policy based on their perceptions of their success (Seabrooke, 2006). Such perceptions of ideas and discourse do not necessarily place politicians as the sole agents with power in the creation of discourses – Schmidt's communicative discourse, for instance, focuses on the way shared narrations of policies develop, in which the media may play an active role. But few explicitly focus on the discursive context that is created by the media, a perfectly understandable outcome of the preference of interests that political analysts will have for politicians.

Through this thesis, I add to this avenue of research by incorporating a focus on media narration. The fact that elites need a 'social source' of legitimacy for their power is put forward by Seabrooke (2006), and it is in this spirit that I likewise wish to focus less on elites themselves than on the public debate which shaped the limits of acceptable political action by such elites. Indeed, to narrow the frame of analysis even further, I wish to focus explicitly on the broadsheet newspaper media. The role of the media in generating understandings and politically acceptable strategies is often cited in ideational understandings of the political, but it rarely gets systematically studied. Rather, choice quotations are often used to demonstrate a wider point, without exploring the extent to which these quotes fit within the general picture or represent fairly the wider beliefs of the press in general. Thus, such selectivity raises questions about the methodological impartiality of the researcher and the extent to which they are capturing a relevant element of political debate, or are merely 'cherry-picking' an example to support a wider point. A wider discussion of this issue will be covered in the below methodological chapter, but for now it is worth briefly dwelling on how an analysis of media discourse fits into the strategic selectivity of a context.

The major analytical intention of this thesis is to work within a constructivist institutionalist framework to suggest potential developments on this theory's understanding of political continuity and change. Thus we take as a starting point Blyth's assumption that in periods of crisis, structures become problematised and entrenched interests are dissolved by the uncertainty of a situation. Thus in a context of limited structural constraints, the public understanding of 'what is to be done?' is important in shaping how these interests re-align. The public debate over crisis is therefore one that is particularly significant, and, from a strategic-relational and realist constructivist standpoint, the development of publicly acceptable norms is crucial in limiting the agential scope of politicians to respond to events in their 'own' desired way. Thus the public acceptability of certain ideas and

norms limits political action in a particularly significant way during times of crisis, and thus by focusing on the arena of this debate, the media, we can begin to understand why certain options became tenable and untenable, credible and incredible. From a realist constructivist standpoint, it is not decipherable whether the media generate public norms which constrain politicians, or whether their norms simply reflect those of politicians. Again, like structure and agency this is an alloy we can only see the outcome of. But regardless, we can deduce that the ideational context was important in shaping the outcome.

3.8 Methodology

As Parsons (2010: 91) notes, the diversity of constructivist methodologies is almost as broad as the topics that constructivism has discussed. Most, however, include some element of process tracing to show how ideas have been adopted over time through persuasion, socialisation, or bricolage (Ibid.: 92). Important here is the dynamic nature of much constructivist study; whilst there is no common methodology, the inclusion of a focus on development over time, as opposed to a static analysis of a given moment, is a common approach. As noted above, one weakness of existing constructivist literature on crisis, however, has been the relatively limited focus on dynamics within crisis discourse, however. This section justifies the dynamic tracing methodological approach adopted to help tackle this weakness, borrowing insights from the framing literature to justify the importance and value of media analysis as a tool in political explanation. Having done this, it will continue by describing the specific methods adopted in the thesis as a whole.

Here, I take the phrase 'dynamic tracing' to refer to holistic analysis of patterns of discursive continuity and change across a medium-term timeframe. Whilst my specific approach may be deemed one of discourse analysis, I prefer the term dynamic tracing to explain the particular discourse analysis approach adopted here. The approach differs, however from 'policy discourse' type analyses such as Schmidt's (2002; 2008) 'discursive institutionalism' which focuses to a greater extent of the coordination and communication of policy approaches by elites. Here I focus more on the 'communicative environment', which I consider a central part of the strategic context (Payne, 2001: 39). A focus on public discourses allows us to understand 'how they legitimise some policy initiatives and marginalise others' (Burnham *et al*, 2008: 250).

There are two central reasons for this. Firstly, discourse analysis, and specifically critical discourse analysis (CDA), often generates connotations of a type of post-structuralist analysis that I am not attempting to engage in here (On CDA, see: Fairclough, 1995a, 1995b; Van Dijk, 2003). Secondly, however, the thesis attempts to move beyond one of the central challenges of undertaking discourse

analysis; namely the question of whether politicians create, or merely react, to discourse (Burnham *et al*, 2008: 254). This is a result of my adoption of a more realist constructivist approach where I focus less on the meanings of communication and language, but rather choose to focus on the emergence and development of certain public discourses which frame situations as necessitating certain action.

The framing literature has itself provided a varied discussion on how media discourses can shape public perceptions on given topics. Whilst demonstrating a causal link between discursive frames of understanding and public opinion as a whole has at times proved problematic, sufficient empirical work has highlighted the ability of specific frames to mould political opinion. Thus I am able to justify a focus on media discourse as an area that plays an important 'part of the process by which cultures are produced' (Gamson and Modigliani, 1989: 3). Public discourse as a form of 'argumentative rationality' is crucially 'linked to the constitutive rather than the regulative role of norms and identities' (Risse, 2000: 2), suggesting the value of media analysis as a realm in which public ideas can be created as well as debated.

A short overview of relevant literature is helpful in displaying the importance of frames for shaping our perceptions of social reality. Whilst the notion of the media as an ideological agent has declined in social studies since the 1970s due to the failure of this research to seriously investigate the ways the public 'read' texts and discourse (Fairclough, 1995a: 47), we should also not be 'beguiled by the economic metaphor of a marketplaces of ideas into assuming that the media automatically provide citizens with exactly the ideas and information they need' (Page, 1996: 8-9). Much modern framing research has therefore taken an explicitly constructivist stance in viewing the media as the shapers and makers of social reality (Chong, 1993; Scheufele, 1999: 105). Whilst Jacobs and Shapiro (2000) argue that public opinion is almost completely obsolete in modern political decision making, the focus of the 'communicative environment' within a realist constructivist position means that media content remains relevant without having to make direct causal assumptions about whether media discourse is top-down or bottom-up in nature (Payne, 2001: 39). Jacoby's empirical study of framing and perceptions of government spending is particularly relevant here, demonstrating that attitudes 'vary markedly with the presentation of the issue' (Jacoby, 2000: 750), but also that 'public opinion is *not* merely an aggregate characteristic of the mass public' but rather that 'public opinion results from the *interaction* between the public and political elites' (Ibid.: 763). From this flows the notion of the media as an important venue for this interaction as a whole.

Frames of understanding are not simply an opinion or topic (Carragee and Roefs, 2004: 218-219), but serve the function of problem definition, causal interpretation, moral evaluation and treatment recommendation (Entman, 1993). They make 'certain practices possible but others unthinkable' (Dunn, 2006: 370). They therefore provide 'a singular interpretation of a particular situation and then indicate appropriate behaviour for that context (Payne, 2001: 39). Klotz (1995: 31) notes that agents tend to demonstrate how their ideas fit in with existing practices, a more sociological-constructivist hybrid understanding which demonstrates the importance of socialisation

alongside the scope for new ideas to be incrementally developed if they are able to describe themselves within existing patterns of discourse.

Whilst some benefits of using framing as a way of explaining the political salience of certain ideas (Stromback and Kiousis, 2010: 287), I intentionally do not engage in a full framing methodology here. Several weaknesses could be noted. Framing research has tended to focus on one directionality and on single frames (see Entman and Herbst, 2001; Zaller, 1992), but this has been criticised for missing the existence of multiple frames in a given context (Borah, 2011: 251; Carragee and Roefs, 2004: 216; Pan and Kosicki, 2001; Sniderman and Theriault, 2004). On a more practical level, there is insufficient space in this thesis for a focus both on media discourse and its effect on public opinion through survey data. Demonstrating the precise effect of particular frames has also been challenging for the existing literature (Rein and Schön, 1993: 151).

I therefore favour a looser 'bricolage' approach to discourse, framing and ideational dynamics (Parsons, 2010: 96). The methodological focus is not on individual frames but the more general discursive context surrounding several topics of state and economy. The 'bricolage' approach I adopt intends to provide a broader overview of the strategic context created by political discourse, and leads me to adopt a more holistic analysis. The advantage of this broader approach is that by focusing on specific discourses or ideas, we miss the 'complex landscape of overlapping realms of action' (Ibid.). Given that a financial crisis impacts and problematises a variety of spheres of governance and policy paradigms, I feel this breadth is a strength.

The thesis engages broadly with the fiscal policy debate within 'the media' in Britain, although it is important to clarify again that the engagement is not with the media in its entirety but with a specific section of this: broadsheet newspapers. The issues of justification here relate to whether broadsheet newspapers provide a representative sample of 'the media' as a whole, and what specifically they can tell us about national policy debates and ideational legitimacy. There are a number of valuable reasons for choosing to use the broadsheet media as a focus for analysis. The representativeness of the general political philosophies of sections of the public are captured very neatly by newspapers, in that they target a specific section of the public in a way that cannot be said of television news media. They are independent of the need to achieve 'balance' in the way that the biggest news media outlet in Britain, the BBC, is, and thus they are able to generate an acceptability of certain ideas, or at least to reproduce them, more so than news media is able to do. For my chosen topic of continuity and change, this provides a much clearer and accessible route into the issue than news media where the need to provide balanced reporting masks the extent to which there is genuine change or continuity in the ideas on a subject.

On a more practical level, broadsheet newspapers as a medium offer a continuity of style and format that is not always be the case in other forms of the media. A single-author analysis of economic events does not change over the period of analysis, ensuring that the comparison between earlier and later time periods is done consistently and does not accidentally produce an 'apples and pears' approach. For television news, specific documentaries, different editors to programmes, and

the multiple voices that come from the range of interviewees means that analytical consistency is much harder to achieve. Likewise, practically, a good national database of archived broadsheet publishing exists in a way that is not specifically the case for television news media.

I likewise chose to ignore the role of social media. The academic utilisation of this as a medium of value was relatively scarce at the start of the project, and the vast development of social media as a feature of daily life over the course of the timeframe for analysis again makes 'apples and pears' type comparisons problematic. Moreover, recent research suggests that a great deal of the 'noise' on social media is made by a very small proportion of the public on it, raising major questions of representativeness (Pew Research Centre, 2013; Tufekci, 2014).

The decision to focus on a single case study differs slightly from the more common constructivist approaches, which often focus on a small-n sample of cases to engage in comparative and counterfactual testing of necessary and sufficient conditions for ideas to be adopted. The decision to focus on a single case stems from the following major factors. Firstly, as above, the issue of space and scope limits the variety of cases I can consider in sufficient detail in the space allowed. Ultimately this creates an option whereby either more cases could be analysed in thinner general detail, or a single case in greater detail. Because one of my central criticisms of existing constructivist work above has related to the lack of detailed attention paid to crisis periods, I decided upon the latter as the more suitable logic of progression. Furthermore, a challenge of cross-country comparison is the inevitable cultural differences between the two that obscure more scientific comparative analysis (Blondel, 1995: 6). As will be mentioned in chapter 8 however, one obvious scope for extension of this work is the expansion into more case studies to provide more general conclusions on ideational stability and change.

Small-n comparative studies are of particular value in situations where the researcher is attempting to establish causality and isolate potential variables causing change. This is often a valuable approach, in constructivist institutionalism as elsewhere. I do not wish to argue that a single case study discussion is more suitable in all contexts. However, given my interest here in starting a conversation about whether constructivist institutionalism has sufficiently discussed continuity and explored the potential reasons for it from an ideational perspective, the above reasons for choosing a comparative approach are less pressing. It seeks to extend the literature by highlighting ideational continuity in a crisis, and using this as a basis for discussion about constructivist institutionalism itself. For such an enterprise, a single case study is sufficient, and indeed the extra depth that undertaking a single case allows for actively works to strengthen the argument that this continuity is present and worthy of discussion.

The time-frame for analysis is the 2007-10 period, beginning in January 2007 and ending at the 2010 General Election. The logic of this is to provide a roughly equal time-frame both before and after the election to study the extent of ideational commonality between the two periods. The election provides a suitable ending point given the definitive policy approach that was adopted by

the Coalition government in its aftermath. As such a similar time period before the crisis is covered to ensure balance.

In order to ensure a breadth and depth of coverage, the analysis has included both left- and rightleaning papers. As such, the Guardian and Independent represent the left-leaning media, the Daily Mail and Daily Telegraph represent the right-leaning media, with the Times representing something of a middle ground in between. The Financial Times is also included, given its prestige in financial circles. Whilst the decision to only focus on the broadsheet media could be critiqued for its narrowness, the inclusion of a variety of titles from different sections of the political spectrum was undertaken with the minimising of this critique in mind. It ensured an analysis of the full depth of major titles in Britain which play an active role in high-level policy debates, and hence avoids some of the questions over selectivity that would stem from focusing more on a couple of titles specifically. The decision to avoid tabloid titles was taken in large part because of the desire to engage with the more detailed intellectual offerings on the subject of fiscal policy, which are more consistently in evidence in broadsheet than tabloid newspapers. Whilst tabloid newspapers can often claim to play a central role in shaping the political spectrum, on the topic of the high-level ideas governing the country, it is the broadsheet media that arguably plays a more active role in the detail of political debate. This is not to say that a study of tabloid newspapers is of no value, but given the relative lack of economic depth to their analyses, such a study would have perhaps required a 'deeper' discourse analysis looking at the way in which language choice affected either public understandings or political reality. I have sought throughout this study not to engage in quite this 'type' of discourse analysis, and have not therefore coded headlines or differentiated between front-page and other articles, but instead sought to look at the major ideas behind broadsheet media analysis.

Many of the methodological choices made within the thesis are forced to straddle the breadth/depth divide, and this is no different. It is hoped that these largely work to balance each other out as a whole; the breadth of media titles chosen helps to balance out the lack of breadth of this news medium within 'the media' as a whole.

Nexis UK was used to identify relevant news articles, and a variety of search terms were used to provide an overview of the arguments being made in each time period. Particular attention was paid to the coverage surrounding budget and pre-budget reports to parliament, set piece acts of political debate which allow for the press' appraisal of the government's role in the economy to be exposed most fully. 'Financial Crisis' and 'Global Financial Crisis' were also used to pre-select relevant articles. Particular attention was paid to 'comment' articles by regular columnists at the paper, and hence the names of each papers' regular economics commentator were used also. 'Fiscal', 'Crisis' 'Fiscal Policy', 'Fiscal Crisis', 'Keynes' 'Hayek' 'Recession, 'Depression' 'sub-prime', 'economic policy', 'economics', and 'banking' were also used to produce relevant articles.

To capture the detailed processes through which discourses evolved across the Global Financial Crisis period under investigation, qualitative notes were made on each about the central economic arguments, the underpinning ideas relating to the proper role of the state in the economy, and the impact of the crisis (for mid- and post-crisis articles) on such roles. While quantitative content analysis has been used in other constructivist studies (e.g. Pirie, 2012), the technique is typically deployed when discrete operational practices or the presence or absence of broad themes are being tracked; in this study, to generate a rich understanding of points of continuity and change, this qualitative approach was more appropriate. That said, after notes had been taken on whole sum of the documents, a rough content analysis of key themes for the pre-crisis period was undertaken. It was then that the content of the pre-crisis analysis began to be structured, and only then that the post-crisis documents were compared with the pre-crisis ones over these themes. In such a way selection bias over the mid- and post-crisis material has been minimised as far as possible. Inevitably there is a question over the problem of selection bias here, and naturally it is one that is impossible to entirely eradicate the potential of, for even the separation of articles into major themes entails a certain amount of human subjectivity. I do not intend to argue that an entirely objective approach is feasible or was undertaken, merely that the content analysis used to bring out the key themes was sufficiently rigorous to minimise it as far as was possible.

The quotes used from these articles in the following empirical chapters is intended to summarise sections of such articles faithfully and objectively, although naturally from a constructivist position, the ability of the researcher to engage such an analysis in a purely objective way is open to question. A central underpinning assumption of the analysis that follows from such empirical work, then, is that this objectivity has been reached as fully as is possible.

3.9 Conclusions

'Through this chapter as a whole, I have added to the foundations provided in Chapter One by outlining the specific engagements with constructivist literature that have been catalysed by my analysis of UK media narrations from 2007-10. In particular, two core extensions have been highlighted. The first extension relates to the value to be gained by using a constructivist framework to navigate a middle-path between historical and sociological institutionalisms. The second to the value to be gained by holistically studying media narration to gain traction over a period of crisis. I recap on these two dimensions below.

Firstly, the chapter review much of the existing 'new institutionalist' work. Here I demonstrated that, by often focusing on the ability of institutions to rapidly reconfigure or to produce stasis, they

were perhaps less comfortable in dealing with crisis situations which produced partial or 'slow' change. Existing work has tended to focus on institutional inertia along historical or sociological institutionalist lines. Constructivist institutionalism, however, was found to be guilty of overemphasising the ability of crisis to create paradigm shifts through its focus on ideational failure and replacement. Importantly, then, this chapter has built upon the existing constructivist literature to create a focus on discussing both continuity and change from a constructivist position without risking tautological findings.

Secondly, the chapter has justified a focus both on the media, and on a dynamic tracing of key ideas over crisis to provide greater empirical examination of their ability to cause ruptures in existing understandings. The focus on the media has been justified from a combination of the strategic-relational and realist constructivist literatures in that it allows us to dissolve the issue of beliefs and persuasion, and instead focus on the significant context created by media discussion in the face of a major economic rupture. We can thus argue that the context create by media discourse creates a strategic context likely to favour certain policy paradigms over others. The empirical chapters of this thesis will utilise this to examine the extent to which we can suggest that the post-crisis context created a strategic advantage for the orthodox austerity discourse. By dynamically tracing key ideas on the role of the state over the time period of the crisis, this thesis intends to overcome existing weaknesses in constructivist institutionalist literature relating to the relatively under-detailed analysis taken to actual crisis discourse dynamics.

However, before this analysis is undertaken, it is necessary to provide greater economic context on the financial crisis itself, and to explain the significance of the two principal schools of economic thought that came to dominate discussion through the 2007-10 period. The following chapter will therefore do this by providing an overview of the economics of the pre- and mid-crisis contexts, and explain how the available raw economic evidence could be interpreted in two different ways by these schools of thought. Again, this feeds into a justification of constructivist thought highlighting the lack of social truths and the importance of ideas to help us explain the world as we observe it.

Chapter 4: Contextualising the Global Financial Crisis

4.1 Introduction

The previous chapter highlighted the significance granted by existing constructivist research to the role of uncertainty in creating a space for new ideas to gain pre-dominance in a crisis context. In particular this literature highlights the capacity of crises to create these episodes of epistemic uncertainty that reduces the structural impediments to change. These 'windows of opportunity' are marked by ideational contestation in which successful ideas come to shape the future institutional order. One of the primary purposes of this thesis is to explore the extent to which the Global Financial Crisis ruptured existing discourses and truly allowed for new ideas to germinate in the public consciousness.

Through this chapter, I bridge between the earlier conceptually-focused chapters and the following empirically-focused chapters in three central ways. Firstly, because of the principal significance of orthodox monetarist and Keynesian schools of thought during the crisis, I provide a broad outline of both and the impact that past crises had on their key ideas. By looking to the Wall Street Crash and Great Depression, the role of crisis in shaping these schools of thought, and in turn how they have been used previously as models of crisis resolution, provides context on the significance of these ideas to post-war British economic governance and tradition. Secondly, the chapter provides a general overview of the economic context of the pre-crisis period and the economic events of the financial crisis itself. It achieves this by outlining the governing orthodoxies of the 'Great Moderation' period, but also highlights some of the key economic data to which political discourses might naturally attach themselves and seek to explain. Again, the purpose here is to provide context and economic detail to the discourses which will be referred to in the following empirical chapters.

The third, and most significant, purpose of the chapter blends the two previous sections to highlight the alternative explanations these discourses held over the economic events in question. The logic here is not to favour one approach over the other as a method of crisis resolution, but precisely the opposite. By demonstrating the capacity of two distinct schools of thought to interpret the same events and suggest alternative solutions, the potential value of both as governance

paradigms can be established. If a pure structural vacuum existed, we would expect neither of these discourses to have a particular pre-selected advantage over the other, and theoretically either could expect to become dominant should they prove discursively popular. Crucial, here, then, is the concept of 'contestability'; the events of the financial crisis represent a contestable context in which events are open to a wider variety of explanations than is generally the case in a period of stability. It is important to state, then, that what separates the two schools of thought is not simply whether one is more economically 'right' than the other, but the ability of either to win over more support more normatively. To re-state Hay's focus, then, the outcome is one that is political, rather than economic, in nature. In making this distinction, this chapter plays a significant role in the development of the thesis as a whole, for it allows the empirical work which follows to question the nature of this existing constructivist tenet, and as such allows the conclusions to offer a development upon current work in this regard.

4.2 Introducing the Keynesian and Orthodox Paradigms

Through this section, I provide a 'broad brushstroke' overview of the contending Keynesian and Monetarist approaches to crisis management, as developed in response to the Wall Street Crash and Great Depression. Contestation of the 'correct' lessons to be drawn from the Depression has been identified as a key battleground in elite policy discourse over the Global Financial Crisis (see Samman, 2011). Through this section, I provide a 'broad brushstroke' overview of the contending Keynesian and Monetarist approaches to crisis management, as developed in response to the Wall Street Crash and Great Depression

After a booming post-war economy had inflated asset prices in America to all-time highs in the late 1920s, a series of shocks hit the economy in late 1929, and again in the early 1930s, knocking such a great value off the American stock market that it took until 1954 for the Dow Jones to return to its pre-1929 level. Whilst Britain had suffered relative post-war stagnation in the 1920s, Western European economies were still sufficiently coupled to the USA's for there to be a severe contagion effect. The crisis created a period of great ideational dispute, leading to a host of different policy programmes being followed in different countries to tackle the crisis' consequences (Dobbin, 1993: 6; Gourevitch, 1984: 95).

Looking firstly at America, often overlooked is the variations within 1930s economic policy. Herbert Hoover often plays the fairy-tale villain in modern representations of the crisis for favouring orthodox policies, focusing on balanced budgets above all else. This overlooks his support for public works schemes during Harding's Presidency (Kuehn, 2012: 156), and his statement that the government was "engaged upon the greatest programme of waterway, harbour,

flood control, public building, highway and airway improvement in all our history" in response to the crash (quoted in Roubini and Mihm, 2011: 159). The 1932 presidential election does not therefore constitute the clear-cut 'austerity versus stimulus' vote as is often suggested. But regardless, Roosevelt unveiled an even greater collection of stimulative policies, labelled the New Deal, to create employment and boost business investment.

Most commonly cited by Keynesians is the attempt of Roosevelt to cut back on such spending during 1936-37 in the hope that the recovery had been sufficiently supported by the New Deal, only for it to become apparent that it hadn't. With the economy tumbling back into recession, Roosevelt opened the spending taps once again leading to a return to growth. Various political coalitions were necessary to undertake such the sweeping changes that the New Deal represented (see, for example, Ferguson, 1984; Kurth, 1979), highlighting how crisis had swept away existing governing consensuses. The Federal Reserve generally opted against an activist stance to prop up the money supply, which as noted below has been attacked by some as a key cause of the depth of the downturn. An important issue to highlight here is that of temporality; it took almost a decade of uncertainty before the crisis was ended, and the best part of half a decade before a reasonably coherent set of policy prescriptions were decided upon to fix the initial crash. As such we can question the inevitability of change in the short term as a response to such shocks.

Britain lacked such a New Deal programme, focusing on orthodox balanced-budget measures (Arndt, 1944: 94). Its Depression was milder than in America (McKibbin, 1975: 104), 'surprisingly vigorous' even (Richardson, 1967: 313), but still saw high unemployment and stagnation until the War forced the economy into gear. There was not a coherent macro-economic strategy that ran through the decade as could be argued with America (Booth, 1987: 499), although the Treasury did show some fiscal, monetary and exchange rate innovations (Howson, 1975: 142; Howson, 1980: 56; Middleton, 1985: 173) Indeed the British left were highly complicit in the economic orthodoxy (Gourevitch, 1984: 121; see also Lyman, 1965: 142-143), and it was not until the Second World War that the Labour Party started to adopt Keynes as an economic figurehead for the party (Barberis, 2006: 145, 161-162). In the 1930s Keynes was seen to be seeking to save capitalism from itself, something the Labour Party at the time was not naturally aligned with (Skidelsky, 1967: 12). Likewise therefore New Deal spending was seen as merely propping up a failed economic system by many on the left (Malament, 1978: 138-139). In its most extreme guise, it was suggested that 'fascism was capitalism in decline, American capitalism was in decline, and therefore the New Deal was a nascent form of fascism' (Ibid.: 139). There was a general concern that counter-cyclical measures would lead to capital flight (McKibbin, 1975: 105), and most of the establishment therefore followed orthodoxy. In both cases, it took the impact of the Second World War to truly wash away the stagnation of the 1930s and return output towards its maximum potential. In both cases also, the window of uncertainty was open for a prolonged period.

As stated above, this is far from a detailed or objective analysis of the key components of the 1930s, but its intention is to highlight some of the central moments which have been used by differing schools of economic thought about macroeconomic management in the post-Depression

era. Two central schools of thought came out of the crisis; the monetarist and Keynesian schools. Two further groups, the 'Austrian school' led by Von Hayek and a Marxist interpretation of the 1930s operated more on the margins of mainstream debate, but argued for more fundamental changes as a result of the Depression and shall be covered briefly.

The monetarist conclusion from the Great Depression was that the Federal Reserve was largely to blame for allowing the money supply to drop so dramatically between 1929-1933. It was this, they argued, that had prevented the economy from recovering quickly from the impact of the crash (see Friedman and Schwartz, 2008). The more classically Keynesian plan of government spending was rejected because it was feared that government spending would drive up inflation and scare business away from investing (Leuchtenburg, 1963: 245; Winch, 1969: 136-137), and didn't influence the recovery as is often suggested (Romer, 1992: 781). The monetarist conclusion was therefore that the central bank, and therefore monetary policy, could solve a crisis by actively and artificially propping up the money supply. Indeed monetary policy was believed to be able to prevent crises happening in the first place by manipulating interest rates to prevent asset price bubbles occurring in the first place; 'taking away the punch bowl just as the party gets going' in the words of former Fed Chairman William Martin (quoted in Born et al, 2011: 254).

This 'lesson' from the depression, perpetuated by Milton Friedman and his supporters, has had a profound impact on American central banking ever since, with current Fed Chairman Ben Bernanke stating at Friedman's 90th birthday celebration that 'you're right. We did it. We're very sorry. But thanks to you, it won't happen again' (quoted in Krugman, 2009). Moreover Robert Lucas in 2003 declared that the developments in knowledge about monetary control since the Wall Street Crash meant that the entire 'central problem of depression-prevention has been solved' (Ibid). As will be discussed later in the chapter, this view of crisis-management is highly contested by the heirs of Keynes, especially when interest rates hit the Zero Lower Bound, but regardless has been perhaps the dominant stream of economic thought amongst economic-policy circles in the USA and beyond for a great majority of the post-Depression era. The stagflation period of the 1970s and the failure of the post-war Keynesian consensus to make sense of it gave these views particular pertinence in the Great Moderation, favouring minimal government interference in the economy and the use of monetary policy to mitigate downturns as the most efficient mode of macro-economic management.

The Keynesian lesson from the Great Depression was somewhat different. Following Keynes' ground-breaking 1936 publication A General Theory of Employment, Interest and Money, and to an extent also built on the work of American economists Irving Fisher and Marriner Eccles, the role of an activist fiscal policy was promoted as central to preventing mass unemployment and a permanent loss of output resulting from Depression, replacing the prior consensus that governments must seek always to balance their budgets in a depression (see Fisher, 1933). Friedman's comment that "we are all Keynesians now" (albeit "only in once sense") highlights the support that this theory developed in the late Depression and post-war eras. Keynesians did not suggest that the monetarist solution of massaging the money supply was pointless, but rather that it was unlikely to solve a crisis alone. Activist fiscal policy would actually pay for itself in a deep depression because the

multiplier effect of government money in getting people back into work, paying taxes, consuming and hence stimulating further job creation actually saved more money; something the Depression appeared to verify (Almunia et al, 2010).

Unlike neo-classical economists, Keynes did not believe economies to be self-correcting, believing that a cycle of decline could set in as job losses reduced consumption, hitting profits and hence leading to further job losses. This was supported by Fisher's notion of debt deflation; in response to those who feared sovereign insolvency if they took on more debt to pay for an economic recovery, Fisher argued that the cost of doing nothing was greater because deflation in the economy increased the relative debt burden of the country (Fisher, 1933). The viewpoint that Keynesian stimulus spending is vital in a downturn has unarguably been a dominant force in economic policymaking ever since, despite Keynes' complaint that 'it seems politically impossible for a capitalistic democracy to organise expenditure of the scale necessary to make the grand experiment that would prove my case – except in war conditions' (quoted in Renshaw, 1985: 231).

The uncertainty that surrounds the best way to interpret the events of the Depression, and indeed the debate over what exactly stabilised the economy, highlights that even with the benefit of hindsight, the events of economic shocks produce a great deal of uncertainty and contestability. Interpretations of them inevitably highlight different levels of ideas that commentators have in regards to political economy, meaning that the debate, like any meta-theoretical one, can never truly be 'won' empirically. This sets the scene for the same contestability being applied to our own time period, and therefore contextualises the discussion later in the chapter of the different economic interpretations of fiscal necessity in relation to the Global Financial Crisis.

Whilst having more radical (and therefore generally less influence in policy circles) approaches to the lessons of the depression, Austrian and Marxist scholars have highlighted two further arguments of macroeconomic matters, both of which have been re-applied to our own crisis. The Austrian school maintain that economies are 'perfect' so long as they are not influenced by government (Giménez-Roche, 2011: 17-18). They take quite the opposite approach to the monetarists in suggesting a complete abolition of central banks such as the Federal Reserve, in the belief that any government interference in the economy will only act to distort the natural impact of a free market, hence making crises more common and dangerous, rather than less. This happens because whilst in a free market information is equally available to all, central planning distorts this, removing 'any semblance of economic efficiency' (Parsons, 2003: 6). Marxists have highlighted how the 1929 Wall Street Crash merely re-enforces Marx's original argument in Capital (and later, Hobson's) that capitalist economies were naturally crisis-prone and tended towards crisis as a result of their inherent make-up. Furthermore they doubt the real impact of Keynesian stimulus spending on the ability of governments to kick-start a troubled economy, highlighting that it took the Second World War to truly generate growth and recovery both in Europe and America (Foster and Magdoff, 2009: 107-108; Mattick, 2011: 54-55). They therefore ironically actually shadow Keynes' own concerns quoted above that democratic states do not realistically have the ability to adopt a stimulus package large enough to have a strong enough effect on the economy.

This section has provided a broad overview of the two central economic approaches which will be identified throughout the empirical chapters below. It has highlighted the importance of recessions and depressions to their development, and indeed demonstrates that it is in these particular episodes that the two theories are at their most distinct. Indeed theoretically they are perfectly compatible in their views of public spending during economic growth periods; as Keynes suggested, "the boom, not the slump, is the time for austerity" (Quoted in Jayadev and Konczal, 2010: 37). They are both recession-fighting toolkits with an important intellectual history within Western states, and hence the discursive battle between the two in our modern crisis context is not surprising. Having provided an overview of their key ideas and conclusions from historical recession episodes, I now turn to the development of macro-economic thought in the Great Moderation to provide further context on the central policymaking ideas of the pre-crisis period.

4.3 The Great Moderation and the Events of the Financial Crisis

Moving on from the debates surrounding the lessons to learn from the Great Depression, this section aims to outline the developing macroeconomic norms of the pre-crisis period, now most commonly called the 'Great Moderation'. In particular it will focus on the development of the credit-based economy in a period of low inflation and the growing dependence on both American and British economies on housing-related debt and finance. This will set the context for the economic theories of crisis-solution which follow, and highlight some of the central empirical links to both the causes and outcomes of the crisis.

The multiple economic and political crises of the 1970s, and the ways in which they were resolved, lead to a rapid change in the composition of the Growth Model in both the UK and USA. As ideologically alike leaders in the two countries led similar transformations, the development of the economic conditions which would ultimately lead to our current financial crisis were unknowingly put in place. This section highlights these transformations. Following a period in which Keynesianism generally dominated the field of macro-economy (although it is questioned to what extent the British and American economies of the post-war period truly were Keynesian to any real extent), the onset of the 1980s led to a period of dramatic changes in the ways in which the global, and in particular the Anglo-American, economies were regulated. The period of greater freedom of capital movement, greater financialisation of the economy, and greater operational independence for central banks with specific inflation targets is now most commonly called the Great Moderation.

4.31 The Leverage Economy

The Great Moderation saw a period of relative decline in manufacturing in the west, as multinational businesses increasingly took advantage of cheaper labour markets and cheaper transport costs to transfer manufacturing to developing countries. As such much of the economic growth in the west in this period was dominated by financial innovation as opposed to rising internal productivity growth. The forces of finance therefore became somewhat distanced from the actual productive capacity of the nation as a whole (Minsky, 1993: 112). There is much academic debate about the lines of causation that led to this situation. Most developed are the proto-Keynesianism of Hyman Minsky and the Marxist interpretations. Minsky highlighted how opportunities for investment and profit diminished over time during a budget cycle, leading to a greater dependence of debt and credit to fund further capital investment. This 'financial instability hypothesis' suggested that financialisation was an inherently unstable economic development, both resulting from and leading to less and less stable endeavours to find profit (Minsky, 1977; 1982). Eventually a crash becomes inevitable (i.e. it is systemic, not accidental to capitalism), as hedge and speculative finance give way to Ponzi finance as returns diminish (Minsky, 1977: 25). Unless government spending offsets it, 'the decline in investment than follows from a reluctance to finance leads to a decline in profits and in the ability to sustain debt' (Ibid.). In fact, for Minsky the capacity of the state to offset depressions by using active fiscal policy has increased in the modern age as the public sphere has become larger, and therefore more able to take up the slack presented by a shortfall from the private sector (Rochon, 2003: 137).

The Marxist interpretation is based on a similar understanding of the tendency towards crisis as Minsky's hypothesis. Financialisation, for Marxists, is an attempt by capital to overcome stagnation in the real economy. But as the housing bubble that developed in the Great Moderation shows, 'it is still dependent on its fortunes' (Foster and Magdoff, 2009: 83). Whilst the post-war period saw a development of monopolies as profits diminished (Baran and Sweezy, 1966), highlighting the overcapacity in the production system, so our current crisis stems from a financialised version of the same tendency (Foster and Magdoff, 2009: 63). Whilst linked to stagnation, this financialisation is in turn a result of the general decline in the purchasing power of the working class, forcing them to turn to credit to maintain consumption levels (Ibid.: 27-28).

Whilst credit has clearly played a larger part in the Anglo-American economies in the post-1979 period, the importance of specific financial innovations in spawning our current financial crisis should not be under-estimated. A period of strong economic growth and rising house prices spawned an era of banks allowing mortgages to those who would previously not have been deemed credit-worthy, so called sub-prime borrowers. This, combined with innovations in which mortgages were bundled together and sold to investors, so-called Collateralised-Debt Obligations (CDOs), meant that the wider financial community became tied to the apparently ever-rising US household market. This process was first adopted by Fannie Mae in 1938 (Thompson, 2012), but the power of

computers allowed for a far greater pace of development of this type of securitised debt in the modern age. Institutional norms are often at least partially a result of the crisis that led them into being, and the development of this highly liquid mortgage-based debt circulation was ironically a response to earlier financial failures. The Savings and Loans crises of the 1980s and 1990s saw a number of lenders go bust because they had kept a large number of mortgage loans on their books. Such illiquid debt was therefore seen to be dangerous to lenders, and securitisation allowed the distribution of loans to those 'better able to shoulder the risk…lessen(ing) the risk of a systemic banking crisis' (Roubini and Mihm, 2011: 64-65).

Whilst in theory it was believed that the packaging of mortgages into financial instruments spread risk out throughout the financial system, the increasingly convoluted chains of debt constructed became 'impossible for anybody to fit...into a single cognitive map' (Tett, 2009: 299). The corollary then was that banks increasingly lost the knowledge of the true value of the assets they were holding, buying and selling, and that growth in the wider financial system became fully dependent on confidence that house prices would increase eternally. The repeal in 1999 of the Glass-Steagall Act in the USA, which had previously separated commercial and investment banks only heightened the diffusion of mortgage-related debt into the banking sector and meant that any crash would have dangerous consequences for the wider economy. In essence, therefore, this constituted a classic bubble, which the wider economy had become dependent on. With debt and credit flowing without viscosity, a wider consumer-led growth model developed based largely on this private debt (Hay, 2011: 4).

4.32 Current Account Imbalances

The developing global aspect of trade flows and current account imbalances should not be underestimated either, something that has greatly enhanced the capacity of crises to spill beyond national borders, certainly more so than was possible with the Great Depression. The West's growing dependence on debt and credit was offset by a 'savings glut' in East Asia, which led to China and Japan developing huge current account surpluses, which directly influenced the current account deficits run by both America and Britain. Economic theory suggests that large imbalances in current accounts are unsustainable over a long period of time, as debt cannot be rolled over indefinitely without investors becoming concerned about the capacity of the economy to continue to supply a solid return on any such investment. Crisis is an inevitable consequence of such imbalances as investors become spooked and the absence of capital controls allows for capital flight which can wreak considerable havoc to an economy, as the Asian Financial Crisis of the late 1990s highlighted. Whilst the importance of sterling protects the UK from quite the same level of these dangers despite high current account deficits, there is some uncertainty about how long foreign

investors will continue to support their deficits, something that the orthodox school has paid particular attention to in its call for spending cuts.

4.33 Monetary Consensus

The period of the Great Moderation also saw remarkable consensus on several elements of macroeconomic governance. In monetary policy, a consensus developed on the intrinsic worth of an autonomous central bank with power to manage inflation. The market liberalist view of the state came to the fore here, with the belief that monetary policy would be more consistent, and hence a country more stable and inviting to foreign investors, if much economic power was taken away from partisan politics. Whilst taking such power away from politicians was deemed to reduce the likelihood of abuses to the power in the search for electoral gain, the handing of power over to a technocratic elite challenged the sense of accountability and the potential for diversity of opinion on economic matters. Much has therefore been made in the academic literature of the epistemic communities which sprung up in central banking circles and the danger of group-think that this entails (amongst others, see, Bean, 2007; Bowman et al, 2012; Gieve and Provost, 2012; Lawson, 2009; Marcussen, 2006, 2009; Ozgercin, 2012). Moreover the confidence in the power of finance to generate economic growth and the importance attached to gaining the trust of the City of London and Wall Street meant that banks developed a strong ideological hold over economic policy during the Great Moderation (Bell and Hindmoor, 2015; Johnson and Kwak, 2010). Moreover it has even been argued that the overlap between finance and academic economics became too great, with professors having too much interest in the vested interests of the status quo to provide sufficient critical understandings of the potentially dangerous effects of the existing governing orthodoxy (Carrick-Hagenberth and Epstein, 2012).

The lack of confidence in politicians to be trustworthy with national finances and to take decisions of long-term sustainability over short-term potential electoral gain spilt over into fiscal policy also. Academic work focused on the 'electoral business cycle, highlighting how politicians artificially created booms before elections, hence skewing economic growth from its 'natural' trajectory (Alesina et al 1989; Alesina et al, 1993). Economic credibility therefore became a central electoral tool in the Great Moderation, with governments aiming to show that their economic decisions were externally supported by experts, and not skewed towards creating artificial short-term bubbles for electoral gain. Fiscal 'rules' therefore developed in popularity, most famously represented by New Labour's two 'Golden Rules' for fiscal policy which were implemented upon coming to power in 1997 (See Clift and Tomlinson, 2012 on the development of such rules

historically). These rules intended to ensure that not only were the public finances sound, but that 'spending and taxation impact fairly both within and across generations...those generation who benefit from public spending also meet, as far as possible, the costs of the services they consume (Budd, 2010: R35). The first golden rule, intended to prevent governments racking up large budget deficits to support growth, allowed the government to borrow only to invest, with the second limiting the debt/GDP ratio to 40% (IFS, 2001: 2; Hodson and Mabbett, 2009: 1047). It was believed that by placing tighter restrictions on government borrowing and spending, that respectability would be gained by international finance and that the nation's finances would be put on a permanently sustainable footing. As with the forecasting of ever-increasing house prices however, the crash of 2008 highlighted that these rules worked only as far as the economy was growing; much structural public spending was constructed on tax receipts that were in hindsight only cyclical, leaving the UK with a large budget deficit when the economy entered recession.

4.4 The Onset of Crisis and its Political Response

'By the spring of 2006, the financial system, with its extraordinary reliance on leverage – and its blind faith that asset prices would only continue to rise – was primed for a breakdown on monumental proportions. Financing increasingly depended on the sort of speculative and Ponzi borrowing that Minsky predicted. Euphoria that began in the housing sector and percolated upward through the entire financial system only encouraged further risk taking, and the few sceptics who raised the alarm were not heard' (Roubini and Mihm, 2011: 88).

This 'Minsky moment' of the financial crisis was actually somewhat of a prolonged affair, with the slow increase in sub-prime delinquencies developing through 2005 and 2006. Lenders in the shadow banking sector began to see their credit lines tighten up as many of their assets went bad. By March 2007, over fifty of these relatively small institutions had gone bankrupt (Ibid.: 90). Slowly, however, the larger institutions began to realise that this was no longer a problem contained in one small section of the industry, but a wider fracture that threatened the entire system. Lenders rushed to try and prop up the divisions of their companies that were most clearly exposed, but this only signified the famous metaphor of shutting the stable door after the horse has bolted. Bear Stearns eventually collapsed in March 2008, with many of its assets taken over by JP Morgan Chase. The Fed began operating as lender of last resort to key institutions such as Goldman Sachs and Morgan Stanley, but provided a central sign of its unwillingness to support the entire edifice by allowing

Lehman Brothers to enter bankruptcy in September 2008. This in turn led to AIG being nationalised as its assets were downgraded. In the UK, the first sign of the knock-on effect of the events in America was the distress caused to the Northern Rock bank, prompting the first run on a British bank since 1866 (Shin, 2009: 102).

This is necessarily a very reduced history of the onset of crisis, for more relevant to our own discussion is that of the impact of it on government debt and borrowing, and the politics of the response. The UK acted to insure bank deposits at the beginning of October 2008, with Washington's Federal Deposit Insurance Corporation (FDIC) insuring a swathe of new debt from selected lenders later in the month, effectively allowing them to borrow and lend at lower rates. The aim at this stage was to prevent the entire credit chain from drying up and to give banks confidence in each other's capacity to repay debt by opening emergency credit lines and lending mechanisms. The US government was also forced to offer emergency aid to Fannie Mae and Freddie Mac, which held a great deal of mortgage related debt. Late 2008 saw governments take an unprecedented amount of debt on board to keep the ailing banking sector afloat. In July, the US Government passed the Housing and Economic Recovery Act, which pumped \$320 billion into helping mortgageowners refinance, and followed this up with the notorious Troubled Assets Relief Program (TARP) which aimed to buy up \$700 billion worth of banks' toxic assets to help clear their balance sheets, and to buy shares in the troubled institutions. With little public accountability, sovereign governments vastly increased their role in the banking economy with such actions. The UK's equivalent of the TARP package allocated £500bn for a combination of re-capitalisations, loans and loan-guarantees, although focused less on the purchasing of toxic assets than did TARP.

The impact of this on budget deficits was understandably huge. Having had a surplus during the first New Labour term in office, Britain had been running a deficit of £32bn in 2006 and £36bn in 2007. This increased to £69bn in 2008 and £156bn in 2010. Public debt-GDP increased from 36% in 2008 to 52% by 2010. US government debt, already high from the Bush administration's combination of spending increases and tax cuts, increased from 69% to 93% over the same period. Net tax revenue for the UK government fell from £514bn in 2007-09 to £500bn in 2008-09 and £485bn in 2009-10. Had it kept up with the pre-crisis trend for these years however, it would have reached around £575bn in 2009-10, i.e. the real drop was actually somewhat closer to £90bn.

The US budget deficit, which stood at \$161bn in 2006-07 (down from \$412bn in 2003-04), rose to \$459bn in 2007-08, \$1.41tr in 2008-09 and then dropped slightly to \$1.29tr in 2009-10. Tax receipts dropped from \$2.56tr in 2006-07 to \$2.54tr in 2007-08, then to \$2.10tr in 2008-09 and then \$2.16tr in 2009-10. Again though, whilst this fall accounts for half of the increase of the budget deficit, taking into account the pre-crisis trend growth it falls around \$1tr short by 2009-10 of the expected revenue for this year, accounting for a great deal of the swollen budget deficit.

Both countries essentially responded to the downturn by implementing some level of stimulus package. In the USA this equated to 4.8% of GDP in the 2008-10 period, whilst in Britain it was lower at 1.5% (Schirm, 2011: 48). In Britain, measures taken included a reduction in the Value Added Tax (VAT) on sales, the impact of automatic stabilisers including unemployment benefits

and a series of smaller schemes targeted at specific industries. In the US, the defining fiscal feature of Obama's first term in the presidency was the American Recovery and Reinvestment Act (ARRA), passed in February 2009 which set aside up to \$787bn to help stimulate an economic recovery. It was estimated by the Congressional Budget Office (CBO) that this created or saved around 3.6million jobs by 2010 (CBO 2010), with the money going to a variety of infrastructure projects, as well as a considerable proportion going to tax breaks.

4.5 The Economic Logic of Stimulus in the Current Context

The Keynesian argument is based on analysis of several core tenets of our current economic situation: the importance of debt; the role of stimulus in open economies, and concurrently the potential size of multipliers; the link between private and public spending and the danger of 'crowding out'; the need for creating confidence in the business community; and the power and scope of monetary policy.

Starting with stimulus and multipliers, the general Keynesian argument of today has in many ways stayed remarkably similar to those first professed almost eighty years ago. The metaphor of 'one person's spending is another's income', (Irvin *et al*, 2010: 12) representing the individual but also the public-private levels, remains at the core of Keynesian logic. Highlighting the central problem of the recession as being one of a dramatic shortfall in private expenditure, the fact that an economy is composed of public and private spending means that for an economy to remain steady then such a shortfall in private spending must be compensated by a Newtonian equal-and-opposite increase in private spending.

Stimulus packages in both the UK and USA were criticised from this perspective for not being large enough (Stiglitz, 2010: 63; Taylor et al, 2012: 189), representing only a fraction of the private shortfall, and hence lacking the capacity to return these economies fully to growth. It remains the assumption therefore that government spending at a time of crisis has the potential to have a large multiplier effect, in essence that spending can pay for itself because of the impact of increased employment, spending and tax revenues that they have, which in turn creates new employment, spending and taxes. Whilst it is noted that the increased openness of economies now compared to the 1930s means that multipliers may not be as large (Armingeon, 2012: 546) in that the money props up foreign economies more due to an increased spending on products which involve an international division of labour, this is taken by the Keynesian case to mean that stimulus should be international more than national in character (e.g. IMF, 2009: 3-4).

Likewise, austerity as a method of becoming more competitive so as to resume growth is doomed to fail if everyone mirrors each other (Van Reenen, 2012). Moreover, the nature of a financial crisis means that multipliers will be larger than during 'normal' economic circumstances (Auerbach and

Gorodnichenko, 2010; Christiano et al, 2009; Corsetti, 2012; Corsetti et al, 2010; DeLong and Summers, 2012; Eggertsson and Krugman, 2012; Jordà and Taylor, 2013). Blinder and Zandi, for instance, argue that without government intervention, US GDP in 2010 would have been 11.5% lower than its actual level (Blinder and Zandi, 2010: 1)

The danger of 'crowding out' private investment is likewise disparaged (Coates, 2012: 152; Eggertsson and Krugman, 2012: 1491), again using much of the same debates on Say's Law, which argues that supply creates its own demand (Sardoni, 2002) and Ricardian Equivalence that Keynes himself focused on. Most stridently, this case was put forward in Paul Krugman's initial response to the crisis and how economists had 'got it so wrong' in believing that many tenets of Keynesian economics no longer applied (Krugman, 2009). The Ricardian argument holds that 'public sector profligacy can be fully offset by private sector prudence if economic agents correctly anticipate that future tax liabilities will rise as a result of a fiscal expansion. It then follows that the contractionary consequences of a fiscal retrenchment will be offset by an increase in private sector spending as 'economic agents correctly anticipate a decline in future tax liabilities' (Islam and Chowdhury, 2010).

In essence therefore any short-term boost to GDP provided by a stimulus will be cancelled out by a medium or long-term decline of equal proportion (Ulrich, 2011: 153). The Keynesian argument either rejects Ricardian Equivalence and Say's Law out of hand (as Krugman generally has), or in a milder form suggests that whilst government spending could lead to reduced output in the future, the nature of such a deep financial crisis means that the long-term reduced output will be smaller than the short-term increase (NIESR, 2009: 43 Stiglitz, 2010: 73). This is not to say that all government spending will be of equal value; capital spending is deemed to have the highest return, and tax cuts a notably lower multiplier (Holland *et al*, 2010: F12; Pollin, 2012: 161).

The Keynesian argument maintains that economies are not made up of perfect, rational, self-calculating agents, but rather are dominated by the 'animal spirits' of confidence and the herd mentality (Keynes, 1936: ?). This applies in busts as well as booms, making the creation of business confidence central to getting private investment back to pre-crisis levels (Irvin *et al*, 2010: 9-10). For Keynesians, the confidence created from a growing economy which knows it will be supported by the state if necessary is far greater than the confidence created by highlighting a willingness to cut expenditure at the expense of anything else (King et al, 2012: 9). By acknowledging that in the modern state the private and public realms are far more inter-connected that is generally believed, the ability of the state to take up the slack of the private sector and to underwrite some of its gambles is the primary developer of confidence.

But the issue of public debt cannot be completely ignored. The Keynesian position on the gross size of the national debt is generally to highlight how it began the crisis in a historically low position, and therefore has the capacity to increase without necessarily leading to solvency crises. The potential for debt to curb economic growth is doubted though, with a '1 per cent of GDP rise in government debt to GDP ratio' being associated on average with only a '2-4 basis point rise in interest rates (Bagaria *et al.*, 2012: F51; See also Pollin, 2012: 161). Furthermore due to the

methodological basis of this study, these figures 'overstate the impacts for non-Euro Area countries', such as the UK (Ibid.). Either way this increase in itself is vastly outweighed by the impact that Quantitative Easing has had on interest rates, with the measures taken by the Fed alone in the aftermath of the crisis estimated to have reduced Treasury yields by around eighty basis points (D'Amico et al, 2012: 29). DeLong estimates it would take debt yields to normalise above 7% for stimulus to become self-defeating (DeLong, 2012), something that central bank independence and the success of QE makes highly unlikely in the UK. The notion of debt-deflation means that increasing government debt and deficits is not a sufficient cause for public sector cuts in the Keynesian viewpoint. Rather, it is growth that is best placed to reduce the debt burden of the economy, re-affirming the need to invest (Taylor et al, 2012: 201). A restoration of growth and a period of moderate inflation provide a template of how to work down debt in the future (Dornbusch and Draghi, 1990: 8; Krugman, 2012).

Whilst monetary policy has the power to limit as sharp a decrease in the money supply and hence the depth of the recession, and as shown above has the power to decrease government debt yields, it is not seen as sufficient for a recovery to Keynesians. Whilst both the Bank of England and the Fed quickly and dramatically slashed interest rates in the wake of the crisis to attempt to reduce the incentive to save and to encourage investment, this has not been sufficient to spur the economy back into growth (Blanchard et al, 2010: 205). Indeed it is estimated that in the US, a further cut in interest rates of 3-5% was necessary to stimulate sufficient investment (Rudebusch, 2009), something which would require negative interest rates to achieve. This introduces the concept of the 'zero lower bound' where monetary policy loses the ability to stimulate growth – typically where interest rates have gone as low as they can. The addition of QE into monetary policy complicated the calculation of where the lower bound existed, but the Keynesian standpoint suggested that by 2010 this has been reached (Blanchard et al, 2010; see also Bernanke, 2009).

Finally the Keynesian analysis posits that governments are not as strait-jacketed by the forces of globalised finance as is often suggested (Aoki, 2002; Boyer and Saillard, 2001; Boyer, 2012; Lodge and Hood, 2012: 79; Posner and Blöndal, 2012: 32); the variety of responses to crisis undermines the argument that governments had little operating space in terms of fiscal policy (Armingeon, 2012: 544). The economic logic for such an approach is therefore multi-pronged: lack of belief in higher debt stocks as a burden in terms of interest rates; belief that monetary policy has reached its limit; belief that government spending will initiate growth, and will not crowd out private investment; belief that this growth will be the real spur to business confidence; and finally a statement that states are not forced by the markets into a single one-size-fits-all macroeconomic strategy.

4.6 The Economic Case for Austerity in the Current Context

In the same way that a Keynesian case for stimulus is based on a variety of understandings of the current context and the relevance of classical economic literature, so does the case for austerity.⁸ Generally they take the opposite view on a number of the central tenets of Keynesianism highlighted above, but are worth dwelling on in greater detail than this statement suggests.

The core logic of the case for austerity is therefore based on: a fear that high debt levels will hamper growth in the future; the fear that market confidence will be lost by failing to immediately tackle spending; the suggestion that spending today will provide an unfair burden on future generations; that the level of spending needed is not achievable or sustainable; the belief that government spending crowds out the private sphere, and that cutting is therefore not as contractionary as Keynesians suggest; and a belief that innovative monetary policy can produce a recovery without fiscal interference.

This 'orthodox' approach takes a very different opinion on the debt burden that the financial crisis has created. With budget deficits in the UK and USA even during the boom times before the crisis, the central argument of the 'austerians' is that this level of spending was clearly unsustainable, and indeed a great amount of it became structural spending (as opposed to cyclical). Growth alone will not therefore solve the problem of governments spending beyond their means, therefore deep cuts need to be made into the existing social contract in terms of what the state can seek to provide for its citizens; 'debt cannot be rolled over indefinitely' (Ulrich, 2011: 160-161). This lack of sustainability is heightened by the rapidly aging population in western states, meaning that future funding problems already exist due to the ratio of tax-payers to dependents falling over the next generation and beyond. Further debt is therefore unwise now, as Western states already were perceived to need to start stockpiling for future funding issues (OBR, 2011: 3).

But of course if you were to accept the Keynesian growth and debt argument, then this alone would not be a case for austerity. The orthodox approach therefore holds very differing views of the expansionary capacity of fiscal stimuli, and of the capacity of such a stimulus to reduce the debt burden. The true impact of multipliers has been highly contested itself academically during the current crisis, with those who have supported austerity arguing that the multiplier from increased spending is low because of crowding-out effects (De Soto, 2011: xvi; also Gros, 2011), and as such the negative multiplier of reduced spending is low also (Alesina and Perotti, 1994; Alesina and Ardagna, 2010; Perroti, 2013). Ramey (2013), using US data, has suggested that multipliers may have been anywhere between 0.4 – 1.5 in the current crisis, which respectively would make fiscal stimulus either fairly worthless, very valuable, or anywhere in between.

In its more extreme form, however, it is suggested that contractionary fiscal policy can actually have an expansionary impact. Based on Say's Law, it is assumed by the orthodox school that 'the

⁸ I will leave it to the discretion of the reader to decide which they consider more compelling...

decrease in government spending lowers aggregate demand...and output decreased. The aggregate supply curve also shifts backwards, reducing the prices of inputs and lowering the level of inflation. In the medium term, due to the economy-wide deflation, productive resources become more attractive as investment options and so investment increases...There are multiplier effects on this increase in investment', which will raise output further (Kinsella, 2012: 230). In other words, 'debt-financed spending can't have any more effect that spending financed by raising taxes', negating any benefit as people alter their behaviour in response to such taxes (Cochrane, 2009: 4).

Again, the true worth of government spending to influence a recession is dependent on the validity of applying Say's law and Ricardian Equivalence in practice (Sawyer, 2012: 208). Whilst the belief in expansionary contractions began with the monetarist revolution of the 1980s, in which a context of high interest rates, stagnation and high public spending were important in creating the conditions for its success (Giavazzi and Pagano, 1990), later examples from the 1990s of Sweden and Canada running successful fiscal consolidations were used in the current era to provide historical support for the theory. A systematic analysis of all fiscal contractions in the 1970-2005 period brought about uncertain results, however (Afonso, 2010), highlighting further the uncertainty regarding the potential outcomes of different policy approaches.

Whilst those in favour of austerity have rarely argued that it provides a complete objective truth and that all cuts will pay for themselves (Cochrane, 2009: 4-5), it has generally been assumed by this school of thought that the multiplier effect of cuts in government spending will be considerably lower than 1. This more common middle ground accepts that Keynesian stimulus might have some beneficial impact on the economy, but that in a globalised world it is too difficult to do so in a way that doesn't merely prop up your competitors, or that it will not be strong enough to pay for itself, with the negative of extra debt and the danger of this in terms of interest rates outweighing the positive of marginally stronger growth.

The zero lower bound of monetary policy, highlighted by Keynesians as the point where fiscal multipliers become most effective, is also confused by recent monetary developments. The innovation of central banks through the crisis, seen most notably through the scheme of Quantitative Easing (QE) but also through a number of other lending facilities and liquidity measures, makes the identification of a lower bound particularly challenging (Alesina and Giavazzi, 2013: 4). As such, for the orthodox school, the potential benefits of stimulus are more uncertain, and the damage more serious, than for Keynesians. The danger relates to the government's medium term debt accumulation and the willingness of markets to finance this.

Debt and interest rates therefore remain primary concerns. The work of Reinhart and Rogoff (at the time accepted but now ridiculed⁹) highlighted a belief that a debt-GDP ratio of over 90% would begin to produce a drag on growth (Reinhart and Rogoff, 2010; for a critique see Baglan and Yoldas, 2013). Moreover the uncertainty over market forces and the impact of growing debt on interest rates and gilt yields proved central to the argument (Ulrich, 2011: 160). In the absence of

⁹ It later emerged that the statistical significance of the results was dependent on a mistake at the data entry stage.

clear empirical results showing that interest rates would not rise, the growing debt burden impacting much of the western world meant that it was deemed safer in the orthodox view to err on the side of caution and provide a clear sign of intent to the markets that governments were serious about making cuts to expenditure in order to tackle debt. The most successful policies in this logic are not those that necessarily produce the most growth at the least cost, but those simply that are keep the country's credit rating intact and pacifies the markets. The slightly counter-intuitive nature of this argument has made it very contentious in academic circles; the empirical likelihood of these market forces leading to spiralling borrowing costs even in major economies like the UK and USA has been doubted by Keynesians, with Krugman in particular mocking both the fear of 'bond vigilantes' and the belief that the 'confidence fairy' of sharp cuts in expenditure would be sufficient to deliver a return to growth (Krugman 2009; 2012).

But either way the gross level of debt is a primary concern. Even if it does not slow down growth prospects, the danger highlighted by proponents of austerity is that we endanger future generations by providing them with such a huge debt burden. Whilst Keynesians aren't generally fazed by the size of this debt, highlighting how in the post-war period even bigger debts were paid down against a backdrop of persistently strong economic growth, the austerians doubt our ability to achieve this record twice. After all, the West, particularly America, was at the peak of its power during that period, with potential economic rivals such as Germany and Japan highly damaged by the War (Roubini and Mihm, 2011: 178). American manufacturing was in demand, providing a huge trade surplus which is no longer the case, with an aging population and the general feel of a nation that, whilst still powerful, is in decline. This is only magnified in the British case. Whilst previously the power of the dollar would have made almost any American debt level safe, perhaps now even the US cannot guarantee its own solvency permanently (Ibid.: 239).

4.7 Ideal Types and Their Political Usage

The above section has intended to highlight some of the central tenets of the two different 'ideal type' schools of thought. The different methodologies used by these different schools of thought allow both to present evidence in favour of their argument. Keynesian models tend to rely on vector auto regression (VAR) models, alternately using case studies to demonstrate that many expansionary contractions were based on export led growth that is not replicable to the world as a whole. The methodology of the orthodox school, led by Alesina and Ardagna, focuses on a comparison of key macroeconomic variables before, during and after a retrenchment to create a dataset of consolidations and their consequences (Perotti, 2013). However, detailed case studies are also used to differentiate between the likely impact of different types of consolidation (Ibid.). The

purpose here is not to discuss these different methodologies in detail and attempt to suggest that one is methodologically more rigorous or comprehensive than the other, but quite the opposite; whilst both can create evidence of the validity of their findings an essential contestability exists. Therefore it is suggested that the economic events of the Global Financial Crisis could logically be explained through either of these schools of thought; neither holds a monopoly on logic in relation to the principle methods of tackling the impact of the Global Financial Crisis. Following from this is the assumption that the policy response to the crisis could have followed either of these two paths, depending on which better fitted existing structures, preferences, discursive understandings etc. The decision of this thesis to focus specifically on the discourse surrounding the two to explain which better resonated with the lived crisis experience has already been justified in previous chapters. The two ideal types rely on different visions of economic imperatives, pathologies of the crisis, and views on economic growth. The essential differences can be summarised along three central and inter-linked themes:

4.71 The Role of Uncertainty

Neither theory doubts the importance of uncertainty in shaping economic events, but they have different views on the way in which it creates economic priorities. From the Keynesian position the central problem in the crisis is the lack of confidence that businesses have to invest again and start recruiting. The longer that economic growth is stagnant or retarded, the worse this will become unless new investment opportunities appear. The role of government economic policy is therefore to help produce investment projects which puts money back into the real economy and hence kick-start this confidence. The knowledge that the government will act as the stimulator of last resort means that businesses have the certainty they need to invest, not having to worry that the economy could potentially continue in a permanently depressed state which would ruin such investments.

However the principal role of uncertainty from the orthodox perspective relates to the willingness of financial markets to fund such government debts. Given the importance attached to transnational finance and the belief that it will cease to fund government deficits permanently, the central issue is how and how large deficits can become before the 'electronic herd' loses confidence in a government's credibility. When this occurs, the high rates of interest of debt the government will have to pay are perceived likely to far outweigh the aggregate demand benefits of greater government spending. As such, the state is perceived as fragile in a globalised financial system, and therefore is forced to play by the rules of a game it is not in control over. How different approaches to uncertainty were understood, and were given precedence over others, provides an important tool in the constructivist understanding of crisis resolution and discursive contestation. The analytical chapters that follow will highlight that fear of government over-spending was prevalent in pre-crisis

discourse in the UK, and the failure of the crisis to alter these important perceptions helped to provide strategic strength to the austerity discourse.

4.72 Stimulating Growth

Likewise the two schools take different positions on growth, which link into this element of uncertainty. From the austerian viewpoint, government debt will produce fear in the business community that it will be taxed higher in the future, preventing investment and growth. The impact of high debt on growth might therefore be a secondary as opposed to a direct correlation of it, but it is significant nonetheless. Moreover, it is impossible to increase spending and hope to cut debt within this viewpoint, because multipliers will not exceed 1 and some of this economic growth will be offset by an increase in interest rates that stem from market concerns about the permanence of such spending. Stimulating growth therefore comes primarily by altering people's medium term expectations on tax levels. A belief that medium term taxation will be lower increases short-term expenditure and investment, and the major way to develop these expectations is to signify a reduction in the size of the state. Alongside this, innovative monetary policy can aid investment through low interest rates.

Overall, then, precedence is placed on supply-side intervention. But from a Keynesian position, the drag on growth comes not from tax expectations but from aggregate demand; if the demand isn't strong enough in the economy then businesses won't invest. Hence the short-term priority is to artificially increase demand through government spending. Cutting spending will always entail a negative multiplier in a multi-national recession, producing a drag on growth which will have a very limited effect on the total debt burden anyway. The scope of short-term fiscal stimulus leading to noticeably higher interest rate payments is doubted in this view, at least in the case of Britain with its monetary independence. Monetary policy is believed to have lost its traction in the case of a recession, meaning that fiscal policy is required to stimulate investment. The following analytical chapters highlight the prevalence of supply side thinking in the pre-crisis period, and again this was not radically altered through the crisis creating a greater discursive strength for the orthodox approach.

4.73 Economic Priorities

As such the two differ on the chief economic concerns plaguing the country in question. For the Orthodox school, the imminent danger is a loss of confidence in the ability of the country to finance its debts, with a sharp spiking in interest rates resulting from market concerns. This will hamper the more medium-term restructuring of the economy, and leave an over-burdened state which itself is dragging the economy down. No consistent growth can occur whilst the state delays the issue of signposting to the markets that reduction of the debt burden is its primary concern. Meanwhile the Keynesian position suggests that this reduction of the deficit cannot take place until solid growth is restored to the economy, and under-taking cuts in a depressed economy prevents this taking place. Business needs the certainty of government support and aggregate demand to persuade it to begin investing again and for economic growth to begin to take off again. Meanwhile the government has a duty to protect and create jobs in a time of depression, so that people do not suffer prolonged unemployment which will damage their skill-sets and have an adverse effect on future productivity. The concern over debt, and the danger of market forces on debt repayments, is hence seen in a very different light, at the same time a less pressing issue that jobs whilst also being better solved in the medium term by a solid growth strategy that comes from retaining investment during the crisis.

4.74 A Case of Contestability

The picture if therefore one of contestability; a solution could have been developed along the lines of either of these, or indeed somewhere in between, depending on a range of discursive factors. This study looks specifically at the discursive development in the public realm from 2007-10, analysing how these ideas were used, developed, understood and debated in order to justify the policy responses chosen, and hence how certain ideas gained primacy over others.

This contestability is demonstrated by the short-lived nature of some the academic work which the uncertainty of crisis produced. The study by Reinhart and Rogoff, cited above, was at the time influential in conservative circles in demonstrating the need to limit the level at which debt/GDP peaked. However, it emerged in April 2013 through the work of Herndon et al that the results on Reinhart and Rogoff came from a selective exclusion of data, but more importantly because of a simple coding error which skewed the results and that had gone unnoticed at the time. The result was that growth rates of highly indebted countries was on average estimated by Reinhart-Rogoff to be two percentage points lower than may actually the case. The question of reverse causation, i.e. whether it is slow growth that causes debt as opposed to vice versa, has also gone some way to discrediting their conclusions (Nersisyan and Wray, 2013).

The lack of consensus in the academic economic community therefore provides a palpable sense of the extent to which previously dominant paradigms were being questioned and how some of the most fundamental questions of macroeconomic policy were now becoming disputed. The lack of consensus in this epistemic community provided the potential for a more overtly public and political debate surrounding this central role of the state that had previously been the case, and allowed for new political coalitions to be built depending on how the issue was discursively narrated.

This chapter has therefore analysed the dominant economic paradigms of the New Labour era and outlined a hypothetical case of contestability to allow for the analysis chapters that follow to investigate whether the discursive playing field was as levelled by the uncertainty of crisis as the existing constructivist literature suggests.

4.8 Conclusions

This chapter has had two central aims. Firstly it has contextualised the Keynesian and Monetarist frames of economic understanding that are reflected in UK media discourse. Secondly, it has introduced key events of the pre-crisis economic context and the Global Financial Crisis itself. Both are referred back to in the commentary and analysis of the following empirical chapters. The purpose of introducing these two frames is also twofold. Firstly, it was from these frames that public narratives had to be woven, as will be seen in the following chapters. Secondly, however, it has sought to demonstrate that the two schools provide alternative but equally rational arguments for specific policy programs and priorities. It is on this basis that we can determine that economic facts alone do not necessitate a particular form of action, but it is through the ideas used to interpret, analyse, and weave these facts into popular discourses that a certain policy programme becomes appropriate and credible. The analysis chapters which follow take up this logic, in order to analyse the extent to which ideas changed over the course of the crisis, and therefore the extent to which popular pre-crisis discourses of the state came to determine post-crisis economic necessities.

Chapter 5: Charting the Pre-Crisis

Economic Discourse

5.1 Introduction

In order for a crisis to be said to have precipitated a change in the ideational understanding of economic events in relation to the state, the frames through which key policy issues are understood, and the central foci of prudent macroeconomic management have to be shown to have shifted over time. As has been outlined above, this work intends to question the extent to which crises necessitate such sweeping change by studying the processes of explanation that accompanied the global financial crisis. It focuses especially on uncovering points of continuity and change in ideas regarding fiscal policy instruments: the role of government; fiscal capacity; the drivers of economic growth; and tax policy. By doing so, the thesis questions the extent to which crisis in one realm (financial stability and monetary policy) can pervade others (fiscal policy) to form more comprehensive ideational paradigm shifts. More generally, the analytical chapters highlight that the analysis of the financial crisis from the British press cannot be unwound from the wider understandings that pervaded the pre-crisis era on the success of New Labour economic philosophy and governance. Thus a considerable amount of 'ideational stickiness' can be observed through the crisis. The purpose of this chapter is to aid the answering of research questions 1.) and 2.) by highlighting the predominant narratives of state and economy in the pre-crisis press so that the reversion to such ideas in the aftermath of the crisis can be most clearly analysed. It analysed the common frames of economic debate in the immediate pre-crisis period, and following on the methodology outlined in chapter 3 discovered central themes of discussion whose development through the crisis is tested in the following two chapters.

The findings of the chapter can be separated into two separate categories. Firstly, in relation to the role of the state in the economy, the legacy of a decade of New Labour governments had created an environment in which the media were sceptical about the purpose and economic and social value of further public expenditure. Here we find common discourses of an oversized state sector and a weariness at the perception that Labour's instinct was to 'throw money' at social problems without any ability to gauge the success of such spending. Thus the belief in a bloated and increasingly inefficient public sector was becoming ubiquitous. Further to this, the lack of trust in government to spend money and account for it accurately had resulted from criticisms of the 'spin' attached to all

policy announcements. The combination of a lack of trust in Gordon Brown's claims in relation to the likely success of his social expenditure, and the sense of an inefficient and over-bearing state sector swamped in 'red tape' led to a framing of reduced spending (and hence offering 'less' to the public) as being more honest, and thus more virtuous. Moreover, discourses of globalisation were used to highlight the relative weakness of the state in relation to external economic forces, suggesting that the role of government in relation to boosting economic growth was to increase incentives for investment on the supply-side, with little room for demand-side stimulation.

The second major category of findings relates to the extent to which crises can truly be considered 'shock' and 'unforeseen' events. A major discourse of the pre-crisis era, stemming from the above discourses of the oversized state, was one of the need for fiscal consolidation. In other words, the austerity discourse actually pre-dated the financial crisis in the British case. Whilst the depth of the crisis was not foreseen at this stage, throughout 2007 the sense of a likely downturn in the global economy led to a pre-diagnosis of the necessary British response from the broadsheet press. Given the belief that Brown had over-spent and made regulatory burdens too high on business, the result was a belief in the need for supply-side initiatives, a reduced tax burden and a cut in public expenditure.

The major empirical arguments of the thesis relate to the reversion to pre-crisis ideas in the post-crisis context. This chapter therefore demonstrates the above major discourses before returning to them in chapter seven to analyse their re-emergence. Whilst such pre-existing ideas need not necessarily have survived the crisis, the fact that it re-emerged with force once the height of the crisis had passed in 2009 suggests that the ability of crises to completely re-configure existing webs of understandings has to date been over-stated by the existing constructivist literature.

5.2 The Size and Role of the State

Political discourse on the role of the state in the national economy centred not only on the ability of government initiatives to produce positive social outcomes, but also on the more general theme of political trust. Here, the growing sense of political apathy and of a political process that was distant from most citizens, the product of which was historic low turnouts at general elections, was linked to the lack of trust people had in their politicians, partly because of the inability of politicians to be seen to speak truthfully on important issues. This lack of political trust was merged in the economic realm with the perception that politicians could not be trusted with public money, and furthermore could not be trusted to report back honestly on the success of their use of such money. This section highlights a prevailing trend of 'anti-statism' in the media's analysis of the state's involvement with the private sphere.

The first element of this discourse relates to the sense of deception that the media expressed as being central to any government announcement. Brown's reliance on people not claiming tax credits to balance the budget, and hence his ability to present a fiscally neutral budget without announcing tax rises on any social group, was seen as a wider signifier of his deceit and his having lost touch with voters through incessant 'spin' (Times, 2007f). Damian Reece in the *Telegraph* argued that 'The Tories are right to call Brown's latest mathematical myth a tax con not a tax cut' (Reece, 2007a). A dichotomy was drawn between the claim to be cutting taxes and the Budget as fiscally neutral. Thus the *Times* reported that 'the Chancellor robbed Peter to pay Paul, and then nicked it all back off Paul to compensate Peter. Peter and Paul are perplexed and their accountants are weeping over their calculators' (Times, 2007c).

The impact of the removal of the 10p tax band was relatively unnoticed, with the *Daily Mail*, *Daily Telegraph* and the *Times* actually complaining most forcefully of its regressive nature that would make one in five families worse off (Daily Mail, 2007c; Daily Telegraph, 2007c; Times, 2007d). The *FT* considered his final Budget fitting in that it 'displayed all the hallmarks of the past decade: a tendency to tinker, an assertion of Treasury power across government and headline initiatives that rely on sleight of hand in the small print' (FT, 2007f). The *Daily Mail* were even more blunt, calling the speech 'no better than a fib', but noting that the same could be said for David Cameron's response (Daily Mail, 2007d). Reece accused Brown of 'using his trademark fiscal fiddles and social engineering' (Reece, 2007a). An existing discourse of political distrust in relation to Brown's stewardship of the national finances was therefore present during this period, something which developed in the summer of 2007 when the media appraised the role of the fiscal rules more broadly as Brown's chancellorship came to an end.

The second major discourse related to an appraisal of Brown's chancellorship as a whole over the previous decade. The metrics of success chosen focused firstly on the lack of any major recession, for which he was deemed to have been successful (see Brummer, 2007a; Daily Mail, 2007e; Glover, 2007; Oborne, 2007), but secondly the overall social value of his increase in government spending, for which the outcome was much more mixed. A common caveat to the assessment of a relatively successful Chancellor was therefore that the net spending of the Brown years was becoming increasingly inefficient, that public services were therefore bloated and full of waste, and that cutting them back would be beneficial in allowing for lower taxes (Bootle, 2007; Brummer, 2007a; Daily Telegraph, 2007e; Johnston, 2008a). Edmund Conway in the Telegraph wrote that 'it has become increasingly clear that the public institutions have not been able effectively to absorb the massive influx of cash arriving in their accounts' (Conway, 2007a). Furthermore a moral element also exists here, in the sense that an overly-intrusive state, whilst being economically undesirable, also prevented individuals from having control over their own personal liberty. Extra investment in public services was no longer deemed to be producing beneficial outcomes, but was now being used as the cure to any negative social indicator (e.g. Conway, 2007a; Randall, 2007; Halligan, 2007a; Daily Telegraph, 2007a).

The FT continued to suggest that Brown's biggest failure was that he has never provided a convincing answer to the biggest question: has the additional spending been wisely used?' (FT, 2007g). The *Independent* used the example of youth unemployment to 'show how hard it had been to meet the laudable aims' set out in the 1997 Budget speech; 'spending state funds does not seem to help and for (Gordon Brown) who has an embedded belief that spending state money is the right way to tackle social problems, this must be a puzzle' (McRae, 2007b).

The corollary of such a discourse was therefore to question the extent to which government could micro-engineer positive social outcomes through spending increases, with the suggestion instead that better social outcomes could only result from a more individualist approach. The size of government was also directly linked to future economic potential because a rise in government spending was blamed for falling productivity, central to growth (Heffer, 2007; Daily Telegraph, 2007h, 2008a). Discussing Brown's record before the budget, Edmund Conway argued that spending too much was Brown's biggest flaw, with the danger of such spending not necessarily being a solvency issue, but one of 'excessive taxes and regulations' which had 'clogged up the arteries of commerce' (Conway, 2007a). From this flowed an assumption, then, that lower spending would actually produce greater medium term growth through a reduced tax burden on citizens and businesses.

The *Daily Mail* was similarly critical of the level of government spending that Brown had overseen, stating that it had 'exploded' in recent years (Daily Mail, 2007f). Again, at this stage the corollary of such spending was seen through problems of red tape for business and high taxation for individuals, rather than as an issue of national borrowing costs. The *Mail* complained on the eve of the 2007 Budget that 'his spending boom has pushed the tax burden to its highest level in two decades' (Daily Mail, 2007b). As such the perceived 'spin' of the 2007 Budget was reflected in Mail headlines reporting 'Brown's hidden £2bn tax grab' and 'forty new taxes hidden in Brown's budget' (Daily Mail, 2007g, 2007h respectively).

The growing expectation of a global economic slowdown over the coming year led to a third discourse emerging through the media, arguing that the state was powerless to counteract the forces of a global economic contraction, with activist policy therefore being deemed worthless. The sense of weakness in the face of an exogenous impact of globalisation and the impossibility of decoupling the British economy from problems in the American housing market meant that for the most part the media were pessimistic about the chances of any policy saving Britain from a recession. Whilst theoretically this evidence needn't have prevented a Keynesian stance from still being viewed as preferable, the agglomeration of helplessness and the analysis of Brown's fiscal legacy into a single frame which tended to be employed by the media meant that few policy alternatives were deemed worth the corollary of the extra debt they would entail. Thus we see here an important element of pre-diagnosis of the necessary response to crisis. Even in 2007, then, the media were outlining the acceptable responses to crisis based upon their existing normative assessments of Brown's economic policy. According to standard constructivist understandings, such beliefs would most

likely not withstand the shift in understandings once a crisis became apparent. The extent to which this was the case will be analysed in the following chapters.

This frame seemed to transcend political preference. Few in the media considered Britain's record decade of uninterrupted economic growth as being caused by Gordon Brown's stewardship from the Treasury, with most calling him 'lucky' to have had such a favourable global economic climate during his tenure as Chancellor (Guardian, 2007c) and with the perception of more constrained economic conditions on the horizon, judged that he had 'got out in time' (Guardian, 2007b). The *Guardian* (2008h) reported that the Budget would be 'overshadowed by growing fears for the economy', suggesting a disunity between the two. Larry Elliott described the government finances as 'maxed out' and the deficit as 'far too high for comfort' (Elliott, 2008a).

Whilst the paper cautioned against sticking to the 40% debt/GDP limit (Guardian, 2008a), it did report a developing consensus view in early 2008 that the fear of recession 'cuts Darling's budget options', giving him 'little money to give away' (Guardian, 2008b). The budget was nick-named the 'Mother Hubbard' budget, given that the cupboard was deemed bare (Guardian, 2008c). The paper reduced the government's options in the medium term to a dichotomy between 'having to raise taxes or cut spending after the next election to address the weakness of the public finances', implicitly redacting the option of doing neither and allowing borrowing to rise (Guardian, 2008g). This wasn't a universal discourse however, with columnists such as William Keegan highlighting the economic logic behind the Augustinian motto of 'make me virtuous, but not yet' (Keegan, 2008a).

Hamish McRae in the *Independent* likewise demonstrated the limited capacity for discretionary fiscal action by stating that 'we have all become aware that the big things that affect our lifestyles are not footling changes to the tax system but the seismic shifts taking place in the world economy' (McRae, 2008a). Steve Richards followed this up by stating that Darling had to 'keep his fingers crossed that proclamations of stability are matched by a period of steadiness', suggesting that the scope of the Chancellor to force the latter to follow the former was limited (Richards, 2008a). The *Daily Mail* and the *Times* called the Budget a 'gamble' (Brummer 2008a; Times, 2008a), with little inclination that the growth figures were to be believed. The *Times* added that things 'could only get worse', and that 'voters believe that Britain's economic position is affected more by the condition of the global economy than by anything the Chancellor does' (Riddell, 2008). Kaletsky offered their most scathing attack, calling Darling 'brave' for presenting 'the British economy's extraordinary stability and resilience as an established fact', when perhaps he should have been 'predicting or forecasting or hoping or praying' instead (Kaletsky, 2008a). The *Telegraph* suggested that Darling's talk of stability had a 'profound air of unreality about it', and was 'out of kilter with reality' (Bootle, 2008c).

This cross-spectrum discourse of relative powerlessness suggested that there was little Britain could do to actively alter the impact of the credit crunch on its economy. This reduced the discursive scope for a radical approach to the downturn involving activist fiscal policy to counteract the forthcoming private-sector contraction. Whilst an acceptance that borrowing might need to go

up did begin to emerge, this was seen more as a consequence of recession rather than a medium for conquering it. The capacity for the nation-state to actually 'solve' such an economic problem was therefore in doubt. Furthermore as the following section demonstrates, this developing sense of powerlessness was combined with a universal media discourse which argued that Britain's budget deficit left it with 'no room for manoeuvre' in fiscal terms to use a stimulus to counteract a recession.

5.3 Fiscal Responsibility and State Capacity

The issue of what was considered responsible in fiscal terms, and of the state's capacity in this realm, was discussed in some detail by the media in the aftermath of the 2007 budget, and again as Gordon Brown became Prime Minister in July, a *fin de siècle* moment that allowed the media to provide a full evaluation on his decade in the Treasury. What is striking in the analysis below is the consistency across the media spectrum of the sense that Britain was on the limits of being fiscally responsible because it was running budget deficits and getting close to breaking the second golden rule relating to government debt/GDP. This section therefore highlights a 'no room for manoeuvre' discourse which argued that fiscal deficits precluded the option of using fiscal stimulus to counter a recession.

In relation to the media coverage of Brown's stewardship of the public finances over the past decade, here already can be seen the historical antecedents of the deficit discourse of 2009-10. Whilst the sense of urgency in relation to the budget deficit is missing during this period compared to the following two years, there was a general consensus across the political spectrum of the mainstream media that the national finances had been mismanaged over the past five years and that in the medium term a good deal of consolidatory repair work was needed so as to place the UK on a more stable plane. The uncertainty regarding the medium-term timeframe, particularly in relation to the demographics of an aging population, was already causing some tension for the newspapers.

The *Daily Telegraph* and the *Daily Mail* had unsurprisingly been the most vocal critics of Brown's fiscal management during this period. They both worried that the increased size of the state was damaging for business and posed a threat in the case of a downturn. Liam Halligan criticised Brown's record as 'the biggest, most sustained increase in government expenditure in this country's history' (Halligan, 2007a), while Jeff Randall talked of the 'pile up' of public debt (Randall, 2007). The *Telegraph* columnists continued that 'though he has raised taxes a 100 times, though low global interest rates have cut the cost of Britain's national debt, though dearer oil has brought a North Sea tax windfall, revenues still cannot keep up with his spending' (Evans-

Pritchard, 2007). Halligan, the *Telegraph*'s primary fiscal hawk, was the first to suggest that the government was 'approaching a fiscal crisis due, entirely, to the Chancellor's mismanagement', something Anatole Kaletsky of the *Times* later echoed, albeit as a less likely scenario (Halligan, 2007b; Kaletsky, 2007a). 'Nowhere is this more apparent than his hapless record on borrowing', Halligan (2007b) continued. The suggestion that the 2003-2007 period represented a bubble in tax receipts which still couldn't keep pace with the increase in expenditure led Halligan to conclude that 'the more I pore over the budget fine print, the more I feel that Brown is seriously damaging the nation's finances' (Halligan, 2007c). This view, initially at least, remained on the margins of mainstream debate, with even Halligan himself admitting that debating government debt as being dangerously high 'sounds obsessive' (Ibid.).

This highlights a further key discourse that permeated the media during this period, that of the limited scope for fiscal expansion in the face of a downturn. In this vein, the FT talked of the fiscally neutral budget as being forced by the 'tight state of the public finances' (FT, 2007b), and suggested that even optimists thought that his planned reductions in budgets over the next spending period was insufficient to repair the fiscal balance sheet (FT, 2007c). However, it was further noted that despite the arguing over minutiae that follows any Budget speech, the key point as far as business was concerned was 'the broad-brush picture of steady growth and unspectacular inflation' (Guthrie, 2007), and as such the national finances were far from being a priority of action for the government. Even here, however, we see evidence of a perception of the relative weakness of government interference to radically alter economic outcomes for its citizens, when compared to the forces of global business (Bootle, 2008b). The FT added that 'he faces no significant fiscal problem to darken his big day' (FT, 2007a), with only one of three economists questioned on the Budget raising the deficit as a worry (FT, 2007e). They concluded that he was a 'competent steward of the nation's finances', but complained that 'he has also been prone to excessive complexity, tireless interference and a plethora of gimmicks' (FT, 2007g). The Times suggested that Alistair Darling would have 'none of the ready fiscal firepower to respond to a downturn with the sort of spending boost that saw his predecessor through' (Duncan, 2007).

On the public finances, the *Times* likewise noted how 'badly stretched' they were despite the 'billions in extra taxes' (Times, 2007b). They continued that 'if the economy were to hit rockier times...it would not have the extra fuel from strong public spending growth that staved off threatened recession at the start of the decade. And with the fiscal rules already at breaking point, statistical accounting changes forcing the Treasury to put more of its concealed public debt on the books could make life harder still' (Ibid.). The *Daily Mail* concurred, calling the public finances 'out of control' and highlighting that strong economic growth was needed to prevent a 'dire' debt outcome (Daily Mail, 2008a).

The decision by Brown to reduce medium term departmental budgets below the rise in inflation was interpreted by the *Guardian* as highlighting that the 'public could expect a period of austerity over the coming years as Mr Brown sought to bring down public borrowing' (Guardian, 2007a).

Their Budget coverage also highlighted the criticisms of Brown over his failure to 'bring public finances under control' (Guardian, 2007d). The *Independent* took a similarly mixed line, calling the public finances 'a shambles' and highlighting that his successor 'will have to find a way of paying it all back' (Independent, 2007a), whilst also claiming that the economy was 'in a reasonably stable position' (Independent, 2007b). Economic commentator Hamish McRae argued that the state of the public finances meant that Brown's successor 'will not realistically be able to borrow much more' (McRae, 2007c), and thrice suggested that any leeway to fight a downturn had gone because of the size of the deficit (McRae, 2007a, 2007b, 2007e). A budget deficit of one per cent, McRae reasoned, would not hamper using fiscal means to counter a slowdown, but that a deficit of three per cent would (McRae, 2007b).

The change of Prime Minister in June 2007, and as such the change of guard in the Treasury, led to a further re-evaluation of the fiscal rules that Gordon Brown had used to guide the management of the public finances in the previous decade. This gives a powerful insight into the interpretations of fiscal policy on the behalf of the media, their sense of economic priorities and medium-term expectations. Again, existing norms of economic understanding which pervade public discourse during the stability period that preceded the crisis can be observed in such media coverage.

Generally, the broadsheets were all agreed that both fiscal rules had severe flaws as frameworks to govern macroeconomic policy. They were deemed 'past their sell-by date' after having been undermined by tweaking in their component parts by Gordon Brown since 1997 (FT, 2007n; also Stewart, 2007). The first fiscal rule was widely considered defunct after the dates of the beginning and end of the cycle was constantly altered by the Treasury for what was deemed to be political purposes (Brummer, 2007b). Thus the *FT*, *Times*, and Vince Cable writing in the *Telegraph*, regarded independent statistics as one of the priorities for Alistair Darling to adopt (Cable, 2007; *FT*, 2007m; Kaletsky, 2007c). Kaletsky added that a 'more sophisticated understanding of public borrowing was therefore needed', adding that in a knowledge economy, future economic returns are not always equivalent with physical investment (Kaletsky, 2007c).

The logic of the second Golden Rule was more widely supported, limiting government debt as it did. However, again, the problem of politicians' trust which arose from the manipulation of expenditure to meet the rule was seen as evidence of its weakness as a measure of credibility for the government (FT, 2007l; Kaletsky, 2007c; Stewart, 2007). Moreover, the inflexibility of the second golden rule risked preventing the country from undertaking expensive but high quality public investments. The *FT* noted the paradox that 'without a change, valuable public investment projects will be ditched and the incentive to pay for ever more exotic off-balance sheet financing will grow. But any change in the rule will lead to charges that the government is relaxing its grip on the public finances' (FT, 2007l; also Kaletsky, 2007c; Stewart, 2007).

Discussions regarding fiscal rules highlight the deeply constructed nature of governing credibility in relation to the public finances and debt. Historically, the 40% threshold for public debt set by Gordon Brown was relatively arbitrary, and yet the credibility of fiscal plans was still being measured against it, despite the recognition that it was fallible. Fiscal hawks in particular noted the

use of PFI deals to shift a great deal of public debt outside the reach of this measure, thus producing fear that the public debt was at a much higher level than was being presented. Yet the legacy of the second golden rule still clearly framed the media's understanding of acceptable changes in fiscal policy. The suggestion of 'no room for manoeuvre' in fiscal terms was deeply tied to the 40% debt limit. It was the fact that public debt was so close to this figure that led the press to suggest that Darling could not use increased spending to counter-act potential recession in 2008. The lack of any clear economic historical evidence for the necessity of keeping public debt below 40% again demonstrates the significance of the metrics of credibility governments and the media generate for the measurement of economic policy in general. The decision of Brown to place public debt as one of these metrics during the early days of his chancellorship therefore helped to shape the media debate on the necessary response to crisis a decade later.

Furthermore the political trust discourse, with Brown regularly being attacked for 'deceit' in not being honest about public debt, also has significant traction here in terms of its economic implications. With reporting on debt being described as too 'political', the desire to remove further the political from the economic was becoming more tangible in the media's reporting. Again, then, 'honesty' was something attached to an acceptance of the poor state of the public finances and an acceptance of the need to deal with this as a political priority. With spending commitments being deemed too political, the depoliticisation discourse logically entailed fiscal consolidation.

In the aftermath of the 2007 Pre-Budget Report (PBR), the *Independent* in particular among the liberal press, and to a lesser extent the *Guardian*, hardened in their analysis of the fiscal deficit as precluding extended stimulus in the face of a downturn. Jeremy Warner, then of the *Independent*, supported the view put forward in the paper earlier in the year by Hamish McRae, argued that 'one of the reasons UK growth survived the last downturn in the world economy so successfully was because the government was able to spend its way out of recession. No such option exists this time around...there's nothing left in the fiscal cannon' (Warner, 2007) McRae continued this theme in two further articles in December (McRae, 2007d, 2007e). The paper also precluded a re-birth of Keynesianism by suggesting that 'the Treasury is in a tight spot – and this time it can't rely on Keynes' (Independent, 2007d).

They also suggested that an opening was being created by the developing economic situation for Cameron to offer a genuine challenge to Labour orthodoxy on the issue of tax and spend, which until now the media had generally assumed was Labour's territory and that the public would not stand for even relatively mild cuts in public services (Independent, 2007e). For a paper that has come to stand as one of the leading opponents of the austerity policy in post-2010 Britain, it is telling of the zeitgeist in late 2007 that it so consistently argued this position. Whilst Keynesianism was still inherently framed as the generic correct response to an economic downturn by most of the media, to offset contracting private demand, there is a silent assumption running throughout such articles that a state can only undertake such a stimulus if it has previously offset such deficits with surpluses at the top of the cycle. The danger that awaited if one failed to do so is left unstated.

The *Guardian* took a similar position, also noting the differences between the current context and that of the dot-com crash earlier in the decade which New Labour had been able to offset with a loosening of fiscal policy to prevent a recession (Guardian, 2007g). Likewise they argued that there was little room for 'Treasury largesse' (Ibid.). Darling's maintenance of Brown's plan to increase spending below inflation, in essence a real-terms cut, was called a return to 'prudence', the framing of budgetary discretion which would come to provide a heavy rhetorical inheritance for Labour during the recession (Ibid.).

The IFS began to play an important role in fiscal discussion during this period that would be maintained up until the 2010 election. Their report that Darling was in danger of breaching the 40% debt-to-GDP limit and that a £13bn gap was apparent in Labour's spending plans was seized by the media as confirming the legacy of Brown of excess spending and the need to either increase taxes or cut spending to bring back balance. Whilst £13bn would soon become a calculation error in regards to the size of the budget deficit, after a decade of relatively stable budgets this was seen by the *Guardian* as a 'black hole' (Guardian, 2007i) and a 'hefty extra allowance' (Guardian, 2007h), while the *FT* quoted Robert Chote, the IFS' then director, as calling the pre-Budget Report a net "unfunded spending increase" (*FT*, 2007o).

For the conservative-leaning press, the fiscal position and the credibility of Labour's economic governance were the only major story of the 2007 PBR. The Daily Mail matched the themes of the Guardian and Independent in highlighting the importance of the projection of competence in Labour's fiscal moderation over the past decade. Thus they argued that the 'debt numbers in the pre-Budget Report stretch (Labour's credibility) to the limit' (Daily Mail, 2007l; also Brummer, 2007c). They argued that Labour's fiscal plans were funded 'on the never never' (Daily Mail, 2007k) and that spending plans needed to be 'slashed' (Mail on Sunday, 2007). Already a narrative was forming that Brown had 'borrowed recklessly' in his time at the Treasury (Daily Mail, 2007j), and that efficiency savings were needed from the government (Daily Mail, 2007i). These were mirrored by the *Telegraph*, where Halligan suggested that Brown had 'mortgaged our future in a vain bid to secure his own', and that 'the real story of the PBR...is borrowing' (Halligan, 2007d). While the Telegraph acknowledged that the shaky nature of the British economy was also a function of private debt, the narrative that 'you can't fight debt with debt' also began to formulate here (Reece, 2007b; Conway, 2007b). Moreover Conway associated the dangers of government debt to that developing in the housing sector, suggesting that 'the longer (the government) carries on borrowing, the greater the risk that eventually it is not just the Pre-Budget Report but the entire economy which is sub-prime' (Conway, 2007b).

The *Times* sarcastically sent a letter to Prudence on behalf of Gordon Brown, saying it was time for them to part: 'I have instead asked your old rivals, Naked Opportunism and Sheer Expediency, to take over from here' (Times, 2007g). Likewise they agreed that the Chancellor 'has none of the ready fiscal firepower to respond to a downturn', and that borrowing left little 'wiggle room' (Duncan, 2007; Times, 2007h respectively). Kaletsky noted that 'political parties who want lower tax – and by implication smaller government no longer seem out of tune with the times' (Kaletsky,

2007d), although the paper added that with the margins of 'prudence' already being deemed so tight that there would be no room for 'pre-election sweeteners' (Times, 2007i).

The discourse of 'no room for manoeuvre' was becoming increasingly ubiquitous by the spring of 2008 (FT, 2008a, 2008c, 2008d). The Daily Mail declared Britain's public finances as 'out of control' (Daily Mail, 2008a; also Dale, 2008) and warned that a 3p rise in income tax might be needed to balance the budget after an election (Daily Mail, 2008b). The Telegraph lamented how 'Gordon Brown failed to use years of strong growth to store up a Government surplus for the years of weaker growth now under way' (Daily Telegraph, 2008a, 2008b) and described the 'bones of a tight fiscal corset' which were digging into Alistair Darling's ribs (Reece, 2008a). Both the Mail and Telegraph framed the Budget as an opportunity to see if the government had learnt its lessons on the economy, with a suggestion that the waste created by 'if you have a problem, throw money at it', was the cause of our problems and that what the economy really needed was a boost to productivity that being set free from the burdensome state would allow (Daily Telegraph, 2008c; also Daley, 2008a; Oborne, 2008a). The combination of fiscal expansion and dishonesty was rolled into a single frame by the *Telegraph*, by arguing that the government had an 'inability – or is it unwillingness? – to face up to the mess it has made with the taxpayers' money' (Daily Telegraph, 2008d). They concluded that 'the focus must shift now to a small state' after a 'decade of waste' (Daily Telegraph, 2008e; also FT, 2008b). Only Roger Bootle offered a more Keynesian position within the paper, suggesting that fiscal orthodoxy risks 'catastrophe' if used too early (Bootle, 2008a).

The lack of clear growth initiatives in the Budget was bemoaned (Times, 2008b; also Independent, 2008d), but with the explicit explanation that they were perhaps impossible because 'Labour mismanaged the country's finances through a decade or more of unusual global growth' and that there was 'little scope for giveaways' (Reece, 2008b; FT, 2008d respectively). Furthermore, Alistair Darling was accused of 'complacency' in failing to prepare Britain for a downturn (Daily Telegraph, 2008e; also Daily Telegraph, 2008b), with the general mood being summed up in the phrase 'you wouldn't start from here' (Walker, 2008). Even the liberal press partially supported this view, with the *Independent* stating that 'there is no money in the pot' and 'no scope for either cutting taxes to any significant degree, or further increasing public spending' to boost the economy (Independent, 2008b; Warner, 2008a respectively; also Daily Mail, 2008c; Independent, 2008c). Whilst some liberal support for a more Keynesian approach continued (Independent, 2008a, Mitchell, 2008).

The decision of the government in July to pave the way for breaking its self-imposed fiscal rules was deemed 'symbolic of the collapse of the (New Labour) regime' (Warner, 2008c). Warner continued that these were a 'humiliating blow' that placed Britain as among the most indebted in the G7 (Warner, 2008b). Kaletsky noted that whilst golden rules may be arbitrary, 'erven arbitrary and illogical rules about public borrowing are better than no rules at all – and the danger of the rules being abandoned is that an open season for unlimited deficits and spending will be declared'

(Kaletsky, 2008b). Kaletsky continued that 'nobody can say for sure where the safe limit for government borrowing may be in today's Britain. But the precedents from the 1970s are fairly horrific' (Ibid.).

5.4 Drivers of Economic Growth

Implicit in much of the macro-economic reporting of 2007, then, was an internal contradiction which criticised the government for not saving money to spend in a recession, whilst at the same time suggesting that government spending was of relatively little significance against the scale of global economic forces. Regardless, such an inconsistency was not criticised or noticed, and this impacted the press' perceptions of the most suitable methods for stimulating growth. Accepting a frame that there was little room for manoeuvre in the national finances, alongside that of an overbearing state, much of the suggestions for forward growth involved supply-side measures to free up businesses from tax and regulation. Whilst at this stage the cynicism towards government spending did not have a deep macroeconomic element to it, this scepticism of the ability of government to create socially desirable ends through spending projects was to be transliterated in the face of the financial crisis to cast doubt on the worth of stimulus spending. This section therefore demonstrates the existence of discourses relating to economic growth which argued that the size of the central government was suffocating business and that stimuli to growth should focus on supply-side measures.

The media coverage of the government's growth plans dealt largely with problems hitting the corporate sector. The importance of competitive tax rates and the reduction in the regulatory burden were the two most commonly discussed themes here. The Budget of 2007 created an opportunity to debate both, for whilst Brown simplified the tax regime, little was done to simplify regulations. Both of these two topics were most commonly discussed in terms of 'competitiveness'. The debate was universally grounded in the issue of international competition against trade rivals, and as such the importance of providing a strong business environment was paramount given the sense that Britain was slipping down the league tables of measures such as tax competitiveness. Whilst the right-wing press were naturally very pro-business in this regard, the left made little in the way of a case against cutting red-tape and taxes.

The business voice was somewhat harmonious about the required strategies to promote future growth during 2007. The focus was on the UK's falling productivity (Templeman, 2007), with the recurring conservative complaints about inflexible labour laws, red tape and tax burdens being the central foci of complaints. The 2007 Budget's intention to cut corporate tax was offset by increases in tax rates for small and medium-sized businesses, which the changes skewed in favour of financial services and against manufacturing (FT, 2007k). The macro impact of these measures on economic growth was therefore questionable, with benefits for multi-national companies potentially

being cancelled out by reduced spending by small, domestically-based firms (Guardian, 2007e). Yet the media focused less on this issue, highlighting the more hidden tax rises on small businesses only to question Brown's personality and his dishonesty (Daily Telegraph, 2007d; 2007e).

A great deal more dialogue took place involving the promising signs this sent to business about the UK's desire to increase competitiveness, and the potential impact this would have on firms' desire to invest (e.g. FT, 2007j). Comparisons were drawn with Lawson's budget of 1984, where a large reduction in corporate tax rates led to a boom in investment (Independent, 2007c, Daily Telegraph, 2007g). The FT dwelled on the 'most important' impact on the UK's standing in international rankings of competitiveness (Guthrie, 2007), and dwelt on the successful impact of the concerted efforts by business to campaign for such changes (FT, 2007d). They cited business as being 'relieved' and 'delighted' by the changes, quoting several who voiced their praise in terms of competitiveness (FT, 2007i; 2007j respectively). But caution was noted about this being an end product, with the CBI calling for further cuts in rates in the future, to fund which the government would need "to get a grip on public spending" (cited in FT, 2007d). Likewise the *Times* noted that the tax change had not moved Britain out of the bottom third in the EU for tax 'competitiveness', and brushed aside the fact that Germany's and France's rates were higher still (Times, 2007e)

Whilst the FT supported the move to greater competitiveness, the Daily Telegraph was more explicit in highlighting potential causes of its apparent loss, dwelling in particular on the size of the public sector as an impediment to growth. A dichotomy between a productive private sector and an unproductive public sector was drawn, implying that true growth in the former had to come at the expense of the latter. Simon Heffer was most explicit here, stating that 'our economy is underperforming because of the bloating of the unproductive sectors at the expense of the productive' (Heffer, 2007). The paper referred to the key areas of taxation, infrastructure, workforce flexibility, regulatory simplicity, skill levels and the size of the state all as problems in which Britain had 'no room for complacency' (Daily Telegraph, 2007f). Trends revealed that 'Britain's poor showing in so many areas has coincided with a huge increase in the size of the state', they reported (Ibid.). Commerce was likewise deemed to have been 'clogged up...with unnecessary and excessive taxes and regulations' (Conway, 2007a). Liam Halligan argued that the state was now accounting for its largest share of national income since 1983, 'threatening UK competitiveness' (Halligan, 2007a). With business now apparently increasingly vocal about the size of the state (Ibid., also Times, 2007a), the *Telegraph* also referred to the increased challenge from China and India as a justification for the need to slim down the state (Daily Telegraph, 2007b).

The likelihood of a future slowdown in growth noted by the media during this period led to a renewed discussion of economic growth, often taken for granted in previous years. The burden of the state noted in above sections was particularly pertinent in relation to future business growth, with common suggestions for cutting regulatory burdens on business. As such, in the pre-crisis period, the necessity of supply-side measures over government-led stimulation dominated media discussion. Whilst this discourse was most notable amongst the right-wing and business-focused media, there was little counter-argument on the issue from the left-leaning press. As such the belief

of an over-burdened business sector struggling through regulations and red tape from an overly bureaucratic government developed during this period.

5.5 The Tax Regime

The issue of tax was intractably bound to the above discussion on the relevant size of the state. The perception that the government was getting too big and stifling enterprise and personal wealth meant that very few were defending the current tax take, with only green taxes being seen on the left as avenues for acceptable tax rises. But the Budget of 2007 resembles somewhat of a watershed in the media analysis of tax in that Brown's set-piece reductions to personal income tax in the speech changed the political environment surrounding it as an electoral issue. The media's response to the cutting of income and corporation tax suggests they saw it as a political opening for tax cuts, which as they openly admitted, they had previously assumed the public not to be in favour of on the assumption that tax cuts meant cuts in public services. They also cited the lack of trust in politicians to actually decrease the tax burden, another impact of the trust discourse. But with Gordon Brown advocating tax cuts openly¹⁰, it was assumed that he would no longer be able to spread fear of 'Tory cuts' at an upcoming election. The issue of the state of the public finances problematised this however, with tax cuts likely to increase the public debt. *This section therefore highlights the perception that the tax take was at its limit and that, whilst reductions in tax for business and individuals would be preferable, the fiscal deficit might prevent this from being possible.*

The *Daily Telegraph, FT, Independent, Observer* and *Times* all either explicitly or implicitly suggested that tax take was at its limit and that there was no room for future stimulus. The Daily *Telegraph* bemoaned the tax burden being due to rise over the 40 per cent threshold in the coming two fiscal years (Daily Telegraph, 2007c), and quoted Peter Spencer of the Ernst and Young Item Club who stated that "five years ago Brown had a big enough surplus to fight recession and a war at the same time. He can't do either now" (quoted in Evans-Pritchard, 2007). They suggested that Britain was 'taxed to the limits' (Daily Telegraph, 2007a). The *Financial Times*' Nicolas Timmins agreed, suggested that 'public spending has now reached the limits that the leading politicians of both major parties believe is permissible in Britain if a government hopes to be re-elected' (Timmins, 2007). The Observer agreed too; reviewing the economics of New Labour's decade in

¹⁰ Again this demonstrates the particular significance of the realist constructivist position. The issue here, from a realist constructivist position, is not whether there was a genuine alignment of ideas behind cutting the gross national level of taxation and state spending, but of whether such ideas were politically feasible and supportable in the public arena. By publicly supporting the ideas of cutting personal levels of taxation, the frame that tax cuts meant dangerous spending cuts was rendered un-supportable.

power it suggested that 'Brown has probably pushed the tax burden to the maximum the electorate will accept' (Observer, 2007a)¹¹. The *FT* was generally praiseworthy of the Chancellor's late attempts to simplify the tax regime in his final Budget (FT, 2007h). Kaletsky in the *Times* agreed, stating that he 'delivered exactly what most economists, tax experts and business lobbyists had long been demanding' in this regard (Kaletsky, 2007b).

In a time of economic stability, the logic of this discourse, as suggested above, was to legitimise tax cuts as opposed to deficit reduction, with the conservative-leaning media initially critical of the Conservative party's hesitation to publicly revoke their commitment to match Labour's spending plans (e.g. Daily Mail, 2007a). Miles Templeman, director of the Institute of Directors (IoD), writing in the *Mail*, said that in the 21st century 'we cannot afford a public sector of the magnitude that is to be seen in parts of this country' (Templeman, 2007; also Daily Telegraph, 2008f). On noting that academic research had suggested that higher social expenditure may be more attractive to businesses than low taxes, the *Mail* argued that such evidence didn't fit with the theory that 'lower tax rates are critical to encouraging both personal and corporate entrepreneurship' (Brummer, 2007b), and therefore dismissed it.

Conway argued that under Brown the UK had transformed from 'being a relatively low-tax economy to being a high-tax economy' (Conway, 2007a). The *FT* agreed that Brown's tinkering had meant that the voters had grown 'weary' of promises of new money for public services, and had ceased to believe that higher taxation would improve the quality of such services (FT, 2007f). Whilst the *Guardian* and *Observer* were not wedded to the importance of tax cuts as a social good in the same way that the conservative newspapers were, they largely reported the Budget's cut in income tax in 'political' rather than 'economic' terms in the sense that its importance was derived largely from the tactical quandary it put David Cameron in, dispelling as it did the idea that Labour were permanently set on increasing state capacity. As such the tax cut was welcomed by the papers, even if they did question the ability of Brown to be a true redistributionist without using tax as the major instrument to undertake this (Guardian, 2007f).

The political implications of the tax cut were debated more vociferously than the actual economic impact, the revenue-neutral element of which had led the media to assume that generally there would be relatively little of. The conservative media in particular jumped on Brown's volte-face as a tax cutter, suggesting that this had opened up political space for the Conservatives to make a more successful case for tax cuts and a reduction in the 'bloated' state sector. Norman Lamont, writing in the *Telegraph*, suggested that given that the Conservatives now spoke of economic stability ahead of tax cuts, they could square the circle by realising that a high-tax environment is destabilising and that there was now political space to 'make both the moral and economic case for tax cuts' (Lamont, 2007). The *Telegraph* also blamed the reduction of personal incomes and the increased private sector debt load as a symptom of this high-tax environment. They also began to suggest that

¹¹ The Conservatives' 'third fiscal rule' of spreading the proceeds of growth between spending increases and tax cuts, warned by Gordon Brown to require a £21bn spending cut at the 2007 Budget, was perhaps now more in line with media orthodoxy.

the current scale of public spending meant that 'it should not be too difficult' to afford tax cuts without harming public services (Daily Telegraph, 2007b). As such a wielding together of discourses of the state, taxation, and economic growth were quite solidified in the conservative press before the financial crisis, with complaints about the size of the state, the burden it was placing on business, and therefore the need to cut it back in order to allow business to grow sufficiently again. State-led growth was not considered an option.

5.6 Conclusions

This chapter has reviewed the major economic discourses of the pre-crisis period to highlight emerging trends in public debate over the role of the state and the economy. In the four separate sections above it has highlighted four central, and over-lapping discourses relating to economic affairs that were prominent a decade on from New Labour's rise to power. It is perhaps not surprising that in a period when, with the exception of Gordon Brown's premiership honeymoon, the Labour Party's approval ratings were steadily declining whilst the Conservatives' saw a revival, the major foci of economic and wider political discourse was increasingly critical of New Labour's economic philosophy and policies. As is demonstrated through the following chapter, these discourses provided the template on which future discursive trends after the arrival of the financial crisis were built.

The first of the four central discourses outlined above relates to a questioning of 'statism' in relation to the appropriate role of the state in the economy, added to a questioning of the size of the state that had developed under Gordon Brown's chancellorship. The media regularly suggested that the size of government bureaucracy and the centralising tendency of the Brown era had gone too far, and that this was providing unnecessary obstacles which stunted private business growth. In particular the desire to bureaucratically micro-manage any measurable problem was criticised. Brown's management of the Treasury and his capacity to report truthfully back to the people was widely questioned, and he was viewed as being underhand, each budget representing more of a political game than of an assessment of wider economic conditions. Whilst Labour were still working on the assumption that the electorate supported any extra government spending, the media now viewed extra initiatives as a budgetary burden and questioned the benefit of extra money on outcomes in relation to social policy.

Secondly, as a consequence of the increasing size of the state and the budget deficit by 2007, there was a universal discourse which questioned the government's 'room for manoeuvre' in the face of a downturn, with the self-imposed second golden rule preventing fiscal expansion. There was widespread criticism of Labour for having shed their 'prudent' management of the public

finances and for having allowed extra spending to have developed beyond what was sustainable. Whilst this was by no means seen as the most pressing issue of the time, it does highlight how the media had begun to pre-diagnose several central national problems before the crisis hit, and as such it is telling how the deficit discourse of 2009 and 2010, analysed in the following chapters, was built upon this major pre-crisis political discourse. For the most part, at this stage, this building up of deficits was not argued to prevent future spending because of the impact it would have on the financial markets and the capacity of the British state to borrow, but merely presented as a normative objection to debt and the belief in the need for cyclical balance.

The beliefs from these more broad perceptions of government involvement and capacity filtered down to the more specific policy levels relating to supporting economic growth and tax. With little support, or scope, for statism, it followed thirdly that the future drivers of economic growth were seen to come from the freeing of private enterprise from the yoke of government interference and the reduction in 'red tape'. Demand-side problems were not mentioned, and as such it was believed that the reduction in supply-side constraints which the over-sized British state was currently imposing was the route to better growth. Little at this stage was mentioned relating to the dependence of the wider economy on credit. Fourthly, regarding tax issues, the size of government was seen as having pushed the tax level to its limits. Here a more notable left-right divide existed than in the previous issues, with the right calling for tax cuts for both businesses and individuals, supported by reductions in government spending, whilst the left did not make such demands.

As such the period before the financial crisis is marked in the British case by an increasing questioning not only of the capacity of the state to generate or support economic growth through spending, but also of the very worth of such an approach in the first place. The British state was perceived to have become burdensome, taxing its citizens too highly and placing too high a regulatory burden on its businesses. Generally, therefore, the calls for reform were generally based in ideas of supply-side reform; free-marketism, competitiveness, reduced regulation and a low tax regime. The left lacked a meaningful critique of the status quo which also was able to posit a meaningful role and justification for government involvement, and indeed often shared the right's view that the state was starting to interfere too far.

This is not to say, however, that the major financial shocks associated with the global financial crisis did not alter the economic beliefs, priorities and narratives of the media in late 2008 and beyond. The constructivist theory suggests that the existing structural constraints to change are removed by the kind of uncertainty that a crisis creates, and that new public understandings can become new political norms as the battle to diagnose the causes of crisis gives them an advantage over existing the existing orthodoxy. However, by demonstrating the reversion back to the discourses outlined in this chapter, the thesis is able to demonstrate a greater level of continuity, and thus greater strategic advantage, for ideas that placed a greater weight on slimming down the state by reducing public spending. The following chapter also demonstrates that the type of structural vacuum commonly utilised in constructivist institutionalist work does not necessarily create a

generalised sense of crisis but rather that this can be limited to certain spheres of the economy. Thus the thesis is able to provide an explanation of political continuity that is still constructivist at heart.

Chapter 6: Political Economic Discourses of the Crisis Period: August 2008 – January 2009

6.1 Introduction

The existing literature leads us to see crisis moments as episodes of structural re-alignment and changing policy paradigms. The previous chapter highlighted the major existing discourses relating to the role of the state in the economy during the immediate pre-crisis period. The purpose of this chapter is to examine the extent to which the severity of the financial crisis shifted these. It finds that this is only partially the case, and has three major findings.

Firstly, during the time that the danger of systemic financial collapse hung over the City of London, extensive state intervention emerged as a politically acceptable course of action. The sense of Britain having 'no room for manoeuvre' in terms of fiscal policy was eroded by the greater threat of financial collapse. As such we can observe the capacity of a crisis to substantially alter the public discourse in relation to macroeconomic affairs. However, contrary to the expectations of the existing literature, this re-imaging of political possibilities remained relatively limited. Rather, the frame of acceptability of intervention always had a time-limited clause added to it, suggesting that whilst fiscal issues were no longer most pressing, they could only be ignored whilst the financial risk was high. Thus, whilst a shifting narrative of state intervention existed, the frame precluded a permanent change in the acceptability of state involvement in the economy.

Secondly, the analysis identifies a siloing of the blame for the crisis within the financial sector. This builds on the initial primary research findings of Pirie (2012: 342), who shows that matters relating to the real financial cause of crisis were increasingly sidelined compared to fiscal issues. Whilst media narrations suggested that the financial sector required new forms of regulation and intervention, views on the relationship between the state and the broader economy were un-altered by the crisis. The lack of diffusion of the concept of crisis from the financial to the corporate sector is therefore a significant finding in relation to how constructivist scholarship should think about crises in general, suggesting that whilst crises do lead to the questioning of existing ideas, this need not necessarily take place on a systemic level.

The third finding highlights that the crisis was deemed to be too large for the UK state to be able to solve by itself. Thus the discourse of the fragile state within a globalised economy remained intact from the pre-crisis period, and suggested that whilst fiscal policy could support the economy in the short-term, it was not capable of supporting the domestic economy until the world economy re-stabilised. Thus again there was a need for supply-side solutions which limited the burden on business and encourage their growth. This therefore demonstrates a considerable level of ideational continuity through the crisis, questioning the ability of major economic ruptures to provoke widespread upheaval in the way that economic governance is perceived.

In general, therefore, the chapter finds that the ability of the crisis to cause radical changes in the ideas governing the political economy was very partial. Whilst the perceived structural barriers to fiscal demand-side intervention were altered by the scale of the financial emergency, the understandings about the medium-term recovery strategy were still significantly influenced by precrisis narratives about the weakness of the state and the sense of the precarious nature of the British public finances. This helps to explain the discursive advantage held by the austerity school over the Keynesian school once the immediate danger of financial collapse had receded in 2009 and 2010, which will be highlighted in the following chapter.

The findings help to explain further some of the extensions to the constructivist institutionalist school suggested in chapter 3, and answer research question 2.) in particular. It suggests firstly that if we loosen our definitions of crisis away from that of a complete structural vacuum and more towards a partial and time-limited one, then political continuity may be expected and explained through a constructivist lens. The idea of crisis being 'siloed' within certain sections of the economy provides an important new way for constructivism to engage with crisis, and helps to explain the difference of outcomes to the 1930s and 70s. In these cases, a more universal sense of economic crisis prevailed, where here it was limited to the financial system.

The chapter begins by analysing the extent to which the whole western growth model was questioned by the media during the onset of the crisis. Here it is shown that whilst the crisis did provoke some questioning about the neoliberal economic model in general, this was limited and somewhat speculative. The following section supports this by covering the siloing of the crisis within the financial sector and the continuation of support for supply-side measures to provide assistance to corporate business. The ability of the crisis to shift the acceptability of fiscal stimulus and limit the significance of the 'no room for manoeuvre' thesis is shown in section 6.4. However, the limited capacity of the state in general is covered in section 6.5, demonstrating that whilst some ideational re-imagining took place during the crisis, this was only a very partial process. The chapter concludes by summarising again the three central findings and their relevance to the thesis as a whole.

6.2 A crisis of capitalism or a crisis of finance?

Whilst the central argument of this chapter is that even during the height of the crisis there was limited change in the ideational frameworks underpinning popular understandings of suitable economic governance, for a short period as the depth and outcomes of the crisis were unknown a discourse did emerge suggesting that the entire neoliberal governance agenda might be reconfigured afterwards. The perception of a crisis beyond that of the financial system, a diffusing of the failure towards the wider neo-liberal project, therefore gained a limited acceptability as the uncertainty increased. However, worth noting are the relative briefness of this ideational crisis moment, and the fact that such predictions remained largely speculative; a suggestion of where policy and public moods might shift as opposed to a critique of failures in the existing governance structure that necessitated a dramatic reconfiguration. Where the crisis was argued to highlight failures in the existing structure, these were related to the financial regulatory regime, and rarely to a wider critique of Western capitalism.

The explicit linking of banking failures to the wider neoliberal economic model remained somewhat limited, therefore. John Plender, writing in the FT, came closest to making this argument, suggesting that the 'triple mantra of privatisation, liberalisation and deregulation' were being questioned by the crisis, arguing that the neoliberal model of wider economic governance was a central cause of crisis (Plender, 2008). He continued that credit market problems 'cast a cloud over the strong market orientation of western policy' and as such that 'fiscal policy could revert to the more redistributive ways of the pre-Reagan/Thatcher era' (Ibid.). With government already becoming used to changing the limits of what it constituted acceptable interference in the private economy through liquidity injections, assurances of support and potential nationalisation for stricken banks, he pondered that it was becoming far more probable that politicians might expand its intervention 'outside the financial sector' and 'claw back power hitherto devolved to the private sector' (Ibid.). Even here, however, what is notable is that it never explicitly supports such a change of ideas, but notes that they may gain political tractability as the public seeks answers and blame for the turmoil. Indeed he concludes by suggesting that such a development of fiscal policy is unlikely, with a lack of a clear alternative brand of capitalism as there was in the 1930s, and the general unpopularity of the political elite and public managerialism preventing a discourse of the 'return of the state' capturing the public imagination (Ibid.).

Chakrabortty and Elliott, in the *Guardian*, produced similar analyses of the failure of capitalism. Chakrabortty described capitalism as a structure that 'lies in shambles', but critiquing the left for having 'gone AWOL' and having 'nothing to say' on the matter (Chakrabortty, 2008). As such, even those who linked the crisis to that of governing ideas failed to see an alternative vision for western capitalism. Larry Elliott continued this message in September, blaming 'deregulation, liberalisation and privatisation' for the crisis and the timidity of the state in the face of financial

capital, but again blaming the left for not having the courage to 'seize the social democratic moment' (Elliott, 2008b).

Once the crisis began to go 'nuclear' (Elliott, 2008c) in September 2008, the suggestions of the need to re-think the entire edifice of financial capitalism continued, but ironically became less pressing because the need to maintain stability today took precedence over re-thinking the world of tomorrow. Will Hutton argued that the financial system needed 'reconfiguring', and that the 'dynamics of British capitalism' needed to be changed; but such talk was for 'later columns...today we must simply avoid financial calamity' (Hutton, 2008a). Even articles from the *Times* declared that the West may well end up 'ditching its faith in free markets and private enterprise' which would 'filter through the global economy' starting from America (Kaletsky, 2008e). The paper asked 'was the left right after all?' (Times, 2008d)

This suggestion that uncertainty would precipitate a change in attitudes towards the free-market economy was not supported by Kaletsky's other articles, however. Whilst he argued that the logic that market forces should be allowed to rectify financial instability was 'profoundly wrong', the notion that Darling's support of the banking system had avoided 'an unavoidable recession (becoming) a one-in-a-lifetime depression' (Kaletsky, 2008g) suggested that such a crisis would be contained within the financial economy and not spill over into the 'real' economy (Kaletsky, 2008c). This dichotomy between the two forms of economy undermined the previous notion that the free market would be universally critiqued and challenged. Crisis, both financial and ideational, was therefore being contained within the banking sector through such political action and discourse.

Slightly more surprisingly for a left-leaning newspaper, the *Independent* was more reticent about the impact of the financial crisis on the relationship between state and economy. Whilst it noted in several articles the likely reconfiguring of financial regulation in the post-crisis period, very few of these suggested the need for a wider re-evaluation of economic governance. In essence therefore, this was viewed as a banking crisis not dissimilar to ones in the past, and which despite necessarily leading to a re-think of banking regulation didn't pose broader questions of the role of finance in society, or the governing ideas relating to markets. Jeremy Warner wrote of the 'bonfire of the vanities' on Wall Street and suggested that 'bankers can in future expect a more tightly regulated environment' (Warner, 2008f). The notable perception here, then, was that the crisis legitimised a much more active role for the government in the financial system. As Alistair Darling declared economic conditions to be the "worst in 60 years", Jeremy Warner called this 'obviously untrue' (Warner, 2008d; also Independent, 2008f), and called 'the banking crisis' 'not so different to previous ones' (Warner, 2008e) hence questioning the extent to which the situation could be described as a wider economic 'crisis' (also FT, 2008f).

The paper accepted the role of the government as the most significant tool in ensuring financial stability (McRae, 2008b), and suggested that the contracting out of their own authority to technocrats needed to change, perhaps including the re-politicisation of monetary policy (Brown, 2008a). Arguing that 'there are things only the state can do', McRae continued that the impact of the crisis would be to 'redefine the relationship between government and finance in the years

ahead' (McRae, 2008e). Only Richards provided a more systemic version of potential change, noting that 'untrammelled free markets got us into this mess', 'cannot get us out of it' and suggesting that 'an uncritical faith in the market and indiscriminate hostility to the state is broken' with government intervention and regulation now becoming 'acceptable – desirable even' (Richards, 2008c). Even here, however, it is not clear exactly to what extent this view of the return of the state is being referred to only in relation to the financial sector and to what extent to the wider economy.

Across a selection of broadsheet papers, then, at the height of the uncertainty surrounding the GFC speculation about the systemic sustainability of capitalism began to emerge. However, the right-leading newspapers provided no coverage questioning the future of British capitalism as a result of the ongoing crisis. With Barack Obama's election victory in the American presidential election leading to suggestions that global capitalism might be re-made and that electorates would now consider radically different policies to those deemed acceptable in the past, Daley warned in the *Telegraph* that the only lesson of the election was in fact that 'tax cuts win votes' (Daley, 2008b). As such he warned against reading the election as representing a realignment to the left and the perception that the state was the cure to current economic problems (Ibid.). The Telegraph reported with concern that the future looks 'profoundly and worryingly anti-capitalist' with failed institutions being taken over by others leading to less competition and less accountability (Bolchover, 2008). This was not only a result of the crisis, however, but a fundamental pre-requisite of it. The underlying cause of the crisis was 'insufficient capitalism' (Ibid.). The problem, in Bolchover's view, was that the cautious risk which made small business profitable was absent in financial investment banking and that hence the need to re-make capitalism should focus on reestablishing proper market discipline to the financial world. The 'massive' upsides for investment banks were no longer balanced by risk and accountability on the downside, meaning a 'complete imbalance of risk and return', which needed to be re-established (Ibid.). Again here, then, the crisis explicitly starts and finishes in the financial sector alone.

The *Daily Mail* was the first paper to place the blame largely on governments rather than market failures as the principal cause, and likewise took an early position that public spending was not an option for recovery. However at this stage it remained somewhat of an outlier in this regard compared to the other newspapers covered above. Hastings praised Cameron's increased focus on spending restraint as evidence that he had grown into a 'PM in waiting' (Hastings, 2008), whilst Oborne (2008c) noted that we were witnessing 'a fundamental change in the role of the state' where 'there will inevitably have to be major cuts to public expenditure'. Oborne echoed Bolchover's concern that the crisis was caused by 'the abolition of individual responsibility' (Oborne, 2008d), and criticised the 'political elite who run our lives' for continuing to 'adhere to these discredited statist ideas' (Ibid.). The paper argued against bank bailouts from a similar perspective, suggesting that the free market would correct itself if allowed and that politicians were simply encouraging reckless behaviour by involving itself in financial affairs (Brummer, 2008a; Oborne, 2008b, 2008c). On public spending, the paper equated the context not wit the 1930s but with the early 1980s, where

'the official wisdom of the day held that billions of pounds of additional public spending was the only possible answer to the problem' (Oborne, 2008f). The paper also began to foretell a major theme in the conservative coverage of 2009 by focusing on the class 'envy' and 'spite' in taxing the rich (Platell, 2008).

The attention paid by the press to the financial crisis therefore demonstrate that the shock of the collapse in the autumn of 2008, and at times during this period the fear that the entire system could still implode, led to a marked shift in the acceptability of state intrusion into the financial system and the questioning of free, self-regulating financial markets. However the blame for the crisis was rarely extended beyond the immediate financial system, meaning that corporate capitalism was not deemed in need of change. This highlights the extent to which blame, and even arguably uncertainty, was contained within the financial sector and did not extend to other areas of the economy. The capacity of the financial crisis to create major alternative discourses challenging the nature of neoliberalism was therefore relatively limited, even during the crux of the crisis. Below, the analysis of the press towards the issues of economic growth and corporate capitalism are covered, to highlight the extent of continuity from the pre-crisis period in this realm.

6.3 Corporate Capitalism in Crisis?

Much of the focus on economic growth revolved around helping small and medium sized enterprises (SMEs), and the ways in which this was debated replicated the existing discourses each paper employed in relation to the private sector and state involvement therein. Right-leaning papers discussed cutting taxes and removing barriers such as red tape and excessive regulations and planning restrictions, whilst left-leaning papers focused on the potential of green growth, manufacturing and the need for assistance to business not to come and the expense of the rights of those employed by them. Little weight was therefore given to fiscal stimulus from either side of the political spectrum.

The *Daily Telegraph* called for tax incentives for new and small businesses (Daily Telegraph, 2008h), and warned that a Labour approach of 'soaking the rich' was a 'recipe for economic stagnation' (Daily Telegraph, 2008j). The British Chambers of Commerce, referenced in the *FT*, explicitly suggested that 'ministerial rhetoric in response to the financial crisis should be seen separately from the government's relations with companies in other sectors of the economy', i.e. that any criticism of capitalism and market failure relating to the world of finance should not pervade into a wider anti-free-market discourse (FT, 2008e). Camilla Cavendish of the *Times* suggested that 'just because the Treasury forced banks to recapitalise does not make state intervention an unalloyed benefit', suggesting that a 'bonfire of regulation' was needed to help

small businesses, which would in turn limit the scale of the recession (Cavendish, 2008). Sister paper the *Sunday Times* called for more government involvement in supporting national industries against foreign competition, with France and Germany 'more aggressive in supporting their own industries' than Britain was (Sunday Times, 2008a).

The *Daily Mail* also focused on credit and regulation as the keys to recovery above that of fiscal policy. Frost (2008) called access to credit 'the biggest problem facing businesses', and the paper suggested that a VAT cut wouldn't be sufficient to tempt consumers into spending more (Mail on Sunday, 2008). Where the government should focus fiscal policy was on tax cuts offset by public spending cuts, rather than increasing public debt, the paper continued (Daily Mail, 2008g)

The *Guardian*, meanwhile, criticised the increasing dependence of finance in the British economy, and suggested that with it the New Labour economic strategy had failed (Guardian, 2008f). The idea of a Green New Deal gained support through the articles of Larry Elliott, in which support for the private sector was aligned to produce a new political economy focused on green innovation and manufacturing (e.g. Elliott, 2008e). It reported the internal Labour debate over flexible working reforms, with Peter Mandelson planning to drop them to limit new costs and regulations for business, and Yvette Cooper warning that business assistance should not come at the expense of their employees (Guardian, 2008g). Will Hutton called for the need to 'convince business and consumer alike that there is a road map for a prosperous future', but that this was not necessarily 'a call for a new statism' (Hutton, 2008b; also Guardian, 2008d). The *Independent* declared that a priority should be not adding 'to the regulatory burden' on small business (Independent, 2008h; also Lawson, 2008).

Press focus on economic recovery did not therefore differ greatly from pre-crisis discourses on the role of the state in the economy. For the right, reduced regulatory and tax burdens on business were the correct stimuli to apply, whilst for the left and some on the right there was support for more active industrial policy without any major support for a new statism (Times, 2009l). Government spending was not seen as a universal benefit, but rather that support should be targeted to key growth industries. For both sides of the spectrum, then, providing support for the economy was best achieved through incentives for private industry, even if the first-order ideas about which were best to apply differed slightly.

6.4 'No room for manoeuvre' and orthodoxy v stimulus

Discourses about the capacity of the state to respond sufficiently to the financial crisis were mixed during the autumn of 2008. In many respects the media maintained the frame which stated that there was no room for manoeuvre and that Labour could not afford to spend their way out of a

recession. However the depth of the banking crisis led even conservative papers to argue that the state must do whatever it took, and more importantly whatever it cost, to prevent bankruptcies among major financial institutions, with few still arguing for immediate cuts in government spending. As such a frame of 'deferred conditionality' became commonplace, with the press accepting the need for government deficits to rise in the short term, but only on the condition that measures were put in place to restore parity and balance in the medium term. We therefore observe that the danger created by the financial crisis was sufficient to shift perceptions about necessities and limits in relation to fiscal policy to some degree. Importantly, however, these pre-crisis discourses were not eradicated by the crisis, but largely put on hold until stability returned. Thus, even during such a crisis, the British media perceived that the central problems it outlined with government policy during the pre-crisis period would have to be returned to once the more pressing financial problems had been ironed out.

In this context of questioning the impact and possibility of government spending in a crisis, the debate about the benefits and pitfalls of Keynesianism became prominent. These debates to a major extent reflect the respective newspapers' opinions on the issue of the capacity of the state in the crisis period. The *FT's* pages reflected some ambivalence about the logic of such a stimulative approach. On the one hand they argued that the 'problems Keynes faced in the 1930s, such as the ineffectiveness of monetary policy and banking failures triggered by falling asset prices, again seem the most pressing' (FT, 2008g). It added that the criticisms relating to stimulus' impact on budget deficits 'are still heard but are increasingly seen as irrelevant' (Ibid.). Three days later, however, the paper suggested that 'it is not yet time for a boost' and that anyway the 'weak public finances means it may not be able to handle it', adding that 'worries will inevitably emerge about fiscal sustainability' (FT, 2008h). The *FT* suggested that automatic stabilisers might be sufficient and that the government, 'a victim of its past excess', 'has left itself little room to do much more' (Ibid.).

The FT's Samuel Brittan defined the debate as between two sides, where the former 'still think in terms of a conventional business cycle' and who believe that 'it is important to worry about the size of the current budget deficit, re-establishing fiscal guidelines in the medium term and the maintenance of an arm's length relationship between governments and central banks' (Brittan, 2008). The second grouping believed 'that a successful demand stimulus can influence the future output trend' and that the usual logic of proceeding cautiously with change has been over-ridden by the uncertainty of the depth of the recession; 'if you are too hidebound to countenance this move you deserve to run the risk of a severe slump' (Ibid.). Despite some level of concern about 'risky' stimulus and deficits (FT, 2008m), the FT still called fiscal policy 'our most potent instrument' and argued that sharp cuts in interest rates from the Bank of England were 'scary, justified, and irrelevant' given that monetary channels were clogged (FT, 2008j; also FT, 2008l). 'Fiscal policy, more than monetary policy, will determine how and when this crisis will be resolved', the paper continued, although it sounded caution about the likelihood of a stimulus give the world's 'fiscal collective action problem (Ibid.). Nonetheless is called Darling's £20bn stimulus 'the correct call' (FT, 2008o).

The Financial Times therefore largely, but not completely, rejected the notion that government deficits were partly to blame for deteriorating conditions, and that they limited its fiscal options. Giles noted that Brown would likely get the blame for a crisis which was not truly his fault, but that whilst tighter fiscal policy in the pre-crisis years would have been preferable, it would not have prevented the current problems and did not influence the scope of the necessary crisis response (Giles, 2008). Martin Wolf in particular wrote of the need for an increase in spending to tackle the downturn; 'the UK should be able to run large deficits for a while' (Wolf, 2008). Whilst he accepted the rhetoric of prudence, he argued that in unique circumstances of the crisis period 'sometimes boldness is prudent' (Ibid.). The general tone of the autumn reporting of the FT, however, was that Britain risked the wrath of foreign investors and that regardless of questions of short-term necessity, there would need to be a period of austerity in the medium term to pay for the increase in sovereign debt (FT, 2008k). The power of the economic circumstances to shorten timeframes and focus minds on how to deal with current problems regardless of their future implications was not therefore complete; even whilst short-term boosts were accepted despite the sense of limited fiscal capacity, the paper maintained a level of concern for the medium-term economic picture should such an approach be enacted.

The *Guardian* threw its support behind the shift to Keynesian thinking most explicitly, but even this left-leaning paper posted a mixture of articles supporting stimulus and worrying about the constraining factor of large budget deficits. Whilst Larry Elliott did suggest that new fiscal rules would need to be drawn up to avoid governments over-extending during the boom times (Elliott, 2008d), the paper generally focused on the need to provide a narrative of the necessity of the active state as a social good (Elliott, 2008a, 2008e; Toynbee, 2008a, 2008b). The Conservatives' talk of "we wouldn't start from here" was deemed 'no answer to the only question that matters – what would you do now?' (Toynbee, 2008b). However even Larry Elliott criticised the notion that the 'way to solve a crisis caused by too much private-sector debt is to boost public-sector debt' (Elliott, 2008f). The paper warned that the 'chasm of debt' could become 'unbridgeable unless economic growth resumes in 2011 at the optimistic 3% rate' (Guardian, 2008k) The *Observer* called the PBR stimulus 'an unprecedented bid to use government cash to kick-start economic recovery' (Observer, 2008).

The *Independent*, meanwhile, was considerably more concerned by fiscal capacity in the early period of the crisis, maintaining in late August that the government had 'no room for manoeuvre' and that scrapping the fiscal rules was not 'a responsible course of action' (Independent, 2008e). Steven Richards argued in early September that the government 'hasn't the resources to reverse the overwhelming tide of a global credit crunch' (Richards, 2008b). However as the crisis entered October and with conditions deteriorating, McRae compared the national debt figures to those at the end of World War Two to suggest that the government could afford to more greatly expand its borrowing without pushing debt levels to unprecedented or dangerous levels (McRae, 2008c). He followed this up by suggesting that the lessons of the Great Depression had been learnt and as such that 'budget deficits will be allowed to rise' (McRae, 2008d).

Even Jeremy Warner described the government's response to the crisis as 'broadly right' and disagreed with the 'doomsters' who argued that governments were running out of ammunition to deal with the crisis (Warner, 2008g). He suggested that 'governments may need to resort to classic, Keynesian-style, reflationary tactics to escape economic calamity' (Warner, 2008h) and suggested that he had 'no problem with the application of Keynesian reflationary thinking' (Warner, 2008k) as long as it was focused on capital spending. Whilst he maintained that some countries had more scope that others in this regard, 'even in Britain' something 'can, and now must, be done' (Ibid.). 'Worrying about the cost is something than can be left to another day', he concluded (Ibid.). The paper suggested that 'no one is advocating Hooverish austerity now' in relation to the American election and likely policy response to the crisis, in so doing implying that the same could be said in Britain (Independent, 2008j).

The *Independent* maintained the diversity of opinion on its pages throughout the crisis, however, with different articles and different commentators appearing to put their support behind mutually exclusive ideas, analyses and responses. Whilst the article above suggest that the paper had shifted its opinion on the state of the public finances and its implications for fiscal policy, its response to the Pre-Budget Report still tended to reflect heavily on the borrowing of the Treasury, and the scope to continue this into the medium-term. Jeremy Warner suggested in early November, *contra* his previous articles, that Gordon Brown had 'got virtually no money to play with and limited scope for borrowing it either', and that a plan to cut VAT to 12.5% at a cost of £24bn would be too much for the markets to handle (Warner, 2008i). But despite this, he accepted that there could be no net reflationary effect from the Conservatives' fiscally neutral plans (Ibid.).

Indeed the paper attacked all parties for suggesting that 'there is a pain-free way of getting us out of recession' and expected the voters to be bribed by 'the prospect of short-term tax cuts and resorting to the national credit card' (Brown, 2008b). Brown continued that the PBR was 'the most irresponsible budget I have ever heard' and called it an attempt to 'bribe the public with their own money' (Brown, 2008c). He attacked the theoretical logic that government borrowing could help end the recovery sooner and in a less painful way (Ibid.). Anderson summed up the paper's general tone by suggesting that whilst an argument could be made for a Keynesian stimulus, 'can we be certain that investors will buy gilts if a further depreciation seems inevitable?' (Anderson, 2008; also Independent, 2008m). Uncertainty still reigned in this period, with the paper admitting that it was still unclear whether the measures announced in the PBR 'will boost the economy or whether it is, ultimately, just huge sums of public money down the drain' (Independent, 2008l). Likewise it suggested that the battle of 'Labour waste' versus 'Tory cuts' was undecided and 'it is impossible to say, at this stage, which argument will be received with more sympathy. Everything hinges on whether this fiscal package is judged to have achieved its aim' (Ibid.).

The *Telegraph* likewise saw itself challenged by the logic of stimulus. On the one hand it maintained its concern about government debt and deficits, but on the other it also wanted to see tax cuts. It tried to reconcile these positions by suggesting that tax cuts could increase revenues as per the Laffer curve, but declined to comment on whether such tax cuts would therefore fund

themselves and be fiscally neutral (Daily Telegraph, 2008j). Liam Halligan maintained his focus on fiscal deficits as a potentially destabilising variable, suggesting that Britain was heading in the direction of America, where 'public liabilities are now so huge (that) the nation's entire economic stability is threatened' (Halligan, 2008a). He called the PBR a 'borrowing binge' which was 'reckless' and threatened the 'UK's credit rating and status as an "advanced economy" (Halligan, 2008b). Jeff Randall agreed, arguing that 'everyone except the government knows we're spending too much' and suggested incoherence in trying to cure a problem caused by too much debt with further borrowing (Randall, 2008a). This discourse gained popular traction throughout the crisis, with the Archbishop of Canterbury Rowan Williams also calling the government's response akin to "the addict returning to the drug" (Independent, 2008n; also Oborne, 2008e).

Randall continued by linking the government's response to the wider conservative discourse opposing the 'something-for-nothing culture' which represented 'a disgraceful conspiracy of shameless layabouts and gutless politicians (Randall, 2008b). The intended 'spending spree' of the government was of little potential benefit because investment would take too long to get through to the real economy and could cause inflation (Daily Telegraph, 2008f). As with earlier in the year, Roger Bootle provided the only Keynesian position analysis on the paper's pages, suggesting that 'we now face Keynesian conditions' in that lower interest rates are unlikely to work and that as a result the nation shouldn't 'be shy about allowing huge increases in government borrowing to stave off depression' (Bootle, 2008d; also Bootle, 2008e). He presented limits to this however, suggesting that the Keynesian framework is 'both valid and truly radical only in depression conditions', providing the grounds for the framework's rejection based on a re-evaluation of the definition of present economic conditions (Ibid.). As such 'if you can do without a major fiscal stimulus you should. Public borrowing can squeeze out borrowing by the private sector' (Bootle, 2008e).

Even the *Daily Mail* in places accepted the maxim that reduced private sector spending should normally call for the public sector to 'take up the slack' (Brummer, 2008b). And whilst Brummer noted that Britain's 'ration of debt to national wealth is relatively low, irrespective of bank rescues', the problem was not so much one of capacity but one of trust: 'in Britain's case the public sector has become so bloated under Labour...that on one really trusts the government to spend taxpayers (sic) money well' (Ibid.). The paper, like others, called the PBR a 'gamble' in relation to the perception that it involved attempting to spend one's way out of crisis (Daily Mail, 2008e). Oborne suggested that the PBR represented a large fiscal boost which, because of the state of the finances, could not be repeated (Oborne, 2008g).

The *Times* began the autumn sceptical of the suggestions of a crisis of capitalism. Kaletsky argued that only Brown and Darling believed the stalling of GDP to be temporary, but noted that 'we should not be too worried' given that US economic and housing statistics showed signs of stabilisation (Kaletsky, 2008c). He suggested that the current trading of bank shares were an exaggeration of their true worth, with their current valuations 'implying that they are almost insolvent' despite the fact that they 'face no conceivable funding problems in any plausible economic scenario' (Ibid.). He suggested Darling had got his figures 'wrong' in suggesting that

economic conditions were the worst for sixty years (Kaletsky, 2008d), calling financial conditions 'relatively benign' despite the headlines. By mid-September, however, Kaletsky was calling the situation an 'amazing crisis' and suggested that 'the only question is how to avoid a catastrophe in the next few weeks' (Kaletsky, 2008f).

The paper was similarly mixed in its approach to the stimulus versus austerity debate, suggesting that 'experts are more divided than ever' by the question (Times, 2008e). The paper questioned the logic of tax cuts on the Ricardian basis that they might foretell future tax rises which undermine their impact (Times, 2008f), and adopted a more Keynesian position in suggesting that monetary policy might be insufficient to cure the crisis (Times, 2008h). In October it maintained that the room for manoeuvre was 'all but non-existent', however (Times, 2008c). Like other papers, it called the PBR a 'gamble' whose outcome would depend on how much confidence was influenced by the depth of 'red ink' and 'whether the Chancellor can convince the markets that his plans are affordable' (Times, 2008g). The electoral chances of the both Labour and Conservative parties depended on whether they could convince the voters of the truth of their respective discourses of cutting too hard and fiscal recklessness on behalf of their opponents (Times, 2008i; Smith, 2008b).

Its own opinion on the issue was as mixed as the other newspapers, suggesting that given that this wasn't a 'normal' downturn 'reinforces the case for turning to the weapons of fiscal policy' (Times, 2009k) but noting the risk that 'by easing fiscal policy further, the Government may merely undermine confidence' (Times, 2008j). It suggested that America's fiscal plans represented 'going for broke', but that the relative modesty of the UK's plans by comparison may mean that 'they are just not large enough to deal with the threat' (Times, 2008k). Kaletsky's response to the PBR was to call the plans 'good economics' and downplay the threat of national bankruptcy (Kaletsky, 2008h). He presented a classic Keynesian argument of demand deficiency to support the general point. However whilst its articles generally supported stimulus, the *Sunday Times* questioned whether Darling's plan was the best one in terms of stimulating activity, and gave most 'bang for the buck' (Sunday Times, 2008b).

Jonathan Guthrie, in the FT, wrote that a stimulus 'must balance the need for boldness against the risk of lumbering the UK with an incapacitating legacy of higher taxes', but added that business was currently in favour of stimulus regardless of its impact on future taxation or sterling (Guthrie, 2008). The Independent showed a similar uncertainty, calling for tax cuts and suggesting that 'worrying about the cost is something that can be left to another day' in October (Warner, 2008g), but then suggesting that a VAT cut costing £24bn would be 'too much for the markets to handle' less than a month later (Warner, 2008i). Several papers saw a Ricardian danger of tax cuts, arguing that temporary tax cuts would be ineffective 'if consumers believe that tax cuts now mean tax increases later' and that they would therefore not do much 'to save the UK economy from recession' (Guardian, 2008h; Independent, 2008k; Times, 2008f). Given the general support of the idea of fiscal stimulus, Gordon Brown took over Conservative territory by again voicing support for tax cuts, no longer at economic odds with increased government involvement, and sought to portray the Conservatives as the only people opposing such an approach (Guardian, 2008i). The Guardian

worried that whilst Brown was correct to borrow more during this period, he 'must not fritter away cash on cynical tax cuts and wasteful spending' (Guardian, 2008j), matching David Cameron's frame of questioning tax cuts by referring to the lack of trust in the government (Cameron, 2008). The *Telegraph* again cited the Laffer curve to suggest that tax cuts and fiscal prudence weren't incompatible objectives, suggesting that tax revenues would actually increase as a result (Daily Telegraph, 2008j).

The onset of a crisis and the call for more fiscal action to help support a weak economy generally over-powered existing beliefs in the danger of a high tax burden. There remained a small element however who warned of the risk that the stimulus would have on medium-term tax rates in Ricardian terms, suggesting that any extra spending now would necessitate tax rises after the next general election. Most of the stimulus debate of the autumn of 2008 took place without regard for this issue however, highlighting how the perception of tax constraints was over-ridden by the immediacy of the economic trouble at hand. Again then, in the short-term of a major economic crisis we see a public shift in the limits of acceptable discretionary fiscal action by the state.

6.5 Medium term decisions

The above sections have highlighted at most a partial re-framing of the belief in the acceptability of fiscal stimulus and greater state involvement in the economy to avoid a deep recession generated by the crisis. To what extent had the crisis allowed for genuinely new ideas to flourish in this regard, however? The contention of this chapter, and the following one, is that these were actually far fewer than the existing literature prepares us for. The press in general had supported the value of greater state borrowing and spending, at the very least through automatic stabilisers, whilst the crisis was at its worst. However, within this shift in belief was a notable time-limitation; when conditions improved, the extent of government intervention should roll-back. Indeed, from late November suggestions began to emerge that the nature of the government's response would necessitate a subsequent period of austerity. These seeds of the post-crisis austerity discourse are reviewed below. Again, this demonstrates the extent to which existing pre-crisis discourses of the state were more resistant to change than would generally be expected.

In this context, the opponents of stimulus feared the scale of future tax rises and an over-bearing presence of the government in the economy. Whilst the *Independent* noted the inevitability of an increase in government borrowing in the short term, it added to this with some caution of the 'the lamentable history of economies in which governments have played an over-active role in

allocating resources' (Independent, 2008i). It argued that 'there will have to be a reckoning at some point' for the short-term increasing of the deficit (Independent, 2008g), and warned that it would require tax rises and possibly lead to an erosion of confidence in sterling (Independent, 2008i; also Brummer, 2008c; Oborne, 2008g). The Telegraph followed a similar logic, printing a letter from 16 economists who warned that Alistair Darling's plans to spend his way out of a recession would put the state in such a "dominant position" that it would "stunt the private sector's recovery once recession is past" (Daily Telegraph, 2008g). For the *Times*, the benefit of stimulus as a crisis response had to be off-set by the inevitability of medium term tax increases (Times, 2008g). The *FT* suggested that the pre-crisis output level and its corresponding tax take was artificially high, and as such future output and taxes would likely be lower, requiring lower future spending to balance the budget. 'Austerity' would be the watchword in Whitehall, it predicted (FT, 2008n).

Thus an explicit separation was erected between the worth of fiscal policy in the crisis itself, and its worth in the aftermath. Whilst the use of fiscal policy during this period of uncertainty was therefore generally accepted as a necessary evil across the political spectrum, such action was explicitly delineated as time-specific and thus a fuller ideational shift had already essentially been ruled out before the crisis had even passed. Hence, the dominant ideas of the pre-crisis period, a sense of fiscal precariousness, an opposition to an over-sized state, and a belief in supply-side assistance to business to boost economic growth, were still maintained throughout the crisis but were considered suspended for this brief period due to the danger of wider financial events.

Even at the epicentre of the crisis, therefore, there was a general acceptance in the British broadsheet press that the capacity of the state to undertake hitherto unacceptably interfering actions to stabilise the economy was a temporary necessary evil. Whilst in constructivist terms we could therefore conclude that the crisis did shift the parameters of acceptability, it does not suggest that it eradicated such existing ideational structures; rather it merely rendered them temporarily insolvent. The medium-term future, as characterised by the media, was therefore believed to entail a continuation of the central ideas of the pre-crisis period. For those with a pre-crisis concern about government spending, such a development merely increased the priority given to spending cuts in a post-crisis world, whilst even on the left there was no rhetorical appetite for a statist post-crisis world in which the state played a renewed central role in the ordering of the economy.

6.6 Cures beyond fiscal policy

Whilst we can see some ideational change in the opening up of a debate about the worth of looser fiscal policy during the crisis despite the existence of a 'no room for manoeuvre' discourse in the pre-crisis period, one element of the discussion over the worth of fiscal policy continued to use the

pre-crisis ideational backdrop to inform it. Namely, this was the argument that fiscal policy was not sufficient to solve the crisis alone, and that Britain therefore needed to be sceptical about spending large amounts of money when it was the wider forces of global capital that would ultimately dictate the nature of the recovery. Such an analysis stemmed directly from two of the central pre-crisis discourses identified in the previous chapter: firstly, the existing lack of trust in Labour's fiscal honesty; and secondly, the power ascribed to globalisation as a dictator of economic success over that of domestic policy.

The lack of political trust in the reporting of the success of government programs led to a belief that giving the government a license to spend more would not lead to money being spent in a sensible way, but that it wold provide the government with an opportunity to undertake political bribes by targeting spending at core voter groups. Here then, the lack of confidence in the ability of political class to target policy towards the areas where it was most 'objectively' needed fed into a wider lack of confidence in fiscal policy more generally. The depth of the crisis also solidified, rather than undermined, the pre-crisis belief in the precarious nature of the state in relation to global forces, suggesting that it was important for Britain to maintain its 'pro-business' appeal firstly in order to help ensure recovery.

Fiscal policy alone, therefore, was not deemed sufficient to restore the economy to health; and indeed a relatively weak tool compared to the much larger forces of global capital. Behind this argument lay the assumption that the state was relatively powerless to stimulate the economy, and that only the health of private business and the ability to attract investment was sufficient to return the economy to normality.

This sense of powerlessness served to undermine the worth of stimulative fiscal policy, on the basis that it risked bankrupting the nation without providing any clear benefits. To make a real difference to GDP, Elliott argued, Darling would have to implement a stimulus worth 1% of GDP, something he called 'out of the question given the state of the public finances' (Ibid.). The *Guardian* only partially followed this discourse during this period, however, and the general tone of the paper remained in support of some level of government spending to aid the economy, pointing out that growth 'would be even weaker without the boost provided' (Guardian, 2008m) and with William Keegan suggesting that there is 'no other way' to expansionary fiscal policy, even if it not sufficient to ensure recovery alone (Keegan, 2008b).

At the height of the crisis, the focus of the media on rescue efforts related mainly to the unblocking of credit channels. David Smith in the *Times* noted that Ben Bernanke was well placed as head of the Federal Reserve to lead such rescue efforts, given that he was a scholar of the Wall Street Crash and Great Depression and therefore knew 'that what caused the problem in the early 1930s was the fact that the normal credit channels closed down' (Smith, 2008a). The focus on credit-easing rather than fiscal policy was replicated in several papers, with the *FT* also calling returning lending levels back to pre-crisis levels 'more pressing than discretionary fiscal action (FT, 2008i). Like other papers, the *FT* supported 'additional direct lending programmes, asset purchases and government guarantees' to maintain liquidity (Magnus, 2008). The *Guardian* also called a

resumption of normal lending 'the important element of the recovery' above the level of government spending (Guardian, 2008l).

The *Times* agreed, calling the PBR 'miserably irrelevant' without any major action of credit lines to businesses (Times, 2008l), which it called the 'fundamental problem throttling the life out of the economy' (Times, 2008m). Warner wrote that the ideas being suggested regarding direct government interference in business funding and credit creation 'would have been thought bonkers, and certainly too interventionist to be remotely worth considering' a few months previously, but not 'nothing is off limits in the search for solutions (Warner, 2008j).

The *Telegraph* voiced open support for Vince Cable's plans for more direct government lending to counter-balance the drop in lending from banks (Daily Telegraph, 2008i). The paper also praised Cable for suggesting that mild cuts or stimulus were not the central determinants of the medium-term economic picture (Daily Telegraph, 2008i). Rather, it was the cutback in direct bank lending that was central, making fiscal stimulus "chicken-feed by comparison" (Ibid.). In doing so, the paper suggested that the impact of the crisis on questions of government involvement in the economy were limited to re-establishing credit and financing. The paper also openly discussed the option of nationalising the banks, marking how far the limits of acceptable action had changed over the course of the year (Corrigan, 2008).

Whilst the PBR produced debates about fiscal policy over and above those relating to credit, the intervention of Mervyn King in December to suggest that "the single most pressing challenge to domestic economic policy is to get the banking system lending in any normal sense" was openly supported by the *Telegraph* as 'rapidly becoming the consensus' (Daily Telegraph, 2008k). The *Daily Mail* agreed, suggesting that it would 'take a powerful global rebound to pull the UK out of its economic funk; (Daily Mail, 2008f), demonstrating the relative weakness of the government to use fiscal policy to improve economic growth. It suggested the VAT cut in the PBR was too meek to have any real effect in enticing customers to spend more (Mail on Sunday, 2008) As with the other papers, it called credit 'the biggest problem currently facing businesses' (Frost, 2008).

Because it was deemed too 'difficult to calibrate the extent to which this crisis will hold back economic growth' in the autumn of 2008 (King, 2008), a greater focus was placed to returning to stability and freeing up the blocked credit markets than on medium or long-term economic restructuring. The *Independent* suggested that 'unblocking the credit markets should come before capital spending' and that large spending projects should not be rolled out quickly before being thought out, for 'when governments take big and costly decisions in a hurry, those decisions often turn out badly' (Independent, 2008h).

Following on from section 6.5 above, this section has highlighted that based on the perception of the limited scope for government spending, and also on wider ideas of the importance of global capital in shaping economic possibilities, the idea of a demand-side stimulus having the capacity to return the economy to growth was at best a marginally supported idea, and that as such support for more government initiatives in returning the economy to growth lay in the monetary and credit

spheres, which as outlined in section 6.2 were more open to ideational change than the fiscal sphere. These ideas remain partially based in the anti-statism and 'no room for manoeuvre' discourses of the pre-crisis period highlighted in the previous chapter, again demonstrating the extent to which the crisis was unable to completely remove the pre-crisis ideational framework upon which much fiscal policy analysis took place. Thus, again, we can note that whilst the crisis opened a political space for new ideas within the financial sector of the economy, a generalised crisis of ideas is still not fully evident even during the period of greatest economic turmoil and uncertainty. Rather, there is a significant drawing on pre-crisis discourses of the fragile state in the global economy, the incapacity of politicians to spend money with integrity, and a sense that pre-crisis developments limit the scope for major fiscal stimulus and government intervention in corporate capitalism.

6.7 Conclusions

This chapter has demonstrated that the financial crisis did produce some forms of uncertainty in the economic thinking of the broadsheet press, but that this uncertainty did not entirely erode much of the dominant economic thinking of the pre-crisis period. Whilst in certain sectors the norms of what was to be considered politically acceptable were transformed very rapidly, particularly in relation to nationalisation, strict supervision of vulnerable financial institutions and the potential to artificially loosen up credit channels, in others there was a more broadly consistent pattern of analysis with the preceding period.

The analysis that this was a major crisis moment for Western financial capitalism was certainly pronounced and commonly held. Thus the dominant underpinning ideas of the pre-crisis period, of relaxed regulation and the belief that financial markets were fully efficient were rapidly reconfigured in the autumn of 2008. The general lack of sympathy with the banks for their plight cemented this acceptability of discourses of change in financial regulation.

The potential for epistemic uncertainty to allow for new economic ideas to flourish, a central tenet of existing constructivist scholarship, is certainly evident to some degree through this period. Particularly in relation to Labour's planned fiscal stimulus in the 2008 PBR, the necessity of all possible means being used to support the fragile economy meant that existing concerns about fiscal prudence became over-ridden through necessity. As such, no clear growth model pre-dominated with acceptability and credibility being passed to anything deemed likely to promote economic stability. In that uncertainty was evident over the likely successes of different growth models, the

situation could be described as conforming to the constructivist model of a structural vacuum allowing for re-configurations of patterns of existing beliefs.

However this chapter has also suggested that this uncertainty was not as deeply rooted as Constructivist scholarship would suggest occurs in the case of a generalised economic crisis. There was uncertainty about how the economy would react to stimulus, and what the nation's scope for wider public borrowing was, but this uncertainty did not permeate into all aspects of political economy, and was indeed largely contained within the financial sector. Here there was a very real separation discursively between financial capitalism and corporate capitalism, with the blame and uncertainty resting primarily in the former, minimising any diffusion of the third order uncertainty into the understanding of other sectors of the economy.

We see in this period a continuation of three of the major discourses of the pre-crisis period in relation to the roles of the state and the economy. The first element of stability related to the fear that government involvement might burden businesses through expensive regulation and red tape was maintained, limiting the scope for alternative discourses arguing in favour of the necessity of state involvement to support businesses. The second element related to the lack of belief in the capacity of the state to efficiently alter economic outcomes; the consequence of this being that the future of the economy was positioned as being dependent on the return to health of the banks and the credit supply, and not any fiscal demand-management on behalf of the state. Thirdly, the lack of trust in Labour's economic management continued to underlie much of the concern about a renewed role for the state in the demand side of the economy, for it was assumed that any attempt that Gordon Brown made to boost spending to support the economy would only be done for narrowly political objectives and not therefore have a wider beneficial economic impact.

The lack of more cross-cutting ideational uncertainty in the wake of the crisis meant that much of the blame for failure, and the search for new ideas to govern the economy into the future, therefore fell on monetary policy and financial regulation. Blame fell both on banks, for having been greedy and too eager to place risk ahead of discretion, and on politicians for failing to regulate them properly. Thus we fail to observe a more systemic crisis in economic ideas during this period, even as uncertainty dominated the short-term economic situation. In the next chapter, as the immediate risk of financial collapse abated through 2009, we will analyse the extent to which the ideas of recovery continued to rest on the ideational underpinnings of the pre-crisis period.

The implications of this for constructivist institutionalist research on crises is that the 'structural vacuum' argued for by Blyth is rarely absolute. The chapter has answered research question 2.) by showing that whilst the crisis did create a shock in which new approaches gained short-lived attention, this was never completely removed from its pre-crisis context, and indeed forecasting in particular remained focused on the same set of objectives as had dominated beforehand. Adopting a looser approach to epistemic uncertainty and the structural vacuum need not go against constructivist principles, and therefore provides a valuable addition, rather than a critique, to this school of thought. The following chapter will continue this theoretical addition to the school by highlighting ideational reversion in the post-crisis context, suggesting again a greater stickiness of

ideas than perhaps to date had been recognised. The continuation of these ideas of the pre-crisis context gave a greater social acceptability to the policy programme that became known as 'austerity', whilst providing a harder-to-justify context for Keynesian stimulative principles.

Chapter 7: Ideational Reversion: January 2009 – May 2010

7.1 Introduction

The previous two chapters have respectively highlighted the dominant political economic narratives of the pre-crisis period in British politics, and the partial shift in the acceptability of state involvement in the economy in the crisis period of 2008. Significantly, however, it was noted in the previous chapter that the press' acceptance of greater state borrowing and spending in the autumn of 2008 involved a time-limited clause, which suggested that once the danger of financial collapse had been avoided expensive state assistance must be wound down. Whilst the crisis caused immediate questioning of the role of financial regulation and the practices of the banking sector, the blame for the crisis was effectively siloed within this segment of the economy, leaving the dominant ideas governing other sectors un-touched.

This chapter answers research question 1.) by highlighting a return to pre-crisis discourses in the aftermath of the crisis, and as such demonstrating how these discourses came to shape the acceptability of a programme of austerity over that of a Keynesian stimulus. The chapter is able to answer research question 3.) by using a constructivist logic to argue that the austerity narrative had a greater social source of legitimacy stemming from its closer alignment to pre-crisis ideas, allowing for it to be adopted in a way that a more Keynesian approach did not. This therefore allows us to arrive at the second major empirical conclusion of the thesis; that a process of 'ideational reversion' can be noted in the post-crisis discourse, whereby the ideas of the pre-crisis period also came to shape the acceptability of governance practices in the post-crisis context.

The victory of the orthodox paradigm was neither complete nor universal; disputes and competing ideas continued to feature across broadsheet coverage. However, the fragmentation of support was not evenly split, with a far greater proportion of the economic writing on the crisis highlighted the need for budget restraint than supported spending, with even those who supported demand-side support suggesting that crisis-resolution was not possible through this route alone, undermining its significance. Moreover, as the dispute on the academic level between Keynesians and the orthodox school rumbled on, so its significance was belittled in the eyes of the media, under the logic that if expert economists couldn't agree on the issue, what chance could the general public have in deciding between competing frames when deciding upon their political preferences? Thus

the absence of economic consensus, often taken in the existing constructivist literature to be a key step in the rise of new paradigms, actually favoured continuity in that political analysis tended to fall back on existing frames of understanding in order to distinguish between the political parties.

This chapter will begin by demonstrating the development of the budget deficit as the primary focus of economic reporting in early 2009, replacing the uncertainty of the financial system which had pre-dominated the previous autumn. It will chart the increasingly central discourse of 'no room for manoeuvre' within such discussions, highlighting the ideational continuity from the pre-crisis period. Whilst a Keynesian counter-narrative did exist in this period, it remained less prominent.

The strategically selective advantage that the 'austerity' discourse appears to have held here is two-fold. Firstly, the discourse was advantaged by being able to tie itself more closely to the politics of the pre-crisis context that its Keynesian alternative. The issues of 'no room for manoeuvre', an over-regulated economy, an over-bearing state sector, the powerlessness of the state in a globalised world, and the lack of ability for politicians to spend money effectively all fit into the justificatory frame of austerity, whilst they ran counter to the intentions and beliefs of the stimulative approach. Secondly, the 'technical' social scientific question over which policy approach would benefit the economy the greatest amount in the short-term actually receded in importance closer to the election as the lack of agreement between economists on the issue - a structural vacuum in constructivist terms – increased the public perception that the importance matters at the up-coming election were actually the more medium-term ideas of the parties about the role of the state in the economy. In this regard, there was greater appeal for the orthodox approach in that it gave discursive primacy to the role of private business and supply-side measures in returning the economy to growth, whilst the Keynesian approach foresaw a greater role for the state in boosting the conditions for growth through spending and the 'picking of winners'. Again, we see a great deal of continuity in crisis here as the orthodox approach maintained its greater appeal than the Keynesian one.

In developing this overall insight, the first section of the chapter highlights the increasing ubiquity of discourses of 'no room for manoeuvre' in the immediate post-crisis period. As such the reversion to pre-crisis ideas to explain and generate policy proposals for dealing with the crisis is demonstrated, highlighting the failure to the crisis to reconfigure existing ideas of the role of the state in society, or of the variables through which economic credibility and feasibility were measured. The second half of the chapter further explores this ideational reversion by comparing evolving discourse in the 2009-10 period with the analysis of pre-crisis discourse presented in Chapter 5. Mirroring the thematic structure of Chapter 5, debates surrounding the need to cut bloated public services to achieve growth are first reviewed. Following on from this, the link between rhetorics of political trust and spending cuts are reviewed. Finally, the chapter focuses on the discourses of globalisation and the weakness of the state. These foci will allow the chapter to conclude that a process of 'ideational reversion' took place in the post-crisis period.

Following from this empirical analysis, two conceptual conclusions are also advanced. The first conceptual insight is that the epistemic uncertainty generated in a time of crisis does not necessarily lead to a paradigmatic shift in existing ideational framework. The second insight is that the successful siloing of the perceived blame for the crisis within the financial and political landscape may have aided this process by limiting the generalised sense of a policy failure within the existing governance structure.

7.2 'No room for manoeuvre' and its critics

As we saw in the previous chapter, the danger of systemic financial collapse in the autumn of 2008 temporarily suspended the view that the state had no fiscal scope to react to economic events. Alistair Darling's 2008 PBR, which provided a small stimulative fiscal boost, saw the media withhold judgement until its full effects were shown. In other words, the existing ideological or ideational commitments of the media were suspended because of the uncertainty that had arisen. However, as the immediate danger of financial collapse receded in early 2009, and as thoughts began to turn to the route back out of recession and indeed towards the necessity of any medium-term realignment of the British economy as a result of the crisis, the discourse of 'no room for manoeuvre' returned.

Rather than having been eroded by the crisis, the existing discursive beliefs were strengthened by it. The belief that cuts would have to be made to state spending became universal, although debates over the timing and structure of these cuts existed. However the depth of the 'budget crisis' that was largely accepted to exist during this period meant that even those of a Keynesian disposition largely limited themselves to opposing immediate cuts, as opposed to calling for a fuller stimulative fiscal expansion. Whilst a social scientific debate existed in the media throughout this period, with Keynesians and monetarists reaching different conclusions about the role of fiscal policy in the crisis based on different perceptions of risk, this debate began to grow stale closer to the election as it became clear that even economists could not agree on the issue. Thus the debate about the worth of state spending fell back on largely normative criteria, which invariably linked to pre-crisis discourses of the value of state spending and the proper size of the state. The most pressing concerns surrounding the state of British governance in the pre-crisis period therefore continued to shape the response of the media to the crisis, rather than having been reconfigured by them. In the paragraphs below, I review the manifestations of these dynamics across broadsheet media.

7.21 'No room for manoeuvre' and the 2009 Budget

The run-up to the 2009 Budget was characterised by discussion in the media about the scope for fiscal policy in helping to support the economy. As the scale of the budget deficit became clearer, the general consensus was that there was now no scope for any further fiscal stimulus despite the severity of the private-sector slump. Thus, rather than having been altered by the financial crisis, the discourse of 'no room for manoeuvre' was strengthened by it. New empirical developments, such as the lack of certainty on the capacity of international credit markets to fund the British deficit, were analysed using the same frame of reference, highlighting the durability of such discourses through the crisis period. Significantly, the breaking of Labour's golden rules was significant in shaping many of the perceptions of the scope for expansionary fiscal policy.

For the *Times*, the lack of room for manoeuvre now meant that there was a danger of international investors 'balk(ing)' at the creditworthiness of the UK's sovereign debt (Times, 2009c), with even the more Keynesian commentators on the paper such as Anatole Kaletsky now 'rejecting the case for tax and spending measures in the Budget' (Times, 2009g). There was now a 'clear danger', the paper argued, 'that spiralling borrowing as revenue wilts will trigger panic in the markets over whether Britain can make ends meet' (Times, 2009p). The paper framed the principal role of the Budget as being 'to convince investors who buy British government debt that the ship will be bailed out eventually' and to 'chart a course back to longer-term national solvency' (Times, 2009m; 2009r respectively), and therefore presented deficit reduction as the principal economic imperative to be tackled, as opposed to economic growth, financial stability or any other criteria. Matthew Parris described the task for the Tories as one of 'how to bring down state spending before it destroys national solvency' (Parris, 2009a), suggesting that the question was no longer one of whether to cut, but of when (Parris, 2009b). It now described the relatively modest fiscal expansion of the 2008 PBR as 'going for broke, allowing Treasury finances to plummet into the red on an almost unprecedented scale' (Times, 2009p).

As with the *Times*, the *Telegraph* gave itself the voice of investors by claiming that 'the international investment community will refuse to fund (Darling's) plans by buying gilts' (Ibid.), and raised the danger of a 'gilt strike' leading to the IMF having to be called in to provide assistance (Daily Telegraph, 2009i). Such a discursive claim was of course particularly poignant in the British case given the need for the previous Labour government to call on the IMF in 1976, something the *Telegraph* made an explicit comparison to (Daily Telegraph, 2009j). The paper continued that the 'markets will punish (Darling) mercilessly' if severe spending cuts were not made (Ibid.), and warned that a ratings downgrade 'would be devastating for the economy' (Daily Telegraph, 2009n; also Conway, 2009a).

The *Daily Mail* similarly focused on the 'mess', 'basket case' and 'wreck' of the national finances (Daily Mail, 2009c, 2009b, 2009g respectively) and the need to send a clear message to the

markets that spending would be cut back. Similarly to the *Telegraph*, it suggested that the primary focus had to be on providing a roadmap to fiscal sustainability (Daily Mail, 2009a), warning that debt would be 'more dangerous to add to' than before (Daily Mail, 2009b). The 'efficiency savings' Darling outlined were deemed insufficient and lacking in credibility given Labour's spending record; rather what was needed was a more 'radical plan' that took 'an axe to the public sector' (Brummer, 2009b). Again, the paper chose historical comparisons of fiscal conservatism to support this. The combination of fiscal consolidation and political trust returns here tangibly, and will be returned to below. The paper argued that austerity was not a 'terrible new age' but 'necessary medicine for an economy that has overdosed on the state' (Daily Mail, 2009e). Brummer noted that 2008's PBR had already represented 'a massive fiscal stimulus', limiting the scope for another one in 2009/10 (Brummer, 2009a).

Whilst the Financial Times was not quite as apocalyptic as either the Daily Mail, Times or Telegraph in early 2009, the paper maintained some of the same foci of discussion and likewise made the deficit and public spending a central concern. The paper called the deterioration in the public finances 'alarming' and highlighted the risk of the UK slipping 'back to its 1970s and 1980s pariah-status in government bond markets' (FT, 2009b, also 2009f). The paper reported 'the British government's biggest challenge is to avoid a funding crisis', and suggested that 'markets will not let (Alistair Darling) boost government spending' (FT, 2009j). Niall Ferguson reported that the bond market was 'quailing' in the face of huge deficits for what was a relatively 'small' recession (Ferguson, 2009). The FT reported that business' view of the Budget was 'largely hostile...concern that the chancellor was taking risk with the public finances - and fear of tax rises - outweighed qualified enthusiasm for specific measures' (FT, 2009g). Goldman Sachs was quoted as saying that 'the public finances have been blown out of the water', and the fiscal thinktank IFS also focused only on the debt as the major issue of the Budget (FT, 2009i). Thus the extent to which external 'experts' were likewise being included in newspapers' narratives about the deficit, and the increasing size of the discursive focus on this, served to reinforce the image of this issue as the primary one of importance.

Even Martin Wolf, generally supportive of looser fiscal policy in the autumn of 2008, was now questioning the capacity of the markets of take on extra government debt. Whether Labour's policies were realistic, he suggested, was now a question for markets, and suggested that whilst the 'markets have been forgiving' it was not certain that they would continue to do so; 'markets tend to behave – until they cease to do so' (Wolf, 2009c). Buiter too, suggested that discretionary fiscal stimulus only made sense if it could 'avoid spooking the markets' (Buiter, 2009a). Thus the uncertainty of the previous autumn in relation to the stability of banks had been replaced by an uncertainty about the length of time that investors would continue to buy government debt. 'There is a chance investors will run scared if they can see only red ink stretching into the future', the *FT* concluded (FT, 2009j). This uncertainty legitimised the call to either cut now, or at least set out a credible timetable for cutting soon, and undermined the acceptability of stimulative policies in the eyes of the media. The emergence of need for greater medium-term clarity of spending plans as a

major discursive theme of this period is expanded upon in 7.5 below. It also demonstrates the primary voice given by the media to the international credit markets as the agents influencing the state's economic policy autonomy.

The *Guardian*, the main anti-austerity voice during this period, was even caught up in the belief that there was little room for manoeuvre for Alistair Darling in the run-up to the Budget. 'There is little doubt' it reported, that Darling 'has very little room for more fiscal stimulus packages in his...budget' (Guardian, 2009a). The level of the deficit was 'start(ing) to become unsustainable because financial markets would lose confidence and refuse to buy any more government debt except at very high rates of interest', it reported (Ibid.). The call from city experts to avoid further stimulus was seen as ending the capacity of Brown to undertake such action (Guardian, 2009b; also Times, 2009f), and Jackie Ashley declared Darling to be 'more realistic' in being cautious about the scope for extra stimulus (Ashley, 2009).

The paper also summed up the budget as signalling a return of bankers as the masters of the universe, with government having little option but to follow their demands for no further stimulus (Roberts, 2009). Larry Elliott, like reporters from the more conservative papers, expressed fear that lack of sustainability 'of the public finances would lead to higher long-term interest rates, choking off recovery', and that the lack of attention being paid to this danger by the government risked 'the wrath of the credit-rating agencies' (Elliott, 2009a). He called the risk of gilt yield increases pushing up interest rates on mortgages and bank loans 'the facts of fiscal life' that 'would risk pushing the economy back into recession' (Elliott, 2009c). The difference to the more conservative press, however, was that Elliott accepted that 'the still fragile state of the economy means there is a case for delaying fiscal stringency until the risks of a double-dip recession have faded' (Ibid.). Regardless, the differences between the papers remained limited to the issue of the timing of cuts, not of their inherent necessity, nor the importance of engendering growth to cut the gap between spending and tax intake.

Thus the medium-term timeframe of a reduced role for the state was developing in ubiquity across the journalistic spectrum. The near universal assumption that budget deficits would lead to large increases in interest rates making public debt unsustainable is also notable in its consistency across the spectrum. The certainty that bond markets would punish 'irresponsible' governments is particularly surprising given the shifting sands of the global economic landscape in the previous year. Again, uncertainty about whether the existing norms of the Great Moderation period still applied in this changed landscape is relatively muted, again displaying ideational continuity through the crisis.

Alongside the significance of international credit markets, the issue of state spending in the precrisis period was also a significant shaper of post-crisis discourse. The scope for Keynesian expansion was often precluded on the basis that it could only be done when sufficient saving has been done in the boom years (Times, 2009a; 2009u), much as several papers had argued before the recession. The *Times* characterised the public attitude as being one 'no longer willing to tolerate government spending that spirals out of control' (Sylvester, 2009a), suggesting implicitly that this is what Gordon Brown's government had allowed to happen. Several papers called the post-Budget era a 'new age of austerity' (Independent, 2009f; Elliott, 2009c; FT, 2009b; Times, 2009u), which was often unquestioningly accepted as a necessity.

Likewise the *Independent's* pre-budget coverage focused primarily on the lack of room for manoeuvre that Alistair Darling had to stimulate the economy (e.g. Independent, 2009a). What had happened to the public finances was not within past experience, argued McRae (McRae, 2009a). Meanwhile Warner highlighted that the 'recession is already self-evidently much more severe than forecast in the pre-Budget report', but argued that this highlighted primarily the weakness in the public finances as opposed to a signal that more support was needed to aid the economy (Warner, 2009a; also McRae, 2009e). The *Independent* now described 'how to limit state spending' as 'the main event in politics' (Independent, 2009g), and like other papers accepted that 'Britain's public finances are out of control' (Independent, 2009j; Independent, 2009k). It also accepted the notion that Britain had spent too much during the boom to be able to afford Keynesian stimulus in the bust (Independent, 2009k). The state of the public finances 'threaten recovery', it argued (Independent, 2009o). Even whilst noting that Labour remained ahead in the polls on economic leadership, the paper called Gordon Brown 'unrepentant' in failing to change his approach (Independent, 2009h).

Asking economists what they would do in the budget, two thirds in the *Telegraph* said there was no room for fiscal expansion (Daily Telegraph, 2009e), and the paper utilised the intervention of Mervyn King, who publicly argued against stimulus, to suggest that such an approach had therefore become untenable (Daily Telegraph, 2009f; also Daily Telegraph, 2009c). The paper regularly employed dramatic metaphor and cataclysmic language to frame the issue of the deficit, describing the nation as 'drowning in government debt' (Daily Telegraph, 2009f) which was an 'underlying cancer' (Sunday Telegraph, 2009c), and equating the application of stimulus as 'analogous to trying to cure alcoholism by going binge drinking' (Daily Telegraph, 2009h). Reece declared that the 'fiscal lifeboat has been holed. It can't be launched, so it won't be launched' (Reece, 2009a). Randall declared in the aftermath of the Budget that 'a ball and chain of spirit-sapping debt has been clamped to the nation's future. We are all serfs now' (Randall, 2009b). As with the pre-crisis opinion of the paper, Reece declared that Keynesianism was only allowed when the government had saved up in the boom time to help pay for stimulus in the bust (Ibid.; also Sunday Telegraph, 2009b, 2009e). As such the central economic problem of the time was already framed as being too much government spending and debt by the conservative media.

By framing existing policy as an ongoing Keynesian experiment, and a rather large one at that, the *Times* set the metrics of success as a 'gamble on growth' (Times, 2009p). In doing so, the previous autumn's growing budget deficit, rather than being seen as an inevitable response to an equal-and-opposite private sector contraction, was constructed as a positive-sum Keynesian expansion that would have to produce a strong economic growth to be gauged a success. Moreover the framing of the central problem as one of the budget deficit meant that the solution had to come from deep cuts; 'efficiency savings will not nearly be enough to bridge the gap' (Times, 2009r). 'The Chancellor's further options look restricted in the extreme', the *Independent* commented,

noting that fiscal stimulus had already been attempted and in doing so implied that this was not a trick that could be repeated (Independent, 2009c).

The discourse that developed in the *Daily Telegraph* through this time also reflected core elements of the *Times*' analysis. In particular, by presenting post-crisis economic policy as a binary choice between cutting spending or increasing taxes, the paper offered a zero-sum analysis in which the latter would risk 'choking off the recovery' (Daily Telegraph, 2009k). The paper therefore began to argue in places that the size of the budget deficit, rather than helping to prevent a further drop in GDP growth, was now a future threat to the recovery because of the likely rise in future taxes to cover it (Times, 2009i). The *Independent* likewise suggested that 'by increasing government debts, we have increased the tax burden on future generations' (Independent, 2009n).

Whilst much of the political debate was internal, the utilisation of external sources to generate credibility for given discourses was also significant. Sometimes these came from external sources of authority like the IMF, at other times the structures of rectitude developed by the New Labour administration were used against them to justify a more limited role for state spending. For instance, the failure of the G20 to agree a global stimulus in early 2009 was implied by the *Telegraph* to mean that it was opposed and that there could not be such an approach in Britain (Daily Telegraph, 2009b). The paper continued to use Gordon Brown's measure of 40% debt/GDP as a gauge of what was 'sustainable', therefore concluding that it would take until 2032 until 'debt freedom day' was reached under Labour's plans (Daily Telegraph, 2009o). This is significant, for it demonstrates that the boundaries of credibility which were drawn by politicians in the Great Moderation era, setting out a variety of indicators to define such credibility, were still taken as the limits of acceptable action despite the changed circumstances and economic conditions to those in which they originated.

7.22 The alternative discourse; maintaining the need for stimulus

During the spring of 2009, there maintained a reasonable opposition discourse to that of 'no room for manoeuvre', which argued that the economic circumstances were still dire enough for the government to take all necessary action to keep the economy afloat, or that the economic logic of cutting was weak anyway. Whilst often not being supportive of government policy in general, it did at least back the economic logic of using government stimulus to aid demand and fear that reducing government spending now would actually damage confidence by removing such demand from the economy. It was most present in the left-leaning *Guardian*, but the more liberal economists of other

papers warned against the economics of fiscal consolidation too, as did the *Financial Times* in general.

Seamus Milne, in the *Guardian*, told readers to 'ignore the Tories. You can't cut your way out of a slump' (Milne, 2009a), calling the deficit hysteria a 'brilliant diversion' from more serious matters (Milne, 2009b; also Hari, 2009). Seager wrote that immediate fiscal tightening 'would be a terrible mistake – the last thing you do in a downturn' (Seager, 2009; also Hutton, 2009a, 2009d). The paper tried alternative ways of framing the economic problem, with Danny Blanchflower, formerly of the BoE writing of a 'jobs crisis' (Blanchflower, 2009a; also Independent, 2009b) and warning that Cameron's alternative risked a 'death spiral of decline' (Blanchflower, 2009b). Will Hutton, in the *Observer*, was particularly prominent in this Keynesian push, warning that cutting now would turn a recession into a depression (Hutton, 2009b, 2009c), although even he accepted the need for cuts at some point in the future (Hutton, 2009c). Likewise Larry Elliott, whilst noting that 'austerity comes after a recovery' still accepted the inevitability of some cuts and put the main divide as being the 'when', rather than the 'if' question (Elliott, 2009b). American economist Robert Reich asked 'why the deficit hysteria?', arguing that 'large deficits this year and next, and even the year after that' might be needed to get the economy back on the growth path (Reich, 2009)

Jonathan Freedland argued that on fiscal policy the conservatives were 'wrong and deserve to lose' the election based on a Keynesian argument (Freedland, 2009; also Elliott, 2009d). Toynbee warned that 'just as the economy is fluttering to life, the Conservatives' one great priority will be to put it all at risk with immediate deep cuts' (Toynbee, 2009a). Even she, however, admitted that Labour's ability to shift the argument towards acceptance of the need for government support for the economy was dependent on 'anyone listening', an implication of the significance of the lost political credibility that Labour had suffered, and the impact of this on their ability to generate credibility for a recovery strategy (Toynbee, 2009b). William Keegan, like Hutton, presented classically Keynesian arguments to maintain support for government stimulus, asking readers 'who do you think is right, Osborne or Keynes? No prizes' (Keegan, 2009a, also 2009b). Again, however, he bemoaned that such a presentation 'does not seem to have penetrated far into the public consciousness' (Keegan, 2009a).

This call for a delay to cuts on a Keynesian basis was most pronounced from the *Guardian* and *Observer* in the autumn as the electoral battle-lines became clearer and economic policy looked likely to play its most important role in several elections. The paper sought external validity for this argument by quoting the IMF as saying that "one of the key lessons from the experiences of similar crises is that withdrawing policy stimulus too early can be very costly" (Guardian, 2009i), and interviewing four economists on the economy's green shoots all of whom called for maintained fiscal support (Guardian, 2009h). They noted that the cost of servicing the national debt had actually decreased in the last year, and that it was better 'to live with horrible risks that the certainty of a crushed economy' (Guardian, 2009j). Keegan maintained that 1930s-style deflation remained the major threat to be tackled, for which stimulus was the obvious answer (Keegan, 2009c), whilst

even Larry Elliott called Cameron's approach 'dangerous and economically illiterate' (Elliott, 2009g). To 'slash spending now would be madness' the headline commented, calling instead for patience whilst the stabilisers took effect (Ibid.).

The paper also harked back to 1937 as a previous episode when fiscal support was withdrawn too early resulting in a lapse back into recession (Guardian, 2009k). The paper also wrote that 'it is hard to find any economists...who believe that the fiscal tightening need be as rapid or as severe as the Conservatives say' (Ibid.). Keegan called the deficit 'an integral part of the solution', as opposed to the problem itself (Keegan, 2009d).

Discussion of the rising yields on government debt was also not universally understood to highlight marked worries about the size of budget deficits. Several noted that rising yields actually represented the more positive phenomenon or investors re-gaining their appetite for private-sector risk and thus reducing the demand for government debt, which increased its yields (Warner, 2009e; also Wolf, 2009d). It was thus a function of improved economic confidence, rather than reduced confidence in national solvency. With fear of the ratings agencies developing through 2009, Warner asked 'what the point' of them was, given that they are merely 'lemmings' who follow trends, and cast doubt on the validity of their figures (Warner, 2009d). He also reported that the 'long-term prognosis for the public finances may not be as bad as widely assumed' because the taxpayer was now due to lose a lot less on bank bailouts than had initially been expected (Warner, 2009f). Even Warner, generally a fiscal conservative, warned that 'deep cuts....might make the economic malaise and therefore (the) state of the public finances even worse' (Ibid.). The real 'long-term fix' needed was 'a sustained economic recovery' which would increase tax revenues (Ibid.).

The paper criticised Cameron's attempt to seek external validity for fiscal consolidation by looking to Canada's previous example (Independent, 2009m) whilst Richards noted that his approach 'leaps over a range of economists on the right as well as the left who fear that big spending cuts imposed too quickly will damage the economy' (Richards, 2009b). Significantly, again though, even those calling for a more Keynesian approach were still obliged to highlight that cuts would still be necessary, and that this was merely an issue of timing (e.g. McRae, 2009g, 2009h; Richards, 2009a), with Richards stating that 'there is not a single economist who does not believe that substantial cuts are required, but there is an important debate about timing and scale' (Richards, 2009b). The paper also expressed that private sector debt was the primary reason holding back growth, and thus the natural response should be to 'drink in the pure milk of Keynesian orthodoxy here... the Government had to take up the slack' (Independent, 2009q). However, reflecting centre-left concerns, Richards repeated the belief that 'away from the economic pages there is a consensus in the British media that unprecedented spending cuts are required and the sooner the better' (Richards, 2009b).

Calls for a more Keynesian approach were not completely absent amongst the more business-focused and right-wing papers either, however, although they were generally limited to one or two correspondents. In the *Times*, as was stated in section 7.21 above, Anatole Kaletsky was starting to shift his position on the value of extra spending, but had still not come to accept immediate cuts

(e.g. Kaletsky, 2009f, 2009g). Describing the Budget speech as 'a good impression of Mr Bean', he nonetheless said that such an achievement on behalf of Darling was 'matched only by David Cameron, who managed in just 15 minutes to prove that Britain under the Tories would fare even worse' (Kaletsky, 2009b). Nonetheless Kaletsky moved from highlighting fiscal policy primarily, to focusing on business confidence by criticising Darling for announcing future tax rises whilst trying to increase investment and confidence and attacking the cyclically most important industry, finance (Ibid.; also Independent, 2009i). He went on to state that given that 'no one can say whether the benign scenario of rapid recovery or malign one of long recession will play out...the best economic policy scenario is simply to wait and see' instead of making drastic cuts (Kaletsky, 2009c). This provides a rare example of the kind of uncertainty posited in 2009 that was used to justify government action during the crisis period of 2008. Kaletsky called large budget deficits the consequence, not the cause, of the nation's economic problems, and warned against cutting spending until adequate growth rates returned (Kaletsky, 2009e). Even in December, he maintained that the 'morally righteous' fiscal hawks were hypocritical in showing faux concern over leaving debt to future generations whilst at the same time being willing to leave the costs of climate change to the same cohorts (Kaletsky, 2009i).

Even in the *Telegraph*, Bootle provided a single more Keynesian voice, arguing that logic demands we 'spend our way out of it' (Bootle, 2009a, also 2009c). Whilst he agreed that government spending in general was too high, in the current context it was no longer relevant given the shortfall of private spending (Bootle, 2009a). He was not consistent in this argument however, suggesting elsewhere that increasing confidence was the only thing the government could do to help the economy (Bootle, 2009b, 2009d, 2009e). Wadhwani criticised Mervyn King's intervention on fiscal policy and argued that it plays an 'important role' in balance sheet recessions of this kind (Wadhwani, 2009). Jeremy Warner, upon moving to the Telegraph, also wrote that 'a longer application of fiscal stimulus than would traditionally be thought wise' might be needed in this context, given the importance of growth to avoid a debt trap (Warner, 2009h, 2009g respectively). Skidelsky also criticised Osborne for offering prescriptions that whilst 'right and sensible in conditions of full employment', were 'wrong and wrong-headed when there is heavy and persisting unemployment' (Skidelsky, 2009). Even the paper's editorial once admitted in October 2009 that 'slamming the fiscal brakes on too hard before the economic recovery has become fully established could push us back into recession' (Daily Telegraph, 2009s), something that was reported as also steadily becoming public opinion (Conway, 2009c; Warner, 2009j).

In the FT, Martin Wolf provided a more consistently Keynesian position on the issue of the deficit and spending cuts, too. 'The government is surely right to borrow its way through the crisis', he wrote in February (Wolf, 2009b). Whilst he noted that the 'Conservatives are right to criticise the government's record as manager of the public sector...they are wrong to take so pre-Keynesian a view of public finances' (Ibid.). The paper also suggested that the zero lower bound of monetary policy was limiting its traction and the 'little difficulty' the government was having in funding its deficits meant that fiscal policy could and should be left loose for longer (FT, 2009d). Wolf agreed,

talking of the 'sustainability of stimulus' whilst the repairing of balance sheets takes place (Wolf, 2009e). As the *Guardian* had done, Clarke invoked 1937 and warned that it 'it is dangerous, as well as tempting, to forget how bad things looked only a short time ago' (Clarke, 2009; also Brittan, 2009a). Thus he highlighted the extent to which the perception had taken hold that the financial crisis was over, and with it the reduced limits of discretionary governmental action. Brittan called the budget crisis 'largely imaginary' (Brittan, 2009a; also Stephens, 2009c). Wolf criticised both the 'Ricardian equivalence' and 'crowding out' theses of the dangers of government deficits, and called deficits the natural 'result of the cutback in the private sector's spending, not a cause of it' (Wolf, 2009f). By the autumn, even Wolf was discussing the dangers of the bond markets but maintained his position that removing stimulus too early was more dangerous (Wolf, 2009g).

Whilst it is therefore important to note that a period of discursive contestation undoubtedly took place during this period, with very different ideas of economic theory being utilised to support different policy approaches, key differences in the framing of problems and different connections with existing political narratives existed between the two approaches. Notable from the above collection of pro-Keynesian media advocacy is the technical nature of the language; the credibility of the approach is derived purely from economic theory, and not from other normative criteria as was the case with the orthodox approach. Significantly, the Keynesian school even partially accepted the frame of public spending as a problem in the medium-term timeframe. As such a distinctive separate Keynesian narrative did not emerge as a popularly accepted paradigm.

The significance of the strategic nature of the political process is also worth noting here briefly. One could suggest that in order to contest fiscal consolidation more concretely, the political left should have opposed the frame of the necessity of medium-term cuts also. However, given the Labour Party's concern with regaining the economic credibility it lost with the banking collapse, such an approach would have been politically unfeasible and risked cementing the charge of being spendthrift that they were trying to avoid. In the following phase of the 2009-10 period, we observe a further consolidation of this ideational reversion as the ubiquity of the orthodox discourse increased and developed both an economic but also a moral nature, whilst the Keynesian alternative lacked this latter element.

1

¹² Interestingly, in the American media, both sides presented their arguments in very technical form relating to economic theory – normative discourses from the fiscal right about the oppressive size of the public sector were much less evident in the post-crisis period. In this different discursive context, a classically Keynesian boost to public spending in the form of President Obama's ARRA was widely supported, and only later did a counter-narrative from the right start to question the basis of the ARRA and hence the wider value of public spending in a recession.

7.3 The Late 2009 - May 2010 period

Whilst there was a developing discourse of 'no room for manoeuvre' in early 2009 based on the high borrowing figures of the 2009 Budget and uncertainties over the bond markets, this was still not completely universal during that period. However it became much more so in late 2009 and early 2010 as the economy began to emerge from recession and the justification of waiting for the recovery therefore became weakened. Commentators who had been willing to give the government the benefit of the doubt began to accept the need for greater action on spending cuts, and the battle of ideas began to be won discursively by those of an orthodox persuasion who highlighted deficit problems as the central problem unconnected to other economic issues. In the paragraphs below, I review these shifting narratives of post-crisis economic management.

The *Times* provides a prominent example of a paper which in early 2009 had provided a mixture of opinions on the issue of spending cuts, but by early 2010 had swung almost entirely over to the side of calling for immediate cuts. It warned of default in the run-up to the 2009 PBR (Times, 2009y), with even Kaletsky lamenting the lack of credibility of the spending and borrowing targets and the undermining of good work on looking for efficiency savings by increasing pension expenditure (Kaletsky, 2009h). In the new year, the paper described a 'most urgent need' to plan cuts in expenditure (Times, 2010a). Whilst Kaletsky accepted that Keynesian policies 'are now almost universally acknowledged to have contributed to economic recovery...higher borrowing and spending that made sense a year ago may no longer be relevant today and (is) almost certain to be dangerously misleading a year from now' (Kaletsky, 2010a). Indeed the paper had gone beyond questioning fiscal consolidation and begun to discuss its makeup. 'The ability of government to support demand with fiscal policy is not there any more', it argued, and turned to focus on the need for fiscal consolidation to entail large cuts in public spending as opposed to an increase in taxation which would 'choke off the prospect of recovery' (Times, 2010b).

The paper now saw evidence in history than consolidation 'does not preclude recovery' (Times, 2010c), and suggested that cuts were inevitable whichever party won the election (Times, 2010d; Kaletsky, 2010c). Whilst timing had often been a central dividing point on the issue of cuts, the paper now also claimed that Britain had 'less room for manoeuvre than other advanced economies' on this issue also (Times, 2010e). The lack of clarity from the political parties in the run-up to the election was also criticised, with spending cuts needing to 'be spelt out to save stumbling sterling' (Times, 2010f; also Sunday Times, 2010a). The paper also explicitly linked the poor control of public spending in the pre-crisis period as the cause of the need to cut the deficit, despite its necessary role in limiting the depth of the recession (Times, 2010g).

¹³ Kaletksy still did not accept the danger of an attack from the bond markets, however, writing that Britain's borrowing is not 'remotely as serious as Pimco' suggested by claiming that Britain debt was resting on a bed of nitroglycerine (Kaletsky, 2010b; Pimco, 2010).

The *Sunday Times* argued before the Budget that the 'parallel that should truly scare us is not with America in 1937, but Greece in 2010' (Sunday Times, 2010b). Again, the uncertainty about when a sovereign debt crisis might erupt was sufficient to make cuts necessary (Ibid.). Kaletsky called for 'greater clarity' in how cuts will be enacted, rather than issues of timing, to give the markets certainty (Kaletsky, 2010e). The markets were asking, the paper reported, whether the Budget would be Gordon Brown's (i.e. profligate), or Alistair Darling's (i.e. prudent) (Times, 2010h). The danger of a sovereign debt or fiscal crisis was still debated, with Kaletsky calling these 'dire prophecies' wrong (Kaletsky, 2010d) and maintaining that there was 'no reason to panic' (Kaletsky, 2010e). Richard Lambert of the CBI, however, still argued that the speech had 'two audiences to aim at'; voters and the bond market (Lambert, 2010).

However as the election neared, the *Times* became increasingly critical of the apocalyptic messages from the Conservative party about the danger of the national debt, reflecting the belief that other questions now dominated such as the future of the role of the state (this theme is expanded upon in section 7.4 below). The paper called the parties 'as bad as each other' in their arguments on the issue of the deficit, suggested that Conservative warnings about the dangers of a ratings downgrade were 'not credible' and said that Cameron's claims that the national debt could double were 'not true' (Times, 2010o). They began to suggest that immediate cuts from a Conservative government could threaten the recovery (Times, 2010p), although the inconsistency of their position was highlighted by articles in both the daily and Sunday versions expressing continued support for immediate spending cuts (Sunday Times, 2010d; Times, 2010q, 2010r, 2010s).

The *Independent* continued to provide a mixture of both generally orthodox and Keynesian opinions on the need to cut spending. On the one hand it agreed that the fear of bond markets was 'overdone' and secondary to the 'risks of slamming on the fiscal brakes too early (Independent, 2010b; also 2010k), noting that even an advisor to the Conservative party was now warning about cutting spending too early (Independent, 2010d; also 2010h). It also suggested that whilst Brown was right about the support being provided by stimulus, the corollary of the importance of such support was that deeper than envisaged spending cuts would have to be made given that there was less underlying growth than previously thought in the private economy (Independent, 2010i). However, this was disputed by McRae (2010f) who noticed that government output was actually falling, undermining the argument that it was behind the recovery. The 'self-healing mechanisms of the private sector are already at work', he suggested, following that the economy didn't therefore need the government to spend in to help it recover (Ibid.). He suggested that there was 'no alternative' to austerity and as elsewhere cited Sweden and Canada as examples of successful expansionary fiscal consolidations (McRae, 2010g).

The FT likewise straddled the divide with a belief in the inevitability of austerity in the medium term and the need to take cuts seriously supported by a concern about cutting support for the private sector too soon. Martin Wolf maintained his Keynesian stance that to believe in expansionary contractions was 'to believe in magic' (Wolf, 2010), and that 'there is a huge risk – in my view, a

certainty, that (fiscal consolidation) would tip much of the world back into recession' (Ibid.). But whilst 'the government is right not to cut too much too fast', this was deemed 'no excuse not to plan' future cuts, with the paper noting that spending cuts 'must do the brunt of the work, not tax rises (FT, 2010a). The paper on the one hand framed the Budget as an opportunity to display Labour's longer term attitude to the role of the state (FT, 2010b), but on the other criticised Darling for not covering measures to reduce the deficit enough in it (FT, 2010c, 2010d, 2010e, 2010f, 2010g).

The *Telegraph*, maintained that the 'case for swifter and more aggressive action on the deficit is pretty much unarguable' (Warner, 2010b), and suggested that markets, not politicians would decide how much debt was too much (Warner, 2010a). Generally, the *Telegraph* maintained its focus on the danger of a sovereign debt crisis, warning that the developing crisis in Greece could be matched in the UK if markets lost confidence in the path the government was taking (Conway, 2010a; Halligan 2010a; Warner, 2010c). Austerity was no longer viewed as an option but as something the markets would impose inevitably, whether the government of the day liked it or not (Warner, 2010c; also Daily Telegraph, 2010b). The agency of governments to decide upon such a decision was therefore not only questioned, but actively undermined (Warner, 2010d). Evans-Pritchard called for looser monetary policy to offset the impact of cutting spending, but that there was no alternative to cuts because of the risk of a gilts strike (Evans-Pritchard, 2010).

Much of the paper's framing of the 2010 Budget was done in relation to its impact on investors and the dangers of a gilt strike. Its front-page lead the day after the Budget called 'the verdict in the city' as being 'swift and damning' (Daily Telegraph, 2010c). It declared the statement a 'seminal point: the moment at which investors signalled that they had lost faith in Labour' (Ibid.). It suggested that 'bond vigilantes and rating agencies gave a frosty reception to the Budget' (Daily Telegraph, 2010e; also Sunday Telegraph, 2010a), alongside the business community which issued a 'damning verdict' in which less than half of those polled actually believed it would damage the economy (Daily Telegraph, 2010g). Halligan gave faint praise to the Chancellor for not offering a spending boost which 'would have caused a sovereign debt downgrade and an instant run on the pound', but maintained his assertion that the 'UK is close to bankrupt', calling Gordon Brown a 'fiscal vandal' (Halligan, 2010a).

The *Daily Mail* similarly continued the theme of British public spending 'testing the limits of the market's confidence' (Daily Mail, 2009i), declaring that 'we're all cutters now' (Daily Mail, 2009l). The paper took external credibility for these positions from the IMF, the EU (somewhat ironically), and think-tanks such as Policy Exchange (Daily Mail, 2009i, 2009n, 2009o respectively). The apparent success of austerity in Ireland was also sold as further justification of this policy (Daily Mail, 2009r). In relation to the PBR, it increased its cataclysmic language, warning of 'financial armageddon' and a 'ticking time bomb' of sovereign debt (Daily Mail, 2009p; Brummer, 2009c respectively). Darling was playing 'a dangerous game with the country's finances', Fleming concluded (Fleming, 2009b).

Whilst for many of the other papers the intricate economic arguments over the timing of deficit reduction and spending cuts was starting to lose its importance in favour of the issue of the vision of the major parties over the role of the state, for the *Guardian* public spending was still the central dividing line. It maintained that this was the 'most important conflict' in the upcoming election (Guardian, 2010f, also 2010d), and continued to maintain a wariness of cutting too much and too soon into a recovery. It developed this argument using both a validity and normative approach.

From the validity perspective it attacked the economic logic of such an approach and sought to quote a variety of sources calling for a slower approach to public spending cuts. Firstly it highlighted that the Liberal Democrats were siding with Labour's argument over this side of the debate (Guardian, 2010b, 2010c), and quoted Treasury sources as denying that any firm evidence existed that 'the markets would demand an increase in interest rates to service government debt if there were no cuts in spending in 2010-11' (Guardian, 2010e). It also quoted Conservative doubts about its own fiscal strategy, reporting some back-tracking on the initial pledge to start cutting as soon as a new government was formed (Ibid.; Elliott, 2010a). Elliott also noted that the IMF's reports were essentially an endorsement of Labour's position (Elliott, 2010b; also Guardian, 2010x). The paper reported high profile economists and businessmen who criticised such an expansive fiscal approach (Guardian, 2010g, 2010h, 2010n), but criticised them in an editorial by stressing that 'the public deficit is essential' in a time of economic weakness (Guardian, 2010i; also Guardian, 2010j). Davies noted the 'great rift' that had opened in the economics profession by the topic, but unlike other papers, the Guardian didn't draw the conclusion that it wasn't necessarily therefore the most important issue (Davies, 2010). It praised the economists opposing immediate spending cuts for their 'more intellectual modesty, and that the danger of high unemployment favours their approach' (Guardian, 2010k). Hutton (2010b) accused those calling for spending cuts for favouring a 'debt morality' over 'good economics'.

Despite this, however, there was also an occasional moral side to the *Guardian*'s argument. Firstly it criticised the rights of the credit rating agencies to be an arbiter of the nation's finances when they were so badly wrong on the finances of major banks before the crisis, and suggested that their lack of impartiality or accountability meant they should have no say on 'how and when the debt is repaid' (Guardian, 2010a). Hutton agreed, noting that 'the fate of millions should not be determined by the not very good economic analysts at the credit-rating agencies' (Hutton, 2010a, also 2010b, 2010c). Keegan added that 'in many cases these advocates of austerity for others are the same people who brought the economy to its knees', and that they had therefore lost credibility to have a voice (Keegan, 2010a).

However, this demonstrates a rather limited re-framing of the significance of institutional financial voices in economic policy credibility from the pre-crisis era. Whilst many of the articles above demonstrate that perceptions of who held economic authority had not been greatly altered by the crisis, those arguing here for a shift in such authority are much more limited. Thus the great majority of the pre-crisis economist elite were to maintain a public perception of authority despite failings in the pre-crisis epistemic communities which held shared beliefs on the economy. It is

noteworthy that the ratings agencies, those closest to the perceived failure of gauging the risk of debt, were the only ones to be tarred by the crisis, whilst others who held similar economic beliefs maintained their perceived authority. Again this demonstrates the limited diffusion of blame, which in turn helps to explain the limited diffusion of crisis and a more radical re-shaping of perceived authority to speak on economic affairs. The wider neo-liberal edifice was therefore not toppled because only certain elements of its structure were discredited through the crisis.

The chancellors' pre-election debate was also used by the paper to highlight the economic logic of Labour's argument by the paper. Talk of dealing with the problem sooner rather than later 'is economic nonsense', it warned (Guardian, 2010s), whilst Toynbee claimed that the Tories' policies got 'an embarrassing raspberry from the world's economists at Davos', and suggested that Labour was now scoring well on the economy 'because it is right: to cut now is dangerous' (Toynbee, 2010a). It focused on the IFS's questioning of making efficiency savings too early, and debate over the 'risk' of Osborne's National Insurance plan (Guardian, 2010t, 2010u, 2010v, 2010w).

However, for the most part the call for a delay to spending cuts stayed largely in the realm of a technical economic analysis of the situation, and only rarely entered into a more moral or normative pitch to the public. Whilst the two sides of this debate held mutually exclusive opinions on the economic priorities of the time there was therefore relatively little room for rapprochement between them. Whilst the austerity discourse was more prevalent in the media than the Keynesian one, there was a sufficient element of each to largely cancel each other out. However whilst the austerity discourse was tied in to other existing normative beliefs within political and economic discourse, this could not be said to the same extent for the Keynesian approach. This argument is expanded upon in the sections below.

We can therefore observe in this time period a notable reversion to the important pre-crisis discourse of 'no room for manoeuvre', demonstrating the lack of ability on the crisis' behalf to radically alter perceptions of economic imperatives. The lack of immediacy for action once the worst of the crisis has passed also removed the window of opportunity for a shift in the role of the state. The beginnings of a shift to more medium-term timeframes also supported the strength of the orthodox discourse, which was able to align with existing preferences for a smaller state and lower taxes better than Keynesianism. Whilst this was by no means a complete process, the blend of economic and normative ideas behind austerity certainly outweighed the more singularly economic arguments of Keynesianism; whilst austerity was able to target both hearts and minds, Keynesianism was only aiming at one.

7.4 Public sector cuts and deficit reduction as a pre-requisite for economic growth

The first way in which austerity was able to tie itself to existing discursive strands was in its link to the sense that the British state had grown too burdensome, and that cutting the size of the state may therefore increase economic growth because of the lower future taxes that would arise from this. This was therefore an explicitly Ricardian narrative; the idea that a fiscal contraction could actually lead to economic growth by improving the medium term tax and interest expectations of investors and hence driving investment and production to increase, counteracting the negative demand-effect of a reduction in state spending. Whilst this argument was debated academically between economists for several years, and is arguably still on-going, its presentation in the media was rarely purely statistical and econometric, but often also a normative judgement on the type of state that was wanted for the future.

Firstly the level of spending was presented as a burden not only to the current taxpayer, but to future generations also. The construction that developed was that without cutting there would be left for future generations a burden of higher taxes. This legitimised austerity by creating a dualism whereby to oppose cutting was to be in favour of leaving such a burden. Randall presented it in exactly these terms, declaring the 2008 budget a 'profusion of unaffordable spending...not just for this generation, but for the next and the one after that'. Likewise the *Independent* claimed that increasing government debts 'increased the tax burden on future generations' (Independent, 2009n). Warner explicitly linked the government's strategy of 'fighting debt with debt' as something that would 'undoubtedly...burden future generations' (Warner, 2010f).

McRae questioned whether constitutional change was needed to prevent a present government creating a debt burden that future generations will 'have to work to pay off' (McRae, 2009f). The *Independent* also linked the issue of burdening future generations with that of re-thinking public services. The need to find efficiencies in what were deemed overly bureaucratic and managerial public services was the only alternative to allowing them to continuously take up an ever greater share of the national income (Independent, 2009p). Again, here, the rhetoric of New Labour's economic governance strategy came back to limit their discursive space for more explicitly Keynesian action. The significance placed by Gordon Brown on not leaving a burden on future generations in his first golden rule clearly still normatively influenced the broadsheet press' understanding of the public finances, and worked to create an innocent victim, future children, to the decision to spend.

A Keynesian counter-argument, prevalent later in the American media through the writings of Paul Krugman, was largely absent in Britain, leaving this claim unchallenged. Such an argument questions the debt burden on future generations, noting that debt/GDP ratios fall over time even with a balanced budget due to inflation. As such, Krugman argued that the western world should

pursue a higher inflation medium term strategy than the 2% limit that was currently the target of many countries, with such a moderate monetising of debt unlikely to impact on the financial credibility of governments (Krugman, 2012).

Secondly, stagnant economic growth and the rise in state spending were used to argue that the medium term development of the British economy was dependent on a reduction in the size of the state. Moreover, stimulus spending was rolled into a single frame with the belief in the development of a larger state *in general*, as opposed to a time-limited stimulus which doesn't necessarily imply such a permanent increase in the size of the state at all. The clear links between such arguments and those of the pre-crisis period are particularly tangible. Randall proclaimed Labour's vision to be 'the triumph of a suffocating state over personal responsibility, self-sufficiency and wealth-creating enterprise' (Randall, 2009b). Halligan declared that 'the mess we're in was largely caused by too much state intervention' (Halligan, 2009a), whilst the *Sunday Telegraph* argued that 'the best way to restore our credibility and our finances is to cut back the size and activities of the state' (Sunday Telegraph, 2009d; also Randall, 2009c).

The Sunday Telegraph continued that a 'sufficiently rigorous shake-up of Labour's bloated client state could well obviate the need for higher taxation' (Ibid.). Reece agreed with the general tone, arguing that what was needed was the 'liberating of the rest of society from the impoverishing effects of an over-bearing and over-funded and wasteful state sector' (Reece, 2009d). Public spending, he therefore argued, 'will serve only to suffocate sustainable, long-term output growth driven by private enterprise' (Ibid.). Warner, by October 2009 a writer for the Telegraph, agreed, stating that 'the public sector has reached an unaffordable and stiflingly large size' (Warner, 2009k).

Likewise in the autumn, the *Telegraph* wrote that 'reducing the deficit is essential if Britain is to thrive in the coming years' (Daily Telegraph, 2009w), and stated that 'a bigger state is not the route to recovery' (Daily Telegraph, 2009z; also Conway, 2009e). Whilst they criticised Brown for only selling 'corporatism' to the CBI, they claimed that David Cameron 'clearly grasps the fundamentals of economic recovery' by highlighting that cutting spending is essential for growth, not damaging to it (Ibid.; also Daily Telegraph, 2009u, 2009x, 2009y). Keynesianism was therefore explicitly being sold as a permanent increase in the power of the state sector over the economy, something incompatible with the discursive norms of the pre-crisis period.

The need to cut back the state in order for private sector growth to flourish therefore partially drew on the discourses of the need to assist corporate capitalism through the usual supply-side measures of the previous autumn. The lack of a crisis of the corporate sector meant that the usual methods were necessary to stimulate them, and that the reduction in state interference was clearly one of the principal measures here. Popular distaste for 'red tape', 'bureaucracy and 'regulation', despite the role of a lack of regulation in the financial crisis, tied in with the potential for fiscal consolidations to increase growth. Again the desire to cut regulation highlights how crisis and the failure of ideas in one sphere of governance does not necessitate that such an idea remains a failure in all other spheres, thus highlighting again the lack of diffusion of ideational turbulence in the

financial system. Given that increased government spending was in itself deemed a form of interference, the necessity of cutting spending to stimulate economic growth was created.

The *Times* suggested that 'an essential part of the (economic) strategy must be a resurgence in economic growth' (Times, 2009z). The best hope for this, it believed, was in exports and business investment. The one area of state spending which should be maintained, they argued, was capital spending, the decision of which to cut in the 2009 Budget would have 'damaging long-term consequences' (Ibid.). Warner presented '10 ways to help restore growth' by promoting largely supply-side measures focused on easing the path for business investment, cutting taxes and red tape (Warner, 2009l). Warner also warned that any planned increase in National Insurance 'threatens to knock the stuffing out of any private sector recovery' (Warner, 2009n). Simon Heffer complained that the 'tax rises on the so-called rich show that Labour has no regard for the creation of wealth, and an understanding that its creation is what makes everybody more prosperous' (Heffer, 2009). Moreover, however, the paper also argued that 'piecemeal restrictions on departmental budgets' would be insufficient to achieve sustainability for the state in the medium term, with major cuts to entire programmes and departments required instead (Times, 2009aa).

The *Telegraph* created an interesting zero-sum dualism by complaining that 'the private sector is taking the full force of the recession, while public services actually grew' (Daily Telegraph, 2009p). Implicit within this is the 'crowding out' assumption that if public services were cut, this would create an equal and opposite positive reaction in the private sector. Halligan called 'Keynesian nonsense – disproved again and again throughout history' the idea that lower spending would 'choke' the recovery (Halligan, 2009g; also Daily Telegraph, 2009q), whilst Warner complemented it by suggesting that 'the Chancellor's economic justification for (delaying fiscal consolidation), that it will support growth, also looks questionable' (Warner, 2009n). Conway jumped on the presentation of evidence of successful expansionary contractions to argue that 'in previous episodes of fiscal consolidation, the deficit cuts promoted rather than hindered the recovery' (Conway, 2009d). Even Roger Bootle, who declared himself 'unconvinced by the applicability' of the relevance of this argument to the current context, maintained that deficit reduction, if done carefully, could still 'aid recovery prospects' by 'increasing the confidence of businesses and households' (Bootle, 2009f).

Finally, for many the route to economic recovery was dependent on the confidence of investors in the British economy, which in turn was dependent on the government showing commitment to deficit reduction. The *Times* reported that 'Alistair Darling is considering fierce public sector spending curbs and deferred tax rises to convince markets that Britain will emerge eventually from its massive debt' (Times, 2009h). The paper also believed that 'any stimulus package is unlikely to have any effect other than to strain an already over-stretched balance sheet' (Times, 2009j). The paper also framed the key task of the 2009 PBR as firstly putting 'the UK's public finances on a pathway to stability over the medium term' (Times, 2009z). They also fell on the side of a fiscal 'correction geared to spending cuts rather than tax increases', something that 'history shows...is more likely to deliver debt reduction (Ibid.).

Likewise for the *Sunday Telegraph*, 'the most important task facing the next government is to maintain investors' confidence in Britain's public finances and restore those finances to order', and concluded that 'there will be no hope of pulling out of recession, or indeed of avoiding a deep and lasting depression, if foreign investors decide they will lend to us only at extortionate rates' (Sunday Telegraph, 2009d). Reece agreed, arguing that 'the state's liabilities are so big and potentially destabilising that they now seriously endanger any nascent recovery' (Reece, 2009d). He described New Labour as 'having gorged on debt to turn us into the world's most financially obese nation' (Reece, 2009d). Warner likewise stated that 'we risk an equally unpalatable world of higher interest rates and ever more ruinous public debt', which he described as the 'most immediate' problem (Warner, 2009i) leading to 'a real chance of a fiscal crisis' (Warner, 2009l).

Thus in the right-wing press a strong discourse was emerging during this period linking the importance of deficit reduction, the danger of the financial markets shunning government debt, the need to reduce the size of the state and the need to ease conditions for corporate businesses to produce again. It is worth stressing that these discourses were most prevalent in the conservative-leaning papers, particularly the *Daily Telegraph*, but nonetheless it highlights the extent to which a uniform picture was developing across this section of the media of the principle problems of the financial crisis for the British state and the necessary solutions to solve them. The analysis of the factors influencing the future growth of the economy bore all the hallmarks of the pre-crisis arguments about the bloating of the public sector and the sapping impact of regulation and red tape on business, and therefore saw a supply-side and small-state solution as necessary. Another of these discourses which re-emerged during this period was that of political trust, which was increasing as an issue of valence due to the 'MPs' expenses scandal'. As is shown in section 7.4 below, the lack of trust in New Labour and with it their approach to governance was drawn into the issue of spending cuts during this period.

7.5 Stimulus, spending cuts and political trust

As was stated above, stimulus spending was often discursively linked to political trust, with questions about Gordon Brown's ability to spend money 'objectively' as opposed to 'politically' being used to question the wider benefit of spending regardless of the recession context. The issue of political trust itself also became more prominent through 2009 with the MPs' expenses scandal dominating non-economic news and the growing sense that there was no clear answer to the austerity versus stimulus debate, which in itself drew other topics back into the analysis for the press when choosing between the merits of the major parties. Thus the increasing lack trust in

Gordon Brown individually, the Labour government in general, and the need for spending cuts coalesced to provide legitimacy for retrenchment.

The growing resentment in the media of the perceived 'spin' from the Blair and Brown governments on all policy announcements led to a common call for 'honesty' in the face of economic crisis. This demand from the media generally included being up-front about the 'true cost' and 'damage' to the public finances, and the need to set out a clear timetable for getting them back under control. Thus whilst commentators continued to critique the government for failing to meet this constructed imperative, the issue of the economic logic behind the government's decisions was downgraded in importance. The frame of the need for more political 'honesty' was transferred over to the issue of justifying spending cuts more than was the case for justifying expansion. The term 'honesty' was equated with being up-front with the public about the true damage to the public finances, and thus also being 'honest' about the scale of the spending cuts necessary to rectify this. Thus a fiscally orthodox position was required in order for a politician to meet this frame, with no alternative in place discursively for 'honesty' to require that politicians spell out the scale of the necessary fiscal expansion in order to counter-act a private-sector downturn. This section will seek to demonstrate the existence of this frame and its prevalence in the media coverage during this time period.

The first element of this discourse was the usage of the pre-crisis discourse of Gordon Brown as an exceptionally 'political' chancellor to critique the use of government spending as a solution for crisis. In its coverage of the 2009 Budget, the *Times* argued that a large scale fiscal boost would just allow sectional interests to take government money for purposes of little public interest (Times, 2009b), thus critiquing the corporatism of Brown's economic management and supporting the notion of the client state posited by the *Sunday Telegraph* (Sunday Telegraph, 2009d). The growth projections used to justify Darling's figures in the 2009 budget were widely questioned as optimistic (Times, 2009q), with the *Times* therefore criticising the lack of honesty about the scale of the problem from the government (Times, 2009s). It claimed that 'New Labour is not credible after this Budget' (Times, 2009t).

Important within this was the claim that the Budget represented class-warfare with an 'old Labour' approach to society. 'Gordon Brown has not been able to detach himself from the tribal prejudices of his party', the paper continued (Ibid.), whilst Simon Heffer in the *Telegraph* opted for a more blunt approach, labelling 'the idiocy, bigotry, tribalism and sheer class hatred of the Budget' (Heffer, 2009). Even Kaletsky, a self-declared supporter of Brown's economic reasoning, recalled the socialism of Labour's 1983 election manifesto in calling his approach to the crisis 'the longest assisted suicide in history', (Kaletsky, 2009d). It was his 'political misjudgements, not his economic decisions (that) lie at the heart of his personal tragedy', he claimed (Ibid.). Instead of honestly outlining the situation with the public and calling for patience whilst the fiscal stabilisers took effect, he tried to set political traps for the Conservatives which simply 'blew up in his face' (Ibid.).

This fitted into a developing theme of the period, expanded upon below in section 7.5, that the economic plans for recovery of the two major parties in this period were not the defining electoral clash that would influence the vote, but rather that other issues would be the central defining topics in the election. As such, whilst many suggested that Labour had taken broadly the right strategy since the beginning of the crisis, this was in itself insufficient to allow the papers to support them in the election, with several therefore falling back on their central complaints of the Brown era from the pre-crisis period in shaping such support. Moreover it demonstrates the significance of the received wisdom that Labour's electoral success stemmed from a shift to a more centrist position under Tony Blair, and thus that Brown's 'class warfare' approach was thus a switch back to the unelectable left-wing pre-1994 Labour Party. Again this serves to highlight the continuation of major political discourses throughout the crisis period, and thus the ideational blockade preventing a wider re-imagining of the political-economic contours of the country post-2008.

Labour's attempt to 'apolitically' run the public finances was also critiqued. Darling's attempt to impose a rule making deficit reduction legally binding was quickly analysed by the *Times* as having 'no economic merit but a transparent political rationale: to win for the Government a credibility in fiscal management that it has lost' (Times, 2009ak). It noted that both of Labour's last golden rules proved 'irrelevant' in the current economic crisis, and that a new one would prove equally so in future crises (Ibid.; also Stephens, 2009b).

The *Sunday Times* critiqued the 2009 PBR for offering 'no credible plan' on getting borrowing down, despite it being at 'levels that led to Greece being downgraded last week' (Sunday Times, 2009a; also Sunday Telegraph, 2009a; Times, 2009x). It called the PBR 'Labour's jumbo deficit deception' (Sunday Times, 2009b, also Sunday Telegraph, 2009f), and quoted economists calling Brown "irresponsible" for failing to set out "even the rudiments" of a consolidation plan (Sunday Times, 2009c). Stephens, writing in the *FT*, was the bluntest about Labour's woes. 'In other circumstances', he wrote, 'Brown's suggestion that the financial crisis has boosted progressive parties and highlighted the need for renewed state activism might be right' (Stephens, 2009a). However, he continued that 'the media has pretty much stopped listening to the government' (Ibid.). This sums up most clearly the lack of trust in government reporting that pervaded media analysis during the period, and highlights the impact of the loss of such political capital on the media analysis of government policy.

The *Telegraph* took a similar view of the honesty of spending projections. It claimed that Darling's reputation was in tatters at the failure of the Budget to offer an honest assessment of growth forecasts (Daily Telegraph, 2009d; also Independent, 2009e). It provided the dualistic options for him at the Budget as being to 'confront the scale of the crisis facing the country honestly and implement the inevitably painful measures needed to prevent a decade or more of stagnation...or he can evade his duty' (Daily Telegraph, 2009j). It similarly provided a class warfare narrative to the Budget, calling it 'shabby cynicism' (Daily Telegraph, 2009m) and

¹⁴ A frame of austerity being the more patriotic option can therefore also be noted, here.

accusing Labour of a sectarian streak in its handling of the Budget (Brogan, 2009a, 2009b). The paper made a connection between the nation's current woes and Brown's lack of connection with Middle England (Martin, 2009). Vince Cable joined the attack by writing an op-ed piece accusing the government of symbolism over solutions (Cable, 2009). Tregfarne wrote that it was only the more honest approach of Osborne, who the markets were already expecting to be the next chancellor, which was keeping the gilt markets from erupting (Tregfarne, 2009). Halligan went even further, calling the danger of deflation a myth dreamed up by the government to allow it to 'pursue wildly expansionary fiscal and monetary policy and perpetuated by the vested interests benefiting from such largesse' (Halligan, 2009b). George Osborne pounced on this perception of the lack of trust in Labour's economic reporting by arguing that Labour's 'electioneering is putting confidence in Britain's economy at risk' (Osborne, 2009). The *FT* took a similar approach, criticising government policy for 'focusing on headline-grabbing initiatives' rather than 'addressing the most important issue in British politics: the fiscal crisis' (FT, 2009m).

Even the *Guardian* had an element of criticism for the focus of government policy, with Simon Jenkins (2009) noting in relation to the extra spending on projects such as the Olympics and Trident, which would far outweigh the efficiency savings promised in the Budget, that 'any fool can raise a tax. But it takes a gutless one to splurge it on this stuff'. Whilst the focus of complaint was more on the power of lobbying than the trust of the government, it produced the same discursive complaint that the focus of government was wrongly aimed at prestige over prudence, and that there was room for cutting unnecessary spending.

Whilst the paper generally supported the government's fiscal logic over the Conservatives' at the time, even they noted that Brown's approach 'looks like a ragbag of ideas designed to score political points' (Elliott, 2009f). Even the *Observer* (2009f) accused the PBR of representing a 'new class warfare' aimed at bolstering Labour's base before the election. Andrew Rawnsley maintained the theme of criticising the political nature of Brown's and Darling's economic statements by noting in the paper that to complain that the PBR was 'political is as futile as deploring bears for defecating in the woods' (Rawnsley, 2009). He described the key audience for the statement as the bond markets, for which it 'so badly failed the credibility test' (Ibid.). The *Independent* took a similar line, calling the speech 'more shaped by politics than economics' (Independent, 2009s). Both papers noted that whilst they supported Darling's argument over the timing of cuts, the credibility of the government had still failed because they had failed to present a medium term timeframe of deficit reduction to reassure the bond markets that fiscal consolidation would take place at some point.

The *Telegraph* also criticised Keynesianism on the basis that it assumed the government knows best on the economy (Daily Telegraph, 2009r), with Halligan also countering the use of Keynes to justify expansion by arguing that 'Keynes' work is being used...for short-term political ends to which, in my humble opinion, the man himself would object; because Keynes wasn't, in fact, a "Keynesian" (Halligan, 2009e). The *Sunday Telegraph* added an interview with the University of

Chicago's Gary Becker who argued that the crisis had 'been overstated to advance a sweeping collectivist agenda' (Sunday Telegraph, 2009f).

The analysis of the 2009 PBR was largely undertaken in the same vein, with Randall accusing Darling of choosing 'electoral expediency over fiscal integrity' (Randall, 2009d) and Halligan suggesting that the PBR was purely electoral, not economic. He accused Labour of 'dishonesty' which was 'leading us down the road to sovereign default' (Halligan, 2009h). 'Brown's prolonged Keynesian boost has exacerbated the long-term fiscal damage', he argued (Ibid.). D'Ancona followed a similar tone, accusing Labour of being 'not a government, just a clunking electoral machine' (D'Ancona, 2009), with the *Telegraph* again referring back to the 1970s, calling Brown's cabinet 'class warriors' (Daily Telegraph, 2009ac).

The *Daily Mail*, whilst relatively quiet on the topic of fiscal consolidation as an engine for growth, focused in particular on the issue of political trust, framing spending cuts as an integral part of a politicians' honesty (Daily Mail, 2009k). Likewise, leaving the issue for the following parliament, or future generations, was a dereliction of public duty (Daily Mail, 2009q. The decision to delay cuts was taken not as an economic decision based on the recovery but as an electioneering stunt by Labour to win votes whilst hoping that 'fantasy' economic growth would do much of the work for them (Fleming, 2009a; also Daily Mail, 2009f). The MPs expenses scandal was rolled further into the political trust frame, with the *Mail* arguing that if 'we can't trust MPs with their expenses, we can't trust them with public spending' (Daily Mail, 2009h). The paper argued that the 2009 PBR was aimed at 'shamelessly' rousing the 'party faithful rather than acknowledging the sorry truth about our national predicament' (Hastings, 2009). Likewise Cameron's rising poll numbers was taken as a reflection on his 'candour about spending cuts (Daily Mail, 2009m). The theme was continued at the 2010 budget, which the paper called a 'naked appeal for Labour votes' (Daily Mail, 2010a; also Daily Mail, 2010c) and joked that the 'fraud squad would have been called in' if Darling's financial statements had been given by a business (Daily Mail, 2010b).

A particular theme developed was to compare to previous Labour governments who had cut spending to legitimise doing again now. Sandbrook highlighted Snowden as a man who had 'put the national interest ahead of party advantage' and whose fiscal consolidation 'did the trick', with 'Britain recover(ing) from the Depression faster than any other major nation' (Sandbrook, 2009). Continuing this theme in 2010, Sandbrook also noted Attlee as a Labour leader who 'put economic discipline ahead of spin and spending', and 'long-term recovery ahead of short-term popularity' (Sandbrook, 2010).

The Conservative Party wasn't immune to criticisms over honesty either, but this criticism tended to re-enforce the existing frame as opposed to countering it. With their poll lead reducing in the face of the discussions of deep cuts to public services, the party avoided presenting their plans as being simply to cut. Thus they too were criticised for dishonesty, with the *Telegraph* suggesting that 'none of our politicians appears to realise what a mess we're in' and adding they 'have to understand not just that they have to cut spending, but that they have to cut it by far more than they imagine' (Daily Telegraph, 2009v). Likewise Halligan suggested that the Tories were 'wilting in

(the) face of economic honesty' (Halligan, 2009f). However, for the media, this lack of honesty didn't alter the fundamental importance of the deficit, merely it represented evidence that the British people were not sufficiently prepared for the likely effects of austerity at this stage. It did not therefore alter the framing of economic importance that the media was employing, but actually re-enforced the suggestion that politicians must be more upfront about the spending cuts to come, which in turn reduced the discursive scope to question this underlying assumption that they were inevitable.

Moving into 2010, the idea of political trust was argued by Kaletsky to be more important than economic arguments. 'What most voters focus on is not economic theory', he argued, 'but broader issues of trust, judgement, character and ideology' (Kaletsky, 2010a). If they did, he suggested, Brown might have a decent chance of winning the election, but realistically this was not the case. Indeed the *Times'* coverage of the 2010 Labour Budget focused almost exclusively on trust, and framed credibility in this regard as not offering a 'political' Budget. It framed Darling's credibility as depending on spending cuts, and criticised the government for avoiding 'a grown-up debate about what government is for' (Times, 2010i). The paper warned him against choosing 'petty politics' within his Budget measures (Ibid.). Ultimately, the speech was deemed to have failed to have met these conditions, with it being called 'nakedly' and 'breathtakingly' political (Times, 2010j; 2010k respectively). The Sunday Times called it a 'blatant exercise in dishonesty' (Sunday Times, 2010c). Likewise the *Telegraph* attacked the Budget for putting party political interests ahead of its perception of the needs of the country at large. Calling it an exercise in 'audacious conceit and ruthless political calculation', it noted that only Roy Jenkins of modern Labour chancellors has put the country before the party, and 'nobody would have expected either Mr Darling or Gordon Brown to act so nobly' (Daily Telegraph, 2010d). The paper called it 'short on policy...made up for by an overload of politics' (Daily Telegraph, 2010f).

There was therefore an explicit link drawn between the need for political honesty and the need to produce evidence of planned spending cuts by the media in the pre-election period. With political honesty having been a major theme in the pre-crisis period, and this having become more so in the face of the MPs' expenses scandal, 'confronting the scale of the problem' became a common frame of reference in relation to the financial crisis and the impact of government budget deficits. Those who provided most detail about what would be cut, where and when, were rewarded with the tag of honesty, and those who avoided talking about the issue in an attempt to turn attention to other topics of political debate were tarred with being 'political'. The developing 'anti-politics' theme tied into such a frame neatly; the scale of the economic crisis called for politicians to put the country over personal political gain, and as such those who acted 'apolitically', or 'objectively' in their acceptance of the stringent effect of the budget deficit were deemed most honest, and implicitly even patriotic. Meanwhile those who refused to be straight with the public about the scale of the problem and the necessity of spending cuts were re-enforced as the dishonest politician whom the public was represented as growing increasingly weary of. Thus, fiscal consolidation was tied in

discursively to the issue of political trust, for which there was no alternative for the alternative economic approach. The policy choice therefore developed an external validity through such an approach that stimulus could not. Moreover, the increasingly prevalence of this discourse in late 2009 and early 2010 increased the dominance of the austerity discourse in the media.

7.6 Non-state foundations of economic growth

One of the developing discourses of the crisis period the previous autumn had been that fiscal policy alone could not solve the problems of the British economy, which served to undermine using it at all. Whilst this was still emergent in late 2008, and had not become so ubiquitous as to make the majority of the press criticise the 2008 PBR's moderate fiscal loosening, by 2009 the discourse of 'no room for manoeuvre' which had accompanied the end of the crisis period for the financial system tied in with it to develop it into a wider lack of belief in the worth of taking what was now considered a 'fiscal risk' by increasing spending with the financial markets apparently so jittery about the prospect. Such a discourse built on common perceptions of globalisation as limiting the capacity of fiscal policy to play a major role in directing and stimulating growth, and as a consequence the belief that the most successful method of state-stimulated growth was by 'creating the conditions for investment' which tended to utilise more supply-side functions. The focus of the problem as one of deficits rather than growth has already been highlighted in the literature (Hay, 2013). This section again demonstrates the significance of the pre-crisis perception of the relative weakness of the state to influence growth rates through fiscal policy as the key explanation for this.

Much of this was based, as has been broached in the sections above, with the central economic need beyond fiscal policy being to satisfy financial markets to avoid increased interest rates on national debt. The centrality of financial markets to the economic debate of 2009, and the uncertainty about how they would act in the face of high budget deficits, justified the discursive importance of cutting spending. Again it is worth noting that there was very little social scientific evidence supporting the notion that financial markets were likely to stop buying British government debt and that there was any specific limit beyond which spending and stimulus could go, but the constructed centrality of the financial markets to the economic debate meant that policy which best pleased them was deemed most credible, and this invariably involved deficit reduction, even if it was delayed until after the recovery had taken hold.

This need to satisfy the markets applied across the spectrum, with Hamish McRae in the *Independent*, describing himself as having been 'critical of the surge in public borrowing' but accepting of its role in 'restricting the scale of the decline' stating that the following summer, as growth returned, 'a clear path will have to be set out for fiscal consolidation' (McRae, 2009b; also

McRae, 2009c). Two of the four economists interviewed by the paper before the Budget about what they would do if they were the chancellor cited the importance of satisfying markets by reducing deficits, with a further one highlighting the need to cut spending without mentioning markets specifically (Independent, 2009d). Only Adam Lent of the TUC called for further stimulus (Ibid.). Jeremy Warner, whilst still with the paper, said that 'bolstering confidence must be the Chancellor's Budget priority' (Warner, 2009b). For the *FT*, Buiter wrote of the tough balancing act between tightening 'too soon and the economy risks stalling again, or you tighten too late and you provoke a sovereign debt and sterling crisis' (Buiter, 2009b). The paper elsewhere talking of the 'tightrope' that Darling was walking, needing to 'reassure investors that the government has a credible plan to restore the public finances' (FT, 2009n).

The concept of 'credibility' in terms of future plans was tied up inextricably with cutting spending, for the other way to close the deficit, high levels of economic growth, had already been rejected as unlikely and out of the government's control by the commentariat. For Buiter, then, the government needed to 'inflict political pain on itself today, as a signal of its long-term commitment to fiscal tightening' (Buiter, 2009b). John Gieve, a former member of the BoE's Monetary Policy Committee also wrote in the paper of the need for markets to know what 'further measures' the government will take 'to get the deficit back to the published path' (Gieve, 2009). The FT highlighted the increased yield on credit default swaps on British government debt, up by 77 basis points by the time of the Budget from summer 2008, as evidence that as long as investors 'continue to fear in a general way that UK debt levels – and bail-out packages – are set to grow...they will remain nervous about the credibility of gilts' (FT, 2009h).

Again, this section ties in most notably with those in the earlier section about the importance of global credit markets in the press' thinking. What we therefore see is a continuation in the fear of the role of global credit markets and an assumption that national economies are dependent on these. Thus there developed a belief that the markets must receive whatever they wish to help them get the economy lending again, and what they required was a cut in the size of the state, because the size of the debt was deemed to be becoming unsustainable and the future growth path of the economy uncertain whilst the size of the state remained so large. The only logical policy given this framing of the impact of globalisation on the state is to set out a clear path to fiscal consolidation to appease the perceived market concerns. The main point of interest then is that such a bowing to the desires of the market is so central in much economic thinking in this time despite the clear failures of the capacity of foresight in the credit markets in the pre-crisis period.

For several papers, the factors influencing any return to growth still focused on the return to normality of the financial sector and credit channels, a continuation from the crisis period itself. 'Without that, any amount of fiscal pumping is unlikely to have much impact', the *Times* reported (Times, 2009d; also Kaletsky, 2009a). It argued that the 'speed at which the (economic) position improves depends on factors outside (the government's) control' (Times, 2009n), and suggested that without increased bank lending, 'Mr Darling's budget measures to boost recovery are likely to fall flat' (Times, 2009o). The indebtedness of the household sector was highlighted as a central

'drag on recovery', with the government unable to 'take up the slack' because of its own debt levels (Times, 2009w). The *Telegraph* agreed that bank lending was the problem, and hence criticised America's own fiscal expansion on the basis that it wouldn't tackle the root cause of the problem, but merely add an extra debt burden (Daily Telegraph, 2009l).

Likewise the *FT* called the reduced availability of credit to businesses 'the greatest threat to recovery' (FT, 2009a, also 2009c). It also pointed to an increase in world trade as the main hope for a recovery; again something that was out of the government's control (FT, 2009e). The *Independent*, too, warned of the lack of lending holding back recovery (Independent, 2009l; also Anderson, 2009a). Even the *Guardian* was largely sceptical of the government's growth figures on the basis that pre-crisis growth came from the city, high house prices and high public spending, all of which were now disappearing and of which only public spending was within the government's control. 'That economic lop-sidedness is Labour's real crime', it noted (Guardian, 2009c).

As such, the 'zombie economy' of loose monetary and fiscal policy had 'arrested the economy's decline' but had failed to offer any real re-balancing (Elliott, 2009e), something several others papers noted and which a reduction in the public spending element of GDP growth chimed in better with than stimulus (Daley, 2009; Rachman, 2009; Reece, 2009e; Taleb and Spitznagel, 2009; Wolf, 2009a). Even government attempts to stimulate the economy through industrial policy were merely creating bubbles which could deflate violently rather than creating last growth, Warner (2009m; also Reece, 2009b) argued. As with the previous autumn, such understandings solidified the role of supply-side assistance to the private sector as the most suitable vector to recovery, as opposed to any stimulus spending.

As was mentioned above, a more limited discourse, the equal and opposite of that highlighted here, suggested that because fiscal policy was not enough to secure recovery, austerity wasn't any more the solution than Keynesianism. Kaletsky called for a 'wait and see' economic policy given that the nature of the recovery was still unknown (Kaletsky, 2009c), whilst Killick called the 'age of austerity' a 'stupid idea', arguing instead that Britain's recovery was dependent on deciding what Britain's selling points were in the global economy (Killick, 2009a). Whilst the *Times* noted a link between reducing deficits and economic growth, it also maintained that exports and business investment were a central factor not to be ignored (Times, 2009z). In the *FT*, Samuel Brittan also highlighted the relative importance of growth over 'deficit hysteria' (Brittan, 2009a). 'If we have a normal recovery the red ink will diminish remarkably quickly', he wrote (Ibid.). And if that didn't occur, the red ink wouldn't diminish but wouldn't need to, given its role in propping up the private sector (Ibid.).

Whilst a technical debate about fiscal policy had existed since early 2009 about the merits of Keynesian-style stimulus versus Ricardian-style retrenchment, several articles tried to downplay the importance of this versus a more general argument about the purposes of the state and the political trust of the major parties. Noting that both major parties had situated themselves to an extent between these two economic theories, the *Times* tried to shift the argument away from the merits of

stimulus by suggesting that 'there is little difference between the parties on the need to restore the public finances' (Times, 2010l). By accepting the need for some cuts, however, it suggested that Darling has legitimised tough action to reduce the deficit, playing into the Conservatives' hands (Times, 2009ab). Moreover it ridiculed the suggestion that the 'difference between recovery and stagnation is how quickly spending cuts start and how wide they are' (Ibid.). It accused Labour of 'misread(ing) the question that the public are asking. It is not "How are we going to get there?" but "Where are we going?" Messrs Brown and Darling seem to want this election to be about the method of government. The people want it to be about the purpose of government' (Times, 2010k; also Times, 2010n). The *Daily Mail* agreed, arguing that all three major parties would 'cut more than Maggie (Thatcher)' if elected (Daily Mail, 2010d), and accused all three of dodging the issue of cuts in the pre-election chancellors' debate (Daily Mail, 2010e).

The *Independent* likewise agreed that the 'three political parties have now moved closer together with regard to the timing of Britain's fiscal consolidation' (Independent, 2010a; also Independent, 2010c), suggesting that 'the rhetorical differences are far greater than the real ones' (Independent, 2010e). McRae suggested that the nature of the recovery would anyway be dictated by global savers, and America's economic success, which the government would have no say in (McRae, 2010a, 2010b, 2010c, 2010d, 2010e). As such there was therefore very little fiscal policy could now do to alter the shape of the recovery; McRae called the 'heated political debate...utterly irrelevant' (McRae, 2010d; also 2010c). 'We tend to overstate the significance of politics in determining economic outcomes', he continued (McRae, 2010f). The paper, like the *Times*, declared the electoral dividing ground as being one of 'the role of government' (Independent, 2010f). Asking the voters to take a decision on fiscal policy which 'even the supposed experts cannot resolve seems almost farcical', the paper commented (Independent, 2010g; also Independent, 2010j). Calling the debate about the timing of cuts 'sterile', it suggested that 'we are all facing a future of austerity' regardless of who gets elected (Independent, 2010k).

The FT, by the time of the budget, was also wary of the argument that 'the heroic efforts of the Labour government are all that stand now between the British economy and oblivion' (FT, 2010h). Thus more important than fiscal policy decisions were the long-term development of the British state, with 'the case against micro-management' needing a key voice in the paper's opinion (Ibid.). The Telegraph agreed that the differences between cutting now and cutting later were now less important than giving markets confidence and certainty by outlining how exactly they would go about cutting (Warner, 2010e, 2010g; also Daily Telegraph, 2010a; Sinclair, 2010).

With how to foster growth now being deemed a key issue, Reece in the *Telegraph* suggested that 'we're not going to spend our way to growth', and that the private sector had to be nurtured in order for a genuine recovery to be made (Reece, 2010). Warner noted that the recovery had 'encouraged all three of the main parties to soften their positions on deficit reduction', suggesting that the differences between the issue from the parties was also reducing (Warner, 2010g). As with several other commentators, he therefore framed the election as 'a clear choice between a conservative vision of small government and low taxation, and a Labour one attached to big government and

free-market intervention' (Ibid.). He also drew a direct parallel between Canada's 'economic renaissance' and 'the brutal fiscal consolidation it undertook' in the 1990s (Ibid.). As with the previous autumn, therefore, much of the economic focus centred on how to help British businesses start growing again, with supply-side measures generally favoured as the most optimal policy tools.

Whilst the Guardian continued to call the spending debate the central issue of the election, it warned against 'fire-fighting and fiscal stimulus' as being the only pillars of Labour's economic manifesto (Guardian, 2010l). Like other papers it questioned the true gap between the major parties on such issues, and suggested that Darling should 'pitch his budget on ground that Labour has barely touched: rebuilding the economy after the crash' (Guardian, 2010m). It noted that the lower borrowing figures than anticipated might allow for an extra targeted giveaway to help the economy, suggesting targeting industry (Kettle, 2010; also Guardian, 2010n, 2010o). It suggested that reforming the City and helping the housing market should be the two central priorities, and calling for investment in the productive capacity and long-term future of the economy (Guardian, 2010p). Milne maintained that only 'public intervention' could continue to rebuild the damaged economy (Milne, 2010). Noting that the budget was a largely responsible, prudent one which failed to spook the City as feared, Keegan even worried that Darling was 'almost being too "responsible" given the scale of the financial crisis (Keegan, 2010b; also Guardian, 2010q, 2010r). Like other papers, the Observer highlighted that the election was 'not just about cuts', but instead of framing it as between a large and small state vision as other did, framed it as about 'values' (Observer, 2010). This alternative framing of the 'role of the state' discourse was much more limited in its utilisation by the media, however.

The extent to which the strategic electoral pressures came to shape their response and likewise the media discussion about such responses can continue to be observed in this period. With splits in opinion over the value of cuts and stimulus, both political parties, and to an extent the media also, came to develop a hybrid-type approach which mixed a fear of the over-burdening state in the medium term with some level of caution about starting cutting too early. Whilst some commentators from the Guardian and Telegraph in particular fell outside this spectrum, the majority of the press clearly lay within it. This in turn worked to minimise it as an issue of major electoral importance, with the modesty of the proposals from both sides leading to the suggestion that ultimately it should not be termed the defining electoral issue. With this switch back to more normative and ideological discussions about the role of the state, political credibility and encouraging business and investment, the weariness of a majority of the press with the New Labour approach favoured a switch in government. Indeed, given the greater priority of medium-term timeframes over immediate ones in regards fiscal policy decisions, the preferences of the media tied in more notably with the Conservatives than the Labour Party as the election neared. Thus the strategic constraints of the context in which the crisis occurred in Britain appear to have in many ways favoured austerity discursively.

7.7 Spending cuts and public services

Building on the discourse of the pre-crisis period which blamed Labour for having allowed the public services to become bloated and wasteful, a developing discourse in the immediate post-crisis period, seeking to tackle the fear that spending cuts generally imbued in the populace, was that such cuts could be undertaken without harming front-line services, and that there was sufficient 'fat' in the public sector for cuts to take place without influencing the 'muscle' upon which their operation was dependent. Of note here is that these discourses ran throughout the political spectrum of the press, with even the more left-leaning *Independent* and *Guardian* as critical of statism and the perceived poor value for money gained from the increase in public funding over the Brown period as the more right-leaning newspapers.

The *Times* declared that there were two public sectors in Britain today: the "front line" that had jobs the public understand, often for low to middling wages, and the "back room", 'that is firmly on the gravy train' (Times, 2009e). The argument held that the 'back room' public servant could easily be cut without damaging such services. The paper also noted that public sector productivity had fallen over ten years of Labour governments, with a 38% increase in funding only leading to a 33.6% increase in services (Times, 2009ad). With 'austerity' rapidly becoming deemed inevitable by mid-2009, they called for a 'progressive' version, suggesting that the party that can best cut back the state without damaging the most needy will stand to gain at the next election (Times, 2009v; also FT, 2009l). Randall, in the *Telegraph*, bemoaned the amount of waste, but added that 'we can't seem to achieve' a reduction in it (Randall, 2009a).

The paper targeted measures which could easily be cut, such as TV licences for wealthy pensioners, child benefit to wealthy couples and reducing the pay of senior doctors, to suggest that such cutting needn't necessarily be challenging (Daily Telegraph, 2009g). Reece suggested that the government's method of throwing money at social problems wasn't working anyway, so a reduction in it wouldn't necessarily decrease results (Reece, 2009c), and the paper bluntly assessed that 'the public sector cannot continue at its present bloated size' without financial markets punishing Britain 'mercilessly' (Daily Telegraph, 2009aa). 'Trimming' spending was not deemed enough, rather the issue required 'responsible surgery' (Elliott, M., 2009). At times, however, even the *Telegraph* admitted that cuts to frontline staff might be required, although this wasn't deemed to reduce the need to pare 'back the public sector significantly' (Daily Telegraph, 2009ab).

Interviewing businesses, the FT reported that they saw public spending cuts as an 'opportunity' to deliver "more efficient and effective public services" (FT, 2009k). A reduction of "20 to 30 per cent" in costs was possible, it reported (Ibid.). The Guardian reported that the 'rise of the state under New Labour is over' (Guardian, 2009d), with the paper almost accepting of the inevitability of sharp public spending cuts given that even Labour envisaged some reduction in its size over the next parliament (Guardian, 2009e). It quoted the CBI as saying that a drop in spending did not necessarily mean an equivalent drop in service quality (Guardian, 2009f), whilst an Observer poll of

businesses found that 91% wanted the size of central government cut and 81% thought increased public spending had not increased efficiency (Observer, 2009). Even commentators in the *Observer* supported the notion that reducing funding wouldn't necessarily mean reduced quality, with Bundred writing that even a cut of £50bn would only take us back to the spending level of 2003/04, 'when services were not noticeably worse than they are now' (Bundred, 2009). 'Don't believe the shroud wavers who tell you grannies will die and children starve if spending is cut', he continued, 'they won't (Ibid.). 'Cuts are inevitable, and perfectly manageable', he concluded (Ibid.).

Deborah Orr, in the *Independent*, likewise saw the recession as a structural adjustment in which the economy 'has shrunk and the state must shrink in proportion', (Orr, 2009). Warner agreed, noting that 'the only realistic solution to a fiscal crisis is to shrink the size of the public sector', and again invoked the power of financial markets saying that 'the taps of international credit can be turned off at the flick of a switch', making it impossible to have a big public sector without a 'thriving private and financial sector' (Warner, 2009c). Even Labour minister Alan Milburn, writing in the *Independent*, warned that there are 'limits to the role of centralised states' and that economic assistance had to take place 'without heralding a new era of economic protectionism and state interventionism' (Milburn, 2009). Anderson provided the stark alternatives between squeezing 'waste and eliminating unnecessary expenditure' in the public sector' or facing 'a sterling crisis, a gilt strike and enforced cuts on the orders of the IMF' (Anderson, 2009b; also Halligan, 2009c, 2009d). The paper argued that 'efficiencies' wouldn't be sufficient and that whole programmes would have to be eliminated, suggesting that 'unless the government puts more flesh on these plans in its November Pre-Budget report it will lack credibility and the potentially destabilising fears of government bond investors will continue to grow' (Independent, 2009r).

One element of the talk of cuts was a much more normative version, with austerity gaining a certain fashion credibility and offering the potential to re-make the state in a better way, with this meaning different things to different commentators. McRae, in the *Independent*, spoke of the 'opportunity to do things better' by changing the managerial set-up of public spending (McRae, 2009d). Top-down directives are 'corrupting', he argued, because 'it makes managers focus on what they have been told to do, rather than what the customer would like them to do' (Ibid.). Meanwhile Dejevsky co-opted Barack Obama's popular election cry of 'yes we can' to offer a cando attitude to tackling the deficit that made it seem like a popular challenge of thinking rather than a damaging problem for people's future quality of life (Dejevsky, 2009). It was possible to cut spending without causing social damage, he argued, suggesting that 'once you begin to look, the cuts just start rolling in', and calling fiscal consolidation a 'recipe (that) is so well tried and tested' (Ibid.).

For others, however, austerity was less an opportunity than a darkening approaching cloud. Papers on both sides of the political divide warned that austerity would not come easily and that it would be hard to achieve it pleasantly. The *Independent* noted that previous government 'efficiency drives' had actually increased costs (Independent, 2009t). The *Times* argued that whether people set out to cut 'nasty' or 'nice', 'there will be blood' (Times, 2009ae), while Fraser Nelson stated that

'herds of sacred cows will have to be led to the slaughterhouse' and that 'brutal cuts' will be needed to satisfy the ratings agencies (Nelson, 2009). Conway called the likely spending cuts 'savage' (Conway, 2009b), whilst Sir Howard Davies was quoted in the *Telegraph* as saying that the British "were ignorant of the scale of the crisis" (Daily Telegraph, 2009t). The *Guardian* framed Osborne's explicit call for cuts as a case of finding out whether 'voters are ready to take their medicine', suggesting that honesty about the scale of cuts to come might backfire because of their sheer scale (Wintour, 2009). They called this honesty their 'biggest gamble in a generation' (Guardian, 2009g). Despite this more cataclysmic framing of the cuts to come – note again here the inevitability of this assumption – few of those predicting the worst actually saw it as a sufficient reason to hold back from it. As such it was presented as an unfortunate and unpleasant inevitability, rather than an optional drastic measure. Given that the ratings agencies 'demanded' such action, in the words of Nelson (2009), the British state was no longer deemed to have any agency over the issue.

7.8 Conclusions

Overall, we can identify in the media reporting of the pre-election period a notable return to many of the key themes of the pre-crisis period, notably those of 'no room for manoeuvre', political trust and the over-sized nature of the British state. The fact that austerity offered a connection to a number of these existing discourses increased its validity against that of the Keynesian alternative, which called for policy measures which went against many of these beliefs. The ideational competition amongst economists about the correct course of action created no clear consensus, demonstrating to some extent the capacity of crisis to cause epistemic uncertainty in policy ideas and discourses. However, whilst constructivist thought to date has utilised such a concept of uncertainty as a key feature in allowing for policy change, the conclusions from this chapter challenge the strength of the relationship between uncertainty and change.

Through the analysis presented above, we see that the narratives which had dominated the precrisis period in fact re-emerged through the post-crisis period. In the absence of a new economic policy consensus, past criticisms of the sustainability and effectiveness of state spending returned at the heart of narratives of economic policy. The continued significance granted to financial markets and the belief that globalisation necessitated moderation of public finances further highlights, moreover, the existence of additional pillars that were returned to through this time of ideational reversion.

The first strand of ideational reversion emerged in relation to the discourse of 'no room for manoeuvre'. This narrative strand, which had temporarily reduced in its visibility the previous autumn, became increasingly prominent in early 2009 as the impact of the financial crisis on the

public finances became clear. Most commentators fell back on previous perceptions of the boundaries of acceptable deficit levels, presenting deficit reduction as the core task to be addressed through economic policy. External constraints, in terms of a perceived high likelihood of Britain falling foul of international investors and the ratings agencies, also featured in these evolving discourses. Whilst this narrative was contested by segments of the left-leaning media which applied Keynesian demand-side arguments, this was from the beginning outweighed by those questioning further fiscal expansion. As such the Keynesian side focused more on the need to prevent immediate cuts than on the need to increase stimulus.

The second strand of ideational reversion related to the continued belief that recovery must come from the private sector and that there was little the government could do to stimulate such a recovery. The need to ensure a return of normal levels of lending to business called for stimuli in the financial sector, but the lack of confidence in the multiplier effect of government fiscal spending meant that a central discourse continued to see the state as impediment rather than a central aid to future economic growth. The major role the government could play in terms of stimulating the financial sector, related to the belief in the danger of large deficits, was therefore to provide it with the confidence that it would tackle its debts in the medium term. This entailed providing a clear timeline of planned deficit reduction, which in turn necessitated spending cuts because of the Ricardian fear of tax increases.

The failure of the government to provide these assurances was tied into a complaint about the honesty of the economic reporting of politicians. Thus, subtly, advocating spending cuts became the 'honest' approach to the situation, whilst the press regularly denounced Labour for its failure to provide an honest assessment of the true state of the national finances and what cuts would need to be made. The pragmatist solution of waiting to gauge the permanent drop in tax revenues and thus the cyclically-adjusted budget deficit was therefore deemed to be a failure to be honest with the electorate and an attempt to turn attention away from the most pressing issue; the deficit. Aided to smaller discourses advocating the role of deficit reduction in future economic growth, and the developing belief in 2010 that the election should be chosen not only on the macroeconomic debate about the demand-side impact on spending cuts versus stimulus, and a discursive patchwork was created in which austerity and its vision of the state provided a much more acceptable policy response to the perceived central economic situation than stimulus did.

Together, the analysis presented through the above sections and the previous two chapters have demonstrated that, overall, the global financial crisis was followed by process of ideational reversion in media narration on the role of the state in economic management. While the immediate post-crisis period was followed by narratives of exceptionality, in which the usual standards of effective economic management were temporarily relaxed, very soon key tropes from the status quo ante were re-adopted. Prominent strands of discourse surrounding private sector growth, the size of the public sector, and political trust served to support the rise to dominance of orthodox ideas through this time. In the final chapter of the thesis, the overarching insights that are developed off the back of this analysis are elaborated.

This chapter has therefore brought all three research questions together to advance the debate on crisis, change, and constructivist institutionalism. It has shown that a full structural vacuum did not occur during the crisis, and that this meant that much of the blame was kept within discussions of the financial sector rather than the wider economy. This allowed the ideas of the pre-crisis period to continue to play a major role in framing post-crisis discourse, and as such provided a selective advantage for the austerity discourse over that of its Keynesian alternative. Moreover, by applying a augmented constructivist institutionalist approach to this issue, I have shown the value of widening definitions of crisis in a way that captures instances of political continuity as well as instances of political change.

Chapter 8: Conclusions

8.1 Introduction

'We revel in every new excuse to label our times revolutionary; ours is the atomic/permissive/electronic/affluent/space age. Attention centres on the glittering pageant and dramatic incident, rather than on the elusive processes that evoke the incidents. Revolutions must be visible, palpable, and immediate, although it is the annual change of only one percent that can produce some of the greatest transformations. Paradoxically, a glib preoccupation with the 'revolutionary' has tended to reduce our sensitivity to change itself (Heclo 1974: 1).

My suggestion in chapter three was that the new institutionalisms, in focusing on the ability of institutions to rapidly reconfigure or to produce stasis, were perhaps less comfortable in dealing with crisis situations which produced partial change. The above quote illustrates a similar analytical concern, that it is 'slow transformations' that most commonly occur. The type of revolutionary third order changes highlighted by Blyth and Hall remain relatively infrequent, and a significant focus on them perhaps leads us to over-estimate the likelihood of them happening again. Whilst this thesis has not attempted to cover a macro picture of multiple countries or medium term timeframes to explain exactly what conditions may produce such change and which may not, it has nonetheless attempted to start this process off with one particular example of relative continuity where existing research would perhaps expect to find revolutionary change. I have attempted to begin a constructivist explanation of the reasons for this evolutionary change by questioning our existing understanding of crises themselves as determining of change and major ruptures to our existing ideational webs of understanding.

To make these conclusions, I have worked within a constructivist logic to analyse the case study. In particular, I have followed the constructivist assumption that in periods of crisis, narratives, or discourses, come to play a particularly significant part in shaping the policy outcome. Because constructivism does not follow a rationalist approach to political analysis, I have stated that any given political event does not *necessitate* a particular action to be taken; all events are in this sense *essentially contestable*. Thus an economic crisis does not necessitate a Bayesian shift in governance where a rational flaw in the existing paradigm is identified and corrected, but rather what the crisis

means for such a paradigm is open to public debate. The public understanding of crisis and how it is presented and argued over is therefore particularly prominent in such a time. As such the case study chosen involved the major form of public argumentation over public policy in Britain; the broadsheet media. Undertaking an analysis utilising constructivist thought and methodology, the intention was to highlight the greater extent of continuity within political discursive trends through a crisis than has currently been postulated by the prevailing opinion of constructivist institutionalist literature.

The purpose of this discussion chapter is firstly to review the empirical lessons of the thesis as a whole, but secondly to contextualise them so as to allow for the wider implications of the findings to be more explicitly stated, and a future research agenda roughly outlined. One of the principal foci for future research stemming from this work is the need to focus on, discover and explain more about the processes of slow change and the ways these have come about. Constructivist institutionalism still, I feel, offers sufficient analytical purchase to provide strong empirically-focused explanations of why certain transformations become tenable in certain crisis situations and not in others.

This chapter will therefore begin by reviewing the major empirical contributions of the thesis. Within this it will also clarify the limitations of these findings and outline what the findings *haven't* told us. Secondly, it will outline the theoretical implications of these findings in relation to existing constructivist work. Thirdly and finally, the chapter will discuss the potential future research agenda which could flow from this work.

8.2 Overview of the Major Conclusions from the Research

This thesis has sought to develop three central arguments about the nature of economic crises and their impact on existing ideational paradigms and political discourse.

The first major empirical finding of the thesis, in relation to research question 1.) demonstrates a process of 'ideational reversion' in British political discourse. Whilst constructivism has often highlighted the capacity of crises to alter existing schools of thought through a process of exposing their faults and weaknesses, in the British case we find that many of the major understandings of the pre-crisis period continued to play a significant role in shaping responses to the crisis. There was therefore a remarkable amount of discursive stability over the role of the state in the economy, and the perceived successes and failures of New Labour and 'Brownism' played a major part in the discursive understanding of the crisis and the response it necessitated. The suggestion that the

British state sector had grown too large and could be cut without damaging front-line services, that Brown had overspent as a Chancellor and that a medium term fiscal retrenchment was needed to balance this out, and a commitment to reducing regulatory impediments to corporate businesses as the most productive mechanism to future economic growth, were all present and largely uncontested in the immediate pre-crisis context. Rather than seeing the extent of the danger posed by the financial crisis re-configure these central priorities in the popular imagination, they re-emerged as central ideational justifications for post-crisis austerity.

Despite this process of ideational reversion, in answer to research question 2.) the thesis found that the discursive barriers to change were largely removed during the period of major epistemic crisis and uncertainty in the autumn of 2008. Whilst the very future of the financial system remained unclear, state action to do whatever was necessary to stabilise the economy became acceptable. It was in this context that the greatest amount of policy ingenuity and change took place. Even here, however, the ideas of the pre-crisis period had not been completely removed from the picture, and the media as a whole maintained that the arguments for fiscal retrenchment would have to return once the immediate danger had passed. The medium-term timeframe had therefore been largely un-altered in fiscal terms by the crisis.

Linked to this is the second major finding about the extent to which the crisis was siloed within the financial sector. With blame focusing on bankers and those who regulated them, there was no apparent perceived need to re-configure the corporate capitalist economy as a result of the crisis. The only major link between the two became the issue of business having difficulty accessing credit, and again it was here that novel policy approaches begun to be developed. However, for the major part the view on incentivising business through the massaging of the supply-side, which gained greater credence through the late Brown period because of the discursive focus on his ability to create 'red tape', continued to be the principal approach to returning the economy to growth. Thus while discursive critique was applied to the notion that Britain had become too dependent on debt, and thus the financial sector, to stimulate growth, the result was not a perceived need for state activism but on freeing up corporate business to more productive ends. With statism as weak as financialisation, fiscal stimulus as a crisis-counteracting measure was not sufficiently tenable.

The consistent belief in the primacy of supply-side measures in stimulating British corporate business, economic growth and job creation was almost completely untouched by the crisis, meaning that the preferred paths to a return to growth in the post-crisis period were largely the function of the same economic ideas that had existed beforehand. Furthermore, the sense of an over-burdened British state, and the normative dislike the media had for this in the context of private and individual liberty, meant that a more activist role for the state was not generally popular in the post-crisis economic context despite the fragility of the private sector and the clear failure of free-market doctrines. This remains largely the case today, despite the continued failure of austerity to promote a strong economic recovery.

Thus we arrive at the answer to the third research question; that the ideational context of the pre, mid- and post-crisis periods produced a strategic advantage for the orthodox austerity discourse

over its Keynesian alternative. The central argument produced here holds that because austerity was able to tie itself more concretely to the discursive norms of the pre-crisis period, and because these had not been eradicated by the crisis, it was able to achieve a higher level of social sustainability than a Keynesian approach. This alternative, despite offering an academically logical approach to crisis resolution, as set out in chapter 4, often ran counter to many of the discursive trends of the time. Indeed, increasingly in the run up to the 2010 election as academic economists could not agree on the relative weight of the two schools of thought, the broadsheet media came to focus on the more medium-term ideals epitomised by the two approaches than their actual crisis-specific content. Thus, when extrapolated to contexts in which its major arguments held no relation 15, the orthodox 'ideal' of the state fitted better with the ideas of an over-sized public sector and business suffering from red tape than Keynesian ideal of an active state managing public demand. Thus, while constructivist work to date tends to posit that crisis situations radically re-shape the discursive context and allow for new ideas to gain prominence, it has been suggested through this argument that the context of the pre-crisis period is still very important in shaping how the crisis is viewed.

The thesis has also sought to clarify that as constructivists we should not fall into the trap of being over-determinist with the link between crisis and change. Because crises can lead to third order policy change, it does not necessarily mean that they will. This is because, following a constructivist position on social reality, there is no such thing as an objectively failed paradigm exposed by a crisis. Thus in the absence of objective failure, there is only perceived failure. Whilst crisis opens up a space for critique of policy norms in a way that there has not been before, a strategic-relational approach to political analysis informs us that the specific political discursive context of a particular polity may still not be conducive to rapid change despite the external economic conditions bringing existing norms increasingly into question.

Constructivists should therefore be careful not to posit that a crisis is a situation in which the status quo cannot continue to explain events in its own terms, but only one in which its dominance of the political narrative is not a given. This is not necessarily something which goes against existing conceptions of crisis in the constructivist institutionalist literature, but it is something that it is important to highlight more explicitly that has hence been the case. The idea of a structural vacuum created by Knightian uncertainty, particularly prevalent in Blyth's (2002) conception of crisis narratives could well be read in such a way, but could also be read to suggest that the precrisis narrative had failed and thus that the argumentation over how to explain the Great Depression was between competing new discourses. I wish to highlight more explicitly that existing discourses, in the absence of their objective failure in a crisis context, need not necessarily be excluded from the post-crisis picture.

The thesis has also attempted to make contributions to existing research by developing a different focus of study and methodological approach to what has commonly been used in constructivist

¹⁵ Keynesianism, after all, is significantly distinctive to orthodoxy in policy terms only in a crisis.

research. Because a critique was made that existing work didn't focus sufficiently on the actual crisis period itself, and thus still had data limitations surrounding ideational stability and instability through a crisis, a more rigorous and dynamic focus on a shorter time frame was chosen. This dynamic tracing methodology sought to capture exactly which discourses common in pre-crisis political discussion lost purchase in the context of a radically altered economic landscape, and which were unaltered. It was through this process that the 'ideational reversion' of the British public discourse could be captured. The more systematic incorporation of media debate as a context within which political actors must operate helps to provide an understanding of why certain political ideas became, or remained, accepted, and why others didn't. Thus it allows for a more detailed explanation of crisis dynamics and ideational stability than has hitherto been common in the literature. This methodology has therefore been important in helping to explain the relative continuity of the British case during the Global Financial Crisis.

Moreover the more general focus on the media was justified by the very public nature of the need for reflection on the governance of the economy that the crisis initiated. Seeing crises as evolving selective contexts, rather than pre-determinants of change, called for a focus on how such a context created acceptability and social sustainability for different ideas. Thus whilst the thesis has not argued that the media are the sole generators of credibility or that they create a structural straitjacket in which political actors have no independent power, it has sought to view the media arena as a neutral one which nonetheless helps to create the conditions for credibility of different ideas and suggestions. I have not attempted to make the case that other foci are less relevant or valuable, however. Indeed a further step for similar research is to combine in more detail the public ideas seen through the lens of the media and the political strategies of those who existed within such an ideational context. Within the limitations of this study, it has been argued that a focus on the media alone is sufficient to explain this context and hence explains why policy stability was the predominant feature.

From the empirical findings of the thesis, some conclusions about the state of constructivist research can also be made. Firstly it has demonstrated the need for constructivists to accept the 'stickiness' of ideas, and remove the determinist suggestion that crises will necessarily remove all ideational impediments to change. As such a central contention has been that economic shocks do not re-order the political-economic discursive context to the extent that has currently been suggested. The ability of political discourse to 'pre-diagnose' the potential limits to state action should not therefore be under-estimated. As such, whilst crises are often seen as 'shock' events unforeseen by the existing paradigm, in this case there was an expectation of a global slowdown before the crisis occurred. The possible role of the state in such a context had already therefore been debated by the media in advance of the financial crisis, allowing for common understandings to be reached. Whilst constructivist work to date would not expect these understandings to survive the confusion of crisis, the conclusions reached here suggest otherwise.

Secondly, a conclusion can be made about why change occurs in certain contexts and not in others. The thesis has not attempted to argue that crises cannot lead to change, but merely that we

should not expect this to always be the case. By highlighting the stickiness of ideas and the discursive contexts within which crises occur, we can begin to explain why certain policy positions became tenable and credible. Within the British crisis experience, the growing loss of confidence in the New Labour project and the pre-crisis belief that future change in Gordon Brown's 'statism' would be required, provided a context in which it became difficult to create a socially sustainable programme featuring fiscal stimulus. As well as these shorter-term discourses, longer term ideas about the value of governmental fiscal prudence, epitomised by Labour's 'Golden Rules' as well as discourses of the fragility of the state in a globalised world, coalesced to add greater credence to austerity arguments than those opposing them.

An applied reading of Hay and Jessop's strategic-relational approach to the structure/agency 'problem' of political science should not necessarily lead to these conclusions being surprising, but it is my contention here that this important conception of the relationship between the political structures of the time and the political agents who live within them has often not been fully applied to research relating to critical junctures and points of change. Rather, this work has often been overly determinist in assuming that existing policy norms will be shunned through an objective analysis of their failure which is actually at odds with a constructivist reading of social reality.

It also highlights in relation to political economy that economic arguments do not exist in a vacuum independent of other strands of political discourse, but that the very political arguments over the 'statism' of Gordon Brown infused themselves into economic arguments over the necessary response to the financial crisis. Short and medium-term timeframes here became confused, with a short-term policy of stimulus being successfully equated with the sort of statism and irresponsible spending that few saw as an acceptable medium-term approach to the British growth model. Here again we see that the political context of the pre-crisis period made it harder to advocate a more classically Keynesian response to the crisis. But moreover it highlights that the ideational context created by the British press through their depiction of economic priorities and acceptable values and ideas created conditions in which it was impossible for New Labour to offer a genuine Keynesian alternative because of their need to attempt to counter the political narratives of the Conservative opposition which the economic context had provided them with. Thus we can rarely expect policy ideas that emerge from a crisis to be fully and logically attached to a specific economic philosophy, but are rather a type of fudge based on the institutional, discursive and electoral contexts of the time. Again, in such circumstances, the tendency is towards a relatively greater amount of stability than of rapid change.

This was perhaps heightened in the British case by the failure of economists to agree on a new policy consensus. Given the lack of academic and 'expert' consensus on the 'meaning' of the crisis for issues such as fiscal policy, I highlight the significant role played by the aforementioned discourses of the state from the pre-crisis period in the post-crisis debates over preferred policy courses. Thus, à la constructivist thought, the debate over economic policy choices in this case study was not a purely technical exercise but a deeply *political* one. However, *contra* constructivist thought, the absence of economic consensus meant that pre-crisis discourses played a central role in

these debates and lead me to conclude that to the extent that the context was favourable to a specific paradigm, it was favourable to the orthodox school in that this had more in common with existing discourses than the Keynesian did. Thus crisis failed to remove completely the ideational and discursive context which favoured a largely neo-liberal approach to the role of the state.

The siloing noted in the blame for the crisis also has important implications for future constructivist research into paradigm failure. Whilst this thesis has not sought to cover the discourses relating to financial regulation and monetary policy in any great detail, it took as a starting point the acceptance that there has been a general shift in some of the ideas relating to such issues. The development of macro-prudential ideas over efficient-market ones in the governance of the financial system is the most notable here, although the extent of change in the wider role of the financial sector in the wider British political economy could well be questioned despite this. Regardless, what is interesting theoretically from this research is that clearly the issue of crisis necessitating change did not 'diffuse' down from the financial system to other areas of government economic policy. Thus the central theoretical conclusion from such an observation is the importance of recognising that a financial crisis does not necessitate a change in all ideas which relate to that paradigm. We may well observe that the pre-crisis ideas in relation to both the financial and corporate sectors were 'neoliberal' in nature, but the ability of a crisis which had its locus in the financial sector to produce a 'crisis of neoliberalism' has clearly been overstated. Like the attempt to prevent financial contagion, such ideational contagion was clearly limited in this case as blame stayed largely within the financial system and has rarely spilled over into a wider blame of the neoliberal political economy.

Thus we must accept that crisis can often be contained, publicly if not academically, within certain spheres of a state's political economy, and it does not necessarily diffuse down to all others, even if they are governed by largely similar ideas. It is my contention that existing constructivist work has over-stated the extent to which such 'failure' must be ubiquitous. Academics may draw a conceptual link between the ideas governing the financial and corporate sectors and suggest that if one has failed, the other must be faulty also, but this need not necessarily permeate public discussion. Whilst one could criticise the inconsistency of changing one set of ideas and keeping an identical set for other sectors of the economy, the nature of constructivism in questioning selfevident objective social truths does not necessitate this in the slightest. We may consider it inconsistent on an academic level, but the nature of a contestable social world does not mean that successful political discourses need be consistent, only popular. Indeed, the fact that the conceptual lines we as academics draw are not hermeneutically common in the population as a whole means that such an inconsistency in ideas need not appear inconsistent in the slightest. Daigneault's (2014: 483) notion of the 'fungible' nature of policy paradigms appears particularly relevant here, highlighting as it does that whilst neat in theory, in practice there are rarely simply dichotomous splits between policy paradigms. The 'coarse tuning' or 'privatised' versions of Keynesianism in

the New Labour era provide valuable pre-existing examples here (Clift and Tomlinson, 2006; Crouch, 2009 respectively).

Whilst the thesis has intended to work within the constructivist tradition, some of the conclusions could also be of value to different perspectives on political institutions. In many senses, given its focus on continuity, the thesis could be argued to offer support for historical institutionalism. The fact that crisis does not always create ideational instability, leading to continuity, could be seen to justify the historical institutionalist focus on institutional inertia, which tends to lead to exogenous factors being used to explain revolutionary change. The potential overlap of the theories was noted in chapter 3, and indeed to the extent that an ideationally-driven form of historical institutionalism is possible the conclusions of the thesis naturally slot into this overlap. Likewise the same could be said of sociological institutionalism, with the thesis explaining that norms and beliefs are more solid under crisis conditions than has perhaps previously been the case. I do not necessary argue that these conclusions prove the primacy of those theories over that of constructivist institutionalism, however, merely that they can offer scope for greater communication between the three and demonstrate the potential for shared research foci. The application of a strategic-relational lens on the findings places my conclusions more naturally within constructivist institutionalist reasoning, I feel, but not at the expense of other institutionalisms.

8.3 Further Research

As has been suggested, the global financial crisis has considerably problematised constructivist understandings of change. The intention of this thesis has been to begin to re-frame these understandings away from looking for examples of revolutionary change and making theoretical links to crisis as an explainer of these, but towards a more general discussion of slow transformations, continuity and change within turbulent economic contexts. From this a few suggestions can be made for fruitful future research along similar lines.

Firstly, as was mentioned in the previous section, this work could benefit from being extended and further supported through an inclusion of political strategies and policies in the context of the discursive environment demonstrated by this thesis. This would help to demonstrate more clearly and dynamically the ways in which political outcomes are shaped by perceptions of what is acceptable in a given selective context. Whilst this thesis has not focused on the elite level, it has not rejected it as a valuable avenue of research. Complementing this would therefore be a qualitative approach utilising elite interviews to gain a greater understanding on the impact of the

crisis on what ideas politicians held over economic governance and fiscal policy, and how they implemented electoral strategies based on this. There was not space within this study to include this extra layer of research, but it is certainly one that would complement it.

Secondly the findings of this thesis could be greater supported by a wider approach entailing a greater number of case studies to enhance our understanding of the ability of economic ruptures to create sustainable new ideas in different contexts. From this, a more generalised understanding of how different 'types' of crisis tend to impact the modern state could be gleaned.

Further theoretical re-thinking and reflection may also help to benefit such work. The return to more structuralist explanations of continuity outlined by Blyth creates a certain amount of confusion about the exact tenets of constructivist institutionalist thought. Whilst previously it has tended to focus only on periods of revolutionary change, the logics of its meta-theoretical positions as outlined by Hay need not be applied only in such circumstances, and indeed have always implicitly made statements about continuity through the corollary of their arguments about change. Whilst here I have argued that strategic-relational analysis can help to inform our expectations of change by contextualising to a greater extent the environment in which crises occur and the differing factors likely to shape whether new ideas can become sustainable, it is quite possible that alternative valid approaches are available to constructivist scholars. Further internal discussion between such scholars will hopefully prove valuable in clarifying what exactly the school as a whole can say about politics, change and crisis.

Abbreviations

Abbreviation Description

ARRA American Recovery and Reinvestment Act

BoE Bank of England

CDA Critical Discourse Analysis

Fed Federal Reserve
FT Financial Times

GFC Global Financial Crisis
GDP Gross Domestic Product
IoD Institute of Directors

IMF International Monetary Fund

IR International Relations
MP Member of Parliament
PBR Pre-Budget Report

SME Small or Medium sized Enterprise
TARP Troubled Assets Relief Program

UK United Kingdom

USA United States of America

VAT Value Added Tax

Bibliography

Abdelal, R. (2007) *Capital Rules: The Construction of Global Finance*, London: Harvard University Press.

Abdelal, R., Blyth, M. and Parsons, C. (2010) 'Introduction: Constructing the International Economy' in R. Abdelal, M. Blyth and C. Parsons (eds), *Constructing the International Economy*, Ithaca, NY: Cornell University Press, 1-19.

Adler, E. (1997) 'Seizing the Middle Ground: Constructivism in World Politics', *European Journal of International Relations*, 3(3), 319-63.

Afonso, A. (2010) 'Expansionary Fiscal Consolidations in Europe: New Evidence', *Applied Economics Letters*, 17(2), 105-09.

Alesina, A., Mirrlees, J. and Neumann, J. (1989) 'Politics and Business Cycles in Industrial Democracies', *Economic Policy*, 4(8), 55-98.

Alesina, A., Cohen, G. and Roubini, N. (1993) 'Electoral Business Cycle (sic) in Industrial Democracies', *European Journal of Political Economy*, 9(1), 1-23.

Alesina, A. and Perotti, R. (1994) 'The Political Economy of Budget Deficits', *NBER Working Paper Series: Working Paper No. 4637*.

Alesina, A. and Ardagna, S. (2010) 'Large Changes is Fiscal Policy: Taxes versus Spending' in J. Brown (ed), *Tax Policy and the Economy*, Chicago: NBER and University of Chicago Press, 35-68. Alesina, A. and Giavazzi, F. (2013) 'Introduction to "Fiscal Policy after the Financial Crisis" in A. Alesina and F. Giavazzi (eds), *Fiscal Policy after the Financial Crisis*, Chicago: NBER and University of Chicago Press, 1-18.

Allcock, J. (2000) 'Constructing the 'Balkans' in J. Allcock and A. Young (eds), *Black Lambs and Grey Falcons: Women Travelling in the Balkans* 2ed, Oxford: Berghahn Books, 217-240.

Almunia, M., Bénétrix, A., Eichengreen, B., O'Rourke, K. and Rua, G. (2010) 'From Great Depression to Great Credit Crisis: Similarities, Differences and Lessons', *Economic Policy*, 25(62), 219-65.

Amenta, E. and Poulson, J. (1996) 'Social Politics in Context: The Institutional Politics Theory and Social Spending at the End of the New Deal', *Social Forces*, 75(1), 33-60.

Anderson, B. (2008) 'To silence his Tory critics, Osborne must do more. Now', *Independent*, November 17th, 32.

Anderson, B. (2009a) 'Credit got us into this mess; now we need more to get us out', *Independent*, August 10th, 25.

Anderson, B. (2009b) 'You can cut waste and save services', *Independent*, August 24th, 24.

Aoki, M. (2002) Toward a Comparative Institutional Analysis, Cambridge, MA: MIT Press.

Archer, M. (1989) Culture and Agency, Cambridge: Cambridge University Press.

Archer, M. (1995) Realist Social Theory: The Morphogenetic Approach, Cambridge: Cambridge University Press.

Archer, M. (1998) 'Social Theory and the Analysis of Society' in T. May and M. Williams (eds), *Knowing the Social World*, Buckingham: Open University Press, 69-85.

Armingeon, K. (2012) 'The Politics of Fiscal Response to the Crisis of 2008-09', *Governance*, 25(4), 543-565.

Arndt, H. (1944) The Economic Lessons of the Nineteen-Thirties, London: Frank Cass & Co.

Ashbee, E. (2011) 'Fiscal Policy Responses to the Economic Crisis in the UK and the US' in T.

Casey (ed), *The Legacy of the Crash: How the Financial Crisis Changed America and Britain*, Basingstoke: Palgrave MacMillan, 79-98.

Ashley, R. and Walker, R. (1990) 'Speaking the Language of Exile: Dissident Thought in International Studies', *International Studies Quarterly*, 34(3), 259-68.

Ashley, J. (2009) 'As the clouds gather, Brown has to hug allies tight', *Guardian*, April 20th, 27.

Auerbach, A. and Gorodnichenko, Y. (2012), 'Fiscal Multipliers in Recession and

Expansion', American Economic Journal: Economic Policy, 4(2), 1–27.

Bagaria, N., Holland, D. and Van Reenen, J. (2012) 'Fiscal Consolidation During a Depression', *National Institute Economic Review*, 221, F42-54.

Baglan, D. and Yoldas, E. (2013) 'Government Debt and Macroeconomic Activity: A Predictive Analysis for Advanced Economies', *Finance and Economics Discussion Series*, Washington, DC: Federal Reserve.

Baker, A. (2012) 'Varieties of Financial Crisis, Varieties of Ideational Change: How and Why Financial Regulation and Macroeconomic Policy Differ', Paper presented at SPERI Inaugural Conference, Sheffield, 16th July.

Baran, P. and Sweezy, P. (1966) 'Monopoly Capital: An Essay on the American Economic and Social Order', *Science and Society*, 30(4), 461-496.

Barberis, P. (2006) 'The Labour Party and Mr Keynes in the 1930s: A Partial Keynesian Revolution Without Keynes', *Labour History Review*, 71(2), 145-166.

Barkin, J. (2003) 'Realist Constructivism', International Studies Review, 5(3), 325-42.

Baumgartner, F. (2013) 'Ideas and Policy Change', Governance, 26(2), 239-58.

Bean, C. (2007) 'Is There a New Consensus in Monetary Policy?', Bank of England, available at http://www.bankofengland.co.uk/publications/Documents/other/monetary/bean070413.pdf, accessed 17/09/2012.

Béland, D. (2009) 'Ideas, Institutions, and Policy Change', *Journal of European Public Policy* 16(5), 701-18.

Béland, D. and Cox, R. (2011) 'Introduction: Ideas and Politics' in D. Béland and R. Cox (eds), *Ideas in Social Science Research*, Oxford: Oxford University Press, 3-22.

Bell, S. (2011) 'Do We Really Need a New 'Constructivist Institutionalism' to Explain Institutional Change?', *British Journal of Political Science*, 41(4), 883-906.

Bell, S. and Hindmoor, A. (2014) 'The Structural Power of Business and the Power of Ideas: The Strange Case of the Australian Mining Tax', *New Political Economy*, 19(3): 470-486.

Bell, S. and Hindmoor, A. (2015) 'Taming the City? Ideas, Structural Power and the Evolution of British Banking Policy Amidst the Great Financial Meltdown', *New Political Economy*, 20(3), 454-74.

Berman, S. (1998) *The Social Democratic Moment: Ideas and Politics in the Making of Interwar Europe*. Cambridge: Harvard University Press.

Berman, S. (2006) *The Primacy of Politics: Social Democracy and the Making of Europe's Twentieth Century*, Cambridge: Cambridge University Press.

Berman, S. (2013) 'Ideational Theorizing in the Social Sciences since "Policy Paradigms, Social Learning and the State", *Governance*, 26(2), 217-37.

Bernanke, B. (2009) 'The Crisis and the Policy Response', Speech at the LSE, January 13th.

Transcript available at: http://www.federalreserve.gov/newsevents/speech/bernanke20090113a.htm, accessed 12/02/2014.

Bevir, M. and Rhodes, R. A.W. (2003) Interpreting British Governance. London: Routledge.

Bieler, A. and Morton, A. (2002) 'The Gordian Knot of Agency-Structure in International Relations: A Neo-Gramscian Perspective', *European Journal of International Relations* 7(1), 5-35.

Blanchard, O., Dell'Ariccia, G. and Mauro, P. (2010) 'Rethinking Macroeconomic Policy', *Journal of Money, Credit and Banking*, 42(6), 199-215.

Blanchflower, D. (2009a) 'A few morsels won't solve the unemployment crisis', *Guardian*, Newsprint Supplement, April 23rd, 7.

Blanchflower, D. (2009b) 'Cameron's dud options: The Tory economic plans have the potential to push Britain's economy into a death spiral of decline', *Guardian*, October 10th, 34.

Blinder, A. and Zandi, M. (2010) 'How the Great Recession was Brought to an End', available at http://www.economy.com/mark-zandi/documents/End-of-Great-Recession.pdf, accessed 08/02/2012.

Blondel, J. (1995) *Comparative Government: An Introduction* 2ed, Hemel Hempstead: Prentice Hall.

Blyth, M. (1997) "Any More Bright Ideas?" The Ideational Turn of Comparative Political Economy', *Comparative Politics*, 29(2), 229-50.

Blyth, M. (2002) *Great Transformations: Economic Ideas and Institutional Change in the Twentieth Century*. Cambridge: Cambridge University Press.

Blyth, M. (2003) 'Structures Do Not Come with an Instruction Sheet: Interests, Ideas, and Progress in Political Science', *Perspectives on Politics*, 1(4), 695-706.

Blyth, M. (2011) 'Ideas, Uncertainty, and Evolution' in D. Béland and R. Cox (eds), *Ideas in Social Science Research*, Oxford: Oxford University Press, 83-103.

Blyth, M. (2013) 'Paradigms and Paradox: The Politics of Economic Ideas in Two Moments of Crisis', *Governance*, 26(2), 197-215.

Bolchover, D. (2008) 'Just when we need real capitalists, the fake ones have taken over', *Sunday Telegraph*, October 12th, 27.

Boli, J., Meyer, J. and Thomas, G. (1989) 'Ontology and Rationalization in the Western Cultural Account' in G. Thomas, J. Meyer, F. Ramirez and J. Boli (eds), *Institutional Structure: Constituting State, Society, and the Individual*, London: Sage, 10-25.

Booth, A. (1987) 'Britain in the 1930s: A Managed Economy?', *Economic History Review*, 40(4), 499-522.

Bootle, R. (2007) 'Gordon's last hurrah sounded so bold, but is only robbing Peter to pay...Peter', *Daily Telegraph*, March 22nd, 9.

Bootle, R. (2008a) 'Now is not the time to bolster the public finances', *Daily Telegraph*, February 4th, City, 2.

Bootle, R. (2008b) 'This Budget will mark the end of the Brown illusion', *Daily Telegraph*, March 10th, 2.

Bootle, R. (2008c) 'Darling does not seem to appreciate the mess we're in', *Daily Telegraph*, March 13th, 6.

Bootle, R. (2008d) 'We now face Keynesian conditions and need truly Keynesian solutions', *Daily Telegraph*, October 27th, City, 2.

Bootle, R. (2008e) 'There'll never be a stronger case for increased government borrowing', *Daily Telegraph*, November 17th, City, 2.

Bootle, R. (2009a) 'This recession demands that we use logic and spend our way out of it', *Daily Telegraph*, January 12th, City, 2.

Bootle, R. (2009b) 'If the budget was a numbers game, the Chancellor definitely lost', April 23rd, 2.

Bootle, R. (2009c) 'Debt reduction is better done by cuts in spending than raising taxes', *Daily Telegraph*, April 27th, City, 2.

Bootle, R. (2009d) 'Let's have a five-year spending freeze to kill off Big Government for good', *Daily Telegraph*, June 8th, City, 2.

Bootle, R. (2009e) 'Footling measures are of no use when dealing with Public Sector debt', *Daily Telegraph*, October 26th, City, 2.

Bootle, R. (2009f) 'As the nights draw In so should the deficit, with the aid of sound policies', *Daily Telegraph*, November 30th, City, 3.

Born, B., Ehrmann, M. and Fratzscher, M. (2011) 'How Should Central Banks Deal with a Financial Stability Objective? The Evolving Role of Communication as a Policy Instrument' in S. Eijffinger and D. Masciandaro (eds), *Handbook of Central Banking, Financial Regulation and Supervision After the Financial Crisis*, Cheltenham: Edward Elgar, 244-67.

Börzel, T. and Risse, T. (2003) 'Conceptualizing the Domestic Impact of Europe' in K. Featherstone and C. Radaelli (eds), *The Politics of Europeanisation*, Oxford: Oxford University Press, 57-81.

Bowman, A., Erturk, I., Froud, J., Law, J., Leaver, A., Moran, M. and Williams, K. (2012) 'Central Bank Led Capitalism?', Paper presented at SPERI Conference, July 17th.

Boyer, R. (2012) 'The Four Fallacies of Contemporary Austerity Policies: The Lost Keynesian Legacy', *Cambridge Journal of Economics*, 36(1), 283-312.

Boyer, R. and Saillard, Y. (2001) Regulation Theory: The State of Art, Cheltenham: Routledge.

Brittan, S. (2008) 'This 'bold' cut is barely adequate', FT, November 7th, 11.

Brittan, S. (2009) 'A cool look at the current deficit hysteria', FT, October 2nd, 9.

Brogan, B. (2009a) 'Desperate and fighting to the end', *Daily Telegraph*, April 23rd, 19.

Brogan, B. (2009b) 'Labour has given up governing and now just wants a class war', *Daily Telegraph*, December 3rd, 26.

Broome, A., Clegg, L. and Rethel, L. (2012) 'Global Governance and the Politics of Crisis', *Global Society*, 26(1), 3-17.

Brown, M. (2008a) 'A crisis that leaves Cameron struggling for a meaningful role', *Independent*, October 1st, 28.

Brown, M. (2008b) 'Cameron must ditch Osborne as Tory shadow chancellor', *Independent*, November 14th, 42.

Brown, M. (2008c) 'The most irresponsible budget I have ever heard', *Independent*, November 25th, 32.

Brummer, A. (2007a) 'The power of lower taxes', *Daily Mail*, March 22nd, 10.

Brummer, A. (2007b) 'Britain hits the high notes', Daily Mail, March 23rd, 83.

Brummer, A. (2007c) 'Labour's credibility gap', Daily Mail, October 11th, 87.

Brummer, A. (2008a) 'The danger of Mr do-nothing', Daily Mail, March 13th, 15.

Brummer, A. (2008b) 'The case for big tax rebates', *Daily Mail*, November 22nd, 109.

Brummer, A. (2008c) 'Taking a risk with sterling', Daily Mail, December 4th, 83.

Brummer, A. (2009a) 'Pitfalls of cutting the budget', Daily Mail, April 7th, 80.

Brummer, A. (2009b) 'This is going to hurt, make no mistake', Daily Mail, April 21st, 84.

Brummer, A. (2009c) 'Britain faces perilous times', *Daily Mail*, December 31st, 67.

Budd, A. (2010) 'Fiscal Policy Under Labour', National Institute Economic Review, 212, R34-R38.

Buiter, W. (2009a) 'Timid Darling is left with little room for manoeuvre', FT, April 23rd, 22.

Buiter, W. (2009b) 'Prevarication and Newspeak will not fix our finances', December 10th, 12.

Bundred, S. (2009) 'We've had years of growth – so let's not be afraid of cuts', *Observer*, July 5th, 9.

Burnham, P., Lutz, K., Grant, W. and Layton-Henry, Z. (2008) *Research Methods in Politics* 2ed, Basingstoke: Palgrave MacMillan.

Cable, V. (2007) 'Darling needs to be bold to fix the mess created by Brown', *Daily Telegraph*, City, 5.

Cable, V. (2009) 'This recession is very far from over', *Independent*, June 17th, 26.

Cameron, D. (2008) 'A borrowing binge', Guardian, November 18th, 34.

Capoccia, G. and Keleman, R. (2007) 'The Study of Critical Junctures: Theory, Narrative and Counterfactuals in Historical Institutionalism', *World Politics*, 59(2): 341-69.

Carlsnaes, W. (1992) 'The Agency-Structure Problem in Foreign Policy Analysis', *International Studies Quarterly*, 6(3), 245-70.

Carragee, K. and Roefs, W. (2004) 'The Neglect of Power in Recent Framing Research', *Journal of Communication*, 54(2), 214-33.

Carrick-Hagenbarth, J. and Epstein, G. (2012) 'Dangerous Interconnectedness: Economists' Conflicts of Interest, Ideology and Financial Crisis', *Cambridge Journal of Economics*, 36(1), 43-63.

Carstensen, M. (2011) 'Ideas are Not as Stable as Political Scientists Want Them to Be: A Theory of Incremental Ideational Change', *Political Studies*, 59(3), 596-615.

Cavendish, C. (2008) 'Forget housebuyers. Help small businesses', Times, October 17th, 31.

CBO (2010) Estimated Impact of the American Recovery and Reinvestment Act on Employment and Economic Output From July 2010 Through September 2010, Publication Number 4229, Available at: https://www.cbo.gov/sites/default/files/111th-congress-2009-2010/reports/11-24-arra.pdf, accessed 10/01/2013.

Chakrabortty, A. (2008) 'Capitalism lies in shambles, and the left has gone awol', *Guardian*, August 7th, 30.

Checkel, J. (1999) 'Social Construction and Integration', *Journal of European Public Policy*, 6(4), 545-60.

Chong, D. (1993) 'How People Think, Reason, and Feel about Rights and Liberties', *American Journal of Political Science*, 37(3), 867–99.

Clarke, P. (2009) 'This is no time to throw away the crutches', August 31st, 7.

Clift, B. and Tomlinson, J. (2012) 'When Rules Started to Rule: The IMF, Neo-Liberal Economic Ideas and Economic Policy Change in Britain', *Review of International Political Economy*, 19(3), 477-500.

Clift, B. and Woll, C. (2012) 'Economic Patriotism: Reinventing Control Over Open Markets', *Journal of European Public Policy*, 19(3), 307-23.

Coates, D. (2012) 'Dire Consequences: The Conservative Recapture of America's Political Narrative', *Cambridge Journal of Economics*, 36(1), 145-53.

Cochrane, J. (2009) 'How did Paul Krugman get it so Wrong?' available at http://ricardo.ecn.wfu.edu/~cottrell/ecn272/cochrane.pdf, accessed 20/01/2013.

Collingwood, R. (1965) Essays in the Philosophy of History, Austin: Texas University Press.

Connolly, W. (1993) The Terms of Political Discourse 3ed, Oxford: Blackwell.

Conway, E. (2007a) 'Too few hits on Brown's record', *Daily Telegraph*, March 20th, 7.

Conway, E. (2007b) 'Fighting debt with debt is a risky move', *Daily Telegraph*, October 10th, 3.

Conway, E. (2009a) 'Tumbling Towards a Sovereign Debt Crisis?', Daily Telegraph, May 26th, 3.

Conway, E. (2009b) 'How bad will it get? Take a look at Latvia; Britain will not escape the savage spending cuts seen elsewhere in Europe', *Daily Telegraph*, October 15th, 3.

Conway, E. (2009c) 'Tory plans for austerity cuts could be a disaster, warns survey', *Daily Telegraph*, November 2nd, 3.

Conway, E. (2009d) 'Harsh warning for next government', Daily Telegraph, November 18th, 28.

Conway, E. (2009e) 'There's only one escape from our debt trap', *Daily Telegraph*, December 17th, p.29.

Conway, E. (2010a) 'We must ward off a catastrophe of our own; it is essential that Britain learns from the crisis that has plunged Greece into turmoil', *Daily Telegraph*, February 4th, 3.

Corrigan, T. (2008) 'Nationalising the banks is an act that should be kept in the wings for now', *Daily Telegraph*, November 28th, 7.

Corsetti, G. (2012) 'Has Austerity Gone Too Far?' VoxEU.org, available at http://www.voxeu.org/article/has-austerity-gone-too-far-new-vox-debate, accessed 08/08/2012.

Corsetti, G., Kuester, K., Meier, A. and Müller, G. (2010), 'Debt Consolidation and Fiscal Stabilization of Deep Recessions', *CEPR Discussion Paper No. DP7649*.

Cox, R. (2001) 'The Social Construction of an Imperative: Why Welfare Reform Happened in Denmark and the Netherlands but Not in Germany', *World Politics*, 53(3), 463-98.

Dahl, R. (1961) Who Governs? Democracy and Power in an American City, New Haven, CT: Yale University Press.

Daigneault, P-M. (2014) 'Puzzling about Policy Paradigms: Precision and Progress', *Journal of European Public Policy*, 21(3), 481-84.

Daily Mail (2007a) 'Tories test the fiscal waters', March 2nd, 89.

Daily Mail (2007b) 'Brown leaves an uncertain legacy', March 21st, 82.

Daily Mail (2007c) 'What Gord giveth, Gord taketh away', March 22nd, 1.

Daily Mail (2007d) 'Packed with befuddlement', March 22nd, 12.

Daily Mail (2007e) 'Final flourish to ten remarkable years', March 22nd, 14.

Daily Mail (2007f) 'Labour's next leader sets his sights on winning votes', March 22nd, 82.

Daily Mail (2007g) 'Brown's hidden £2bn tax grab', March 23rd, 6.

Daily Mail (2007h) 'Forty new taxes hidden in Brown's Budget, say Tories', March 24th, 6.

Daily Mail (2007i) 'Credit Crunch could clip Darling's wings', October 8th, 67.

Daily Mail (2007i) 'Labour's smash and grab raid', October 10th, 11.

Daily Mail (2007k) 'Darling stands up to scrutiny', October 11th, 14.

Daily Mail (2007l) 'Labour's credibility gap', October 11th, 87.

Daily Mail (2008a) 'Darling's gamble will decide the election', March 13th, 16.

Daily Mail (2008b) 'Tax rises? You ain't seen nothing yet', March 14th, 20.

Daily Mail (2008c) 'Osborne: I might not have room for tax cuts', March 15th, 10.

Daily Mail (2008d) 'Cuts and tax rises to follow recession', October 30th, 2.

Daily Mail (2008e) 'A gamble to make or break Britain', November 25th, 14.

Daily Mail (2008f) 'UK growth hopes could be shattered by global storm', November 25th, 75.

Daily Mail (2008g) 'Prudent Policy', December 29th, 14.

Daily Mail (2009a) 'Darling's chance to pick up the pieces' April 7th, 3.

Daily Mail (2009b) '£1250-a-year tax rise 'needed to save basket case Britain", April 7th. 22.

Daily Mail (2009c) 'Alistair in Wonderland', April 23rd, 78.

Daily Mail (2009d) 'Borrowing like this could push us over the edge', April 23rd, 64.

Daily Mail (2009e) 'But is Alistair really getting tough?', April 24th, 57.

Daily Mail (2009f) 'Cuts delayed until after the election', April 24th, 10.

Daily Mail (2009g) 'Public debt 'could wreck economy', May 21st, 3.

Daily Mail (2009h) 'Now take the public finances out of MPs' hands', June 9th, 20.

Daily Mail (2009i) 'IMF sounds alarm over scale of UK debt', July 17th, 23.

Daily Mail (2009j) 'Why we're lagging behind in the recovery race', August 14th, 59.

Daily Mail (2009k) 'Time for some honesty from Labour', August 21st, 12.

Daily Mail (20091) 'Why we're all cutters now – even the cat', September 16th, 48.

Daily Mail (2009m) 'Cameron's great gamble pays off', October 10th, 44.

Daily Mail (2009n) 'Britain's in danger of going bust, warns EU', October 15th, 6.

Daily Mail (2009o) 'Battle lines drawn over Britain's debt', November 24th, 9.

Daily Mail (2009p) 'On the brink of financial armageddon', November 28th, 7.

Daily Mail (2009q) 'History will be withering about Darling's abject dereliction of duty', December 10th, 67.

Daily Mail (2010a) 'Cuts? Wait until after the poll!', March 25th, 32.

Daily Mail (2010b) 'If this financial statements had been delivered by a major company, the fraud squad would have been called in', March 25th, 61.

Daily Mail (2010c) 'George Osborne and a test of confidence', March 29th, 23.

Daily Mail (2010d) 'They'll all cut more than Maggie', March 30th, 65.

Daily Mail (2010e) 'Crisis debate? It was the three dodgers show', March 31st, 58.

Daily Telegraph (2007a) 'Tories should tell us which taxes they'll cut', March 13th, 23.

Daily Telegraph (2007b) 'Try these, Gordon: education, education, and less taxation', March 19th, 25.

Daily Telegraph (2007c) 'Brown's tax cut trick', March 22nd, 1.

Daily Telegraph (2007d) 'The tactics: Chancellor's political fingerprints are all over it', March 22nd, 3.

Daily Telegraph (2007e) 'Brown's sleight of hand', March 22nd, 4.

Daily Telegraph (2007f) 'It's a start but UK is a lap behind', March 22nd, 8.

Daily Telegraph (2007g) 'Whitehall will suffer, but the Chancellor may have made companies eager to invest', March 23rd, 5.

Daily Telegraph (2007h) 'The Tories' solid economic legacy has been squandered', September 17th, 22.

Daily Telegraph (2008a) 'Darling would need £240 tax rise to fund budget shortfall', March 10th, 1.

Daily Telegraph (2008b) 'Stormy seas forecast on the economic horizon', March 10th, 25.

Daily Telegraph (2008c) 'Now the time has arrived for the real Alistair Darling to stand up', March 12th, City, 5.

Daily Telegraph (2008d) 'Chancellor at home on Planet Brown', March 13th, 7.

Daily Telegraph (2008e) 'The Chancellor Darling accused of complacency over slowdown', March 15th, 2.

Daily Telegraph (2008f) 'Is this Keynesian spending spree really the way out?', October 20th, 4.

Daily Telegraph (2008g) 'Economists attack Darling's 'misguided' spending plans', October 26th, 1.

Daily Telegraph (2008h) 'Help small firms now, say Cobra founder', November 11th, City, 1.

Daily Telegraph (2008i) 'ministers, ignore him at your peril', November 24th, City, 4.

Daily Telegraph (2008j) 'Cut tax, you know it makes sense', November 26th, 6.

Daily Telegraph (2008k) 'Banks must lend or crisis will worsen', December 19th, 8.

Daily Telegraph, (2009a) 'Kaletsky: Savers bad. Profligacy good', January 8th, 12.

Daily Telegraph (2009b) 'Reality dawns for struggling Labour', April 4th, 10.

Daily Telegraph (2009c) 'Tricky Budget', April 7th, City, 3.

Daily Telegraph (2009d) 'Darling needs to be honest and transparent', April 7th, 21.

Daily Telegraph (2009e) 'Some Budget thoughts, now over to you, Chancellor', April 20th, 5.

Daily Telegraph (2009f) 'Chancellor seeks to position Britain for life after the recession', April 20th, 4.

Daily Telegraph (2009g) 'Darling urged to cut spending by £30bn', April 20th, 4.

Daily Telegraph (2009h) 'Too much stimulus could spell disaster for Dr Darling's ailing patient', April 21st, City, 5.

Daily Telegraph (2009i) 'Markets braced for historic £200bn deficit', April 21st, City, 1.

Daily Telegraph (2009j) 'Darling's moment to face the painful truth', April 21st, 21.

Daily Telegraph (2009k) 'Darling faces twin pressures over public spending plans', April 22nd, 4.

Daily Telegraph (2009l) 'Growing, growing, gone?', April 22nd, 15.

Daily Telegraph (2009m) 'Britain deserves better than this shabby cynicism', April 23rd, 21.

Daily Telegraph (2009n) 'Borrowing puts UK's AAA rating in danger', April 24th, 5.

Daily Telegraph (2009o) 'Britain's debt will not be under control until 2032', April 24th, 1.

Daily Telegraph (2009p) 'UK economy shrinks at fastest pace', April 25th, 27.

Daily Telegraph (2009q) 'Bold first steps towards a Tory culture of thrift', April 27th, 21.

Daily Telegraph (2009r) 'Case for borrowing born out of the Great Depression', September 3rd, 5.

Daily Telegraph (2009s) 'No ordinary squeeze, but these are no ordinary times', October 7th, 4.

Daily Telegraph (2009t) 'British 'ignorant to scale of crisis', October 16th, 1.

Daily Telegraph (2009u) 'CBI urges Darling to act fast to avoid crisis', October 19th, 1.

Daily Telegraph (2009v) 'None of our politicians appears to realise what a mess we're in', October 28th, 24.

Daily Telegraph (2009w) 'You don't need a Bill to cut the budget deficit', November 19th, 29.

Daily Telegraph (2009x) 'CBI boss says act now to cut public deficit', November 20th, City, 2.

Daily Telegraph (2009y) 'Britain has run out of money, the CBI is told', November 24th, 3.

Daily Telegraph (2009z) 'A bigger state is not the route to recovery', November 24th, 23.

Daily Telegraph (2009aa) 'Only tough public sector reform will ease our debt', December 8th, 25.

Daily Telegraph (2009ab) 'Ignore the promises of cuts – the state is set up to spend', December 9th, 28.

Daily Telegraph (2009ac) 'Brown's class warriors take up back to the 1970s', December 19th, 25.

Daily Telegraph (2010a) 'We must arm ourselves for a class war', February 25th, 21.

Daily Telegraph (2010b) 'Markets fear indecision over debt could see investors flee', March 2nd, 4.

Daily Telegraph (2010c) 'The verdict in the City was swift...and damning', March 25th, 1.

Daily Telegraph (2010d) 'Audacious conceit and ruthless political calculation', March 25th, 23.

Daily Telegraph (2010e) 'Little to safeguard UK's triple-A rating', March 25th, 3.

Daily Telegraph (2010f) 'Short on policy and long on politics, the election Budget', March 25th, 4-5.

Daily Telegraph (2010g) 'Business gives the thumbs-down to Darling's Budget', March 27th, 31.

Dale, I. (2008) 'Tories must grab the business opportunity', Daily Telegraph, 14th March, 23.

Daley, J. (2008a) 'Brown can't fix our economy until he admits that he broke it', *Daily Telegraph*, March 10th, 24.

Daley, J. (2008b) 'There's only one thing to learn from Obama: tax cuts win votes', *Daily Telegraph*, November 10th, 20.

Daley, J. (2009) 'The Tories must not throw away Labour's gift', *Daily Telegraph*, *December* 13th, 28.

D'Amico, S., English, W., Lopez-Salido, D. and Nelson, E. (2012) 'The Federal Reserve's Large-Scale Asset Purchase Programs: Rationale and Effects', *Finance and Economics Discussion Series*, Washington, DC: Federal Reserve.

D'Ancona, M. (2009) 'This is not a Government, just a clunking electoral machine', *Sunday Telegraph*, December 13th, 28.

Davies, G. (2010) 'The great rift opens', Guardian, February 20th, 32.

Dejevsky, M. (2009) 'Yes we can! (Slash the budget deficit, that is)', *Independent*, November 17th, 30.

DeLong, J.B. (2012) 'Spending Cuts to Improve Confidence? No, the Arithmetic Goes the Wrong Way', VoxEU.org, Available at: http://www.voxeu.org/article/spending-cuts-improve-confidence-no-arithmetic-goes-wrong-way, accessed 08/08/2012.

DeLong, J.B. and Summers, L.H. (2012), 'Fiscal Policy in a Depressed Economy', *Brookings Papers on Economic Activity 2012*.

Denzau, A. and North, D. (1994) 'Shared Mental Models: Ideologies and Institutions', *Kyklos*, 47(1), 3-31.

Der Derian, J. (1990) 'The (S)pace of International Relations: Simulation, Surveillance, and Speed', *International Studies Quarterly*, 34(3), 295-310.

De Soto, J. (2011) 'Foreword' in D. Howden, (ed), *Institutions in Crisis: European Perspectives on the Recession*, Cheltenham: Edward Elgar, x-xxi.

Dobbin, F. (1993) 'The Social Construction of the Great Depression: Industrial Policy During the 1930s in the United States, Britain and France', *Theory and Society*, 22(1), 1-56.

Dornbusch, R. and Draghi, M. (1990) 'Introduction' in R. Dornbusch and M. Draghi (eds), *Public Debt Management: Theory and History*, Cambridge: Cambridge University Press, 1-13.

Dryzek, J. (1994) Discursive Democracy, Cambridge: Cambridge University Press.

Duncan, G. (2007) 'Will Darling opt to ditch prudence', Times, August 20th, 37.

Dunn, K. (2006) 'Examining Historical Representations', *International Studies Review*, 8(2), 370-78.

Eggertsson, G. and Krugman, P. (2012) 'Debt, Deleveraging, and the Liquidity Trap: A Fisher-Minsky-Koo Approach', *The Quarterly Journal of Economics*, 127(3), 1469-1513.

Elliott, L. (2008a) 'For the government and the consumer, the party's over', *Guardian*, March 11th, 38.

Elliott, L. (2008b) 'Saving Fannie and Freddie was nationalisation pure and simple', *Guardian*, September 9th, 32.

Elliott, L. (2008c) 'This week the crash went nuclear, and Britain will feel the worst of the fallout', *Guardian*, September 16th, 30.

Elliott, L. (2008d) 'to avoid future disaster, those who guide the economy must learn to lean against the wind', *Guardian*, October 10th, 34.

Elliott, L. (2008e) 'Beyond the plummeting markets and whingeing banks is a ray of hope – for a Green New Deal', *Guardian*, October 17th, 39.

Elliott, L. (2008f) 'We just can't afford tax cuts', Guardian, November 12th, 30.

Elliott, L. (2009a) 'Britain's credit rating is on the danger list – and so is Gordon Brown's credibility', *Guardian*, June 22nd, 22.

Elliott, L. (2009b) 'Cuts are not the only option. Taxes can be raised, too', Guardian, July 22nd, 28.

Elliott, L. (2009c) 'The dawning of an Age of Austerity: ballooning budget deficit will usher in a prolonged period of belt-tightening over the next decade', *The Guardian*, August 24th, 26.

Elliott, L. (2009d) 'It's only big government that got us out of the crisis – so why isn't Labour benefiting?', *Guardian*, October 12th, 28.

Elliott, L. (2009e) 'This zombie economy promises frights in store for us all', *Guardian*, November 6th, 31.

Elliott, L. (2009f) 'Don't despair – Brown may be on to a winner', Guardian, November 23rd, 26.

Elliott, L. (2009g) 'A deficit of patience: Cameron isn't all wrong about Britain's finances. But to slash spending now would be madness', *Guardian*, November 26th, 38.

Elliott, L. (2010a) 'Deficit dilemma: slow growth exposes opposition headache over spending cuts', *Guardian*, February 10th, 12.

Elliott, L. (2010b) 'Austerity can wait: world's economy too fragile to rein in debts, says IMF', *Guardian*, February 24th, 24.

Elster, Jon (1982) 'The Case for Methodological Individualism', *Theory and Society*, 11(4), 453-82.

Entman, R. (1993) 'Framing: Toward Clarification of a Fractured Paradigm', *Journal of Communication*, 43(1), 51-58.

Entman, R. and Herbst, S. (2001) 'Reframing Public Opinion as We Have Known It' in W. Bennett and R. Entman (eds), *Mediated Politics*, New York: Cambridge University Press, 203-25.

Evans-Pritchard, A. (2007) 'As Mr Brown will find next door, his is not quite the miracle that is proclaimed', *Daily Telegraph*, March 22nd, 8.

Evans-Pritchard, A. (2010) 'Don't go wobbly on us now, Ben Bernanke, or deflation will claim us all', *Daily Telegraph*, March 1st, 8.

Fairclough, N. (1995a) Critical Discourse Analysis, Harlow: Longman.

Fairclough, N. (1995b) Media Discourse, London: Edward Arnold.

Ferguson, T. (1984) 'From Normalcy to New Deal: Industrial Structure, Party Competition, and American Public Policy in the Great Depression', *International Organisation*, 38(1), 41-94.

Ferguson, N. (2009) 'A history lesson for economists in thrall to Keynes', FT, May 30th, 7.

Fierke, K. (2005) Diplomatic Interventions: Conflict and Change in a Globalizing World, New York: Palgrave.

Finnemore, M. (1996) *National Interest in International Society*, Ithaca, NY: Cornell University Press.

Finnemore, M. and Sikkink, K. (2001) 'Taking Stock: The Constructivist Program in International Relations and Comparative Politics', *Annual Review of Political Science*, 4, 391-416.

Fisher, I. (1933) 'The Debt-Deflation Theory of Great Depressions', Econometrica, 1(4), 337-57.

Foster, J. and Magdoff, F. (2009) *The Great Financial Crisis: Causes and Consequences*, New York: Monthly Review Press.

Fleming, S. (2009a) 'Recovery next year? It's a fantasy!', Daily Mail, April 23rd, p.?

Fleming, S. (2009b) 'Darling's dangerous game with the country's finances', December 10th, p.?

Foucault, M. (1977) Discipline and Punish, Harmondsworth: Penguin.

Freedland, J. (2009) 'In the great argument of 2010, the Tories are wrong and deserve to lose', *Guardian*, September 9th, 33.

Friedman, M. and Schwartz, A. (2008) [1963] *A Monetary History of the United States, 1867-1960*, Princeton, NJ: Princeton University Press.

Frost, D. (2008) 'More must be done to pull us back from brink', *Daily Mail*, December 1st, 50.

FT (2007a) 'Messages to expect from the chancellor', March 19th, 3.

FT (2007b) 'Tories rap 'con trick' as Brown steps out on his road to No 10', March 22nd, 1.

FT (2007c) 'From profligacy back to prudence in search of the sunny uplands', March 22nd, 2.

FT (2007d) 'Corporation tax campaign pays off', March 22nd, 7.

FT (2007e) 'Three economists give their viewpoint', March 22nd, 11.

FT (2007f) 'Brown tidies up before he takes his leave', March 22nd, 22.

FT (2007g) 'A display of virtues and vices', March 22nd, 22.

FT (2007h) 'Fiscally neutral swings and roundabouts', March 23rd, 2.

FT (2007i) 'Business relieved its voice is heard at last', March 23rd, 3.

FT (2007j) 'Corporation tax cut welcomed to help keep UK's edge as global centre for business', March 23rd, 4.

```
FT (2007k) 'UK corporate tax', March 22<sup>nd</sup>, 28.
```

FT (2007l) 'Issues in the chancellor's red boxes', June 29th, 2.

FT (2007m) 'A new chancellor and a new chance', June 29th, 15.

FT (2007n) 'Two fiscal rules', July 27th, 3.

FT (2007o) 'Fiscal rules under threat, warns IFS', October 11th, 4.

FT (2008a) 'Brown accused over scope for fiscal policy', January 29th, 2.

FT (2008b) 'Grasping the nettle', Asia Edition, February 1st, 8.

FT (2008c) 'Stuck with no room for manoeuvre', February 27th, 3.

FT (2008d) 'Budget unlikely to offer many giveaways', March 8th, 2.

FT (2008e) 'Negatives outweigh positives for Darling', March 11th, 2.

FT (2008f) 'Nationalise to save the free market', October 14th, 16.

FT (2008g) 'A prophet reborn', October 18th, 7.

FT (2008g) 'Darling's limited treatment options', October 21st, 16.

FT (2008h) 'What the British authorities should try now', October 31st, 11.

FT (2008i) 'Fiscal policy is our most potent instrument', November 10th, 9.

FT (2008k) 'Austerity must follow a stimulus', November 13th, 14.

FT (20081) 'Why the Brown critics are wrong', November 21st, 11.

FT (2008m) 'Darling's risky fiscal stimulus', November 24th, 14.

FT (2008n) 'Long trek from Pollyanna to Prudence', November 25th, 3.

FT (2008o) 'Say goodbye to New Labour', November 25th, 10.

FT (2009a) 'Credit and confidence threaten recovery', January 2nd, 3.

FT (2009b) 'Public debt levels threaten sharp tax increases and severe austerity', March 23rd, 1.

FT (2009c) 'A reality check from the governor', March 26th, 12.

FT (2009d) 'The folly of hoping for the fiscal best', April 20th, 6.

FT (2009e) 'A Budget that remakes the landscape', April 22nd, 7.

FT (2009f) 'Darling gambles on growth', April 23rd, 1.

FT (2009g) 'Concerns voiced over public finance risks', April 23rd, 6.

FT (2009h) 'Bittersweet bet on credibility', April 23rd, 11.

FT (2009i) 'Economists' view', April 23rd, 11.

FT (2009j) 'Owe, Darling', April 23rd, 28.

FT (2009k) 'Public spending cuts are viewed as 'opportunity", April 25th, 2.

FT (20091) 'Progressive austerity: an agenda to protect the poor', May 22nd, 11.

FT (2009m) 'Less a relaunch than a reheating', June 30th, 12.

FT (2009n) 'Darling walks tightrope in pre-Budget report', December 7th, 32.

FT (2010a) 'Darling must give us a reality Budget', March 23rd, 10.

FT (2010b) 'Darling ditches crisis mode for future vision', March 24th, 2.

FT (2010c) 'Darling ducks deficit challenge', March 25th, 1.

FT (2010d) 'Pain deferred until the polls close', March 25th, 3.

FT (2010e) 'Call for more details on cuts', March 25th, 11.

FT (2010f) 'Economists unconvinced by degree of optimism', March 25th, 12.

FT (2010g) 'Debtors prism', March 25th, 14.

FT (2010h) 'Fine, if you forget the public finances', March 25th, 20.

Gamble, A. (2013) 'Neo-Liberalism and Fiscal Conservatism' in V. Schmidt and M. Thatcher (eds), *Resilient Liberalism in Europe's Political Economy*, Cambridge: Cambridge University Press, 53-76.

Gamson, W. and Modigliani, A. (1989) 'Media Discourse and Public Opinion on Nuclear Power: A Constructionist Approach', *American Journal of Sociology*, 95(1), 1-37.

George, J. and Campbell, D. (1990) 'Critical Social Theory and International Relations', *International Studies Quarterly*, 34(3), 269-93.

Giavazzi, F. and Pagano, M. (1990) 'Can Severe Fiscal Contractions Be Expansionary? Tales of Two Small European Countries', *NBER Macroeconomics Annual*, 5(1), 75-122.

Giddens, A. (1979) Central Problems in Social Theory, London: Macmillan.

Giddens, A. (1984) The Constitution of Society, Cambridge: Polity.

Gieve, J. (2009) 'Joined-up recovery depends on an agreed framework', FT, December 9th, 15.

Gieve, J. and Provost, C. (2012) 'Ideas and Coordination in Policymaking: The Financial Crisis of 2007-2009', *Governance*, 25(1), 61-77.

Giles, C. (2008) 'Is Brown to blame?', FT, August 9th, 9.

Gill, S. (1995) 'Theorizing the Interregnum: the Double Movement of Global Politics in the 1990s' in B. Hettne (ed), *International Political Economy: Understanding Global Disorder*, Halifax: Fernwood, 65-99.

Gill, S. (2003) Power and Resistance in the New World Order, Basingstoke: Palgrave MacMillan.

Giménez-Roche, G. (2011) 'Institutional Illusion and Financial Entrepreneurship in the European Debt Scheme' in D. Howden, (2011) *Institutions in Crisis: European Perspectives on the Recession*, Cheltenham: Edward Elgar, 1-21.

Glarbo, K. (1999) 'Wide-awake Diplomacy: Reconstructing the Common Foreign and Security Policy of the European Union', *Journal of European Public Policy*, 6(4), 634-51.

Glover, S. (2007) 'Despite his carping, Brown has been a great Chancellor', *Daily Mail*, March 22nd, 14.

Goldstein, J. and Keohane, R. (1993) (eds), *Ideas and Foreign Policy: Beliefs, Institutions and Political Change*, Ithaca, NY: Cornell University Press.

Green, D. and Shapiro, I. (1994) *Pathologies of Rational Choice* Theory, New Haven, CT: Yale University Press.

Gros, D. (2011) 'Can Austerity Be Self-Defeating?', VoxEU.org, Available at: http://www.voxeu.org/article/can-austerity-be-self-defeating, accessed 08/08/2012.

Guardian (2007a) 'Brown forced to defend record as thinktank finds £40bn tax rise', January 31st, 21.

Guardian (2007b) 'Brown bucks the Labour trend...and gets out in time', March 20th, 31.

Guardian (2007c) 'Lucky chancellor has beaten Labour's economic voodoo', March 21st, 28.

Guardian (2007d) 'Most people will be unaffected – but Brown is accused of allowing Britain to slide further and further into the red', March 22nd, 10.

Guardian (2007e) 'Corporation tax rise is 'another slap in the face', March 22nd, Newsprint Supplement, 12.

Guardian (2007f) 'How red is Brown?', March 23rd, 40.

Guardian (2007g) 'Darling turns from floss to prudence', October 8th, 30.

Guardian (2007h) 'Something borrowed, something blue', October 10th, 26.

Guardian (2007i) 'Darling faces growing black hole, says IFS', October 11th, 26.

Guardian (2008a) 'The budget: hard times', March 11th, 40.

Guardian (2008b) 'Recession fear cuts Darling's budget options', March 12th, 1.

Guardian (2008c) 'Darling: the hesitant debutant', March 13th, 1.

Guardian (2008d) 'Plans to cut red tape and improve access to finance and public sector contracts get lukewarm reception', March 13th.

Guardian (2008e) 'Recession fear cuts Darling's budget options', March 12th, 1.

Guardian (2008f) 'End of the swaggering City...and New Labour's economics', October 9th, 8.

Guardian (2008g) 'Mandelson under fire over costcutting plan to drop flexible working reforms', October 21st, 6.

Guardian (2008h) 'The effect of tax cuts', November 11th, 4.

Guardian (2008i) 'Brown claims Tories are alone in opposing tax cuts', November 18th, 14.

Guardian (2008j) 'Battle begins: The politics of the recession', November 19th, 38.

Guardian (2008k) 'The government is abandoning its fiscal rules, but whether this pays off is beyond its control', November 25th, 4.

Guardian (20081) 'Crisis economics', November 25th, 30.

Guardian (2008m) 'Rethink suggests economy will shrink by 1% in 2009, producing first year of recession since early 1990s', November 25th, Newsprint Supplement, 5.

Guardian (2009a) 'Chancellor's annual deficit forecast likely to be spectacularly shattered', March 26th, 6.

Guardian (2009b) 'Brown spending retreat as City sounds warning', March 26th, 1.

Guardian (2009c) 'Budget economy: cost of a crisis', April 23rd, 30.

Guardian (2009d) 'Rise of the State under New Labour is over', April 24th, 12.

Guardian (2009e) 'Public services: The rise and fall of the state', April 24th, 38.

Guardian, (2009f) 'Liberal Democrats: progressive austerity', July 23rd, 30.

Guardian (2009g) 'Osborne lifts veil on his age of austerity', October 7th, 1.

Guardian (2009h) 'Guardian round table: Are those shoots any greener?', October 23rd, 38.

Guardian (2009i) 'No exit: Economic life support must stay – Darling', November 7th, 39.

Guardian (2009j) 'Public finances: broke but not bust yet', November 20th, 44.

Guardian (2009k) 'The shadow of 1937', December 2nd, 30.

Guardian (2010a) 'Government debt: Judging the judges', January 4th, 28.

Guardian (2010b) 'Lib Dems back away from Tories with attack on Cameron's deficit-reduction proposals', January 11th, 4.

Guardian (2010c) 'Osborne identifies cuts for first weeks of Tory government', January 15th, 12.

Guardian (2010d) 'Tentative recovery comes with pitfalls for both main parties', January 28th, 10.

Guardian (2010e) 'Tories retreat on pledges to tear up Labour spending plans after election', February 2^{nd} , 4.

Guardian (2010f) 'Public spending: battle of the budgets', February 8th, 32.

Guardian (2010g) 'Top economists attack Labour plans to tackle budget deficit', February 15th, 6.

Guardian (2010h) 'Branson lifts Tories with call for immediate public spending cuts', February 17th, 15.

Guardian (2010i) 'Beware calls for hasty cuts', February 18th, 30.

Guardian (2010k) 'The cautious versus the cutters', February 20th, 34.

Guardian (2010l) 'Economy: Don't get your hopes up', February 27th, 36.

Guardian (2010m) 'The Budget: how to make a difference', March 17th, 30.

Guardian (2010n) 'Borrowing figures raise hopes of Budget giveaway – despite Darling denials', March 19th, 28.

Guardian (2010o) 'The Budget: What Wednesday must bring', March 20th, 32.

Guardian (2010p) 'Take on the City with a 'people's Budget', March 22nd, 28.

Guardian (2010q) 'Political but not profligate', March 25th, 34.

Guardian (2010r) 'Britain's AAA rating gets stay of execution', March 26th, 37.

Guardian (2010s) 'Economic debate: A night of rattling good fun at the undertakers reunion', March 30th, 4.

Guardian (2010t) 'And they're off...Tory plan to cut NI 'a terrible risk", March 30th, 1.

Guardian (2010u) 'Proposal is 'clearly irresponsible", March 30th, 6.

Guardian (2010v) 'Thinktank's verdict: risk to economic recovery', March 30th, 6.

Guardian (2010w) 'Conservatives: deficiencies of efficiencies', April 9th, 36.

Guardian (2010x) 'Don't make spending cuts yet, warns IMF', April 22nd, 32.

Gourevitch, P. (1984) 'Breaking with Orthodoxy: The Politics of Economic Policy Responses to the Depression of the 1930's', *International Organisation*, 38(1), 95-129.

Gourevitch, P. (1986) Politics in Hard Times, Ithaca, NY: Cornell University Press.

Guthrie, J. (2007) 'The chancellor giveth and taketh away', FT, March 22nd, 23.

Guthrie, J. (2008) 'How Darling can revive ailing businesses', FT, November 20th, 17.

Guzzini, S. (2005) 'The Concept of Power: A Constructivist Analysis', Millenium, 33(3), 495-521.

Haas, P. (1992) 'Introduction: Epistemic Communities and International Policy Coordination', *International Organization*, 46(1), 1-35.

Hacking, I. (1999) The Social Construction of What?, Cambridge, MA: Harvard University Press.

Hall, P. (1986) Governing the Economy: The Politics of State Intervention in Britain and France, Cambridge: Polity.

Hall, P. (1993) 'Policy Paradigms, Social Learning, and the State: The Case of Economic Policymaking in Britain', *Comparative Politics*, 25 (3), 275–96.

Hall, P. (2013) 'Brother, Can You Paradigm?', Governance, 26(2), 189-92.

Hall, P. and Taylor, R. (1996) 'Political Science and the three New Institutionalisms', *Political Studies*, 44(4), 936-57.

Halligan, L. (2007a) 'Brown Delays the Pain', Daily Telegraph, March 19th, City, 6.

Halligan, L. (2007b) 'Who will Prime Minister Brown blame when it all goes wrong? The Chancellor, of course', *Sunday Telegraph*, March 25th, 22.

Halligan, L. (2007c) 'The full horror of the Chancellor's debt binge', *Sunday Telegraph*, City, April 1st, 4.

Halligan, L. (2007d) 'Brown mortgaged our future in a vain bid to secure his own', *Daily Telegraph*, October 14th, City, 4.

Halligan, L. (2008a) 'Tories need to get real on the economy', *Sunday Telegraph*, September 28th, 7.

Halligan, L. (2008b) 'Brown's UK borrowing binge', Sunday Telegraph, November 30th, 4.

Halligan, L. (2009a) 'Time to examine tough truths rather than talk green shoots', *Sunday Telegraph*, June 7th, 4.

Halligan, L. (2009b) 'It's time to end the grotesque fiscal bail-outs and grapple with reality', *Sunday Telegraph*, June 21st, 4.

Halligan, L. (2009c) 'If the Government fails to make cuts now it will be racked with gilts', *Sunday Telegraph*, July 19th, 4.

Halligan, L. (2009d) 'Huge gilts row barely registers as the UK sleepwalks into stagnation', *Sunday Telegraph*, July 26th, 4.

Halligan, L. (2009e) 'Stop using Keynes' name to justify wild fiscal experiments', *Sunday Telegraph*, September 6th, 4.

Halligan, L. (2009f) 'This isn't even the end of the beginning for our economic woes', *Sunday Telegraph*, November 8th, 4.

Halligan, L. (2009g) 'Time for the Tories to skewer this wild belief in Keynesian nonsense', $Sunday\ Telegraph$, December 6^{th} , 4.

Halligan, L. (2009h) 'Labour's dishonesty is leading us down the road to sovereign default', *Sunday Telegraph*, December 13th, 6.

Halligan, L. (2010a) 'Darling's box of tricks could have been worse, but we're still in trouble' *Sunday Telegraph*, March 28th, 4.

Hari, J. (2009) 'Britain's not bust. So don't use it as an excuse to impose cuts', *Independent*, October 8th, 31.

Hassdorf, W. (2005) 'Emperor without Clothes: Financial Market Sentiment and the Limits of British Currency Machismo in the ERM Crisis', *Millenium*, 33(3), 691-722.

Hastings, M. (2008) 'A year ago I thought him doomed. But yesterday Mr Cameron showed he's a PM in waiting', *Daily Mail*, October 2nd, 14.

Hastings, M. (2009) 'Mr Brown isn't a bad man. But he's cursed by a profound belief that he, rather than we citizens, knows how best to spend our money', *Daily Mail*, September 30th, p.?

Hay, C. (1999) 'Crisis and the Structural Transformation of the State: Interrogating the Process of Change', *British Journal of Politics and International Relations*, 1(3), 317-44.

Hay, C. (2001) 'The "Crisis" of Keynesianism and the Rise of Neo-Liberalism in Britain: An Ideational Institutionalist Perspective', in J. Campbell and O. Pedersen (eds), *The Second Movement in Institutional Analysis*, Princeton, NJ: Princeton University Press.

Hay, C. (2002) Political analysis: A Critical Introduction, Basingstoke: Palgrave.

Hay, C. and Rosamond, B. (2002) 'Globalization, European Integration and the Discursive Construction of Economic Imperatives', *Journal of European Public Policy*, 9(2), 147-67.

Hay, C. (2004) 'The Normalizing Role of Rationalist Assumptions in the Institutional Embedding of Neoliberalism', *Economy and Society*, 33(4), 500-27.

Hay, C. (2005) 'Making Hay...or Clutching at Ontological Straws? Notes on Realism, 'As-If-Realism' and Actualism', *Politics*, 25(1), 39-45.

Hay, C. (2006) 'Constructivist Institutionalism' in R. Rhodes, S. Binder and B. Rockman (eds) *The Oxford Handbook of Political Institutions*, Oxford: Oxford University Press, 56-74.

Hay, C. (2010) 'Chronicles of a Death Foretold: the Winter of Discontent and Construction of the Crisis of British Keynesianism', *Parliamentary Affairs* 63(3), 446-70.

Hay, C. (2011) 'Ideas and the Construction of Interests' in D. Béland and R. Cox (eds), *Ideas in Social Science Research*, Oxford: Oxford University Press, 65-82.

Hay, C. (2013) 'Treating the Sympton not the Condition: Crisis Definition, Deficit Reduction and the Search for a New British Growth Model', *British Journal of Politics and International Relations*, 15(1), 23-37.

Hay, C. (2014) 'A Crisis of Politics in the Politics of Crisis' in D. Richards, M. Smith and C. Hay (eds), *Institutional Crisis in 21st-Century Britain*, Basingstoke: Palgrave Macmillan, 60-78.

Hay, C. and Wincott, D. (1998) 'Structure, Agency and Historical Institutionalism', *Political Studies*, 46(5), 951-57.

Hayward, J. (1986) 'The Political Science of Muddling through: The De Facto Paradigm?' in J.

Hayward and P. Norton (eds), *The Political Science of British Politics*, Brighton: Wheatsheaf Books, 145-67.

Heclo, H. (1974) *Modern Social Politics in Britain and Sweden*, New Haven, CT: Yale University Press.

Heffer, S. (2007) 'Headline-grabbing Budget that leaves the Tories floundering', *Daily Telegraph*, March 22nd, 22.

Heffer, S. (2009) 'It was a savage and pointless attack on middle England', *Daily Telegraph*, April 23rd, 20.

Hodson, D. and Mabbett, D. (2009) 'UK Economic Policy and the Global Financial Crisis: Paradigm Lost?', *Journal of Common Market Studies*, 47(5), 1041-61.

Hoffman, D. (1991) 'Restructuring, Reconstruction, Reinscription, Rearticulation: Four Voices in Critical International Theory', *Millenium*, 20(2), 169-85.

Holland, D., Barrell, R., Fic, T., Hurst, I., Liadze, I., Orazgani, A. and Whitworth, R. (2010) 'Fiscal Multipliers to Assess Consolidation Plans', *National Institute Economic Review*, 213, F9-12.

Hopf, T. (1998) 'The Promise of Constructivism in International Relations Theory', *International Security*, 23(1), 171-200.

Howson, S. (1975) *Domestic Monetary Management in Britain, 1919-38*, Cambridge University Press.

Howson, S. (1980) Sterling's Management Float: The Operations of the Exchange Equalisation Account, 1932-39, Princeton, NJ: Princeton University Press.

Hutton, W. (2008a) 'Had I been at last night's crunch No 10 meeting, I'd have made the case for these urgent steps', *Guardian*, October 8th, 31.

Hutton, W. (2008b) 'The fallacy of the fix', Guardian, November 19th, 36.

Hutton, W. (2009a) 'Darling did well in a very tight spot. To obsess about reducing the deficit would backfire', *Guardian*, April 23rd, 28.

Hutton, W. (2009b) 'Hail the man who argues that Britain should stop worrying about its debt', *Observer*, July 5th, 24.

Hutton, W. (2009c) 'At last, Brown is getting it right. His tragedy is that no one can see it', *Observer*, July 26th, 26.

Hutton, W. (2009d) 'Sorry, David, if you roll back the state, you invite disaster', *Observer*, October 11th, 27.

Hutton, W. (2010a) 'The Tories have the answers, but not the strength to deliver', *Observer*, February 7th, 33.

Hutton, W. (2010b) 'To thrive we need to distinguish between morality and economics', *Observer*, February 21st, 38.

Hutton, W. (2010c) 'The march to sanity begins', Guardian, March 25th, 33.

IFS (2001) 'The Government's Fiscal Rules', *Briefing Note No. 16*, available at: http://www.ifs.org.uk/bns/bn16.pdf, accessed 16/10/2012.

IMF (2009) 'The Case for Global Fiscal Stimulus' Staff Position Note SPN/09/03, March 6th.

Independent (2007a) 'A Chancellor whose record has divided the economists, March 22nd, 22.

Independent (2007b) 'Gordon Brown goes out with a surprise – just as he began his time at the Treasury', March 22^{nd} , 34.

Independent (2007c) 'Echoes of Nigel Lawson in Brown's company tax reform', March 23rd, 58.

Independent (2007d) 'the Treasury is in a tight spot – and this time it can't rely on Keynes', October 10th, 90.

Independent (2007e) 'It's time for Cameron to challenge Labour orthodoxy', October 10th, 92.

Independent (2008a) 'Market convulsions will lead to the return of the state as a major economic force', January 28th, 40.

Independent (2008b) 'A different Chancellor, but the same mantra', March 8th, 10.

Independent (2008c) 'A lacklustre Budget, in the shadow of Mr Brown', March 13th, 38.

Independent (2008d) 'Budget will worsen the slowdown, say business leaders', March 14th, 6.

Independent (2008e) 'Britain bears the full force of the economic storm', August 23rd, 32.

Independent (2008f) 'I know economic times are bad, Mr Darling, but are you sure that they are the worst for 60 years?', August 31st, 6.

Independent (2008g) 'higher borrowing is the least bad of all the options', September 23rd, 36.

Independent (2008h) 'The Government needs to get its priorities straight', October 21st, 28.

Independent (2008i) 'Mr Brown should beware of sounding too triumphant', October 28th, 28.

Independent (2008j) 'Economy is the shadow that looms over the 44th Presidency', November 6th, 24.

Independent (2008k) 'Tax cuts bounce back on to the political agenda', November 11th, 28.

Independent (2008l) 'A gamble that will decide Britain's political future', November 25th, 30.

Independent (2008m) 'A nation in danger of drowning in a sea of debt', November 29th, 40.

Independent (2008n) 'An economic morality lesson in a time of deep uncertainty', December 20th, 32.

Independent (2009a) 'Balancing the books', April 16th, 42.

Independent (2009b) 'MPC's Blanchflower demands 'Budget for jobs', April 18th, 44.

Independent (2009c) 'Mr Darling must make the best of very limited options', April 22nd, 26.

Independent (2009d) 'So, what would you be doing if you were the Chancellor? Four leading economists outline their prescription to aid Britain's recovery in today's Budget', April 22nd, 6.

Independent (2009e) 'Gilts slide as Darling's forecasts prompt doubt', April 24th, 40.

Independent (2009f) 'Britain's new age of austerity', April 24th, 1.

Independent (2009g) 'The moment that finally cost Labour the general election', April 25th, 12.

Independent (2009h) 'Unrepentant Brown defends Budget as a 'plan for recovery', April 25th, 4.

Independent (2009i) 'The stark questions posed by a desperate Budget', April 25th, 36.

Independent (2009i) 'A timely warning on our finances', May 22nd, 36.

Independent (2009k) 'The economic mess began when profligacy replaced prudence', June 1st, 36.

Independent (2009l) 'Bank warns of credit supply threat to economic fightback', June 19th, 42.

Independent (2009m) 'David Cameron thinks that Canada can show us how to slash public debt. Is he right?', July 11th, 14.

Independent (2009n) 'As the global economy pivots, we face years of looming austerity', August 17th, 32.

Independent (2009o) 'UK's 'truly horrible' public finances threaten recovery', August 21st, 42.

Independent (2009p) 'The end of the phoney war on public expenditure', August 25th, 22.

Independent (2009q) 'Our debt hangover is stopping us joining the global growth party', September 4th, 40.

Independent (2009r) 'The phoney political war on spending is now over', September 16th, 30.

Independent (2009s) 'The glaring hole in this pre-Budget report', December 10th, 38.

Independent (2009t) 'Figures show Government's previous 'efficiency' drives drove up staff and budgets', December 13th, 22.

Independent (2010a) 'The political battle over the unkindest cuts', February 2nd, 24.

Independent (2010b) 'The argument of the fiscal doves is more compelling', February 20th, 34.

Independent (2010c) 'The Tory-Labour divide is a political, not economic, one', March 2nd, 40.

Independent (2010d) 'Tories' economist criticises party's plan for cuts', March 8th, 4.

Independent (2010e) 'The Janus face of recession politics', March 11th, 36.

Independent (2010f) 'This was a sparse Budget with a big political message', March 25th, 35.

Independent (2010g) 'The truth about spending cuts', March 29th, 38.

Independent (2010h) 'Tories' pledge on cuts criticised', March 30th, 8.

Independent (2010i) 'Our political leaders all need to face up to the reality of meagre growth', April 24th, 42.

Independent (2010j) 'This deafening political silence on cuts must end', April 25th, 30.

Independent (2010k) 'Forget the sterile debates, we are all facing a future of austerity and sacrifice', May 3rd, 32.

Irvin, G., Reed, H. and Gannon, Z. (2010) *The £100Billion Gamble on Growth Without a State*, London: Compass.

Islam, I. and Chowdhury A. (2010) 'Fiscal Consolidation, Growth and Employment: What do we Know? VoxEU.org, Available at: http://www.voxeu.org/debates/commentaries/fiscal-consolidation-growth-and-employment-what-do-we-know, accessed 08/08/2012.

Jacobs, L. and Shapiro, R. (2000) *Politicians don't Pander: Political Manipulation and the Loss of Democratic Responsiveness*, Chicago: University of Chicago Press.

Jacoby, W. (2000)'Issue Framing and Public Opinion on Government Spending', *American Journal of Political Science*, 44(4), 750-67.

Jayadev, A. and Konczal, M. (2010) 'When is austerity Right? In Boom, Not Bust', *Challenge*, 53(6), 37-53.

Jenkins, S. (2009) 'Any fool can raise a tax. But it takes a gutless one to splurge it on this stuff', *Guardian*, April 24th, 35.

Jessop, B. (1996) 'Interpretative Sociology and the Dialectic of Structure and Agency', *Theory, Culture and Society*, 13(1), 119-28.

Johnson, S. and Kwak, J. (2010) *13 Bankers: The Wall Street Takeover and the Next Financial Meltdown*, New York: Pantheon.

Jordà, Ò. And Taylor, A. (2013) 'The Time for Austerity: Estimating the Average Treatment Effect of Fiscal Policy', *NBER Working Paper Series*, No.19414, Cambridge, MA: NBER.

Kaletsky, A. (2007a) 'A fitting climax, but...', Times, March 22nd, 1.

Kaletsky, A. (2007b) 'The Budget was timid but not a con', Times, March 26th, 39.

Kaletsky, A. (2007c) 'Four challenges facing the new Chancellor', *Times*, July 2nd, 43.

Kaletsky, A. (2007d) 'The beginning of the end for new Labour', *Times*, October 11th, 17.

Kaletsky, A. (2008a) 'Fingers crossed, Darling', *Times*, March 13th, 1.

Kaletsky, A. (2008b) 'The Treasury needs a new golden rule fast', *Times*, July 24th, 26.

Kaletsky, A. (2008c) 'Britain in no danger of 'sick man of Europe' relapse', Times, August 25th, 35.

Kaletsky, A. (2008d) 'Crisis, what crisis? Darling's just got his figures wrong', *Times*, September 1st, 13.

Kaletsky, A. (2008e) 'We are all capitalists now? Not any longer', *Times*, September 12th, 32.

Kaletsky, A. (2008f) 'If this fails, it will take down all Britain's banks', Times, September 18th, 28.

Kaletsky, A. (2008g) 'Darling's innovations may stave off depression', Times, October 9th, 26.

Kaletsky, A. (2008h) 'A week of stupid politics but good economics', *Times*, November 27th, 36.

Kaletsky, A. (2009a) 'Carry on borrowing – and hope for the best', *Times*, April 16th, 24.

Kaletsky, A. (2009b) 'Bad economics, timing and politics: but a good impression of Mr Bean', *Times*, April 23rd, 1.

Kaletsky, A. (2009c) 'Bullets meant for bankers could kill the welfare state', Times, April 27th, 37.

Kaletsky, A. (2009d) 'Now for the longest assisted suicide in history', Times, April 30th, 32.

Kaletsky, A. (2009e) 'To put things right, you must know what went wrong', *Times*, June 15th, 37.

Kaletsky, A. (2009f) 'Don't worry about rate rises, fear stagflation', *Times*, July 16th, 18.

Kaletsky, A. (2009g) 'Headline-chasing is harming Tory credibility', Times, July 30th, 18.

Kaletsky, A. (2009h) 'Oh no Darling: you've failed', Times, December 10th, 1,3.

Kaletsky, A. (2009i) 'We simply cannot afford a live now, pay later policy', *Times*, December 14th, 37.

Kaletsky, A. (2010a) 'It's not the economy – and voters aren't stupid', *Times*, January 7th, 24.

Kaletsky, A. (2010b) 'Britain can relax on its bed of nitroglycerine', *Times*, January 28th, 30.

Kaletsky, A. (2010c) 'Whoever you vote for, painful cuts will come', *Times*, February 18th, 28.

Kaletsky, A. (2010d) 'Stand by, and watch 1992 happen all over again', *Times*, March 18th, 21.

Kaletsky, A. (2010e) 'Deficits, rates and the savage circle', *The Times*, March 22nd, 43.

Keck, M. and Sikkink, K. (1999) *Activists Beyond Borders: Advocacy Networks in International Politics*, Ithaca, NY: Cornell University Press.

Keegan, W. (2008a) 'Lord, make us virtuous – but not now. We can't stop borrowing yet', *Observer*, 16th March, 10.

Keegan, W. (2008b) 'We shouldn't have started from here, but we might avoid the worst', *Observer*, November 30th, 8.

Keegan, W. (2009a) 'Who do you think is right, Osborne or Keynes? No prizes', *Observer*, September 20th, 10.

Keegan, W. (2009b) 'Labour saved capitalism. Would the Tories have done so well?', *Observer*, October 4th, 6.

Keegan, W. (2009c) 'Inflated optimism and a deficit of clear thinking', *Observer*, November 22nd, 6.

Keegan, W. (2009d) 'Darling shows how to be a good chancellor for bad times', *Observer*, December 13th, 6.

Keegan, W. (2010a) 'Hysteria or fetishism? There's certainly no deficit of gloom', *Observer*, February 14th, 6.

Keegan, W. (2010b) 'Political, clearly. Responsible, very. Shameless? Certainly not', *Observer*, March 28th, 41.

Keohane, R. (1989) International Organisations and State Power, Boulder, CO: Westview Press.

Kettle, M. (2010) 'Darling's budget could point the way for Britain's renewal', *Guardian*, March 19th, 35.

Keynes, J. (1936) The General Theory of Employment, Interest and Money, London: Macmillan.

Killick, R. (2009) 'Timid Britain must look to its risk-takers; The 'age of austerity' is a stupid idea', *Times*, May 13th, 26.

King, G., Keohane, R. and Verba, S. (1994) *Designing Social Inquiry: Scientific Inference in Qualitative Research*, Princeton, NJ: Princeton University Press.

King, S. (2008a) 'Lessons from the Great Depression of the 1930s have not been learnt', *Independent*, October 13th, 36.

King, L., Kitson, M., Konzelmann, S. and Wilkinson, F. (2012) 'Making the Same Mistake Again – or is This Time Different', *Cambridge Journal of Economics*, 36(1), 1-15.

Kingdon, J. (1984) Agendas, Alternatives, and Public Policies, Boston: Little Brown.

Kinsella, S. (2012) 'Is Ireland Really the Role Model for Austerity?', *Cambridge Journal of Economics*, 36(1), 223-35.

Klotz, A. (1995) *Norms in International Relations: The Struggle Against Apartheid*, Ithaca, NY: Cornell University Press.

Krebs, R. and Jackson, P. (2007) 'Twisting Tongues and Twisting Arms: The Power of Political Rhetoric', *European Journal of International Relations*, 13(1), 35-66.

Krehbiel, K. (1988) 'Spatial Models of Legislative Choice', *Legislative Studies Quarterly*, 13(3), 259-319.

Krugman, P. (2009) The Return of Depression Economics and the Crisis of 2008, London: Penguin.

Krugman, P. (2012) End This Depression Now, New York: W.W. Norton and Company.

Kuehn, D. (2012) 'A Note on America's 1920-21 Depression as an Argument for Austerity', *Cambridge Journal of Economics*, 36(1), 155-60.

Kurth, J. (1979) 'The Political Consequences of the Product Cycle: Industrial History and Political Outcomes', *International Organisation*, 33(1), 1-34.

Lambert, R. (2010) 'Give us stability, then let business do the rest', *Sunday Telegraph*, February 7th, 7.

Lamont, N. (2007) 'Cameron please note: Brown has made the case for tax cuts', *Daily Telegraph*, March 23rd, 26.

Lawson, D. (2008) 'Keynes is not enough, Mr Darling', *Independent*, October 21st, 28.

Lawson, T. (2009) 'The Current economic Crisis: Its Nature and the Course of Academic Economics', *Cambridge Journal of Economics*, 33(4), 759-77.

Leuchtenburg, W. (1963) Franklin D. Roosevelt and the New Deal: 1932-1940, London: Harper and Row.

Levi, M. (1997) 'A Model, a Method, a Map: Rational Choice in Comparative and Historical Analysis' in M. Lichbach and A. Zuckerman (eds), *Comparative Politics: Rationality, Culture, and Structure*, Cambridge: Cambridge University Press, 19-41.

Lodge, M. and Hood, C. (2012) 'Into an Age of Multiple Austerities? Public Management and Public Service Bargains across OECD Countries', *Governance*, 25(1), 79-101.

Lyman, R. (1965) 'The British Labour Party: The Conflict between Socialist Ideals and Practical Politics between the Wars', *Journal of British Studies*, 5(1), 140-52.

McAnulla, S. (2005) 'Making Hay with Actualism? The Need for a Realist Concept of Structure', *Politics*, 25(1), 31-38.

McCarty, N. (2012) 'The Politics of the Pop: The U.S. Response to the Financial Crisis and the Great Recession' in N. Bermeo and J. Pontusson (eds), *Coping with Crisis: Government Reactions to the Great Recession*, New York, NY: Russell Sage Foundation, 201-32.

McKibbin, R. (1975) 'The Economic Policy of the Second Labour Government 1929-1931', *Past and Present*, 68 (1), 95-123.

McRae, H. (2007a) 'We're not prepared for the downturn', *Independent*, March 7th, 32.

McRae, H. (2007b) 'Gordon Brown has achieved much of what he set out to do. So why is he not more popular?', *Independent*, March 21st, 40.

McRae, H. (2007c) 'A Budget that fails to confront the difficult days that lie ahead', *Independent*, March 22nd, 2.

McRae, H. (2007d) 'I've changed my mind: the Bank must lower interest rates if we are to avoid recession', *Independent*, December 5th, 28.

McRae, H. (2007e) '2008: A recession in Britain, and oil at \$175 a barrel. Or is that too outrageous?', *Independent*, December 20th, 44.

McRae, H. (2008a) 'In the face of a downturn, Darling will only make matters worse if he pushes up taxes', *Independent*, March 12th, 36.

McRae, H. (2008b) 'Let the financial recovery begin', *Independent*, September 20th, 40.

McRae, H. (2008c) 'Amid all the panic, there is some good news', *Independent*, October 8th, 24.

McRae, H. (2008d) 'It's bad, may get worse, but it's no Great Depression', *Independent*, October 9th, 8.

McRae, H. (2008e) 'There are things only the state can do', Independent, October 15th, 32.

McRae, H. (2009a) 'The grim shape of budgets to come', *Independent*, April 15th, 26.

McRae, H. (2009b) 'We must seize the opportunity to radically rethink our government', *Independent*, June 7th, 84.

McRae, H. (2009c) 'Britain will not default despite the dreadful state of public finances', *Independent*, August 21st, 46.

McRae, H. (2009d) 'Cuts hurt – but they can offer us an opportunity to do things better too', *Independent*, September 20th, 80.

McRae, H. (2009e) 'In the end it's the deficit that counts', *Independent*, December 9th, 32.

McRae, H. (2009f) 'It's time we had some rules – and stuck to them', *Independent*, December 10th, 37.

McRae, H. (2009g) 'Fiscal credibility is on a knife edge – and this spring could be nasty indeed', *Independent*, December 13th, 86.

McRae, H. (2009h) 'We have a lot to learn from Ireland', *Independent*, December 16th, 28.

McRae, H. (2010a) 'It's the economy, Mr Cameron. It can't suddenly be 'transformed', Independent on Sunday, May 30th, 82.

McRae, H. (2010b) 'Recovery was never going to be easy', *Independent*, February 24th, 32.

McRae, H. (2010c) 'A double-dip, then recovery. But what will drive our future growth?', *Independent*, February 26th.

McRae, H. (2010d) 'Do few things but do them better', *Independent*, March 17th, 32.

McRae, H. (2010e) 'Pity the man who must inherit the red box', *Independent*, April 8th, 26.

McRae, H. (2010f) 'The election will not ultimately determine the route of Britain's recovery', April 23rd, 48.

McRae, H. (2010g) 'It will be a bumpy ride but, politics apart, we are headed for a recovery', *Independent on Sunday*, May 9th, 84.

Magnus, G. (2008) 'Five ways to start the world economic recovery', FT, December 19th, 9.

Mahoney, J. (2000) 'Path Dependence in Historical Sociology', Theory and Society, 29(4), 507-48.

Mail on Sunday (2007) 'Chancellor will be 'Cruelly Exposed' by weak tax revenues and big spending pledges', October 14th, 5.

Mail on Sunday (2008) 'Budget help misses out real reforms', November 30th, 73.

Malament, B. (1978) 'British Labour and Roosevelt's New Deal: The Response of the Left and the Unions', *Journal of British Studies*, 17(2), 136-67.

March, J. and Olsen, J. (1998) 'The Institutional Dynamics of International Political Orders', *International Organization*, 52(4), 943-69.

Marcussen, M. (2006) 'Institutional Transformation? The Scientization of Central Banking as a Case Study' in T. Christensen and P. Lægreid (eds), *Autonomy and Regulation: Coping with Agencies in the Modern State*, Cheltenham: Edward Elgar, 81-109.

Marcussen, M. (2009) "Scientization' of Central Banking: The Politics of A-Politicization' in M.

Marcussen and K. Dyson (eds), *Central Banks in the Age of Euro Europeanization, Convergence, and Power*, Oxford: Oxford University Press, 373-90.

Marsh, D. (2009) 'Keeping Ideas in their Place: In Praise of Thin Constructivism', *Australian Journal of Political Science*, 44(4), 679-96.

Martin, I. (2009) 'New Labour is dead – now it's up to Cameron to revive Britain', *Daily Telegraph*, April 24th, 26.

Matlary, J. (1997) 'Epilogue: New Bottles for New Wine' in K. Jorgensen (ed), *Reflective Approaches to European Governance*, New York: St. Martin's Press, 201-13.

Mattick, P. (2011) Business as Usual: the Economic Crisis and the Failure of Capitalism, London: Reaktion Books.

Middleton, R. (1985) *Towards the Managed Economy: Keynes, the Treasury and the Fiscal Policy Debate of the 1930s*, London: Methuen.

Milburn, A. (2009) 'More state is not the answer', *Independent*, May 7th, 28.

Milne, S. (2009a) 'Ignore the Tories. You can't cut your way out of a slump', *Guardian*, April 9th, 33.

Milne, S. (2009b) 'The cuts debate is a brilliant diversion from the real crisis', *Guardian*, September 17th, 29.

Milne, S. (2010) 'Only public intervention can rebuild a wrecked economy', *Guardian*, March 25th, 33.

Minsky, H. (1977) 'The Financial Instability Hypothesis: An Interpretation of Keynes and an Alternative to "Standard" Theory', *Challenge*, 20(1), 20-27.

Minsky, H. (1982) Can it Happen Again? Essays in Instability and Finance, New York: M.E. Sharpe.

Minsky, H. (1993) 'Schumpeter and Finance' in S. Biasco, A. Roncaglia and M. Salvati (eds), *Market and Institutions in Economic Development*, Basingstoke: St. Martin's Press, 103-15.

Mitchell, A. (2008) 'Enough Micawberism, we need a Keynesian cure', *Independent*, March 10th, 36.

Moravcsik, A. (1998) The Choice for Europe, Ithaca, NY: Cornell University Press.

Morrow, J. (1994) Game Theory for Political Scientists, Princeton, NJ: Princeton University Press.

Moschella, M. and Tsingou, E. (2013) 'Introduction: the financial crisis and the politics of reform: explaining incremental change' in M. Moschella and E. Tsingou (eds) *Great Expectations, Slow Transformations: Incremental Change in Post-Crisis Regulation*, Colchester: ECPR Press, 1-34.

Mueller, J. (1995) *Quiet Cataclysm: Reflections on the Recent Transformation of World Politics*, New York: Harper Collins.

Nelson, F. (2009) 'Cameron will need a scowl and a hatchet to stop us going bust', *Daily Telegraph*, May 28th, 20.

Nersisyan, Y. and Wray, L. (2010) 'Does Excessive Sovereign Debt Really Hurt Growth? A Critique of *This Time is Different*, by Reinhart and Rogoff', *Levy Economics Institute Working Paper No.* 603.

NIESR (2009) 'Coordinated Fiscal Stimulus in the Euro Area', *National Institute Economic Review*, 207, 27-38.

OBR (2011) Fiscal Sustainability Report, July 2011.

Oborne, P. (2007) 'the Brown era: What I believe history will say', *Daily Mail*, March 22nd, 13.

Oborne, P. (2008a) 'The real world and a man in denial', *Daily Mail*, March 13th, 15.

Oborne, P. (2008b) 'Beware! Our Politicians – both Tory and Labour – have Sold Out to the Spivs', *Daily Mail*, September 30th, 12.

Oborne, P. (2008c) 'Day that Britain changed for ever', *Daily Mail*, October 9th, 11.

Oborne, P. (2008d) 'Should the state or free choice rule?', Daily Mail, October 11th, 31.

Oborne, P. (2008e) 'All politicians are in denial. It's the state that needs cutting, not taxes', *Daily Mail*, November 12th, 12.

Oborne, P. (2008f) 'Have the Tories the guts to be the voice of financial sanity?', *Daily Mail*, November 15th, 29.

Oborne, P. (2008g) 'Darling's last roll of the dice and the spectre of wheelbarrows of banknotes', *Daily Mail*, December 20th, 29.

Observer (2007a) 'The audit: What has the Labour government achieved during 10 years in power: The Economy', Newsprint Supplement, April 8th, 42.

Observer (2008) 'Labour sets out strategy for economic rescue', November 23rd, 4.

Observer (2009) 'Britain in good shape, declares top economist', June 14th, 2.

Observer (2010b) 'It's not just about cuts – it's about values', March 28th, 26.

Orr, D. (2009) 'What's wrong with us needing to grow a whole new economy?', *Independent*, April 25th, 16.

Osborne, G. (2009) 'Electioneering is putting confidence in Britain's economy at risk', *Daily Telegraph*, September 5th, 4.

Ozgercin, K. (2012) 'Seeing like the BIS on Capital Rules: Institutionalising Self-Regulation in Global Finance', *New Political Economy*, 17(1), 97-116.

Page, B. (1996) Who Deliberates? Mass Media in Modern Democracy, Chicago: University of Chicago Press.

Pan, Z. and Kosicki, G. (2001) 'Framing as a strategic action in public deliberation' in

S. Reese, O. Gandy, and A. Grant (eds), *Framing Public Life: Perspectives on Media and our Understanding of the Social World*, Mahwah, NJ: Erlbaum, 35-65.

Parris, M. (2009a) 'this is a fight the Tories cannot afford to duck', *Times*, April 23rd, 26.

Parris, M. (2009b) 'The torch has passed. Give Osborne the credit', *Times*, September 19th, 19.

Parsons, C. (2002) 'Showing Ideas as Causes: The Origins of the European Union', *International Organisation*, 56(1), 47-84.

Parsons, C. (2010) 'Constructivism and Interpretive Theory' in D. Marsh and G. Stoker (eds), *Theory and Methods in Political Science* 3ed, Basingstoke: Palgrave MacMillan, 80-98.

Parsons, S. (2003) 'Austrian School of Economics' in J. King (ed), *The Elgar Companion to Post Keynesian Economics*, Cheltenham: Edward Elgar, 5-10.

Payne, R. (2001) 'Persuasion, Frames and Norm Construction', *European Journal of International Relations*, 7(1), 37-61.

Perotti, R. (2013) 'The "Austerity Myth": Gain without Pain?' in A. Alesina and F. Giavazzi (eds), Fiscal Policy after the Financial Crisis, Chicago: NBER and University of Chicago Press, 307-54.

Pew Research Centre (2013) 'Twitter Reaction to Events Often at Odds with Overall Public Opinion', http://www.pewresearch.org/2013/03/04/twitter-reaction-to-events-often-at-odds-with-overall-public-opinion/, accessed 27/4/16.

Pirie, I. (2012) 'Representations of Economic Crisis in Contemporary Britain', *British Politics*, 7(4), pp.341-364.

Platell, A. (2008) 'The joker who's brought back spite and envy', *Daily Mail*, November 29th, 19.

Plender, J. (2008) 'The return of the state', FT, August 22nd, 9.

Pollin, R. (2012) 'US Government Deficits and Debt amid the Great Recession: What the Evidence Shows', *Cambridge Journal of Economics*, 36(1), 161-87.

Posner, P. and Blöndal, J. (2012) 'Democracies and Deficits: Prospects for Fiscal Responsibility in Democratic Nations', *Governance*, 25(1), 11-34.

Pouliot, V. (2007) "Sobjectivism": Toward a Constructivist Methodology', *International Studies Quarterly*, 51(2), 359-84.

Price, R. and Reus-Smit, C. (1998) 'Dangerous Liaisons? Critical International Theory and Constructivism', *European Journal of International Relations*, 4(3), 259-94.

Rachman, G. (2009) 'When austerity does not come easily', FT, May 26th, 13.

Ramey, V. (2013) 'Government Spending and Private Activity' in A. Alesina and F. Giavazzi (eds), *Fiscal Policy after the Financial Crisis*, Chicago: NBER and University of Chicago Press: 19-55.

Randall, J. (2007) 'As our debts pile up, it's too late for Brown to get out in time', *Daily Telegraph*, January 26th, 24.

Randall, J. (2008a) 'Everyone except the Government knows we're spending too much', *Daily Telegraph*, November 28th, 26.

Randall, J. (2008b) 'Shameless layabouts and gutless politicians defraud the taxpayers', *Daily Telegraph*, December 12th, 26.

Randall, J. (2009a) 'Here are just a few things Darling should do in his Budget', *Daily Telegraph*, April 17th, 20.

Randall, J. (2009b) 'Now we are all up to our ears in it', Daily Telegraph, April 24th, 25.

Randall, J. (2009c) 'As we head for the rocks, will Captain Darling change course?', *Daily Telegraph*, December 4th, 28.

Randall, J. (2009d) 'The Treasury won't tell us the true cost of this debt disaster', *Daily Telegraph*, December 11th, 28.

Rawnsley, A. (2009) 'Not so much an election springboard as a death trap', *Observer*, December 13th, 31.

Reece, D. (2007a) 'Brown's duplicity strikes at the heart of Britain', *Daily Telegraph*, March 22nd, 2.

Reece, D. (2007b) 'Personal debt could sink the good ship Great Britain', *Daily Telegraph*, October 23rd, 2.

Reece, D. (2008a) 'More Trouble in Store as Labour's Luck Runs Out', *Daily Telegraph*, March 12th, City, 2.

Reece, D. (2008b) 'Numpty of No 11 and his Budget of unintended consequences', *Daily Telegraph*, March 13th, City, 2.

Reece, D. (2009a) "Can't act, won't act' is set to become Labour's sorry catchphrase', *Daily Telegraph*, April 21st, 3.

Reece, D. (2009b) 'There are just two options remaining for Labour: austerity or inflation', *Daily Telegraph*, April 22nd, 3.

Reece, D. (2009c) 'Lambert's thumbs down to Darling', *Daily Telegraph*, April 25th, City, 31.

Reece, D. (2009d) 'Osborne's path from austerity to prosperity is in the right direction', *Daily Telegraph*, September 16th, 3.

Reece, D. (2009e) 'Preserving the UK's credibility will determine whether we see recovery', *Daily Telegraph*, November 3rd, 3.

Reece, D. (2010) 'An end to recession is not the same as being the start of growth', *Daily Telegraph*, January 26th, 3.

Reich, R. (2009) 'Why the deficit hysteria? I only wish we'd borrow more', *Guardian*, September 1st, 29.

Rein, M. and Schön, D. (1993) 'Reframing Policy Discourse' in F. Fisher and J. Forester (eds), *The Argumentative Turn in Policy Analysis and Planning*, Durham, NC: Duke University Press, 145-66. Reinhart, C. and Rogoff, K. (2010) 'Growth in a Time of Debt', *NBER Working Paper No. 15639*, January 2010.

Renshaw, P. (1985) 'The Labour Movement' in S. Baskerville and R. Willett (eds) *Nothing Else to Fear: New Perspectives on America in the Thirties*, Manchester: Manchester University Press, 214-35.

Richards, S. (2008a) 'A dull Budget but one that sets out the political battle lines of the next election', *Independent*, March 13th, 38.

Richards, S. (2008b) 'Clumsy, reckless and a shadow of his former self, *Independent*, September 4th, 30.

Richards, S. (2008c) 'Darling took his time – but he has ushered in a new age', *Independent*, October 9th, 26.

Richards, S. (2009a) 'Cameron has the tone but he still doesn't have the policies', *Independent*, June 12th, 25.

Richards, S. (2009b) 'Will Cameron be tempted to cut with too much severity?', *Independent*, July 28th, 28.

Richardson, H. (1967) *The Economic Recovery in Britain: 1932-1939*, London: Weidenfeld and Nicolson.

Riddell, P. (2008) 'The mood is fatalistic: whoever's in charge, things can only get worse', *Times*, March 14th, 2.

Riker, W. (1990) 'Political science and rational choice' in J. Alt and K. Shepsle (eds), *Perspectives on Positive Political Economy*, Cambridge: Cambridge University Press, 163-81.

Risse, T. (2000) "Let's Argue!" Communicative Action in World Politics', *International Organisation*, 54(1), 1-39.

Roberts, D. (2009) 'Bankers are back for age of gilts and anxiety', *Guardian*, Newsprint Supplement, April 23rd, 11.

Rochon, L. (2003) 'On Money and Endogenous Money: Post-Keynesian and Circulation Approaches' in L. Rochon and S. Rossie (eds), *Modern Theories of Money*, Cheltenham: Edward Elgar, 115-41.

Rogers, C. (2012) 'Crisis, Ideas and Economic Policy-making in Britain during the 1970s Stagflation', *New Political Economy*, 18(1), 1-20.

Romer, C. (1992) 'What Ended the Great Depression?', *Journal of Economic History*, 52(4), 757-84.

Roubini, N. and Mihm, S. (2011) Crisis Economics, London: Penguin.

Rudebusch, G. (2009) 'the Fed's Monetary Policy Response to the Current Crisis', *Federal Reserve Bank of San Francisco Economic Letter Number 2009-17*, Available at: http://www.frbsf.org/economic-research/files/el2009-17.pdf, accessed 12/04/2013.

Ruggie, J. (1998) 'What Makes the World Hang Together? Neo-Utilitarianism and the Social Constructivist Challenge', *International Organization*, 52(4), 855-85.

Sabatier, P. and Jenkins-Smith, H. (1993) *Policy Change and Learning: An Advocacy Coalition Approach*, Boulder, CO: Westview Press.

Samman, A. (2013) 'The 1930s as Black Mirror: Visions of Historical Repetition in the Global Financial Press, 2007-2009', *Journal of Cultural Economy*, 5(2), 213-29.

Sandbrook, D. (2009) 'Man who needs a history lesson', *Daily Mail*, April 23rd, p.?

Sandbrook, D. (2010) 'Austerity Attlee', *Daily Mail*, February 20th, p.?

Sanders, D. (2010) 'Behavioural Analysis' in D. Marsh and G. Stoker (eds), *Theory and Methods in Political Science* 3ed, Basingstoke: Palgrave MacMillan, 23-41.

Sawyer, M. (2012) 'The Tragedy of UK Fiscal Policy in the Aftermath of the Financial Crisis', *Cambridge Journal of Economics*, 36(1), 205-21.

Scheufele, D. (1999) 'Framing as a Theory of Media Effects', *Journal of Communication*, 49(1), 103-22.

Sardoni, C. (2002) 'On the Microeconomic Foundations of Macroeconomics: A Keynesian Perspective' in P. Arestis, M. Desai and S. Dow (eds), *Methodology, Microeconomics and Keynes*, London: Routledge, 4-14.

Schirm, S. (2011) 'Varieties of Strategies: Social Influences on British and German Responses to the Global Economic Crisis', *Journal of Contemporary European Studies*, 19(1), 47-62.

Schmidt, V. (2002) The Futures of European Capitalism, Oxford: Oxford University Press.

Schmidt, V. (2008) 'Discursive Institutionalism: The Explanatory Power of Ideas and Discourse', *Annual Review of Political Science*, 11(1), 303-26.

Schmidt, V. (2011) 'Speaking of Change: Why Discourse is Key to the Dynamics of Policy Transformation', *Critical Policy Studies*, 5(2), 106-26.

Scott, J. (1998) Seeing Like a State: How Certain Schemes to Improve the Human Condition Have Failed, London: Yale University Press.

Scott, J. (2000) 'Rational Choice Theory' in G. Halcli and F. Webster (eds), *Understanding Contemporary Society: Theories of the Present*, London: Sage, 126-38.

Seabrooke, L. (2006) The Social Sources of Financial Power: Domestic Legitimacy and International Financial Orders, Ithaca, NY: Cornell University Press.

Seager, A. (2009) 'We need a more educated approach to save young people from the dole queue', *Guardian*, April 13th, 24.

Shepsle, K. (2006) 'Rational Choice Institutionalism' in S. Binder, R. Rhodes and B. Rockman (eds), *The Oxford Handbook of Political Institutions*, Oxford: Oxford University Press, 23-38.

Shin, H. (2009) 'Reflections on Northern Rock: The Bank Run That Heralded the Global Financial Crisis', *Journal of Economic Perspectives*, 23(1), 101-20.

Sikkink, K. (1991) *Ideas and Institutions: Developmentalism in Argentina and Brazil*, Ithaca, NY: Cornell University Press.

Sinclair, M. (2010a) 'The longer we wait, the more the cuts will hurt', *Daily Telegraph*, March 23rd, 25.

Skidelsky, R. (1967) *Politicians and the Slump: The Labour Government of 1929-1931*, London: Pelican.

Skidelsky, R. (2009) 'Osborne failing to mind the output gap', *Sunday Telegraph*, September 20th, 4.

Skocpol, T. (1979) States and Social Revolutions, Cambridge: Cambridge University Press.

Skocpol, T. and Weir, M. (1985) 'State Structures and the Possibilities for Keynesian Responses to the Depression in Sweden, Britain and the United States', in P. Evans, D. Rueschemeyer and T. Skocpol (eds), *Bringing the State Back In*, Cambridge: Cambridge University Press, 107-68.

Smith, D. (2008a) 'Enjoy it while you can: hard times are coming', Sunday Times, October 5th, 2.

Smith, D. (2008b) 'Zis will hurt', Sunday Times, December 14th, 19.

Smith, W. (1993) 'Review: International Economy and State Strategies: Recent Work In Comparative Political Economy', *Comparative Politics*, 25(3), 351-72.

Sniderman, P. and Theriault, S. (2004) 'The Structure of Political Argument and the Logic of Issue Framing' in W. Saris and P. Sniderman (eds), *Studies in Public Opinion: Attitudes, Nonattitudes, Measurement Error, and Change*, Princeton, NJ: Princeton University

Press, 133–65.

Soifer, H. (2012) 'The Causal Logic of Critical Junctures', *Comparative Political Studies*, 45(12): 1572-97.

Stephens, P. (2009a) 'Time for Cameron to write his prospectus', FT, April 21st, 13.

Stephens, P. (2009b) 'Cameron in sight of Number 10', FT, April 23rd, 23.

Stephens, P. (2009c) 'Enough of the politics of pessimism', FT, October 9th, 13.

Stewart, H. (2007) 'Can Darling escape the stormy weather ahead?', *Observer*, July 1st, 6.

Stiglitz, J. (2010) Freefall: Free Markets and the Sinking of the Global Economy, London: Penguin.

Strömbäck, J. and Kiousis, S. (2010) 'A New Look at Agenda-Setting Effects - Comparing the

Predictive Power of Overall Political News Consumption and Specific News Media Consumption

Across Different Media Channels and Media Types', Journal of Communication, 60(2), 271-92.

Sunday Telegraph (2009a) 'Shadow of downgrade looms large', May 24th, 6.

Sunday Telegraph (2009b) 'Britain can't fund another fiscal rescue, warns IMF', July 12th, 3.

Sunday Telegraph (2009c) 'The fiscal ruin of the Western world', July 19th, 4.

Sunday Telegraph (2009d) 'It's time for tough decisions on state spending', August 9th, 21.

Sunday Telegraph (2009e) 'Our 25-year penance is only just beginning', August 30th, 4.

Sunday Telegraph (2009f) 'Spending more is no solution', November 29th, 21.

Sunday Telegraph (2010a) 'Labour's latest election pledges fail to convince', March 28th, 23.

Sunday Times (2008a) 'Brown must help business – big and small', November 23rd, 6.

Sunday Times (2008b) 'Pulling back from the brink', November 30th, 6.

Sunday Times (2009a) 'A party no longer fit to govern', December 13th, 20.

Sunday Times (2009b) 'Labour's jumbo deficit deception', December 13th, 20-21.

Sunday Times (2009c) 'Brown pounded over deficit', December 27th, 2.

Sunday Times (2010a) 'Political stalemate produces a vote against the pound', March 7th, 4.

Sunday Times (2010b) 'Do the right thing before you go, Darling', March 21st, 22.

Sunday Times (2010c) 'Come on, chaps: Try honesty this time', March 28th, 22.

Sunday Times (2010d) 'There's still a mountain to climb on economic growth', April 25th, 4.

Sylvester, R. (2009) 'Red is the new black and nasty the new nice', *Times*, April 21st, 19.

Taleb, N. and Spitznagel, M. (2009) 'Time to tackle the real evil: too much debt', FT, July 14th, 9.

Taylor, L., Proaño, C., de Carvalho, L. and Barbosa, N. (2012) 'Fiscal Deficits, Economic Growth and Government Debt in the USA', *Cambridge Journal of Economics*, 36(1), 189-204.

Templeman, M. (2007) 'Nice tax cuts but Brown still gets a thumbs down', *Daily Mail*, March 22nd, 84.

Tett, G. (2009) Fool's Gold: The Inside Story of J.P. Morgan and How Wall St. Greed Corrupted Its Bold Dream and Created a Financial Catastrophe, London: Free Press.

Thelen, K. and Steinmo, S. (1992) 'Historical Institutionalism in Comparative Politics', *Annual Review of Political Science*, 2, 369-404.

Thompson, H. (2012) 'The Limits of Blaming Neo-Liberalism: Fannie Mae and Freddie Mac, the American State and the Financial Crisis', *New Political Economy*, 17(4), 399-419.

Tilly, C. (1994) 'The Time of States', Social Research, 61(2), 269-95.

Times (2007a) 'Chancellor who entertained Prudence and Lady Bountiful', March 17th, 70.

Times (2007b) 'Brown has concocted the ultimate poisoned chalice', March 19th, 39.

Times (2007c) 'Robbing Peter and 'demonetarising' Paul', March 22nd, 4.

Times (2007d) 'Brown comes under attack on two fronts for hurting the poor', March 23rd, 10.

Times (2007e) 'Lowering of corporation tax will still fail to win over firms', March 23rd, 52.

Times (2007f) 'The Great Gordo's best trick yet', March 24th, 3.

Times (2007g) 'I adored you once, Prudence. But now we must part', September 28th, 19.

Times (2007h) 'State borrowing to rise as Treasury left with little 'wiggle room', November 21st, 52.

Times (2007i) 'Scandals are damaging, but economy is the threat', November 27th, 31.

Times (2008a) 'The hangover Budget', March 13th, 1.

Times (2008b) 'Coming crunch', March 13th, 16.

Times (2008c) 'From sub-prime to pump-prime', October 20th, 2.

Times (2008d) 'The boot's on the other foot: was the Left right after all?, October 21st, Times2, 3.

Times (2008e) 'From Dear Prudence to beg, steal or borrow', November 11th, 23.

Times (2008f) 'Tax and spend moves sway verdict on rates', November 17th, 40.

Times (2008g) 'Tax-cut gamble will let Britain buy now – but we'll certainly pay later', November 20th, 26.

Times (2008h) 'Mixing medicines may help people to cough up', November 24th, 6.

Times (2008i) 'Tories hope for jackpot as Cameron gambles on tax', November 24th, 7.

Times (2008j) 'A crisis of confidence', November 24th, 2.

Times (2008k) 'Why is Britain not emulating the US in going for broke?', November 25th, 14.

Times (20081) 'Shock therapy won't cure the banks' ills', November 28th, 40.

Times (2008m) 'Biggest mistake was failure to address the funding crisis', November 29th, Money, 3.

Times (2009a) 'Imagination the key if the Keynesian resurrection is to succeed', March 6th, 63.

Times (2009b) 'The economics of stimulus', March 23rd, 2.

Times (2009c) 'The old lady doth protest', March 25th, 2.

Times (2009d) 'Green shoots of recovery starting to appear, but they will need much nurturing', April 1st, 1.

Times (2009e) 'We face a tough choice – cut pay or cut jobs', April 8th, 19.

Times (2009f) 'City experts doubt recovery before 2010', April 9th, 50.

Times (2009g) 'Times MPC backs Bank stand against fiscal stimulus', April 9th, 50.

Times (2009h) 'Darling poised to slash spending and lay ground for tax rises to tackle debt', April 16th, 16.

Times (2009i) 'Foundations of recovery', April 18th, 2.

Times (2009j) 'Two facts to turn you into an optimist', April 20th, 24.

Times (2009k) 'Darling urged to get a grip on soaring government spending', April 20th, 6-7.

Times (2009l) 'Government prepares to give more state help as industry recovers from global recession', April 20th, 40.

Times (2009m) 'Most Budgets don't add up to much: this one does', April 22nd, 1.

Times (2009n) 'Reading between the lines, this is a moment of truth for the Chancellor', April 22nd, 7.

Times (2009o) 'The debt mountain is yet to be scaled', April 22nd, 35.

Times (2009p) 'Think of a Number and Double it, then Double it Again', April 23rd, 21.

Times (2009q) 'A fig lead to cover the real crisis', April 23rd, 4.

Times (2009r) 'Idea is to get the bad news out of the way now', April 23rd, 7.

Times (2009s) 'Dishonesty and the Budget', April 25th, 2.

Times (2009t) 'Forget the vision of '97. New Labour is old hat', April 27th, 23.

Times (2009u) 'Age of austerity', July 27th, 2.

Times (2009v) 'Progressive austerity', August 12th, 2.

Times (2009w) 'Growing pains', August 18th, 2.

Times (2009x) 'Deadpan Darling cannot hide the grim reality', August 21st, 41.

Times (2009y) 'Darling should beware of debt default', December 1st, 43.

Times (2009z) 'At the very least, the Chancellor must do no harm', December 7th, 37.

Times (2009aa) 'Credibility deficit', December 9th, 2.

Times (2009ab) 'Now they are all agreed, the Tories can dare to be different', December 10th, 9.

Times (2010a) 'The £178 billion question', January 4th, 2.

Times (2010b) 'Unknown unknowns', January 8th, 2.

Times (2010c) 'Four key questions on which recovery depends', January 14th, 56.

Times (2010d) 'Five years of cuts regardless of who is elected, says think-tank', February 4th, 5.

Times (2010e) 'Unkindest cuts', February 18th, 2.

Times (2010f) 'Spending cuts must be spelt out to save stumbling sterling', February 23rd, 54.

Times (2010g) 'Debt obligation', February 25th, 2.

Times (2010h) 'Can Darling be his own man so near an election?', March 22nd, 7.

Times (2010i) 'On a budget', March 24th, 2.

Times (2010j) 'Nakedly political', March 25th, 1,6.

Times (2010k) 'The Darling Budget of May', March 25th, 2.

Times (2010l) 'A political message, strong on rhetoric', March 25th, 12-13.

Times (2010m) 'Business groups open fire on Labour', April 2nd, 3.

Times (2010n) 'Britain in 2015; the real question facing our country is not what we think of the political parties, but what we want for Britain over the next five years', April 6th, 2.

Times (2010o) 'As bad as each other', April 14th, 35.

Times (2010p) 'Economists say £6bn Tory cuts threaten the recovery', April 15th, 9.

Times (2010q) 'The condition of Britain', April 26th, 2.

Times (2010r) 'Things left unsaid'. April 30th, 2.

Times (2010s) 'The economy and the election', May 4^{th} , 2.

Timmins, N. (2007) 'Britain's paradox of public spending', FT, March 22nd, 23.

Toynbee, P. (2008a) 'Osborne's ideas are way off, but his guns are well-aimed', *Guardian*, September 30th, 35.

Toynbee, P. (2008b) 'The epic battle has begun', Guardian, October 28th, 31.

Toynbee, P. (2009a) 'Cameron's Basic Error Will Cost This Country Dearly', *Guardian*, September 12th, 31.

Toynbee, P. (2009b) 'Can Labour shift the terms of the argument? Is anyone listening?', *Guardian*, September 16th, 5.

Trefgarne, G. (2009) 'Brown's spree has tarnished our greatest asset', *Sunday Telegraph*, May 24th, 28.

Tufekci, Z. (2014) 'Big Questions for Social Media Big Data: Representativeness, Validity and Other Methodological Pitfalls', *Proceedings of the 8th International AAAI Conference on Weblogs and Social Media*.

Ulrich, F. (2011) 'Fiscal Stimulus, Financial Ruin' in D. Howden, (2011) *Institutions in Crisis:* European Perspectives on the Recession, Cheltenham: Edward Elgar, 142-63.

Van Dijk, T. (2003) 'Critical Discourse Analysis' in D. Schiffrin, D. Tannen and H. Hamilton (eds), *The Handbook of Discourse Analysis*, Oxford: Blackwell, 352-71.

Van Hooren, F., Kaasch, A. and Starke, P. (2014) 'The Shock Routine: Economic Crisis and the Nature of Social Policy Responses', *Journal of European Public Policy*, 21(4), 605-23.

Van Reener, J. (2012) 'Timing is Everything: Fiscal Consolidation During Depression', Interview for VoxEU, available at: http://www.voxeu.org/vox-talks/timing-everything-fiscal-consolidation-during-depression, accessed 12/01/2013.

Wadhwani, S. (2009) 'Will we come to regret the Governor's intervention on fiscal policy?', *Daily Telegraph*, April 2nd, City, 10.

Walker, R. (1993) *Inside/Outside: International Relations as Political Theory*, Cambridge: Cambridge University Press.

Walker, R. (2008) 'Economy stable? It's empty and the horses have bolted', *Daily Telegraph*, March 13th, City, 7.

Warner, J. (2007) 'If public finances look like this at the top of the cycle, think how bad they are at the bottom', *Independent*, October 10th, 94.

Warner, J. (2008a) 'Chancellor remains optimistic on outlook', *Independent*, March 11th, 48.

Warner, J. (2008b) 'Now there's a surprise. Government is being forced to suspend its fiscal rules', *Independent*, July 19th, 48.

Warner, J. (2008c) 'As the economy wilts, prudence on public borrowing is exposed as a sham', *Independent*, July 26th, 50.

Warner, J. (2008d) 'The economy looks bad, but not nearly as bad as the blundering Government', *Independent*, September 2nd, 40.

Warner, J. (2008e) 'The banking crisis is not so different to previous ones, but policy response is', *Independent*, September 6th, 46.

Warner, J. (2008f) 'Bonfire of the vanities as Wall Street and the City get their comeuppance', *Independent*, September 16th, 34.

Warner, J. (2008g) 'It may not work, but at last policymakers are on the right track', *Independent*, October 9th, 38.

Warner, J. (2008h) 'Governments must be bolder to prevent economic calamity', *Independent*, October 17th, 52.

Warner, J. (2008i) 'It's tough reflating the economy when there's no money', *Independent*, November 11th, 38.

Warner, J. (2008j) 'Government is guilty of double-talk over bank lending', *Independent*, November 27th, 50.

Warner, J. (2008k) 'Crass Keynesianism should be replaced with pure Keynes', *Independent*, December 12th, 52.

Warner, J. (2009a) 'Few Budget options as public finances spiral out of control', *Independent*, April 7th, 38.

Warner, J. (2009b) 'Bolstering confidence must be Chancellor's Budget priority', *Independent*, April 18th, 46.

Warner, J. (2008c) 'Competence, not ideology, is all that matters in economics', *Independent*, April 28th, 38.

Warner, J. (2009d) 'What's the point of the rating agencies?', *Independent*, May 22nd, 46.

Warner, J. (2009e) 'Rising bond yields point the way to economic recovery', *Independent*, June 12th, 36.

Warner, J. (2009f) 'Banking crisis may end up costing taxpayers nothing', *Independent*, June 16th, 40.

Warner, J. (2009g) 'How Britain can escape its economic quagmire', *Daily Telegraph*, September 5th, 9.

Warner, J. (2009h) 'Growth holds the key to reversing Britain's fiscal ruination', *Daily Telegraph*, September 17th, 5.

Warner, J. (2009i) 'Gordon Brown's Global Quest for Re-election; Prime Minister is Calling the Tune, but will G20 Leaders Listen?' *Daily Telegraph*, September 19th, 24.

Warner, J. (2009j) 'Back to Madness and Spreads Return to Pre-Crisis Levels', *Daily Telegraph*, October 15th, 20.

Warner, J. (2009k) 'At last there are signs of life to lift the gloom', *Daily Telegraph*, October 17th, 27.

Warner, J. (2009l) 'Britain Isn't Working. Here's 10 Ways to Help Restore Growth', *Daily Telegraph*, October 27th, 27.

Warner, J. (2009m) 'How I accelerated the economic downturn', *Daily Telegraph*, November 7th, 27.

Warner, J. (2009n) 'Political Fudge Does Nothing to get to Grips with Calamitous Levels of Debt', *Daily Telegraph*, December 10th, 1.

Warner, J. (2010a) 'A make-believe debate – but a real choice', *Daily Telegraph*, April 10th, 19.

Warner, J. (2010b) 'Time is Running Out for Make-believe Economics; The Markets have Already Realised that Britain's Deficit is Unsustainable', *Daily Telegraph*, January 9th, 22.

Warner, J. (2010c) 'Eurozone 'pigs' could lead us all to slaughter', *Daily Telegraph*, February 6th, 25.

Warner, J. (2010d) 'Policy looks impotent in calming the storm', Daily Telegraph, February 18th, 5.

Warner, J. (2010e) 'It's the how to cut that matters – not when', Daily Telegraph, February 20th, 21.

Warner, J. (2010f) 'Recovery? Who are you trying to kid?', Daily Telegraph, February 27th, 23.

Warner, J. (2010g) 'Greece is a harbinger of austerity for all; the effects of the stimulus are wearing off, leaving us with a nasty hangover', *Daily Telegraph*, March 6th, 24.

Wendt, A. (1999) Social Theory of International Politics, Cambridge: Cambridge University Press.

Winch, D. (1969) Economics and Policy: A Historical Study, London: Hodder and Stoughton.

Wincott, D. (2004) 'Policy Change and Discourse in Europe: Can the EU Make a 'Square Meal out of a Stew of Paradox'?', *West European Politics*, 27(2), 354-63.

Wintour, P. (2009) 'Osborne's Gamble – are the Voters Ready to Take Their Medicine?', Guardian, October 7th, 6.

Wolf, M. (2008) 'What Britain must do in the crisis', FT, October 3rd, 13.

Wolf, M. (2009a) 'Political game of blind man's bluff', FT, January 9th, 11.

Wolf, M. (2009b) 'It is always the economy, stupid', FT, February 6th, 13.

Wolf, M. (2009c) 'Indebted UK is flying on a wing and a prayer', FT, April 23rd, 22.

Wolf, M. (2009d) 'Rising government bond rates prove policy is working', FT, June 3rd, 9.

Wolf, M. (2009e) 'How today's global recession tracks the Great Depression', FT, June 17th, 9.

Wolf, M. (2009f) 'Private behaviour will shape our path to fiscal stability', November 4th, 13.

Wolf, M. (2009g) 'Give us austerity and fiscal rectitude, but not quite yet', FT, November 25th, 13.

Wolf, M. (2010) 'How to walk the fiscal tightrope that lies before us', FT, February 17th, 9.

Zaller, J. (1992) The Nature and Origins of Mass Opinion, New York: Cambridge University Press.

Zehfuss, M. (2001) 'Constructivism and Identity: A Dangerous Liaison', *European Journal of International Relations*, 7(3), 315-48.